

de Wet, Carl (2017) A mixed-methods study of the implementation of the Trigger Review Method in general medical practice. PhD thesis.

http://theses.gla.ac.uk/8349/

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten:Theses http://theses.gla.ac.uk/ theses@gla.ac.uk

A mixed-methods study of the implementation of the Trigger Review Method in general medical practice

Dr Carl de Wet MBChB DRCOG FRACGP FRCGP MMed

Submitted in fulfilment of the requirements for the Degree of PhD School of General Practice & Primary Care College of MVLS, University of Glasgow University of Glasgow

Deposition to the Library: August 2017

Abstract

Introduction

There is now compelling evidence that a significant minority of patients suffer preventable iatrogenic harm during their interactions with health care, including in UK general practice. While our understanding of the extent of the problem and the contributing factors continues to increase, it remains incomplete. Further patient safety research is therefore urgently required, particularly to develop, test and successfully implement effective improvement strategies, methods and tools. Of the main approaches currently available for improving patient safety, the general practice Trigger Review Method (TRM) is of particular interest and the main focus of this study.

The TRM is, quite simply, a structured way to rapidly screen samples of random electronic patient records for undetected patient safety incidents (PSIs). It is essentially an adaptation of clinical record review, with the same underlying principles of learning from error and improving care. Development of the TRM commenced in 2007 in Scottish general practice, with subsequent testing in The Health Foundation-funded Safety and Improvement in Primary Care (SIPC) programme. In 2013, the TRM was included as one of the three core components of the Scottish Government's Patient Safety Programme for Primary Care (SPSP-PC). Scottish general practices were also financially incentivised through the Quality and Outcomes Framework (QOF) to routinely apply the TRM and report their findings.

However, despite the increasing and national interest in the TRM, many unanswered questions remained: what is its potential value, how acceptable and feasible is it and to what extent (if any) will, or should, it become part of routine general practice? The aims of this study were therefore to: (i) describe the patient safety perceptions of general practice clinicians and staff; (ii) determine the usefulness of the TRM; (iii) explain how the TRM worked; and (iv) identify the main factors that facilitated or hindered its implementation.

Methods

This study has a mixed-methods design. It was undertaken in the West of Scotland region in two NHS Health Boards: Greater Glasgow and Clyde (GGC) and Ayrshire and Arran (A&A). Convenience samples of 12 general practice teams and 25 GP Specialty Trainees (GPST) were recruited. Data were collected through: semi-structured interviews (n=62) with a range of general practice clinicians and staff; and cross-sectional trigger reviews of selected electronic patient records.

Normalisation Process Theory (NPT) underpinned all stages of the research. NPT is a socio-technical, middle-range theory about the 'work' people do collectively and as individuals to implement and sustain complex health care interventions such as the TRM. The majority of the qualitative data were analyzed thematically and a NPT framework was applied to the remaining data. Quantitative data were analysed using recognised statistical tests.

Results

A total of 47 primary care clinicians reviewed 1659 electronic patient records and detected 216 PSIs. A substantial minority of these were considered to have led to moderate or more substantial harm (29.2%), while the majority (54.8%) were rated as being preventable or potentially preventable. The most common type of PSI related to 'medication' (40.7%) and the most commonly implicated drug was Warfarin. The participants reported considering or undertaking specific improvement actions during and after approximately two thirds of trigger reviews. The most common action was 'feedback to colleagues'. More specific actions included: undertaking significant event analyses (SEAs) and clinical audits, designing or redesigning practice protocols and including their findings in their appraisal documentation.

The vast majority of participants identified four main factors as being particularly important for the successful implementation of the TRM, and by extension its potential normalisation. The first and most important factor was provision of adequate resources and protected time to conduct trigger reviews. The second factor was whether senior leaders in the practice teams, the government and professional bodies practically demonstrated their support for the TRM through, for example, contextually integrating it into existing general practice processes. The third and fourth factors related to the characteristics of participants. Successful implementation required knowledgeable clinicians to remain engaged with the TRM, and to perceive it as useful, acceptable and feasible - which the vast majority of participants were, and did.

Discussion

This study is the first known attempt to investigate how the TRM is implemented and perceived from the perspective of general practice clinicians and staff. The main findings are that most participants experienced the method as acceptable, feasible and useful. It is clear that the TRM is uncovering important patient safety concerns and also driving improvements in related care systems and processes at the individual practice level. The implication is that this is making significant and demonstrable differences to patient care, while impacting positively on local safety culture. On the evidence presented, normalisation of the TRM in general practice can therefore be recommended.

However, while the usefulness of an intervention is an important factor in determining whether it is normalised or not, the study findings also clearly indicate - consistent with the international literature - that there are other factors that are at least equally important for normalisation. At the time of writing, there are no formal mandates or financial incentives for general practice clinicians or teams to perform regular trigger reviews. It therefore seems likely that normalisation of the TRM in Scottish general practice will be gradual and piecemeal, if it happens at all. Nevertheless, the lessons learnt from this study can be incorporated in the ongoing efforts to further improve the safety of care in general medical practice. In particular, researchers and policy makers should pro-actively identify and address the main factors that are known to facilitate or hinder the implementation of improvement initiatives; the existing knowledge and 'engagement' of clinicians should be recognised and harnessed; and the lessons learnt from PSIs should be more widely disseminated.

Table of contents

No	Description	Page
	Abstract	2
	List of Tables	7
	List of Figures	9
	List of Appendices	10
	Acknowledgements	11
	Author's declaration	12
	Abbreviations	13
	Definitions	15
Ch 1	Introduction	17
Ch 2	Patient safety and general practice	32
2.1	The terminology of patient safety	32
2.2	The incidence of error and harm in general practice (GP)	35
2.3	Common types of patient safety incidents (PSIs)	38
2.4	Patient safety improvement programmes and initiatives	47
2.5	Safety culture	57
Ch 3	The Trigger Review Method (TRM)	61
3.1	Methods for measuring and improving patient safety in GP	62
3.2	Practical application of the TRM	76
3.3	Potential value of the TRM	81
Ch 4	Implementation science and Normalisation Process Theory	87
4.1	Implementation science	87
4.2	Models, frameworks and theories of change in health care	89
4.3	Normalisation Process Theory (NPT)	93
4.4	Study aims	105
Ch 5	Methods	107
5.1	Study design	107
5.2	Data collection	113
5.3	Data analysis	119
5.4	Ethical approval and considerations	135

No	Description	6 Page
Ch 6	Participants' perceptions of patient safety	137
6.1	Perceptions and experiences of patient safety	139
6.2	The main contributing factors to PSIs	147
6.3	Existing improvement actions and potential for future interventions	156
6.4	Participants' perceptions of safety culture	160
6.5	Different perspectives of the causes and contributing factors to PSIs	163
Ch 7	Main outcomes from implementing the TRM	171
7.1	Characteristics of clinician reviewers and trigger reviews	171
7.2	Triggers	175
7.3	Patient safety incidents	179
7.4	Actions and intended actions of the reviewers and practice teams	190
Ch 8	How the TRM works	200
8.1	The TRM enables the detection of PSIs	202
8.2	The importance of ownership of PSIs	210
8.3	Outcomes from PSI ownership	215
Ch 9	Factors that facilitated or hindered implementation of the TRM	223
9.1	Coherence - The sense-making work individuals and teams did to understand the TRM	223
9.2	Collective action - The operational work of enacting the TRM and integrating it with existing practices and contexts	228
9.3	Cognitive participation - The relational work of engaging participants and establishing a community of practice around the TRM	235
9.4	Reflexive monitoring - The appraisal work of adapting and evaluating the TRM	239
9.5	Discussion of the main study findings in Chapter 9	243
Ch10	Discussion	251
10.1	Should the TRM be normalised?	254
10.2	Will the TRM be normalised?	259
10.3	Strengths and limitations of the study	269
10.4	Recommendations	279
	Appendices	291
	References	3 U8

List of Tables

No	Description	Page
2.1	A timeline of the recent history of patient safety in the UK	48
2.2	QOF Quality and Safety indicators for 2013-14	56
3.1	The five sections of primary care records and associated, pre- defined triggers	77
4.1	The NPT Framework	96
5.1	Summary of the number and duration of individual interviews	115
5.2	Severity and preventability rating scales for detected PSIs	120
5.3	Summary of the quantitative data derived from the trigger review summary sheets	121
5.4	Summary of the study's qualitative data analysis strategies and processes	125
6.1	Participants' patient safety narrative perspectives	140
6.2	The four main patient safety themes participants described	143
6.3	Factors perceived as contributing to PSIs	150
6.4	Examples of improvement methods participants already used	157
6.5	Examples of the potential effects (positive or negative) when different cultures interact in health care	161
7.1	Characteristics of the participating general practices	173
7.2	Number, type and comparison of triggers detected by reviewers in this study and the QOF study	176
7.3	PSI severity and preventability ratings recorded by reviewers	182
7.4	Preliminary classification of types of PSIs	188
7.5	The 10 most common medications associated with PSIs	191
7.6	The types of actions reviewers undertook during trigger reviews	195
7.7	Specific actions relating to patient care and practice processes that were taken or intended after completion of trigger reviews	197

No	Description	Page
8.1	Educational value of the TRM: selected examples of individual and practice teams' learning needs and points	217
9.1	A description of the four components of the 'coherence' construct with selected quotes	224
9.2	A description of the four components of the 'collective action' construct with selected quotes	229
9.3	A description of the four components of the 'cognitive participation' construct with selected quotes	236
9.4	A description of the four components of the 'reflexive monitoring' construct with selected quotes	241
9.5	Implementation success factors, quality improvement challenges and the NPT framework	246
10.1	Summary of the aims and main findings of this study	252

8

List of Figures

No	Description	Page
1.1	A timeline of this study, the history of patient safety in Scotland and the Trigger Review Method (TRM)	27
8.1	How the TRM works and the related chapter sections	201
10.1	The association between 'workload' and 'resources' in general practice	265

List of Appendices

No	Description	Page
A1	Ethical approval letter	291
A2	Invitation letter to general practices	292
A3	Invitation letter to GP specialist trainees	294
A4	Consent form for general practices	296
A5	Consent form for individual study participants	297
A6	Schedule for interview one	298
Α7	Schedule for interview two	299
A8	The Trigger Review Summary Sheet	300
A9	The NPT framework and exemplar quotes from this study	302
A10	Publications related to this study	304
A11	Presentations related to this study	305

Acknowledgements

I would like to express my gratitude and appreciation to my supervisors, Dr Paul Bowie and Prof Kate O'Donnell. Paul has been one of the most important influences in my life for more than a decade now. He is my gold standard for mentoring and a true friend. Paul taught me that true expertise is tempered by humility. Kate inspired me through her consistently kind and positive feedback and helped me understand the practical value of theory in research.

I would also like to acknowledge the current and previous directors of postgraduate General Practice, NHS Education for Scotland (West region), Prof Moya Kelly and Prof Stuart Murray. Their vision of building future research capacity, in conjunction with their practical support, greatly facilitated my professional development.

My sincere thanks to all the participants, co-authors and collaborators who were involved in this study. The generous contributions of their professional knowledge and skills, time and support were crucial success determinants. I am particularly indebted to Ms Hazel Baillie and Ms Kirsty Lampitt for their administrative and IT support.

Finally, I would like to express my thanks to my family: my wife Patricia, daughter Annalize and parents Callie and Pietta. Your unconditional support, loyalty and love are my strong foundation and greatest blessings.

Author's declaration

I declare that, except where explicit reference is made to the contribution of others, that this dissertation is the result of my own work and has not been submitted for any other degree at the University of Glasgow or any other institution.

Dr Carl de Wet

Definitions

Term	Definition
Electronic medical record	Comprises the electronic patient record sections relating to: clinical encounter entries, repeat and acutely prescribed medication, correspondence with secondary care and other relevant organizations, clinical investigations (such as blood test results) and clinical codes for diseases and allergies.
Error	The result of choosing the wrong plan to achieve an aim, or not initiating or completing the right plan as intended (1). There are at least three different types of errors: <i>slips</i> (incorrectly executed plans), <i>lapses</i> (a plan or part of plan is not executed) and <i>mistakes</i> (choosing or executing the wrong plan). Not all errors will lead to harm, just as not all harms are caused by error. However, there is an association between error and harm (2). Errors are unintentional and should not be confused with violations, negligence or recklessness.
Harm	Impairment of structure or function of the body and/or any deleterious effect arising there from (3). A pragmatic interpretation is 'anything' that you would not want to happen to you or your relatives while receiving care. Although some incidents of harm are preventable, others are recognised complications of care. The severity of harm ranges from transient inconvenience and self-limiting symptoms, through prolonged admissions, disabling injuries, permanent functional impairment and even death.
Inappropriate medication prescribing	The prescription(s) that introduce a significant risk of an adverse drug related event when there is evidence for an equally or more effective alternative medication (4).
Near miss	An incident which did not reach the patient (3). It is also sometimes referred to as a 'close call' or 'free lesson'
Normalisation Process	A middle-range theory about the required work a complex health care intervention's intended users have to do to

- Theory implement, embed, integrate and normalise it as part of routine practice (5).
- Patient safety The reduction of risk of unnecessary harm associated with healthcare to an acceptable minimum (3).
- Patient safety An event or circumstance which could have resulted, or did incident result, in unnecessary harm to a patient (3). It has become the preferred term when discussing adverse events, near misses and significant events.
- Positive The ability of some health care staff and teams to deliver deviance exceptional performance by overcoming common problems through uncommon or different behaviours and using only existing resources within their communities (6).

SafetyA method of monitoring the occurrence of incidents andlearningdeveloping improvement strategies to address the cause of thesystem (SLS)incidents (7).

- SignificantAny event thought by anyone in the team to be significant ineventthe care of patients or the conduct of the organization (8).
- Violations Deliberate deviation from an operating procedure, standard or rule (3). They are inconsistent with rules or recommended practice familiar to a health care worker. Violations are sometimes adaptive behaviour in response to complex, challenging or demanding situations. It has been argued that violations cannot be eliminated, but that they can be managed (9).
- Work Purposive social action that involves the investment of personal and group resources to achieve goals (5).

Abbreviations

Abbreviation	Description
--------------	-------------

Supervisors, study collaborators and administrative suppor
--

CdW	Dr Carl de Wet
CB	Dr Chris Black
PB	Dr Paul Bowie
HB	Ms Hazel Bailey
JF	Dr Julie Ferguson
JM	Dr John McKay
COD	Prof Catherine (Kate) O'Donnell
SL	Dr Sarah Luty

Common study terms

GP	General practice / general practitioner
GPST	GP specialist training / trainee
NPT	Normalisation Process Theory
PDSA	Plan-do-study-act
PLT	Protected learning time
PM	Practice manager
PN	Practice nurse
PSI	Patient safety incident
SS	Trigger Review Summary Sheet
SEA	Significant event analysis
TRM	Trigger Review Method

Normalisation Process Theory terms

Coherence

- DI Differentiation
- IS Individual specification
- CS Communal specification
- IT Internalization
- CA Collective action
- IN Initiation
- EN Enrolment

- AC Activation
- LE Legitimation
- CP Cognitive participation
- IW Interactional workability
- SW Skill-set workability
- CI Contextual integration
- RI Relational integration
- RM Reflexive monitoring
- SY Systematization
- IA Individual appraisal
- CM Communal appraisal
- RE Reconfiguration

Health care related terms

AA	NHS Ayrshire & Arran Health Board
ES	Enhanced Service
GGC	NHS Greater Glasgow & Clyde Health Board
HF	The Health Foundation
IHI	Institute for Healthcare Improvement
IT	Information technology
NES	NHS Education for Scotland
NHS	National Health Service
QI	Quality improvement
QOF	Quality and Outcomes Framework
RCGP	Royal College of General Practitioners
SIPC	Safety Improvement in Primary Care programme
SPSP	Scottish Patient Safety Programme
SPSP-PC	Scottish Patient Safety Programme in Primary care
UK	United Kingdom
USA	United States of America

Chapter 1. Introduction

The mantra 'first, do no harm' has been a fundamental principle of health care for hundreds, if not thousands of years. It succinctly describes the duty of clinicians to fulfil the reasonable desire of any patient to receive care that is safe. However, around the turn of the century a series of landmark studies in the United states of America (USA) (10, 11), Australian secondary (12) and primary care settings (13) and the United Kingdom (UK) (14) provided irrefutable evidence that a significant minority of patients suffer preventable iatrogenic harm during their interactions with health care systems.

This finding has since been replicated by a large number of studies in these countries (15-22) and worldwide (23, 24). The importance of this research has been emphasized further by influential reports such as '*To Err is Human*' (25) '*Crossing the Quality Chasm*' (26) and '*An Organisation with a Memory*' (27) by leading institutions and government departments which sounded a clarion call to national governments and everyone else involved in health care to address the patient safety problem. The key messages from the reports are still relevant fifteen years later: health care should learn lessons from adverse events, take preventative action and that formal patient safety improvement initiatives are necessary.

The other key driver of the patient safety movement has been - and continues to be - the media. The in-depth and sustained reporting of selected examples of iatrogenic harm through multiple media sources increases the profile of certain patient safety incidents (PSIs) and raises public and patient awareness of this issue. Unfortunately there are many potential examples that could be provided. In the context of the UK National Health Service (NHS) the two most widely publicised examples are arguably the 'scandals' of the Bristol Royal Infirmary paediatric cardiac surgery service (28) and Mid Staffordshire NHS Foundation Trust (29). In both instances endemic and systemic failures in care resulted in preventable harm and death. At other times, the poignant story of one patient who suffered a catastrophic error has had the power to capture public and professional attention alike and galvanize improvement efforts. An example that reverberated through UK general practice was the preventable death of Mr. David Gray from an inadvertent administration of a lethal dose of Diamorphine during what should have been a routine out-of-hours consultation (30).

The responses from national governments, health care organisations, regulatory bodies and health care professionals, managers and staff to the reported deficiencies in quality and safety of health care vary widely. Some acknowledge the problem but perceive it as relatively insignificant or as an intractable, intrinsic and unavoidable element of health care. On the other hand, some national governments and regulatory bodies and many health care organisations, units, teams and individuals acknowledge and prioritise the problem and are taking steps to ensure they deliver care of a consistently high standard. As the focus on patient safety began to grow in modern health care systems internationally, it became an increasingly acceptable area for research and has duly been prioritised in some institutions and at a national policy level. As a result our knowledge and understanding of patient safety continue to increase rapidly.

Literature review

Chapters two to four provide the background and rationale for this study. They also introduce and define key concepts through a narrative review of the international patient safety literature. The literature that is presented was initially identified through a systematic search of English language papers that was conducted in November 2011. The only limits were 'human' studies and 'English' language. The search terms and how they were combined are summarised in Box 1.1 on page 19 and the databases and periods of time that were included are listed in Box 1.2, page 20.

The initial search identified approximately 3500 articles. The abstracts of all studies identified during the search were screened for relevance. When abstracts were deemed relevant, full texts were obtained. A 'snowballing' technique was used to scan the references of the most relevant articles identified in the search to identify other potential useful papers and reports not identified by the formal search process.

Box 1.1. Literature search strategy: search terms

				Adverse event\$		
				OR		
				Adverse drug event\$		
		General practice		OR		
		OR		Significant event\$		
Patient safety	AND	Family medicine	AND	OR	AND	Trigger tool\$
		OR		Harm		
		Primary care		OR		
				Error\$		
				OR		
				Patient safety incident\$		

Box 1.2. Literature search strategy: databases

Medline (1996+) Embase (1996+) British Nursing Index (1994+) Psychinfo (2002+) Health Management improvement Consortium (HMIC)* Maternity and Infant Care (MIDIRS)* EBM reviews (all)*

*No date limits were selected for these databases

At the time of the initial literature search, there was considerably less evidence about patient safety in primary care settings compared with six years later, when this study formally concluded. In 2006 Charles Vincent wrote in the first edition of Patient Safety that '*primary care is more or less virgin territory in patient safety terms*' (31). Three years later, in 'Health Care Error and Patient Safety' the authors dedicated five lines in 288 pages specifically to the general practice setting (32).

Since then, there has been a substantial increase in the number of patient safety-related studies conducted in primary care, with several recent attempts to synthesize the rapidly expanding evidence through literature 'scans' and other review methods (22, 33-35). Given the sheer volume of the available literature the potential risks of selection bias and oversimplifying or omitting key patient safety principles or findings in this study therefore have to be acknowledged. These risks were at least partly addressed through three strategies.

The first strategy was to apply the procedure proposed by Whitlock et al, whereby the potential relevance of literature reviews (where available) are determined by considering whether they are 'on topic' and of sufficient quality to provide confidence in the conclusions (36). The second strategy was to search for 'pivotal' studies that had been conducted after the reviews and consider whether they provided significantly different or new insights (37). The third strategy was to continually update the selected literature for this study through: my practical experience of working in this discipline for the last ten years; the invaluable feedback and suggestions from colleagues, study participants and stakeholders during this time; and through regular electronic notifications of selected publications from the Patient Safety Network (38).

1.1. The Trigger Review Method

The focus of this thesis is the implementation of the Trigger Review Method (TRM) in general medical practice. This is discussed in detail in Chapter 2; however, a brief summary of its development, testing and implementation is outlined here, by means of an introduction.

The terms 'trigger review method' or 'trigger tool' are often used interchangeably and may be new or unfamiliar to some, but the underlying principle and approach of learning from error and improving care are not new. It is essentially an adaptation of clinical record review or 'case note audit' which should be familiar to most primary care clinicians. The TRM is, quite simply, a structured way to rapidly screen samples of random electronic patient records for *undetected* patient safety incidents (PSIs) (39).

1.1.1. Development and testing of the TRM

The TRM was adapted from the Institute for Healthcare Improvement's (IHI) Global Trigger Tool for secondary care in 2007 and subsequently further developed and validated in Scottish general practices. The aim at that time was to determine whether the TRM had value as a metric to quantify harm in general practice. In the pilot study trained clinicians reviewed a one year period in a random selection of 500 patient records. They detected an overall harm rate of 9.4%, although this included some incidents that had originated in secondary care settings and not all incidents were judged to have been preventable (40). However, the study provided evidence that the trigger tool approach was transferable to the general practice setting, and that it may potentially detect more and different types of harm than any other available method.

The TRM was subsequently tested as part of The Health Foundation-funded Safety and Improvement in Primary Care (SIPC) programme in around 70 general practices in seven territorial Health Boards in Scotland over a two year period, commencing in May 2010. The SIPC programme evaluation suggested that the TRM had important educational and improvement value by enabling previously undetected threats to patients to be uncovered in clinical records, thereby providing the general practice team with a new perspective and opportunity to make patient care safer (41). However, it also found that some reviewers were uncomfortable with the term 'harm' or (rightly) pointed out that it unnecessarily restrained their findings. For example, by focusing on harm, potential opportunities for improvement suggested by errors, 'near misses' or 'acts of omission' were not recorded. As a result, the TRM was modified so that its main aim became the detection of PSIs. In addition, the aim of the TRM was expanded. The detection of PSIs was no longer the only outcome of interest. Rather, detecting and recording PSIs should explicitly be considered by clinician reviewers and their practice teams as opportunities to identify potential learning needs and act as 'prompts' to implement the necessary corrective changes and improvements to care processes when judged appropriate.

1.1.2. Implementation of the TRM

In 2013, at the same time this study was nearing the end of its second year, two developments occurred that dramatically influenced the implementation, adoption and potential value of the TRM. The first of these was that the TRM was included as one of the three core components of the Scottish Government's Patient Safety Programme for Primary Care (SPSP-PC) that launched in March 2013 (42). The other was that all Scottish general practices were financially incentivised through the Quality and Outcomes Framework (QOF) to routinely apply the TRM and report their findings (43). As a result, the TRM was implemented across most general practices in Scotland, even though the findings of this study in relation to its acceptability, feasibility and potential usefulness had not yet been confirmed or shared.

Despite the increasing and national interest in the TRM, many unanswered questions therefore remained: what is its potential value, how acceptable and feasible is it and to what extent (if any) will, or should, it become part of routine general practice? It is also important to acknowledge the assumption underlying much of the aforementioned initiatives, e.g. that all clinicians possess the requisite knowledge, skills and attitudes necessary to routinely apply safety improvement interventions, including the TRM. And that in doing this they are able to apply the technique correctly, produce robust data, evaluate their findings and then plan and implement meaningful and sustainable improvements. A further assumption is that healthcare and educational authorities are able to up-skill the GP workforce in the use of the TRM on the scale necessary to support the proposed implementation of this approach on a national basis.

The main aims of this study are to test these assumptions and thereby provide evidence with which to answer these questions. The study aims are formally stated and discussed in Chapter 4, page 105, but are summarised for now in Box 1.3, page 25.

1.1.3. Complex Healthcare Interventions

The vast majority of quality improvement interventions in health care are described as 'complex', and the TRM is no exception. Complex health care interventions are defined as '*deliberately initiated attempts to introduce new or modify existing patterns of collective action in health care*' (44). The words 'deliberate initiation' are important and imply that the intervention is: institutionally sanctioned; formally or informally defined; consciously planned; and intended to lead to a change in outcomes (44). Complex interventions are characterised by emergent activities, multi-layered hierarchies and adaptive capacity which occurs within complex environments and are affected by internal and external feedback (45).

Complex health care interventions consist of three interlinked components: (i) 'actors', e.g. the people implementing and using the intervention; (ii) 'objects', e.g. the intervention and the materials it relates to; and (iii) 'contexts' in which they are implemented (46). All three of these components will be considered in detail. In particular: Chapter 6 will describe the 'contexts' within which the TRM was implemented; Chapter 7 will provide information about the 'actors' who implemented and interacted with the TRM; and Chapters 8 and 9 will examine the 'object' component further by considering how the TRM works and the barriers and facilitators to its implementation.

- To describe the perceptions and understanding of general practice clinicians and staff of patient safety (Chapter 6)
- To determine the usefulness of the TRM by describing the outcomes from its implementation (Chapter 7)
- To explain how the TRM works (Chapter 8)
- To identify and describe the main factors that facilitated or hindered the implementation of the TRM in general practice (Chapter 9)

The effectiveness of complex health care interventions such as the TRM are determined not only by their utility, but also by the unique and dynamic 'contexts' within which they are implemented, and the characteristics and knowledge of the 'actors' who are responsible for using them. So, while the TRM proposes a potential solution to help address the patient safety problem, it should be acknowledged that an intervention by itself has limited value, unless it can be successfully implemented in practice.

The rest of this Chapter will provide an overview of the study and contents of this thesis. Figure 1.1 provides a graphical summary of the timeline of this study, the history of patient safety in Scotland and the Trigger Review Method (TRM).

1.2. General practice and patient safety (Chapter 2)

The patient safety literature can be classified into one of three interlinked groups, depending on what type of knowledge they generate (47). The first group of studies increases our understanding of the 'problems' associated with patient safety and includes knowledge about the epidemiology and nature of patient safety incidents (PSIs). This group of knowledge will be discussed in Chapter two. A chronological summary of sentinel events, reports and national patient safety improvement initiatives that have been implemented in the UK, and particularly in Scottish general practice will be provided and the concept of safety culture will be introduced.

1.3. The Trigger Review Method (Chapter 3)

The second of the three groups of studies are those concerned with developing and testing complex health care interventions, including different improvement methods and tools. These studies produce evidence that increases our understanding about potential 'solutions' to patient safety problems. Chapter three will describe the strengths and limitations of five different types of patient safety improvement methods that are currently available for use in the general practice setting. One of these, the Trigger Review Method (TRM), is the main focus of this study.





P1 - P11 are listed as Appendix 10 (page 304)

1.4. Implementation science and Normalisation Process Theory (Chapter 4)

Building on the knowledge of the previous two groups, we need to understand and evaluate how complex health care interventions - e.g. the potential 'solutions' to problems such as suboptimal healthcare safety - are implemented and why they are effective (or not). This group of literature also includes knowledge about the factors that hinder or facilitate implementation and normalisation processes. The narrative in Chapter four will therefore be about the science of implementation and the many different theories that are currently available for describing and explaining implementation processes and phenomena, some of which are summarised. Of these, Normalisation Process Theory (NPT) was selected to be the theoretical underpinning for much of this study.

NPT was developed in and for the UK primary care setting, which makes it eminently suitable for the purpose of this study. As a middle-range social theory, NPT enables researchers to describe the work participants do individually and as teams to implement, integrate and embed an intervention, which is the main focus of this study. Chapter four concludes with a dedicated section about the NPT framework and a description of the main study aims.

1.5. Methods (Chapter 5)

Chapter 5 describes the mixed-methods study design, sampling and recruitment strategies. The different methods that were used to collect and analyse the qualitative and quantitative data are then described. Finally, specific ethical considerations are discussed.

1.6. The study participants' perceptions of patient safety (Chapter 6)

The main study findings are described in chapters 6 to 9. In each chapter the findings are compared with the international patient safety literature where relevant, some of the important practical implications are described and the main points are summarised.

The first of the results chapters (Chapter 6) describes the perceptions of a range of general practice staff about the concept 'patient safety'. In particular, participants considered safety to be an important and integral part of the care they routinely deliver. However, most participants perceived a proportion of PSIs as inevitable, and therefore not preventable. Despite this, they unanimously agreed that many high-risk processes and systems *are* amenable to improvement efforts.

The other sections of the chapter describe the factors participants perceived as most important in relation to PSIs; the wide range of formal and informal improvement methods that participants were aware of or already using; and the importance of the prevailing safety culture at the time of the study.

1.7. Main outcomes from implementing the TRM (Chapter 7)

The concept of 'complex health care interventions' and their three main components: 'actors', 'objects' and 'contexts' has already been introduced (Page 24). Chapter 7 focuses mainly on the third component - the 'object' which in this study is the TRM. In general terms the chapter describes how the TRM was enacted and the outcomes that resulted from this. From a NPT perspective, the main work that is considered is that of 'collective action' and in particular its component of 'interactional workability'. In other words, how did the clinician reviewers apply the TRM, were they able to detect triggers and patient safety incidents (PSIs) and what actions (if any) did they subsequently take?

Throughout the chapter the main findings are discussed when they are reported in order to raise a number of directly relevant issues, and they are compared with the patient safety evidence-base where applicable. In addition, the main findings are compared with the aggregated data from the general practices in three Scottish NHS Health Boards who implemented the TRM subsequent to this study as a Quality and Outcomes Framework (QOF) requirement for the financial year April 2013 to March 2014 (48). These data are referred to as the 'QOF study' and provide additional context for interpreting and discussing the main findings of this study.

1.8. How the TRM worked (Chapter 8)

Chapter 8 explains how the TRM worked. In essence, trained clinicians performed structured trigger reviews of samples of patient records with a patient safety 'mindset', ideally as a single uninterrupted activity during protected time allocated specifically for this purpose. The importance of detecting PSIs was that it created potential 'learning moments' (49). The value of learning moments was in turn determined by clinicians consciously or unconsciously choosing to either accept 'ownership' of PSIs or not. This pivotal decision was crucial to the eventual outcomes of the TRM. The potential outcomes from detecting PSIs and accepting ownership for them are described and include: clinicians and teams taking specific actions to help reduce similar PSIs in the future; increased vigilance and awareness of potential safety threats; and identifying learning needs and points.

1.9. Factors that facilitated or hindered the implementation of the TRM (Chapter 9)

This chapter describes the factors that facilitated or hindered the implementation of the TRM in general practice. The results are presented according to the four main constructs of the Normalisation Process Theory (NPT) framework, which are: understanding (coherence); the work required (collective action); engagement of key actors (cognitive participation); and monitoring and appraisal (reflexive monitoring). The chapter also considers the potential implications of the barriers and facilitators that were identified and compare and contrast them where relevant with the international literature.

1.10. Discussion (Chapter 10)

In the final chapter, four questions in relation to the study are considered. The first two questions are directly relevant to the study aims: 'Should the TRM be normalised and What is the likelihood of it being normalised in general practice in Scotland? In answering these questions, the potential implications of the main study findings will also be described.

The third question is: *What are the strengths and limitations of this study?* In answering this question, the implications of selecting NPT as a theoretical framework will be considered as a separate subsection. The relative strengths and weaknesses of the TRM will also be described. The fourth question is: Can recommendations be made as a result of this study? The short answer is 'yes' - a number of specific recommendations will be made in relation to implementation processes and improvement interventions; the TRM; and current and future research priorities.

Chapter 2. Patient safety and general practice

The aim of this chapter is to distil the literature and practical experiences of the last two decades and provide a short yet cohesive summary of the state of patient safety in UK general practice. The chapter begins with the terminology of patient safety and defines common terms in this study, including 'error', 'harm' and 'patient safety incident' (PSI). Next, the incidence of error and harm in general practice is considered.

Current estimates of preventable harm vary widely and are dependent on study designs and methods, but provide compelling evidence of room for improvement. The two high-priority areas in general practice which are implicated in the vast majority of PSIs are identified and discussed, e.g. medication and diagnoses-related processes. Two other types of patient safety incidents (PSIs) are also considered: ordering and processing of investigations and communication, clinical handover and care transitions. The chapter then summarises influential UK patient safety programmes and initiatives, reports and sentinel events in chronological order before discussing those most relevant to this study, including: the Safety Improvement in Primary Care programme (SIPC); the Scottish Patient Safety Programme (SPSP) and the Quality and Outcomes Framework (QOF). The chapter concludes by explaining the importance of safety culture and how it affects all improvement efforts and initiatives.

2.1. The terminology of patient safety

Patient safety has its own vocabulary, just as any other scientific discipline. However, because it is a relatively new field of research, the same terms often have different definitions and different words are used interchangeably. New terms are regularly being added and the meanings of words evolve. As an example, a review of 'medication safety' in 2005 found 25 different terms relating to the same concept, with 119 different definitions for these terms (50). Even everyday terms that may seem self-explanatory and simple can be challenging to definitively define as demonstrated by a qualitative survey of general practitioners (GPs) who provided 25 different definitions of the common word 'error'. The lack of consensus was attributed to them considering the term from different perspectives relating to three sets of factors: process vs. outcome errors, rare vs. common occurrences and system vs. individual responsibility for errors (51). The different perspectives of participants in this study will be discussed in Chapter 6.

This ambiguity can be frustrating for researchers who feel patient safety terms are 'difficult to pin down' and also give the wrong impression that we are investigating and dealing with unbounded 'relative concepts' (52). Semantic differences in terms and lack of coherence around their functional meanings make direct comparison between studies difficult and reduce the potential of evidence to be generalized wider than its initial setting (53). Developing and validating an international, standardized patient safety vocabulary was therefore recognised as an essential requirement for effective patient safety research and improvement (54).

In response, the World Health Organisation (WHO) developed and published an International Classification for Patient Safety (ICPS) in 2009 (3). The definitions of common patient safety terms that are used throughout this thesis are taken from the ICPS and they are listed on page 13. The four terms that are most relevant to this study are defined and discussed below. They are: 'patient safety', the discipline within which this study was conducted; 'patient safety incidents', one of the study's main outcome measures; 'harm' and more specifically avoidable harm; and 'error' because, despite its strong correlation with harm, important differences between the two terms should be explicitly distinguished.

Patient safety: The reduction of the risk of unnecessary harm associated with healthcare to an acceptable minimum (3). A short, practical explanation is that when things go right, nothing bad happens.

Patient safety incident (PSI): An event or circumstance which could have resulted, or did result, in unnecessary harm to a patient (3). This broad definition includes a large number of related terms such as: adverse events, adverse drug events, adverse incidents, near misses, error, harm event and significant events. Harm is defined as the impairment of structure or function of the body and/or any deleterious effect arising there from (3). It can be considered to have occurred if a patient's health or quality of life is negatively affected by any aspect of their interaction with health care. A pragmatic interpretation of harm is that it is anything that you would not want to happen to you or your relatives while receiving care. The severity of harm ranges from transient inconvenience and self-limiting symptoms, through prolonged admissions, disabling injuries, permanent functional impairment and even death. It is also crucial to distinguish between preventable harm and harm as a recognised complication of evidencebased care. In practice it is often challenging to make these subjective distinctions between different degrees of harm severity and whether harm was avoidable or not. This is further complicated by the lack of a validated and reliable classification system or definitions of either concept (55).

Error is the failure to carry out a planned action as intended or application of an incorrect plan. Errors are always *unintentional* and therefore differentiated from violations, negligence or recklessness. These terms are not included or implied in the following chapters unless expressly indicated otherwise. Three main types of errors have been described: slips (incorrectly executed plans), lapses (a plan or part of plan is not executed) and mistakes (choosing or executing the wrong plan) (1).

Observation studies in high-risk medical specialties such as anaesthesia and cardiac surgery have shown that health care errors are common, but that most errors are recognised and ameliorated before harm occurs (56). However, the same errors may go undetected on a different day and then result in harm to varying degrees. The current consensus is therefore that error and harm are clearly associated, but the type of error and the severity of harm are usually not; only some errors lead to harm; and harm does not necessarily imply an error (57). A practical example is the study of the association between the severity of intercepted medication errors in an accident and emergency setting and the probability of patient harm. 82% of errors were rated as 'significant' or 'serious' yet the probability of harm was considered 'medium' (17%) or 'high' (1%) for a minority only (58).

2.2. The incidence of error and harm in general practice

Until recently, there have been no large-scale epidemiological studies to reliably quantify the harm rate in primary care, although a number of these studies are now being planned or have just been concluded (21, 22, 59, 60). Our knowledge of patient safety in the general practice setting were therefore about specific care processes and systems or types of error and harm and were derived from small studies. Consequently, the task of identifying how big an issue safety really is has been described as being akin to 'looking at mountains in the clouds - it is hard to tell where one thing begins and another ends' (61). Just as it can be difficult to separate mountains and clouds, there are significant methodological, cultural and practical challenges in understanding and differentiating between, for example: contributing factors, causative factors and chance; preventable and non-preventable harm and perceptions; and intentions and objective outcome measures. The challenges include: a historic perception that general practice is low-risk and by implication not a research priority; a lack of validated methods that can feasibly and reliably detect all or most PSIs; procuring and allocating the required time and resources to undertake the necessary research; and reaching consensus about what the research priorities should be in general, and as described above reliably differentiating between 'harm' and 'preventable harm'.

Irrespective of these and other challenges, the importance of assessing and improving patient safety in general practice is increasingly being recognised and understood. There are at least three reasons for this. The first reason is that the vast majority of all health care is delivered in this setting. There were more than 300 million face-to-face consultations in general practice in the UK in 2009, which is approximately 90% of all patient encounters in the NHS (62). Since then, there has been a substantial increase in consultation rates, with average consultation duration and total clinical workloads recently described as *'reaching saturation point'* in English general practice (63). The same pattern of year on year increases in workload can be observed in NHS Scotland's general practices, with more than 24 million face-to-face consultations in 2012 alone (64). Primary care clinicians and their representative bodies are arguing strongly that the significant increases in workload have not been matched with
commensurate increases in the workforce or resources and in some instances have declined (65). This argument strongly influenced the context and findings of this study and will be examined in detail in chapters 6, 9 and 10.

The second reason why patient safety in general practice is important is because of significant changes over the past 20 years or so in the way health care is delivered, which in turn have increased the risk for PSIs to occur in this setting (66). For example, patients are discharged from hospitals earlier than in the past; GPs increasingly prescribe and monitor high-risk medications; time pressures in consultations are increasing with complexity relating to an ageing population, multimorbidity and polypharmacy; and services and continuity of care are fragmenting (63, 67-70). The third reason relates to the nature of general practice as a discipline concerned with '*incremental longitudinal processes*' that is '*founded on decisions concerned with managing uncertainty and marginalizing risk*' (52).

There has been a substantial increase in the last five to ten years in the number and quality of studies about the incidence and nature of safety threats and deficiencies in general practice. As a result, it is now widely accepted that healthcare errors are relatively common in primary care settings worldwide and that a substantial minority result in preventable harm to patients (20). We also know that the majority of harm incidents are minor or moderate in severity, but that some have serious consequences, including hospital admissions and even death. (17) Five examples are provided below as further evidence in relation to these general findings. The first two studies were selected because they provided the first known estimates of error rates in UK primary care, the third example was selected because it suggested that the patient safety problem in primary care was much more serious than previously suggested, and the fourth and fifth example was selected because of its relevance to this study.

The first example is the literature review by Sandars and Esmail who aimed to describe the frequency and nature of medical error in primary care in 2003 (71). Their study was the first known review of its kind, and they found only 12 suitable studies for inclusion. The estimated rate of error was between 5 and 80 per 100 000 consultations and the authors described how different methods and

terminology made it challenging to understand the state of patient safety at that time. The outcomes or potential preventability of the errors were not described. The authors acknowledged that this rate was likely an underestimation of the true error rate as most of the studies had used voluntary incident reporting by clinicians. This was confirmed in that same year when Rubin et al (example two) reported their findings of an error rate almost 100 times higher (75.6 errors per 1000 consultations) in UK primary care. This rate was derived from a voluntary incident reporting system they had implemented in UK primary care (72). These early studies highlighted the need for further research, rigorous methodology and reliable metrics.

The third example is the controversial study by Woods et al who reviewed the discharge documentation of patients in the USA in order to detect the number of preventable adverse events originating in ambulatory care and the severity of their outcomes (73). They estimated that these preventable adverse events resulted annually in as many as 75 000 hospital admissions and 2 587 deaths in the USA. While this study has been widely cited since, it is important to acknowledge its potential limitations and assumptions. The authors reviewed a large sample of 14 700 hospital discharge records and detected 587 adverse events. However, only 31 of these events were judged to be preventable *and* originating from ambulatory care, which they defined as 'outpatient settings' and included family practice, internal and emergency medicine. It was from this small number of cases that they generalized and calculated rates for the wider USA context. It has since been argued that '*implausible estimates of deaths due to medical error will do more to erode the cause of patient safety than headline-friendly figures will do to help it' (74).*

The fourth example is the study by de Wet and Bowie who performed a trigger review of a sample (n=500) of electronic patient records from Scottish practices in 2008 (40). They found that harm occurred on average at a rate of one event per 48 consultations. Of these, 42% of adverse events were judged to potentially be avoidable. While the estimated harm rate of 9.4% in general practice was comparable to that in hospitals, it was derived from a small sample of records, with no attempt to calculate inter-rater reliability or verify the adverse events. In addition, some of the detected incidents had originated in secondary care.j

The fifth and final example is the systematic review (n = 109 studies) undertaken in 2016 by Panesar et al in order to answer the question: 'how safe is primary care (22)?' They found rates of from <1 to 24 PSIs per 100 consultations, of which a small minority were associated with severe harm.

2.3. Common types of PSIs

Most PSIs in general practice belong to one of the following four category types: (i) medication and medication-related issues; (ii) diagnoses and diagnostics; (iii) organisational, which includes systems, procedures and managing investigations; and (iv) communication, which includes clinical handover and patients' care transitions between different health care providers and agencies (75). In many instances more than one category may be relevant as demonstrated through the summarised example in Box 2.1 of a PSI that was submitted to NHS Education for Scotland (NES) as a significant event analysis (SEA) report. Each of the four patient safety areas are discussed below and illustrated further through a small selection of studies. Whenever possible, systematic reviews or 'pivotal' studies were selected. However, some examples are included because of their particular interest or relevance to this study. In all instances however, the selected studies are comparable to the wider literature about that topic.

2.3.1. Medication errors

Adverse drug events (ADE) are estimated to affect up to 20% of patients worldwide in primary care settings (19, 76-78). The majority of prescribing errors cause only minor or moderate harm (79), but the small proportion that results in more serious harm, including hospital admissions, is still considerable, given the sheer volume of many millions of prescriptions issued (80). Unfortunately, the true incidence of avoidable medication-related PSIs is not yet clear. Depending on the study design, definition of ADE and specific patient population investigated, the proportions of these events that are potentially avoidable have been reported from as low as 10% to as much as 50%. A patient presented with progressive shortness of breath to her general practitioner and was admitted to hospital with presumed acute exacerbation of her known heart failure. After extensive investigations she was diagnosed with 'nitrofurantoin lung', a rare pulmonary complication of the antibiotic (nitrofurantoin) she had been prescribed as prophylaxis against urinary tract infections. The hospital discharge summary did not contain this information or the results of the investigations they had conducted, the patient was unaware of the cause of her deterioration and the medication was not discontinued in general practice. As a result, her symptoms returned and she was readmitted to hospital. The patient's experience and this admittedly rare complication demonstrate medication-related, diagnostic, organisational and communication deficiencies.

While the exact number of medication-related errors or the proportions of avoidable ADEs may not yet be known, we do have high quality evidence about the causes and contributing factors to ADEs. Garfield et al were the first to map out the whole UK primary care medicines management system and to systematically review the evidence of cumulative medication errors in it (81).

They found that ADEs are usually caused by complex and multifaceted errors that commonly occur in all stages of the medication systems, including prescribing, dispensing, administration and monitoring. According to them the implication is that only a minority of patients (4 to 21%) achieved optimum benefit from their medication. They recommended routine monitoring of adherence, clinical effectiveness and related hospital admissions as the first steps to improve the quality of the medication system.

This may seem like a daunting task - and it is. However, a useful starting point may be the handful of high-risk medications that are associated with the majority of all ADEs (82). These are the cardiovascular drugs, non-steroidal antiinflammatory drugs (NSAIDs) and anticoagulants (83). Other groups of drugs that are also commonly associated with ADEs include the anti-infective agents, drugs used for the treatment of diabetes mellitus and analgesia (84-86). Conversely, some drugs such as anti-anginals and asthma preventers may cause preventable hospital admissions if they are *not prescribed*. An alternative starting point may be to consider the specific risk factors that increase the likelihood of suffering ADEs. These are: increasing age; female gender; very young age (< 4years old); polypharmacy (e.g. multiple prescription items); number of daily doses; multimorbidity; and high consultation rates (78, 79, 83, 84, 86-89).

The risk factors of medication-related PSIs may be well known but there is still a paucity of validated interventions to reliably and cost-effectively reduce them. Zermansky et al conducted a randomized controlled trial in English general practice more than fifteen years ago to determine the potential impact of an experienced pharmacist conducting clinical medication reviews of elderly patients with repeat items. Adverse drug events were not measured but there was no significant impact on mortality, hospital admissions or number of consultations (90). A subsequent systematic review and meta-analysis found

'weak' evidence to indicate that pharmacist-led medication reviews may be effective in reducing hospital admissions but this benefit was not demonstrated when analyses were restricted to RCTs only (91). More recent systematic reviews were conducted in 2013 (92) and again in 2015 (93). They found that some information technology (IT) interventions were able to successfully reduce medication errors in two specific instances: (i) when pharmacists effectively collaborate with prescribers about unsafe prescribing; and (ii) when clinical decision support (CDS) systems (informed by prescribing-safety indicators) target only a limited number of clearly defined medications and specific patient groups. They also found that at least half of IT interventions were unsuccessful or paradoxically associated with e-iatrogenesis (94).

2.3.2. Diagnostic errors

Despite the relative frequency and potentially devastating consequences of diagnostic errors for patients and clinicians alike they remain underreported by doctors and have only recently been acknowledged by the wider patient safety research community as an important area of study (95, 96). Current estimates of diagnostic error rates are highly dependent on the study design and methods. Trigger reviews of medical records detected rates ranging from 5% (97) to 16% (98) of primary care consultations while patients have reported errors in up to 30% of their and their families' diagnoses (99). Diagnostic errors are the most common reason for medico-legal claims against GPs in modern health care systems, including the UK (100).

A review of 25 years' worth of settled primary care medico-legal claims in the USA found that only a small minority were due to negligence (15) and no single clinical condition accounted for more than 5% of the claims. However, some conditions were associated with a disproportionately higher risk of generating a complaint compared with the relative frequency with which they occur in general practice. For example, a missed diagnosis of appendicitis was 25 times more likely to generate a claim than a delayed diagnosis of breast cancer (101). It is unclear why this is the case, but may be due to patient and public perceptions that some diagnoses are 'easier' to make than others. Studies of medico-legal claims also tend to focus only on delayed or wrong diagnoses.

While these are important subtypes of diagnostic error, the potential harm from over-diagnoses are increasingly being recognised (102, 103).

Incomplete history taking and clinical examinations contribute to the majority of diagnostic errors (98). Unfortunately, while some of the contributing factors may seem simple and apparent - especially in retrospect - diagnostic errors are even more challenging to understand and reduce compared with medication, organisational and communication related issues (104, 105).

There are at least three reasons for this (106). The first reason is that diagnostic errors are not condition specific. They only appear that way because certain conditions such as cancers, myocardial infarctions and meningitis (100) are more memorable or lead to litigation more often than others. The second reason is that various patient and condition-specific characteristics combine in infinitely unique clinical scenarios. Some of the more important factors are atypical and non-specific presentations; very low prevalence of some diseases; multimorbidity; polypharmacy; and physiological variation (107). The third reason is the inherent susceptibility of all clinicians to err. The more obvious and better known reasons include: availability bias (108); physical and psychological limitations to human performance; cognitive errors (109); and lack of knowledge (110). However, another crucial reason is that the very same cognitive processes and adaptive behaviours that enable them to provide efficient care most of the time, occasionally and paradoxically predispose them to err.

It is worth examining this final challenge to reducing diagnostic error in a little more detail because of its relevance to this study and as an example of the practical implication of the influential principle of Efficiency-Thoroughness-Tradeoffs (ETTOs) in patient safety. The ETTO principle provides an important perspective to understand the causes and contributing factors of PSIs and will be described in detail in Chapter 6. For now, it is considered from the perspective of clinicians and diagnostic errors.

The 'right' diagnosis is mainly the product of clinicians' conscious and unconscious cognitive processes. Many different processes have been identified and described. Two of these are sufficient to help illustrate the ETTO principle, namely 'hypothesis generation' and 'pattern matching' (also known as 'satisficing'). When clinicians use 'hypothesis generation' they list a wide range of possible diagnoses and then systematically consider and investigate each one in turn to either confirm or reject it. It is a time-consuming and resourceintensive strategy but the benefit is that it helps to reduce diagnostic errors. On the other hand, when clinicians choose to 'pattern match' they will accept the first diagnosis that satisfactorily explains the majority of the available clinical information at that point in time. Pattern matching is time and resource efficient but associated with a higher diagnostic error rate compared with the hypothesis generation approach. The ETTO principle recognises that neither approach is essentially 'right' or 'wrong' and both may be needed in different clinical scenarios. It is only once a diagnostic error had been made and resulted in harm that the need for greater thoroughness becomes apparent in retrospect. In these instances, it is true that the conscious or unconscious decision to select one approach over the other lies with the clinician. Also, contextual, cultural and organisational factors are more often the crucial drivers of clinicians' choices and strongly incentivise efficiency (pattern matching) over thoroughness (hypothesis generation). However, consciously slowing down and reflecting on every case will not necessarily eliminate diagnostic error either (111).

From this perspective, it may therefore be more accurate and helpful to conceptualize and refer to diagnostic errors as 'missed opportunities in diagnoses' (112). For the purpose of this study the more conventional term 'diagnostic error' will still be used but include this expanded definition, unless stated otherwise.

2.3.3. Investigation-related errors

Laboratory tests and imaging are common and essential tools in modern primary health care and help clinicians to diagnose, monitor and screen a very wide range of treatments, symptoms and conditions. Unfortunately, some recommended tests for monitoring chronic conditions and medications are not requested (113) or, once requested, significant results are lost, missed or not followed-up (114). Three examples are provided to help illustrate this safety problem. The first is Casalino et al who reviewed more than 5000 primary care records and found 7.1% of clinically significant outpatient test results had either not been relayed to patients or if they had been discussed, this had not been documented (115).

The second example is Elder et al who used a multi-method approach to intensively study the management of test results in family medicine. They also found that patients had not been notified about the results of a proportion of tests (0 to 13%). However, in their study only 28 to 55% of abnormal results had been followed-up (116). According to them the likelihood of investigations being managed appropriately were associated with two main factors: safety awareness (leadership, communication, teamwork and having policies and procedures in place) and technological adoption (electronic health records and connections between the practice and testing facilities, forcing functions in the software and facilitation of communication with patients through technology).

The third example is Walsh et al's survey of primary care clinicians in which 37% reported consulting with at least one patient in the two week study period who either had a missing test result, or whose treatment in response to a test result had been delayed (117).

From these three examples it might seem that the most common error - or at least the most easily identifiable error - is failing to follow up on significant results. However, this is only one process in the larger system of managing investigations. Other processes include: ordering tests; implementing tests (e.g. taking the right samples at the right time); reporting results to clinicians; clinicians responding to results; notifying patients of results; and general administration (e.g. coding and filing results). Errors have been detected in every single process and are associated with specific system hazards (118, 119). Bowie et al identified four hazard types that are particularly relevant in UK primary care: system variations and weaknesses (e.g., lack of a tracking process); unclear communication between clinicians and administrators; administrators informing patients of test results; and challenges in maintaining patient confidentiality (120). The most error-prone processes were 'implementing' tests (one third of errors), reporting test results to clinicians (one fifth of

errors). This is why laboratory testing in general practice has recently been referred to as the 'blind spot' (121) of patient safety and why it is a priority to standardize and enhance the ordering, tracking, patient notification and follow-up processes to help reduce avoidable investigation-related errors (114). Important progress has been made in this regard with the development and recent publication of a safe laboratory testing 'good practice guidance' by the pan-European LINNAEUS collaboration. It contains 77 statements relating to 10 safety domains and provides a practical starting point for primary care teams who wish to implement safe systems and processes for ordering tests and managing results (122).

2.3.4. Interface and transition-of-care errors

The peri-discharge period has been shown to be a particularly high-risk time for PSIs to occur, of which many are potentially ameliorable or preventable (123, 124). They include: adverse drug reactions, drug omissions and inconsistencies, unjustified medications and drug discrepancies when discharge items are compared with those prescribed before admission or with what was intended by the hospital clinicians (125). Errors and harm relating to procedures and the results of investigations being lost or not followed-up have also been reported, but the most common type of errors are related to medication management and incomplete documentation (126).

There is some evidence that medication discrepancies can be reduced significantly if primary care providers communicate with patients within 24 hours of their discharge from hospital (127). However, a recent systematic review found most unintentional medication discrepancies had 'no clinical significance' and medication reconciliation did not reduce hospital use within the first 30 days post-discharge (128), although significant reductions in readmissions and emergency department visits were found during longer postdischarge follow-up periods (129).

Notwithstanding the lack of consensus about medication reconciliation's usefulness, primary care clinicians still require accurate and timely information about individual patients when they are discharged from hospital settings to ensure care continues and is safe. The discharge documentation should therefore be legible, adequate and include medication lists, relevant investigation results and information about follow-up arrangements (130). It has been argued that effective clinical handover requires not only such standardized information exchanges but that there should also be direct interaction between clinicians (131). Ideally, this would occur with every transition and actively involve patients and their families/carers (132, 133). If we are to achieve this care aspiration there is still much to do as even basic notifications do not reliably happen for every patient. For example, Bell et al found in their multi-centre trial in the USA that only 42% of primary care clinicians had received discharge documentation within a fortnight of their patient being discharged and concluded that there was 'much room for improvement' in communication between hospitals and GPs (134).

The intuitive association between improving hospital-GP communication and improved patient safety outcomes including mortality, hospital readmission and emergency department visits has also not been demonstrated (134). In a recent, large study Oduyebo et al reviewed more than 6000 hospitalizations. Direct communication between inpatient and outpatient providers was reported for 36.7% of patients but again not associated with decreased 30-day readmission rates (135). In fact, Coller et al reports that improving communication may paradoxically have increased readmission rates in a cohort of paediatric patients (136).

In addition to secondary care, patients from general practices also interface with and transition to and from out-of-hours (OOH) services. It is well known that errors and harm do occur in this setting, and it is an important area of research (137, 138). However, for the purposes of this study, interface and transition-of-care errors are arguably less important than the other three error types that were described. This is because, from a general practice perspective, they are usually not preventable. Also, most studies about interface-errors have focused on hospital-GP communication, rather than GP-hospital communication. A recent review (n=20 studies) aimed to address this issue and found no association between the quality of GP communications with hospitals when admitting patients and the subsequent 30-day admission rates (139).

2.4. Patient safety improvement programmes and initiatives

The patient safety literature that have been described so far indicate the need for health care organisations to better identify and act on PSIs. In recognition of this need, there have been several organised attempts in the last decade to develop structured programmes of patient safety in the UK. The key programmes and initiatives will now be discussed.

Table 2.1 chronologically lists improvement programmes and initiatives, reports and sentinel events relating to patient safety in the UK and especially NHS Scotland and Scottish general practice. The initiatives or programmes that are most relevant to this study are then described in more detail. They are: the Health Foundation's Safer Patients Initiative (SPI) and Network (SPN); the Scottish Safety Improvement in Primary Care (SIPC) programme; the Scottish Patient Safety Programme (SPSP); and the Scottish Quality and Outcomes Framework (QOF) for general practice.

2.4.1. The Safer Patients Initiative

The Safer Patients Initiative (SPI) was the first large-scale, complex improvement programme aiming to improve patient safety in the UK. It was commissioned by the Health Foundation and focused on reducing preventable harm in secondary care. Twenty four hospital sites participated during two consecutive phases that ran from 2004 until 2008. The aim of phase one (n=4 hospitals) was to reduce the number of adverse events in the pilot hospital sites by 50%. The initial four sites were to achieve this aim through a 'change package' designed by the Institute for Healthcare Improvement (IHI). This included three core components: (i) implementation of evidence-based interventions in specific clinical areas that are known to be high-risk for PSIs, e.g. critical and peri-operative care and medicines management; (ii) training staff in quality and safety improvement methodologies; and (iii) establishing specific roles for the Chief Executives and senior executive teams in relation to patient safety. The aims of phase two (n=20 hospitals) were to reduce the

Table 2.1. A timeline of the recent history of patient safety in the UK	
---	--

Year	Patient safety improvement programmes, initiatives, reports and		
	sentinel events		
1998 -	The public inquiry into the Bristol Royal Infirmary paediatric cardiac		
2001	surgery services led by Prof Ian Kennedy QC makes 200		
	recommendations for improving patient care and condemns the		
	prevailing 'club culture' which contributed to iatrogenic harm at that		
	time (140)		
2000	The Department of Health (DOH) publishes 'An Organisation with a		
	Memory' in England. The key theme of the report is that health care		
	organisations and staff should identify and learn from PSIs. (27)		
2001	The DOH publishes 'Building a safer NHS for patients' which describes		
	the practical steps to improve care safety. The report identifies		
	specific, high-priority safety threats, proposes the first steps to		
	implement a formal incident reporting system and acknowledges the		
	importance of patient safety research. (141)		
2001 -	The National Patient Safety Agency (NPSA) is established and tasked		
present	with improving patient safety in the NHS in England and Wales. In		
	2012 the NPSA's key functions were transferred to the NHS		
	Commissioning Board Special Health Authority. Patient safety related		
	work was transferred again in April 2016 to NHS Improvement.		
2001 -	The DOH launches the Patient Safety Research Portfolio (PSRP), a		
2009	national program that funds research (n=36 studies) about healthcare		
	error, including how it can be measured and prevented in secondary		
	and primary care settings in England (142).		
2003 -	NHS Quality Improvement Scotland (NHS QIS) is established with the		
present	formal remit to improve the quality and safety of healthcare in		
	Scotland. QIS was renamed in April 2011 to NHS Healthcare		
	Improvement Scotland (HIS)		
2003	The Scottish Executive Health department publishes 'Learning from		
	Experience: How to Improve Safety for Patients in Scotland'. The		
	report recommends raising awareness of iatrogenic error and harm		
	and building a positive, blame-free safety culture. (143)		

2004 The NPSA publishes 'Seven steps to patient safety'. They are: 1. Build

a safety culture; 2. Lead and support your staff; 3. Integrate your risk management activity; 4. Promote reporting; 5. Involve and communicate with patients and the public; 6. Learn and share safety lessons; 7. Implement solutions to prevent harm (144).

2004 - The Health Foundation implements the 'Safer Patients Initiative'

- 2008 (145).
- 2005 The House of Commons Committee of Public Accounts publishes 'A Safer Place for Patients: Learning to improve patient safety'. It concludes that some progress had been made in learning from PSIs and building a safety culture, but that there is much room for further improvement (146).
- 2006 NHS QIS publishes 'Safe Today, Safer Tomorrow'. It recognises the need for co-ordinated senior organisational leadership, additional resources and the active involvement of patients and clinicians in improvement initiatives (147).
- 2008 The Scottish Patient Safety Programme (SPSP) is introduced in
- present selected hospital specialties. It is subsequently expanded into other settings which are described in more detail below (148).
- 2008 Mr. David Gray dies after an inadvertent administration of a lethal dose of Diamorphine during a routine out-of-hours consultation (30).
- 2009 The Health Foundation establishes the 'Safer Patients Network' (149).2011
- 2010 The Health Foundation funded Safety Improvement in Primary Care
- 2012 (SIPC) programme is implemented in Scotland (41).
- 2010 The public inquiry into the Mid Staffordshire NHS Education Trust care
- 2013 failings led by Robert Francis QC makes 290 recommendations (29).
- 2013 The Scottish Patient Safety Programme in Primary Care (SPSP-PC) is
- present launched with a focus on general practice. In 2014 it is extended further to include pharmacy and dentistry (42).
- 2013 The Scottish Quality and Outcomes Framework (QOF) specify 'quality
- 2016 and safety' as one of its four domains. General practices are financially incentivised to comply with specific patient-safety related indicators including the Trigger Review Method (TRM) (43, 150).

mortality rate by 15% and adverse events by 30% over a twenty month period from 2006 to 2008 (145).

The internal evaluation found evidence that the SPI had an overall positive influence on patient care. The initial four sites measurably improved on a range of processes. As a result, the rates of ventilator associated pneumonia, central line catheter bloodstream infections and surgical site infections all reduced. The twenty sites in the second phase all demonstrated improvements in at least some process and outcomes measures but with substantial inter-site variation.

The external evaluation found evidence that the organisational climate had improved but concluded that there had been no significant reduction in mortality or morbidity and that the quality of prescribing remained unchanged during the study period compared with the control group (145). They acknowledged this might be due to the relatively short study period and because interventions were implemented in individual units while the main outcomes were measured at the organisational level.

Despite the lack of quantitative evidence that the SPI significantly improved patient safety, it remains an important and influential programme that provided valuable experience and lessons about complex health care interventions and implementing change at an organisational level in the UK. The evaluation identified important factors that hinder or facilitate successful implementation. Of these, the most important factor is arguably allocating adequate and appropriate resources. This issue will be considered again in Chapters 9 and 10.

The evaluation also identified important learning points, including:

- It is essential to raise awareness of the patient safety problem amongst staff before implementing improvement programmes, i.e. assess the readiness of an organisation for change and ensure clinicians' engagement;
- The majority of health care workers will likely require additional training before they are able to contribute effectively to quality improvement initiatives, i.e. it is necessary to build capacity and capability;

- It is important for researchers and policy makers to understand the unique contexts and environments within which they implement their initiatives and accordingly adapt their improvement methods;
- The right data need to be collected reliably to evaluate the impact of an intervention and this may require developing or adapting existing reporting systems. The SPI used the Global Trigger Tool (GTT) for this purpose and this method will be discussed further in Chapter 3.

2.4.2. The Safer Patients Network

The Health Foundation envisaged an organic network of likeminded individuals and organisations who would independently and collectively continue to build on the work of the SPI and share their experiences and learning. They therefore launched the Safer Patients Network (SPN) in June 2009. The vision for the network was to create a self-sustaining, member-driven community that would spread improvement throughout their organisations and further afield. One of the key findings of the in-depth evaluation of the SPN is that this vision was unfortunately not adequately promoted and potential members therefore did not understand the value of joining or remaining in the network. The launch of the network also coincided with a number of other national and regional initiatives which created competing priorities for organisations. In addition, inadequate resources were provided for maintaining the network and members were dissatisfied with the seemingly unstructured nature of the network compared with the clear plan and processes of the SPI. Consequently, there was a gradual attrition of sites after the SPI concluded. Eighteen of the original 24 sites joined the network. Of these, 10 participated in 'pass it on' collaboratives and only four took part in Innovation projects. The SPN experience is a powerful example of the many challenges inherent in sustaining complex health care interventions for a sufficient period of time to allow their integration and embedding into routine care. This important issue will be considered further in chapters 9 and 10 (149).

2.4.3. The Scottish Patient Safety Programme (SPSP)

The SPSP was launched in 2008 with the ambitious aims to reduce secondary care mortality and harm by 15% and 30% over a five year period. The secondary aims were to reduce healthcare associated infections, adverse surgical incidents and adverse drug events, improve critical care outcomes and build a strong and positive safety culture. The internal evaluation found a reduction in the Hospital Standardised Mortality Ratio (HSMR) of 16.5% between 2007 and 2016. There was also evidence of harm reduction relating to specific conditions or processes. One example is the number of cardiac arrests during hospital admissions that were reduced by 19% (n=11 hospitals) between 2012 and 2015 (148).

The programme design was a collaborative approach with regular regional and local events for staff to learn about quality improvement and share their experiences. For the first few years the programme was supported by the Institute for Healthcare Improvement (IHI) and much of their methodology were incorporated in the SPSP, including the global Trigger Tool and PDSA method. Other important partners were NHS Scotland, NHS Quality Improvement Scotland (now NHS Healthcare Improvement Scotland); NHS Education for Scotland (NES); and the Health Foundation. The initial focus of the programme was acute hospitals in Scotland. However, as the programme became more established it was expanded into primary care (SPSP-PC). The SPSP-PC is described in its chronological place below.

2.4.4. Safety in Primary Care (SIPC) programme

The Health Foundation-funded Safety in Primary Care (SIPC) programme was launched in May 2010 and concluded in June 2012. In the first phase (year one) twenty general practice teams in three regional health authorities in Scotland participated. The second phase commenced in May 2011 with the recruitment of a further four health boards and another 50 practices. One of the main aims of the programme was to test and evaluate a number of potential quality and safety improvement methods and tools in a general practice context. The four main methods and tools were: (i) a proto-type version of the Trigger Review Method (TRM); (ii) GP-SafeQuest, a validated instrument enabling serial measurements of perceptions of safety culture; (iii) the plan-do-study-act (PDSA) method; and (iv) a care bundle approach to improve the reliability of chronic disease management (41).

The SIPC programme was positively received by the majority of participants who gained new theoretical and experiential safety knowledge and reported improvements in practice safety systems, team working and communication with colleagues and patients. However, while the evaluation found anecdotal examples of benefits there was no quantifiable evidence of significant improvements in patient safety (41). Many participants also reported a significant mismatch between their multiple competing workload priorities, the larger than expected amount of time and resources that was required to participate in the programme and the comparatively inadequate levels of backfill funding for staff participation. As a result, three practices disengaged from the programme citing lack of time and increased staff stress due to unmanageable workloads. For them, participation in the programme paradoxically and unacceptably compromised their time to deliver a clinical service. Other practices expressed doubts about future participation in similar initiatives unless their resource concerns were addressed.

2.4.5. Scottish Patient Safety Programme in Primary Care (SPSP-PC)

The Scottish Patient Safety Programme in Primary Care (SPSP-PC) was developed from the SIPC programme by using, and adapting, the same collaborative approach, tools and methods. It was implemented in March 2013 with the aim to measurably improve the quality and safety of care in those general practices who volunteered to participate. At that time SPSP-PC had three core components: (i) detecting, learning from and reducing PSIs by applying the Trigger Review Method (TRM) to samples of patient records; (ii) measuring and building a strong and positive safety culture; and (iii) improving chronic disease and medication management by using a care bundle approach. The TRM component was subsequently removed from the SPSP-PC because Scottish practices were financially incentivised to implement the TRM through the QOF which meant duplication of effort. This will be discussed in more detail below (35). An internal evaluation found that the vast majority of general practices are now involved to some degree in the SPSP-PC and have started implementing the different methods (42). There is emerging evidence that medication management may be improving but it is still too early for a definitive assessment of the programme's overall utility. The intention is to expand the SPSP-PC to dentistry, community pharmacy, district nursing and care homes from late 2016 and to optometry in 2017.

2.4.6. GP Quality and Outcomes Framework

The Quality & Outcomes Framework (QOF) was a major component of the General Medical Services (GMS) contract between UK general practices and the NHS. It was introduced in April 2004 to help address longstanding variation in the quality of primary care provision (151). The QOF was the most ambitious, comprehensive and largest pay-for-performance scheme in international healthcare and one of the most important, influential but also controversial initiatives ever to be implemented in UK general practice. The QOF was essentially a pay-for-performance scheme that financially incentivised practices to reliably provide standardized, high-quality care. The QOF measured participating practices' performances annually against a range of evidencebased or pre-agreed 'point-in-time' indicators. Practices 'earned' points according to their level of achievement for each indicator, with payment starting at a minimum threshold (usually 40%) rising to a maximum (usually 90%). Points were weighted according to the practice list size and were worth from tens to hundreds of pounds each. From 2004 until 2013 practices could achieve a maximum of 1000 points a year. In the 2013/14 financial year the maximum number of points was reduced to 923 points. In the 2014/15 and 2015/16 financial years the number of points were reduced further to a maximum of 659. (43, 150)

Although the QOF has been studied extensively, there is still no consensus about its overall utility, its effect on clinical outcomes or whether it had a significant and positive impact on patient safety (152). Only 'small and inconsistent' associations have been found between QOF indicators, all-cause mortality and emergency admissions (153). On the other hand, there is evidence of improved quality of care for patients with specific chronic diseases and a trend towards improvement in some processes and outcomes which can at least partly be attributed to QOF (154-156). However, there is no evidence of any discernible effect on hypertension-related outcomes (157) and depression indicators failed to improve disease detection or treatment (158). Overall, however, the QOF was 'likely' a cost-effective use of resources (159).

Participation in the QOF was voluntary but in reality most practices would not have been viable business concerns if they had opted out. As a result essentially all Scottish general practices participated in the QOF as it was one of their main potential sources of income. However, following increasing dissatisfaction with the contract and its impact on workload, it was decommissioned in Scotland in 2016 (160). From April 2016 the financial investments previously associated with QOF were transferred into Scottish general practices' core funding. Performance data will still be extracted to support the new peer led GP Cluster Continuous Quality Improvement process as part of the latest GMS contract agreement but will not be used for payment purposes. The other UK home countries are currently considering its future viability. From the perspective of this study only the Scottish QOF will therefore be considered.

The QOF indicators changed considerably over time. In the 2013-14 financial year a 'quality and safety' domain was created with specific indicators financially incentivizing practices to improve this aspect of their service (43, 150). In particular, they were tasked with: reviewing outpatient referrals, reducing avoidable emergency hospital admissions, creating anticipatory care plans for high-risk patients and implementing specific safety-related tools and methods (see Table 2.2, page 56). One of the new quality and safety indicators - Indicator 'QI001(S)' - is of particular importance to this study. Practices who complied with this indicator could earn a maximum of six points. In 2014-15 the indicator was renamed 'QS007' but the requirements remained the same. The 'validated tool' the description refers to was the Trigger Review Method (TRM).

Table 2.2. QOF Quality and Safety indicators for 2013-14

Indicator	Description	Points
Q1001(S)	The practice conducts two case note reviews , using a	6
	validated tool, to detect patient safety incidents, meets to	
	discuss the results, and shares a reflective report on	
	actions and themes that arise from this with the NHS Board	
QP001(S)*	Review data on secondary care outpatient referral	5
QP002(S)*	Participates in an external peer review to compare its	5
	secondary care outpatient referral data with that of the	
	other contractors	
QP003(S)*	follows 3 agreed care pathwaysto avoid inappropriate	11
	outpatient referrals	
QP004(S)*	Review data on emergency admissions	7
QP005(S)*	Participates in an external peer reviewto compare its	17
	data on emergency admissions and to share the learning	
	from at least 25 per cent of the Anticipatory Care Plans	
	(ACPs)	
QP006(S)*	Produces a list of 5% of patients in the practice, who are	5
	predicted to be at significant risk of emergency admission	
	or unscheduled care	
QP007(S)*	Creates Anticipatory Care Plans	30
QP008(S)*	Holds at least 4 meetings during the year to review the	10
	needs of the relevant patients in the practice ACP cohort	
QP009(S)*	Produces and submits a report to the Board on internal	10
	practice and wider Board system changes that may benefit	
	patients with ACPs	
QS002(S)	The practice conducts a safety climate survey with all staff,	5
	clinical and non-clinical, using a validated tool, meets to	
	discuss the results, and shares a reflective report on actions	
	that arise from this with the NHS Board.	

The QOF was initially received with great enthusiasm by general practices but with time there were increasing concerns about the validity of some of the additional and modified indicators that were added annually and the perceived incessant tightening of targets by policy makers. Other concerns were that it created a de-personalising 'box ticking culture'; that it was vulnerable to data distortion and potential gaming; that it accelerated a transition to nurse-led primary care; there were tensions between the different QOF roles as quality improvement tool, regulatory framework or remuneration mechanism; and that the Framework promoted simplicity over complexity and measurability over meaningfulness (161-163). The concerns and perceptions of general practice staff about the QOF are important because they could influence their understanding about new quality initiatives and whether and to what degree new indicators such as 'QS007' are 'enacted' in practice. This has important implications for this study and will be considered in detail in Chapter 9.

2.5. Safety culture

There is strong agreement that safety culture is a highly important concept, not least because it is such a decisive factor in many organisational performance and safety failures (164). Safety culture is important because organisations with a positive safety culture are more likely to learn openly and effectively from failure and adapt their working practices appropriately (165). The converse is true for a weak safety culture, which has been implicated as a causal factor in many catastrophic organisational incidents, for example the Piper Alpha oilplatform explosion, the space shuttle Challenger disaster, and the Zeebrugge ferry incident (164). Comparable NHS incidents where a poorly developed safety culture was cited as a contributory factor would include the failings highlighted in Stafford hospital (high mortality rates from emergency admissions) (29), Bristol Royal Infirmary (high infant surgical mortality rates) in England (28) and the Vale of Leven hospital (potentially avoidable deaths associated with Clostridium difficile) in Scotland (166).

What the aforementioned major incidents have shown us is the influence of high level organisational factors on safety performance and related failures, rather than much of the previous (simplistic) focus in decades past which was often on improving local systems or technical designs or eradicating frontline human error issues (164). It is really only in the past decade that we have begun to look seriously at how we can assess safety culture in healthcare settings to identify related issues of critical importance (such as the strength and effectiveness of team working, communication, leadership, and commitment to safety improvement and so on) and consider their implications in relation to practice systems, performance and safer care.

Safety culture can be defined as 'the sub-facet of organisational culture that is thought to affect members' attitudes and behaviour in relation to an organisation's ongoing health and safety performance' (165). The concept is organic in that culture evolves over time and is influenced by many factors, including the working environment, health & safety policies and practices, the workforce and management as well as leadership characteristics (164). The measurable 'surface' components of safety culture are collectively referred to as the safety climate - 'the individual and group values, attitudes, perceptions and patterns of behaviour that determine how seriously safety management is taken in the workplace' (167).

Assessing the perceptions of the workforce about patient safety can provide a 'snapshot' of the prevailing culture at a given moment in time and is now a widely used and accepted approach amongst many diverse high risk industries, including nuclear power, aviation and healthcare (167). Safety climate instruments are used to measure the values, attitudes, norms, behaviours and perceptions of individual members of a workforce. In this way, the implicit and shared understandings about '*the way we do things around here*' can be rendered visible to the team in the first instance, but also potentially to others such as safety managers and clinical leaders in the organisation. Three examples of validated instruments that are commonly used are: (i) the Manchester Patient Safety and Assessment Framework (MaPSaF) in English primary care (168); (ii) the Safety Attitudes Questionnaire (SAQ) which has been translated and applied in many international secondary care settings (169); and (iii) SafeQuest, a survey with 30 questions (170).

SafeQuest was developed and tested in Scottish general practices, underwent further testing in the SIPC programme (41) and is now one of two key components of the SPSP-PC (42). All participating Scottish general practice teams periodically complete the surveys, meet in groups to discuss and act upon the survey findings, and then submit a summary report of the improvement outcomes to the local health authority.

The issue of safety culture will be considered again in more detail in chapters 6 and 10. For now, the important implication is that the prevailing safety culture (or conversely the lack of a safety culture) influences clinicians and staff to choose behaviours that enhance - or compromise - safety practices and thinking (171). Consequently, safety culture is not only an important contributing factor to PSIs but also to successful improvement efforts, interventions and initiatives.

Conclusion

A thought-provoking editorial by Dr Wachter asked the following, pithy question just over a decade ago now: 'Is ambulatory patient safety just like hospital safety, only without the "stat" (172)?' The answer to this question is an emphatic 'no'. There are many important differences, including: the nature and incidence of errors; patient-clinician relationships; organisational structures and resources; the types of activities and outcome measures that are regulated and rewarded; and the capacity and potential for change. While we can learn much from the patient safety research and improvement initiatives in secondary care settings, it is therefore important to remain mindful that general practice is in some ways a 'whole different world' (172). The aim of this chapter, consistent with this perspective, was to provide an introduction to key patient safety concepts in relation to general medical practice in the UK only, rather than the whole discipline.

The first half of this chapter summarised the epidemiology of patient safety in general practice. More specifically, four common patient safety terms that were particularly relevant to this study were defined. Next, the incidence of error and harm in this setting were reported and the most common types of PSIs were described. The key messages from this section were the wide variation in the

available estimates of preventable harm and the absence of effective interventions to reduce well-known safety risks. However, while there may be reliability concerns about some of the studies' results, there are also incontrovertible evidence of unacceptable levels of preventable, iatrogenic harm in general practice.

The second half of the chapter summarised the patient safety improvement programmes, initiatives, reports and sentinel events in the UK. Selected initiatives or programmes were described in more detail because of their relevance to and strong influence on this study, including: the SIPC program; the SPSP and SPSP-PC; and the Scottish QOF. This section also introduced the concept and importance of safety culture. Safety culture is a core component or outcome measure of many safety improvement programs and is thought to be one of the essential success factors when implementing complex health care interventions. The key message from this section was that the safety of health care, including in the general practice setting, has been and remains a national priority in Scotland. However, despite considerable investment and effort, there has been no reliable evidence so far that care is becoming significantly safer.

Overall then, this chapter described the current state of patient safety in UK general practice and formulated it as a discrete 'problem' of sufficient importance to justify increased efforts and additional research to improve care standards. However, it is unclear how this may be achieved in a cost-effective and feasible manner, or which methods are effective and acceptable for this purpose. The next chapter will therefore continue the narrative by considering the second of three main groups of knowledge we can derive from patient safety research, which are potentially reliable metrics and effective improvement tools and methods with which to address the patient safety problem.

Chapter 3. The Trigger Review Method

Chapter 2 began the narrative summary of the patient safety literature from the perspective of UK general practice. Common patient safety terms were defined, estimates of the incidence of error and harm were provided and the main types of patient safety incidents (PSI) were described. A chronological summary of sentinel events, reports and influential improvement initiatives and programmes in the UK were provided and the concept of safety culture was introduced.

The main message of chapter 2 was that there is irrefutable evidence of avoidable iatrogenic harm in general practice affecting a substantial proportion of patients. Our understanding of the extent of the problem and potential contributing factors continues to increase but remains incomplete and there is a need for further research in this discipline. Until recently, the aim of the majority of patient safety research has been to produce epidemiological knowledge and increase our understanding of why PSIs occur. This is an appropriate focus for initial research in health care, because effective interventions are derived from agreement, understanding and prioritising of a problem. However, while large-scale epidemiological studies are important there is now arguably an even greater need for research to develop effective improvement strategies, methods and tools to measurably improve safety or mitigate error in all health care settings.

This chapter continues the narrative by considering the second main group of patient safety knowledge, which is about the specific methods and tools that are available for measuring and improving health care in general practice. There are four main sections. The first section describes the practical application, relative strengths and weaknesses and intended utility of five main approaches of measuring and improving patient safety. They are: (i) safety learning systems with the examples of Incident Reporting Systems (IRS), Significant Event Analysis (SEA) and Failure Mode and Effects Analysis (FMEA); (ii) Patient safety checklists; (iii) Cyclical improvement methods, e.g. clinical audit, care bundles and the plan-do-study-act (PDSA) method; (iv) mortality data and medico-legal claims analyses; and (v) clinical record review, including Trigger Tools and the Trigger Review Method (TRM).

61

Of these methods, the TRM is the main focus of this study and the rest of the chapter. Section two explains the practical application of the TRM and describes its three sequential steps and minimum implementation requirements. Section three describes the potential value of the TRM and how its main purpose was intentionally conceptualized as improvement rather than measurement of health care performance. This shift in emphasis was strongly supported by original research that formed a part of this study and is therefore described in some detail in the fourth and final section.

3.1. Methods for measuring and improving patient safety in general practice

3.1.1. Safety learning systems

A safety learning system is a method of monitoring the occurrence of PSIs and developing improvement strategies to address the contributing factors (7). Safety learning systems include different types of incident reporting systems, but also methods that allow analyses and learning from adverse events. The two examples of analyses-type methods that will be considered are significant event analysis (SEA) and Failure Modes and Effects Analysis (FMEA), but first a brief overview of incident reporting systems (IRS).

Voluntary IRS have been a popular patient safety research method for many years (173). They harness the unique and personal perspectives of many different health care staff groups but also patients about the care they deliver and receive. IRS are also comparatively cheaper to implement and maintain than some of the other quality improvement methods such as clinical record review. However, they do have a number of well-recognised limitations which include: selective disclosure of incidents (49); variable and poor clinician engagement (21); a perceived disconnect between the reporting, learning and improvement aspects of IRS (174, 175); and the estimated harm rates lack reliability and validity. This will be illustrated further by one secondary care and two primary care examples. The secondary care example is the mixed-method study by Olsen et al who compared the number of adverse events detected through incident report, pharmacist surveillance and 'real-time' record review of 288 hospital admissions (176). The IRS received 11 reports. None caused significant harm and only one was reported by a doctor. Pharmacists identified 30 potential adverse events in the same cohort of patients while the record review method detected 26 adverse events and 40 potential adverse events. There was very little overlap between the three methods; each method mainly detected different PSIs, but the record review method detected the most incidents, the widest range of issues and provided the most detail about contributing factors.

The second example is a mixed methods study to identify adverse events in 4095 patients who visited their general practices during a five month period in the Netherlands (177). General practitioners reported 20 events or approximately one adverse event for every 200 patients who visited their practices. Pharmacists reported six adverse events. There were four adverse events associated with the 28 deaths during the study period. Finally, a retrospective assessment of 150 medical records by external reviewers found 11 adverse events which is approximately one adverse event for every 15 patients who visited the GPs. Again, there was little or no overlap between the detected adverse events but the record review method detected significantly more incidents.

The third example is the implementation of a general practice safety learning system in Alberta, Canada (7). Participating practices (n=19) submitted an average of 1.4 reports per month, with the number of reports decreasing over the study period. Unlike secondary care, the majority of reports were submitted by doctors, and 50% of incidents were associated with patient harm. The authors concluded that IRS provided only a 'glimpse of the scope of the [patient safety] problem' and was not an effective method to determine the types and frequency of PSIs in general practice. They also identified a number of important barriers and several reasons for the 'low reporting rates', including: lack of time to submit reports; clinicians not recognising or interpreting incidents as reportable; practice teams not understanding the value of reporting; and the prevailing practice culture not being conducive to reporting.

3.1.1.1. Significant event analysis

Significant event analysis (SEA) is well established as a safety improvement intervention in general practice in Scotland, where its importance to the patient safety agenda cannot be overstated (8). Documentary evidence of participation was a contractual requirement of the Scottish Quality and Outcomes Framework; it remains a key element of the GP specialty training curriculum; and it is a core element of professional appraisal and medical revalidation (178). SEA has also been successfully introduced in other countries. A recent, notable example would be the inclusion of SEA as one of the core components of the RCGP patient safety toolkit (179, 180).

The SEA method is applied by clinicians and care teams to investigate and learn from sub-optimal care or any other issues of 'significance' that are highlighted for attention, but in reality the great majority of significant events are patient safety incidents (181). The process requires care teams to hold structured meetings to analyse significant events and helps them to identify learning needs and plan effective remedial action together. When undertaken constructively, these meetings provide a forum for meaningful reflection, discussion and analysis in what should be a non-threatening and empathic environment. If done well, SEA can enhance team working and morale, and improve communication and understanding between team members, all of which helps to build a more positive safety culture in GP surgeries (182).

However, there is strong evidence to suggest that many investigations into safety incidents in primary care are poorly conducted (178, 181, 182). There are several key issues that contribute to this problem. For clinicians and others, being involved in a significant event is similar to receiving a form of negative feedback. The subsequent emotional reaction to this type of feedback can interfere with the personal ability to assimilate and process the information beyond the 'self' level (183), thereby potentially impeding an objective and constructive approach to significant events and their analyses. There is also evidence that the health and emotional wellbeing of clinicians involved in these types of events can suffer (the so-called 'second-victim' syndrome) leading to increased stress and anxiety levels and feelings of guilt, helplessness, frustration and anger (184-186). This will be discussed in more detail in Chapter 8, page 217. Additionally, a prevailing 'blame culture' is still widely perceived within health care, which impacts on the preparedness of clinicians to highlight patient safety issues because of concerns about punitive action and professional embarrassment (187). The implications are that many clinicians are highly selective in the types of safety incidents they raise for team-based discussion and analysis, potentially ignoring those of a complex, serious or highly sensitive nature and opting instead for less controversial examples, or even for non-engagement in this learning activity overall (188).

Therefore, while there is strong engagement with SEA in UK primary care, the evidence for its impact on improving the quality and safety of patient care remain mixed (178). The standard of reflection and critical analyses of such events is poor in a substantial proportion of SEAs with many teams seemingly lacking an understanding of the systems and organisational factors contributing to these incidents. Indeed, a recent review of SEA reports showed that most clinicians tend to view the causes of incidents as being mainly attributable to their own actions (181). This will be discussed in more detail in Chapter 6. For now, the important implications are that there may be numerous missed opportunities for individual, team-based and wider organisational learning from PSIs in order to minimise their risks of recurrence and there is wasted time and financial resources associated with participating in (frequently predictable) sub-optimal learning and improvement efforts.

3.1.1.2. Failure mode and effects analysis

Incident reporting systems, SEA and root cause analysis (RCA) are all examples of health care risk assessments using a structured *retrospective* review of PSIs. A complementary approach in high-risk industries is *prospective* risk assessment of complex processes to systematically identify possible ways in which they may fail, the likely impact of such failures, and proactively mitigate these risks. One of the most widely used prospective risk assessment approach in health care is Failure Mode and Effects Analysis (FMEA). FMEA includes the multidisciplinary activities of mapping out processes and systems, quantifying risks according to their probability, severity and detectability, calculating risk priority numbers (RPN) and then intervening to reduce the RPNs. FMEA is increasingly being used as a research and quality improvement tool and serial RPN measures are held as proof that care processes are becoming safer. While this application of FMEA makes sense conceptually, a recent non-systematic review of FMEA found a lack of standardisation, 'no evidence that its outcomes are valid and reliable' and that it is very time consuming. The review also described several mathematical challenges associated with ordinal scales and warned that 'the use of numerical scores gives an unwarranted impression of objectivity and precision' (189).

Retrospective and prospective risk assessment approaches such as incident reporting systems, SEA and FMEA can therefore not be recommended for quantitative assessment of risk or as a reliably metric of harm. However, they do have great potential benefits as qualitative approaches to improve patient safety, particularly when multidisciplinary teams are involved and contribute to analyses and mapping of shared processes and systems.

3.1.2. Patient safety checklists

Checklists are strongly promoted as an effective way to standardize processes, increase the reliability of health care delivery and as cognitive aids to ensure task completion by clinicians and care teams (190). The expectation is that this will support workforce safety performance and provide further systemic defences against error and preventable harm to patients (191). In addition to widespread support from clinical champions the checklist approach is also underpinned by a more robust evidence base compared with many other quality improvement methods (192). A large group of international patient safety experts recently published the findings of their comprehensive assessment of the literature to identify potential patient safety strategies. They conducted a three-stage review over a four year period and identified ten patient safety strategies that are ready for immediate adoption in health care (193). Three of the ten measures hospitals were encouraged to implement relate to checklists of some kind: preoperative checklists and anaesthesia checklists; checklists to prevent central line associated bloodstream infections; and 'do-not-use' lists for hazardous abbreviations. The other seven strategies apply to specific clinical conditions and secondary care settings: interventions to reduce urinary catheter

use; care bundles to reduce ventilator-associated pneumonia; hand hygiene; multi-component interventions to reduce pressure ulcers; barrier precautions to prevent health care associated infections; real-time ultrasonography for central line placement; and interventions to improve prophylaxis for venous thromboembolisms.

At first glance, none of these strategies or specific checklists seems directly applicable to UK general practice. However, the conceptual benefits of a checklist approach are clearly desirable: a safety checklist provides a method to engage front-line staff in the timely and consistent checking of important issues that can impact on the safety, health, and wellbeing of people and practice performance. Bowie et al recently took the first step to this goal by developing and validating a preliminary safety checklist for general practices. The list has 78 items and six domains: medication management; housekeeping; information systems; practice team; patient access and identification; and health and safety. The preliminary checklist has potential as an intervention to measure, monitor, and improve elements of general practice safety and performance, but it is too early to know what the uptake or utility of this list will be, whether it will be associated with improved patient outcomes or the specific barriers to its implementation. However, it seems reasonable that the same barriers (i.e. competing priorities for time, inadequate training and lack of incentives) and facilitators (i.e. a mandate for compulsory use) identified by Shapiro et al to the implementation of checklists in USA procedural primary care would at least have to be considered (194).

In addition to the new general practice safety checklist, there is at least one type of secondary care checklist that may also be useful for the general practice setting - the 'Never Event' list. Never Events are '*serious, largely preventable patient safety incidents that should not occur if the available preventable measures were implemented by healthcare workers*' (195). An unambiguous example of a Never Event in the acute hospital context is performing a surgical procedure on the wrong limb. The rationale for devising and implementing lists of Never Events in healthcare, therefore, is to mitigate or eliminate the risks associated with these types of serious but preventable occurrences. There is now emerging evidence that Never Event lists and policies are delivering on this promise in selected hospital settings (196). Consequently, a preliminary list of ten Never Events was recently developed for the UK general practice setting with the intention to help improve patient safety (Box 3.1) (197).

Before the list is implemented it may be prudent to first consider and address three important and unresolved concerns about secondary care Never Event policies in case they are equally applicable to general practice. The first concern is that some 'Never Events' currently included on lists may not in fact be preventable in every instance despite the best efforts, intentions and adherence to clinical guidelines of the healthcare workforce (198). A second concern pertains to the proliferation and 'broadening' of the Never Events concept with additional items periodically being added to lists (199). The potential risk is that the core essence of the Never Event concept as a means of focusing attention on relatively rare but serious patient safety incidents may be diluted in this process. The third concern is that Never Event policies may have unintended and unwanted consequences, not least being that Never Event lists may become synonymous with medical negligence (200).

Even if these concerns are addressed, there remain a number of important challenges in addition to those mentioned already for the safety checklist approach that will first have to be overcome for the Never Event approach to be successfully implemented in general practice. For example, how do you enforce mandatory reporting in general practice settings where engagement in voluntary incident reporting systems is minimal and inconsistent? A second challenge is whether the specified Never Events can be detected reliably by every general practice. Recent research on the effort required to identify adverse events shows, unsurprisingly, that the rarer the event, the greater the number of patient records to be reviewed to identify such events (201). Other challenges include determining who should be responsible for implementation of such a policy and how this would be resourced, promoted and prioritised.

Box 3.1. Preliminary list of Never Events for UK general practice (197)

- 1. Prescribing a drug to a patient that is recorded in the practice system as having previously caused her/him a severe adverse reaction
- 2. A planned referral of a patient, prompted by clinical suspicion of cancer, is not sent
- 3. Prescribing a teratogenic drug to a patient known to be pregnant (unless initiated by a clinical specialist)
- 4. Emergency transport is not discussed or arranged when admitting a patient as an emergency
- 5. An abnormal investigation result is received by a practice but is not reviewed by a clinician
- 6. Prescribing aspirin for a patient <12 years old (unless recommended by a specialist for specific clinical conditions for example, Kawasaki's disease)
- 7. Prescribing systemic oestrogen-only hormone replacement therapy for a patient with an intact uterus
- 8. Prescribing methotrexate daily rather than weekly (unless initiated by a specialist for a specific clinical condition, for example, leukaemia)
- 9. A needle-stick injury caused by a failure to dispose of 'sharps' in compliance with national guidance and regulations
- 10. Adrenaline (or equivalent) is NOT available when clinically indicated for a medical emergency in the practice or GP home visit

3.1.3. Cyclical improvement methods

Criterion audit, care bundles and PDSA are three quality improvement methods that may appear similar to healthcare professionals because of their cyclical nature. Criterion audit and care bundles are both methods that can be used to measure clinical performance and quality of care. If performance is found to fall short of a set standard, the individual or team is encouraged to implement change in order to improve care. The PDSA method is one possible method to achieve this aim. It requires measurement as part of this process, which is underpinned by the same considerations as criterion audit and care bundles. Each of these three methods are briefly discussed below.

3.1.3.1. Clinical audit

Criterion-based clinical audit is a widely accepted method for monitoring, assessing and improving care quality (202), particularly in UK general practice where a defined method was developed and successfully implemented by Lough and colleagues (203). A Cochrane systematic review of audit (n=140 studies) concluded that the overall strength of evidence for this type of approach to improve quality and standards of care was moderate. In terms of 'success' as a quality improvement (QI) intervention this ranked highly as an effective strategy (>10% increase in appropriate care or equivalent measure) compared with many other approaches (204, 205).

Research strongly suggests that small, highly focussed audits often lead to a much better chance of meaningful improvements in patient care being implemented and sustained (204). In general practice, there is good evidence via external review by trained peer colleagues that many GPs can implement a defined method of criterion based audit and successfully demonstrate potentially sustainable improvements in patient care, although a significant proportion can struggle with this approach (206). A review of the standard of 336 criterion audit reports found a number of '*application of method*' issues which impacted on the potential for effective change and care improvement. For example, 119 projects (35.4%) were judged to have at least one methodological

deficiency in the data analysis and change management stages of the audit (207).

3.1.3.2. Care bundles

The 'care bundle' approach is promoted as a method that may be particularly useful to achieve the aims of improving the reliability of evidence-based care delivery and hence clinical outcomes (208). A care bundle is simply a small number of health care interventions grouped together which normally has a synergistic relationship that impacts on clinical outcome for patients. Bundles usually contain three to six components which may include clinical interventions such as care processes, procedures, or diagnostic tests, but are not deemed suitable to act as comprehensive lists of all possible care. Selection of appropriate bundle components is based on best evidence and/or local considerations and may change with time and experience (209).

There are many similarities between the care bundle and the criterion-based audit method. In fact, a simple way to conceptualize a care bundle is to imagine it as a group of audit criteria (209). However, there are also a number of key differences:

- The care bundle method typically focuses on specific clinical areas or conditions, while the focus of audit is typically on specific processes of care.
- The care bundle involves a composite 'all or nothing' compliance measure, while criterion audits typically report singular compliance measures for individual criteria.
- Every individual component in the bundle should be recognised as an intervention that is routinely delivered or considered for every patient within a specified time period. Compliance with a care bundle and its components is measured on an 'all or nothing' basis, whereas the performance achieved for any given audit criterion does not affect the result of any other criterion (assuming more than one criterion was specified).

Specific care bundles have been implemented in a range of secondary care settings such as paediatric and adult ICU, medical and surgical wards and Accident and Emergency departments in North America and the UK (210, 211).
Reported clinical outcomes have included: significant reductions in health care acquired infections; lower condition-specific and all-cause mortality; reduced re-admission rates of elderly patients; and decreased length of ICU stay and number of ventilation days (212). Although higher compliance rates with bundles are associated with improved outcomes (213), these are difficult to sustain because of organizational and human-system performance factors which often result in rates below 50% (214). In UK secondary care settings, reported compliance with a variety of clinical care bundles ranges from 19-52% (214, 215). Low compliance rates have important safety implications, as a positive and significant association has been found between compliance rates and clinical outcomes such as mortality (210, 215). The care bundle approach has recently been piloted in UK general practice and is included as a core component in the SPSP-PC (216). There is emerging evidence of improvements in specific, high-risk processes such as drug-monitoring but not (yet) of significant reductions in iatrogenic harm in this setting (42).

3.1.3.3. The PDSA method

The Institute of Healthcare Improvement popularized the PDSA method as a key intervention in quality and safety improvement collaboration in the early 2000s. Since then it has been used as a standalone intervention or as a key component of many different quality improvement (QI) programmes in healthcare settings worldwide (217-220).

The PDSA method is an improvement tool that can be used by individuals or teams to plan and test multiple, small and incremental changes to their everyday work practices and systems in a structured manner and then evaluate the impact over time to determine whether improvements in work quality are apparent. A single PDSA cycle consists of the four steps 'Plan, Do, Study and Act' which are performed sequentially. Any number of PDSA cycles can be undertaken sequentially, with the aims of trying different or adapted change interventions (potential improvements) and to increase the number of patients affected by the change per cycle. Cycles often build on the results of previous efforts so that improvement gains accumulate in an incremental manner (221). The PDSA method has a number of potential benefits. Using this approach potentially enables frontline staff to test out planned care or system changes in their own local environment. This increases understanding of the potential feasibility, costs and impact of an intervention before change is implemented on a larger scale, and allows for the opportunity to adapt or abandon the changes if they do not work as planned. As each test (cycle) is small and rapid, the method can provide 'real time' feedback and is therefore relatively safe and resource efficient. The PDSA method may also help to overcome initial resistance to change from other practice colleagues who may be sceptical about the planned change.

However, like all other QI methods, the PDSA approach is not a panacea. A recent systematic literature review found that the reported application of PDSA cycles varied significantly, with many studies failing to comply with the basic principles of the method (222). A key conclusion was that there was much room for improvement in the application and use of the PDSA method and it remains unclear why, when, for whom and in what contexts it is effective (223, 224).

3.1.4. Mortality data and claims analysis

3.1.4.1. Mortality data

Baker et al reviewed the literature (n=53 studies) to determine how mortality data are currently used in the general practice setting (225). They found evidence of increasing interest in this approach in quality improvement activities with exploratory studies conducted about the roles of practice 'death' registers, monitoring and audit and critical incident reviews. However, these activities are hindered by a lack of access to timely and relevant data. In addition, no association between mortality data, quality improvement initiatives and subsequent reductions in mortality rates have been found in general practice (so far). This may be because: the relative numbers of deaths per practice are small; only a tiny proportion of deaths may be avoidable; and quality improvement efforts focusing on structures and processes may not directly affect mortality, or the effect may only become evident after longer periods of study. A large peer-reviewed study (n= 26,126 cases) of settled primary care medicolegal claims in the USA have important implications for patient safety and UK general practice (101). The main finding was that different proportions of claims can be due to negligence (in this case 23%), clinical errors or are the unavoidable result of appropriate care. Conversely, many errors (especially those with trivial or no consequences for the patient) and instances of negligent care are not included in medico-legal data estimates. The distinction between negligence, error and unavoidable harm is critical as there are different potential solutions and approaches depending on this differentiation. Medicolegal data analyses and studies therefore provide estimates of 'claims risk' and not 'error risk'. However, medico-legal claims data remain valuable in extending our understanding of patient safety, especially when combined with data from other methods. In particular, these types of studies have significantly increased our understanding about the incidence, nature and contributing factors to diagnostic errors, as was discussed in Chapter 2. They also provide evidence of the relative frequency of error and harm in general practice settings and the complexity of the associated and contributing individual and systems factors.

3.1.5. Clinical record review

Clinical record review (CRR) is a well-established approach of detecting and quantifying sub-optimal care issues. In fact, it is considered the gold standard approach in patient safety (226). It allows estimation of harm rates for specific patient populations at given points in time and, if repeated, allows comparisons to detect significant changes across time, whether they be deterioration or improvement. The key strength of CRR compared with the other available research, measuring and improvement methods is that it detects a significantly greater proportion of all harm incidents (227). In fact, CRR harm rate estimates are currently the closest approximation a single method can provide of the 'real' epidemiological state of patient safety.

Unsurprisingly, the original landmark studies about the prevalence of adverse events in hospitals in the USA (11), UK (14), Australia (12), Canada (228) and

New Zealand (229) therefore all used some form of CRR. However, while the studies all used CRR, their methods were typically adapted to include a 'stepped' process and included the application of various predefined screening criteria and different groups of clinician reviewers with specific tasks commensurate with their clinical experience (227).

A further strength of CRR is therefore its flexibility. Over the years there have been many examples of different successful adaptations. Different versions of CRR may involve an automated or manual process; retrospective or prospective reviews; reviewing every section or only some sections of medical records; using internal or external reviewers to the organisation; and having one, two or a team of clinicians review records. There is no single 'correct' adaptation (230).

It is important to acknowledge that using the CRR approach to detect patient safety incidents (PSIs) and measure harm does not provide information about their aetiology and contributing factors, nor does it automatically lead to improvement. It is also time consuming and expensive (177, 231, 232). In addition, the reliability (and therefore potential usefulness) of the harm rates detected with any adaptation of CRR is dependent on the method's many constituent parameters (or factors) such as: the quality of the clinical records; individual reviewer factors; and specific characteristics of the review process (227). The parameters will be considered again in more detail later. However, of all the currently available methods, the clinical record review (CRR) is, arguably, the least affected by these barriers and therefore has the most potential as a metric. This will be considered in more detail in the following sections of this chapter.

One of the better known examples of an adaptation of the CRR approach is the 'global trigger tool' popularized by the Institute for Healthcare Improvement (IHI) in recent years as a means for frontline clinicians to estimate harm rates using a rapid, focused and structured approach to record review. Their rationale for the trigger tool method in secondary care settings is its ability to quantify harm accurately with relatively small samples of medical records and to longitudinally track changes in harm rates (233). However, the term 'trigger tool' predates the IHI and was first used in 1974 to describe sentinel words that

may help to identify adverse events in medical records (234). Then, in the early 1990s it was adopted by Classen et al to describe their method of using trigger phrases to search for adverse drug events in patients' electronic records and the hospital pharmacy system (235). Specific trigger tools are now routinely used in many hospital settings worldwide (85, 233, 236, 237) and in ambulatory primary care settings, including Scottish general practice, where it is known as the Trigger Review Method (TRM). Given the relevance the TRM has for this study, the following sections will describe its practical application and potential value.

3.2. Practical application of the TRM

The TRM allows primary care clinicians (e.g. GPs, GP trainees, practice nurses and pharmacists) to review small samples of patient records for previously undetected patient safety incidents (PSI) in a structured, focused, rapid and active manner:

- Structured clinical reviewers consider each of the five sections of a primary care record in turn (Table 3.1.).
- Focused a specific search for pre-defined 'triggers' (e.g. sentinel phrases or words) is conducted. Triggers are prompts or 'signs' in the record that *may* indicate the occurrence of PSIs (Table 3.1.).
- Rapid a maximum of 20 minutes is allocated per record and only a prespecified period in each record is reviewed (usually three calendar months) (39).
- Active clinical reviewers are encouraged to reconstruct each patient journey and should probe, analyse and critically appraise the record for evidence of PSIs and latent risks hidden in it.

Table 3.1. The five sections of primary	care records and associated, pre-
defined triggers (39)	

Saction of the record	Trigger	
Section of the record	(must be present during the review period)	
Clinical encounters	 ≥3 clinical encounters in any given 7 day 	
(face-to-face, telephone or	period	
house calls)		
Medication	'Repeat' medication item discontinued	
	• Optional triggers, e.g. acute prescription of NSAIDs or opiates	
Clinical codes	 A clinical READ code for an adverse drug event and/or allergy was added 	
	• Any new 'high priority' clinical code added	
Correspondence	• Any out-of-hours health care contact (out-	
(referrals, clinic letters,	of-hours service or Accident & Emergency)	
discharge summaries, reports)	• Emergency hospital admission for ≥ 1 day	
Investigations	• Haemoglobin ≤ 10,0 g/dl.	
(imaging, laboratory)	 Optional triggers, e.g. INR > 5 or < 1.7 or AST/ALT > 100 IU/L 	

Clinical reviewers are encouraged to record their findings, reflections and actions on a '*Trigger Review Summary Sheet*' (SS). The SS is a double-sided template for collecting and summarizing data on the number of detected 'triggers', the details of any PSIs uncovered, any learning needs identified and actions that were or should be taken as a result of the review process. There have been many versions of the SS because it is periodically adapted to incorporate the feedback from its users. The version of the SS that was used in this study is included as Appendix 10. The definition of a PSI is provided on the SS to remind reviewers that the TRM's key focus is on detecting a circumstance where harm occurred (physically or psychologically and regardless of severity) or could have happened but was prevented (a near miss) or could happen at some point in the future.

The TRM has three consecutive steps which are described in more detail below:

- (1) Planning and preparation;
- (2) Review of records; and
- (3) Reflection and action.

3.2.1. Planning and preparation

The first step is to define the specific patient population or medical condition from which a small random sample of clinical records will be sampled for review. Although *any* patient population or medical condition could conceivably be selected, the records of frail elderly patients and those with multiple comorbidities and polypharmacy are more likely to contain evidence of PSIs and latent risks. Examples of potential high-yield patient sub-populations are shown in Box 3.2. Specific patient population characteristics may suggest optional triggers. For example, a trigger of 'INR >5' would be suitable for a sample of patients prescribed Warfarin.

The next step is to decide the number of clinical records to sample and what period of time is to be reviewed in each record. Practical experience suggests that reviewing three recent consecutive calendar months in each of the 25 records (randomly sampled from the chosen patient population) is feasible for the vast majority of clinicians.

Box 3.2. Examples of specific 'high risk' patient groups that could be selected for review (39)

1. Specific, Shared Patient Characteristics	2. Chronic Disease Areas	3. High Risk Medications
Nursing Home Patients	COPD	Insulin
>75 years	Stroke/TIA	Morphine
Last 25 Attending Out-of-Hours	CVD	Warfarin
Last 25 Hospital Referrals	Diabetes	NSAIDs
Housebound patients	Heart failure	Diuretics (x2)
Last 25 Hospital Admissions	CKD	>5 Repeat medication items
4. Combinations of Groups 1 to 3 e.g. Patients over 75 years, taking 5+ medications, who attended in previous 12-weeks; nursing home patients prescribed NSAIDs; patients with heart failure and who are prescribed 2 or more diuretics.		

5. Choose Your Own Sub-Populations

e.g. Patients discharged after emergency hospital admission (review the period before and after admission); a random selection of any 25 patients registered with the practice.

3.2.2. Reviewing records

Once the sample of records has been identified, a clinical reviewer screens each record, searching for previously validated pre-defined 'triggers' (Table 3.1.) which may point to the existence of an unknown PSI or latent risk to the patient. For example, the reviewer finds an INR >5 (trigger) and, on further examination of the record, detects that the patient was treated for an associated bleed in secondary care. There is no 'correct' number of triggers. Instead, the number of triggers should be decided by considering the available time and resources (less triggers) and the number of PSIs to detect (more triggers). In practice, only a few triggers will be 'positive', e.g. lead to the detection of a PSI. On the other hand, one trigger may lead to the detection of more than one PSI. A maximum time of 20 minutes is allocated per record and only three calendar months are reviewed in each record. Trigger review studies have consistently found that this amount of time is sufficient (85).

When PSIs are detected, a brief summary of the event should be recorded on the SS. Reviewers are also encouraged to rate the perceived severity and preventability of each detected PSI on a scale from 1 to 4. The dual scoring system was developed by CdW and PB in response to the lack of published guidance on how to judge the 'preventability' of detected PSIs and is described further in Chapter 5. Unfortunately, but inevitably, some patients will be unavoidably harmed as a result of their interactions with healthcare for a range of highly complex reasons. This is a critical and often overlooked issue in the patient safety literature. The key focus from the patient's and the clinician's perspective should therefore be on detecting and learning from those incidents which are judged to be *preventable* e.g. there is consensus that they should not have occurred if the appropriate preventative strategies had been in place.

3.2.3. Reflection and action

In the third step of the TRM, reviewers document any clinical actions they performed during the review (this option is left to their own discretion) and indicate which further actions they intend to take on their SS. A selection of possible actions is offered (Box 3.3) but with the flexibility to consider any

action they considered appropriate. They are also encouraged to reflect on their findings and write down any learning points and needs on a professional and practice level.

Once the SS is completed, the reviewer should consider who they should share the findings with. The ideal forum for sharing the finding is during a practice meeting involving all staff. Finally, the reviewer should consider when they are going to conduct another trigger review. At present, the requirements of QOF are for two trigger reviews during a 12-month period. The first time the TRM is used, a clinician reviewer requires, on average, about two hours of protected time to complete all three stages and the SS - although this may range from around one hour to a maximum of four hours.

3.3. Potential value of the TRM

3.3.1. Appraisal and revalidation

In terms of regulatory and educational policy in the United Kingdom, 'safety and quality' is one of four professional domains describing the expected duties and standards of every doctor registered with the General Medical Council (GMC). Specifically, registered doctors are expected to '*take part in and respond constructively to the outcome of systematic quality improvement activities (eg audit), appraisals and performance reviews*' (238). The TRM is perfectly aligned with this expectation and could therefore play an important role in helping to achieve this standard.

The TRM was therefore included in the GP Appraisal process in Scotland in 2012 as a potential Quality Improvement Activity (239). In Scotland, GPs' participation in the annual appraisal process is mandatory and a prerequisite for their successful revalidation every five years. GPs have to submit supporting

Box 3.3. Suggested actions that could be performed after the review (39)

- Significant event analysis
- Criterion audit
- Implement change for improvement and how this will be achieved
- Provide feedback to a colleague
- Add SS to appraisal documentation
- Submit a formal incident report
- Update or develop a protocol
- Discuss with a GP educational supervisor

evidence of ongoing learning and 'good practice' in four domains: (i) knowledge, skills and performance; (ii) safety and quality; (iii) communication, partnership and teamwork; and (iv) maintaining trust. Each domain has core elements and minimum requirements. The minimum requirements for the 'safety and quality' domain are that GPs will submit ten SEAs and three Quality Improvement Activities in a five year cycle (239). The effect of including the TRM in the GP appraisal process is discussed further in Chapter 9.

3.3.2. Educational value

In response to the recognised patient safety problem, medical educators have started to integrate safety-related topics and issues into undergraduate education and specialty training programmes. The UK Royal College of General Practitioners (RCGP) - which has responsibility for the content of the specialty training curriculum - has developed a curriculum statement on 'patient safety' (240). They also defined specific learning objectives which require GPSTs to demonstrate a whole range of problem-solving skills aimed at improving the management of clinical risk and enhancing the patient experience of care. The TRM is one of the potential quality improvement activities they could undertake that would fulfil these requirements. The TRM also has potential educational value for non-GPST general practice staff. The findings from trigger reviews may help to pinpoint individual and team learning needs and points where patient safety may have been avoidably compromised. The educational value of the TMR is discussed in more detail in Chapter 8.

3.3.3. Improving patient safety

In general practice, safety incidents are typically reported by patients, identified directly by clinicians or highlighted by colleagues as part of routine practice. However, some incident types are not detected so easily - in fact, there is evidence that the majority of incidents remain undetected because they are not reported by patients or clinicians. Systematically reviewing clinical records for previously undetected incidents and potential threats using the Trigger Review Method (TRM) therefore provide care teams with a new perspective on patient safety, by offering valuable opportunities to take preemptive action before harm occurs. The potential value of the TRM as an improvement approach to care in general practice will be discussed in detail in Chapters 7 and 10.

3.3.4. Measuring health care performance

There are at least two important reasons why harm *should* be measured and why harm rates should be calculated in health care, including in the general practice setting. The first reason is that knowledge of the scale of the patient safety problem helps to guide decisions about the amount of resources required to invest in or re-allocate to the potential problem and can inform the design and implementation of improvement initiatives. The second reason is that safety improvement initiatives and interventions need to be formally evaluated to determine whether they are cost-effective and leading to safer care (or not) and robust evaluation requires reliable serial measurements.

The original purpose of the 'trigger tool' approach as applied in secondary care settings was to reliably measure rates of harm detected in the records of specific groups of hospitalised patients over time. There, external reviewers 'objectively' determine and monitor harm rates for individual clinical wards, units and hospitals. Only the aggregated results (harm rates) are typically shared with the care teams. This application of the trigger method has two major drawbacks: the first is that frontline staff do not have ownership of the data and the second is that no attempt is made to investigate why the detected incidents occurred or how they may best be prevented in future. The general practice TRM was adapted to address both these drawbacks.

The TRM has the potential to reliably measure harm in general practice at the regional and national level, providing certain caveats that will be described shortly are observed. However, measuring harm rates at the individual practice level is a potential distraction from the arguably greater benefits to be accrued from applying the method to enhance learning and facilitate improvement activities. This approach is consistent with the explicit understanding that Trigger Tool data should not be used for 'benchmarking' purposes between institutions (241).

There are three important reasons for shifting the focus of the TRM from 'measurement' to 'improvement'. The first reason is that, when this study commenced in 2011, the available resources at the time seemed inadequate to recruit sufficient practices and clinician reviewers to review enough records in order to estimate harm rates with adequate precision or to detect changes in harm rates over time with acceptable power. As a result the decision was taken by CdW to prioritise the TRM's improvement potential over its measurement function.

The second reason relates to the philosophical question of 'how much harm is too much?' In 2011 there was already sufficient evidence to suggest that the safety of care in all health care settings, including general practice, was suboptimal and that there was great potential for improvement (174). A simple clinical analogy would be to consider morbidly obese patients. While measuring and recording an exact weight before implementing reduction strategies is recommended and would be helpful, the absence of a scale should not be used as an excuse for inaction. Similarly, detecting PSIs and becoming aware of patient safety threats in your own practice should be enough motivation for health care professionals to take remedial action, irrespective of what the true incidence of harm may be. These issues will be considered again in Chapter 10.

The third reason was of a more personal nature and informed by anecdotal feedback from clinicians and colleagues during the development and testing of the TRM. While many clinicians understood the importance of measuring harm, they reported a 'disconnect' between 'measurement' activities and what they perceived as the more important task - improving the standards of care they deliver. From their perspective, the overall number of harm incidents, the reliability of estimates and how harm rates compared across practices were unimportant or even irrelevant.

The most common feedback we received during TRM testing can be paraphrased as a question: 'what do we do after we detect harm incidents?' The answer to this question is considered in more detail in Chapter 7. It seemed to me that unless the TRM could be adapted and made more relevant to the individual clinicians and practices who were expected to implement it, it would have no realistic chance of ever becoming part of normal care. The best way to achieve relevance seemed to be through simplifying the method further, emphasizing the TRM's potential for learning and improvement, encouraging clinician reviewers to use their own initiative when applying the method and for them to accept 'ownership' of their findings. This approach was recently endorsed by the authors of a systematic review of the Global Trigger Tool (GTT). They recommended that the purpose of the TRM should be re-framed, as it was in this study, to understand and characterize PSIs, rather than just count them (242).

Conclusion

This chapter summarized the second group of patient safety knowledge, focusing on the specific methods and tools that are available for measuring and improving health care in general practice. The chapter began by describing five methods that can be used to measure and improve patient safety in general practice. Of these, the TRM is the main focus of this study and its practical application through three consecutive steps were described. Next, the potential value of the TRM was considered in relation to GP appraisal and revalidation, GP specialist training, as an approach to improve the safety and quality of care and as a metric for estimating harm rates.

In conclusion - Chapter 3 described the potential value and practical application of the TRM and provided a rationale for allocating time and resources to its implementation and study. However, successfully developing and testing a method - while important first steps - are not sufficient to ensure its successful implementation or eventual 'normalisation' into routine care. The next chapter will therefore review the different theoretical perspectives and approaches to implementation. In addition, a rationale for selecting normalisation process theory (NPT) as this study's theoretical framework to describe, understand and explain the implementation of the TRM will be provided. Chapter 4 will conclude with a description of the main study aims.

Chapter 4. Implementation Science and Normalisation Process Theory

Introduction

This chapter begins by introducing Implementation Science and summarizing some of the different models, frameworks and theories applied to health care, with specific reference to Normalisation Process Theory (NPT). Next, the main constructs, components and potential utility of NPT are described and illustrated with a small selection of practical examples of its use in other studies and the rationale for choosing NPT as the theoretical framework for this study is provided. The chapter concludes by formulating the main study aims.

4.1. Implementation Science

Conducting rigorous research, building a sound evidence base to inform highquality care and designing and developing useful interventions, methods or tools for complex health care settings are all challenging tasks. However, successfully disseminating this evidence into practice or implementing an intervention and ensuring it is used long enough to become embedded into routine care processes are arguably even greater challenges. In fact, there are often 'translational gaps' as much research and many interventions are never implemented at all, or only partially adopted or not sustained despite their potential utility (243, 244). This is true across health care, but a particular exemplar in respect of quality and safety initiatives (45). As a result, precious time and resources are often squandered on unsuccessful projects while the alluring promise of efficiency and quality gains remain unfulfilled.

Unsurprisingly, researchers and policy makers are very keen to avoid this waste and are interested in identifying the facilitators of improvement initiatives as well as overcoming the multiple barriers to the transfer of knowledge in order to build research-policy-practice links (245). Implementation science was borne out of this desire and is defined as the scientific study of methods to promote the systematic uptake of research findings and other evidence based practices into routine practice to improve the quality and effectiveness of health services and care (246). A more practical description of this 'relatively young science' (247) is that it aims to open the 'black box of change' (248).

Our current knowledge about implementation dates back more than half a century and has its roots in many disciplines, including sociology, behavioural economics and psychology. Over the years, a diverse range of implementation models, frameworks and theories have been developed and proposed, offering researchers a rich selection to choose from. However, much of the early research in health care was empirically driven without consideration of the theoretical underpinnings of implementation. This has been likened to 'an expensive version of trial-and-error' and made it difficult to understand or explain how and why interventions succeed or fail, or to identify the determining factors of successful implementation efforts (249).

There are at least two main reasons why explicit theoretical underpinnings are desirable for designing and implementing interventions. First, theory provides a generalizable framework enabling the comparison of effectiveness of interventions in different contexts and settings and opportunities for incremental accumulation of knowledge. Second, there are almost always multiple factors at different levels that determine health care outcomes. Applying theoretical frameworks may help to reduce the risk of important determining factors being overlooked (248).

Seeking to address this 'theoretical vacuum' (250) in the intervention designs of empirical studies, the MRC issued guidelines which strongly recommend the explicit and active application of theory in order to proactively enhance the transfer of research findings into clinical practice (251). The implications of the MRC recommendations are that researchers should aim to: establish the nature of associations between an intervention and observed outcomes; recognise the potential interactions between an intervention and the setting in which it is introduced; and consider the mechanisms through which the intervention and interactions improve care or, conversely, fail to improve care. In other words, determining whether and to what extent new methods, tools, guidelines or interventions are successful in particular health care settings requires that the factors hindering and facilitating their implementation be identified and understood. These necessary processes are greatly facilitated by judiciously selecting and applying appropriate models, frameworks and theories from which interventions' main determining factors can be derived and/or described (248).

Before considering the potential options, it is necessary to first define four common terms in this chapter: theory; model; framework; and implementation. A common definition of 'theory' is a system (or set) of analytical ideas, principles or statements held as an *explanation* or account of a group of facts or phenomena (252, 253). The terms 'models' and 'frameworks' are often and inappropriately used interchangeably with 'theories'. Models are deliberate simplifications of phenomena and are descriptive, but not explanatory. Frameworks consists of descriptive categories that provide outlines, structures or overviews of the concepts, constructs and variables that presumably give rise to specific phenomena but without explanations (254). 'Implementation' is defined and understood for the purposes of this study as 'the process of putting to use or integrating new practices within a setting' (254). It is part of a diffusion-dissemination-implementation continuum, but should be differentiated from these related terms. Diffusion refers to the passive, untargeted and unplanned spread of new practices, while dissemination implies the active spread of new information to target audiences using specific strategies.

4.2. Models, frameworks and theories of change in health care

A review in 2004 of the available theories relating to innovation in health care at that time concluded that the literature was complex, diverse and large and articulated the challenge of describing and understanding change as the product of multiple, unpredictable interactions between interventions, specific contexts and settings (255). The different theories, models and frameworks of change in health care can be classified into five different groups, depending on whether their main focus is on: (i) stages of change; (ii) individuals; (iii) social contexts; (iv) organisations; or (v) political and economic contexts (248).

From the perspective of the 'stages-of-change' models and theories, change happens as a result of individuals and teams taking 'steps' to progress through consecutive stages. Each stage is characterized by different determining factors and therefore requires unique strategies for change. In Roger's innovationdiffusion theory there are: innovators; early adopters; early majority; late majority; and laggards (256) while the stage-of-readiness-to-change model describes levels of motivation by distinguishing between: precontemplation; contemplation; preparation; action; maintenance; and completion (257).

The different theoretical approaches focussing on individuals as the main agents of change can be further subdivided into cognitive, educational and motivational theories.

- Cognitive theories: Rational decision-making theories assume the behaviour of clinicians is the result of considering evidence and balancing the relative advantages and disadvantages of different choices in an objective and impartial manner. In reality, the decisions of many clinicians may not be rational but are instead based on contextual information, previous experience and cognitive structures which combine to create so called 'illness scripts' (258).
- Educational theories: Adult learning theories state people are more motivated to change problems they identify themselves, compared with those that are presented to them. However, clinicians have different learning styles, e.g. activist; reflective; theoretical; and pragmatic learning styles (259). Consequently, not all clinicians will have the inclination to undertake self-directed learning. Even when clinicians do undertake self-directed learning, self-assessment is notoriously challenging (260).
- Motivational theories: The theory of planned behaviour states behaviour is influenced by intentions, and intentions are in turn influenced by perceptions of social norms (e.g. peers), self-efficacy (e.g. perceived control in relation to the behaviour) and the individuals' attitudes (261). Attitudes are determined by the expected outcomes of the behaviour and the clinicians' appraisal of these outcomes.

From a social context perspective, there are models and theories of communication (e.g. the persuasion-communication model; elaboration likelihood model; and heuristic systematic model), professional development and leadership, social network and influence theories and social learning theories. The different types of theories focusing on organisations include: theories of organisational culture and integrated care, organisational learning culture, complexity theory and the theory of quality management. Finally, examples of theories with an economic /political focus include reimbursement theories and the theory of contracting.

An alternative taxonomy with five categories was recently proposed based on the application of the different types of models, frameworks and theories in implementation science. The categories are: (i) process models; (ii) determinant frameworks; (iii) classic theories; (iv) implementation theories; and (v) evaluation frameworks (254). However, the author acknowledges that there is considerable overlap between these categories.

Taxonomies of models, frameworks and theories are interesting and satisfying from a scientific perspective, as is recognising and applying specific 'labels' to individual implementation approaches. However, there is a more important and serious rationale for this explicit differentiation, which is the acknowledgement that models, frameworks and theories all have different assumptions, aims and characteristics which strongly influence their potential relevance and usefulness for specific research designs.

A third and admittedly oversimplified taxonomy of the different theories, models and frameworks is therefore to categorize them into one of two main groups depending on whether they consider implementation as the product of 'institutionalization' or 'individual action' (5). 'Institutionalization' theories are those with the theoretical perspective whereby implementation is understood as the product of organisational activity. From this perspective, the 'actors' and 'contexts' are considered to interact in a predictable and rational manner during implementation processes. In contrast to 'institutionalization', theories with an 'individualized' perspective describe implementation and implementation outcomes from the perspective of the 'actors'. A central assumption of these theories is that individuals have 'free will' and actively choose to implement an intervention (or not). Current evidence supports this perspective - at least to a degree - with a general consensus that 'individual' factors are indeed important determinants of successful implementation, but that they only account for an estimated quarter of the observed outcomes (262, 263).

Several decades of research and the anecdotal experiences of many clinicians strongly suggest that, while these two perspectives of implementation are helpful, neither fully describe or 'capture' the complexities of developing, successfully implementing and evaluating complex health care interventions. Organisational innovations and new health care interventions are often imposed and individuals and teams have to work creatively to flexibly configure their existing practices to accommodate the changes according to their own specific requirements and local contexts. If there are workability issues with an intervention that they cannot resolve it leads to problems with integration. The majority of models, frameworks and theories also only describe and explain implementation processes retrospectively, e.g. they either lack predictive power or it may be that they have not been used in this way.

Given the relative limitations of existing theoretical perspectives about implementation, the need for a new and different theory of implementation became apparent that can mediate between macro (e.g. diffusion, institutionalization and organisational level factors) and micro (cognitive and individual action) levels of analysis (46). In response to this challenge Normalisation Process Theory (NPT) was developed. While there are many similarities between NPT and other, existing theories, NPT expands our understanding of implementation by offering a third potential perspective: successful implementation is the product of the 'work' health care staff have to do individually and collectively to implement research, a new method or a complex health care intervention. This, arguably, makes NPT a useful theoretical approach with which to explore the implementation of patient safety initiatives, such as the TRM. Before considering this, NPT is first described in more detail below.

4.3. Normalisation Process Theory (NPT)

The Normalisation Process Model (NPM) was developed and validated while conducting telemedicine and chronic disease management research in UK primary care settings in order to better understand the work required to implement such interventions in primary care across multiple studies conducted between 1995 and 2005 (44). This model consisted of four components that allowed barriers and facilitators in relation to the work of implementation of complex healthcare interventions to be identified, described and evaluated. However, it does not explicitly consider what the intervention is trying to achieve; who needs to be involved; and how the impact of the intervention is monitored. Innovation and change are recognised as often arising from external sources or being imported into local contexts but, because the departing point of the model is 'normalisation', the focus is on the '*creativity imbued in everyday professional work*' (46).

Recognising that implementation work is greater than the actual activities and resources required, NPM was subsequently further developed into the Normalisation Process Theory (NPT) (5, 243, 264). The most significant changes to the model were grouping the four original components together as a single construct - 'collective action' - and then adding three additional constructs focused on understanding, engagement and monitoring (264). NPT is a formal and verifiable theory with the purpose of empirical application rather than abstract critique. It is defined as an '*explanatory framework for investigation of the routine embedding of material practices in social contexts*' (5). Material practices are regarded as all of the things people do when they implement complex health care interventions (245).

NPT is a middle-range theory of social action making it amenable to the development of testable questions and propositions. Theories are considered 'high', 'middle' or 'low' depending on their place in an abstraction level continuum. The 'high' level theories have a universal or almost unlimited scope of application, while 'middle' level theories explain limited sets of phenomena (253). There is currently no 'high' level theory of implementation. It is possible in some disciplines to build a higher level theory from lower abstraction level

theories (analogous to building a wall from bricks). While a general theory of implementation has recently been proposed (265) the general consensus is that it is very unlikely there will ever be a grand theory of implementation because implementation is too multifaceted and complex to allow universal explanations (254).

NPT is a theory about the 'work' people do collectively and as individuals to implement and sustain an intervention - in the instance of this study, the intervention is the TRM and the main agents of change are primary care clinicians. From this perspective, 'work' is defined as '*purposive social action that involves the investment of personal and group resources to achieve goals*' (5). In other words, NPT is more concerned with understanding what people do than in their attitudes or beliefs.

The term 'normalisation' is defined as the embedding of a technique, technology or organisational change as a routine and taken-for-granted element of clinical practice (46). It includes all of the stages from design, development and testing of an intervention, through to its implementation, embedding and finally integration (5). Normalisation should be differentiated from adoption (e.g. an intervention is accepted and is used from time to time) and rejection (e.g. an intervention is spurned). Just because some innovations and interventions become normalised do not necessarily imply that they were effective in achieving their intended outcomes, nor that they are of high quality or that they are permanent, e.g. they may become de-normalised with time (44). The converse is also true - an intervention may be useful and meet the organisational criteria for success and yet not become normalised (266). This is why 'proof of concept' studies add little value to the evidence base in implementation science and are unlikely to do so (267, 268). However, despite this conceptual problem much research still focuses on 'can it work/does it work?' guestions rather than asking 'how' interventions should be implemented.

4.3.1. NPT constructs and components

NPT focuses attention on the different types of implementation work as described and categorized according to four main interactive constructs termed:

coherence; cognitive participation; collective action; and reflexive monitoring. NPT postulates that for evidence and innovation to become routine practice ('normalised'), work has to be done to understand and organise the method (coherence), staff have to be enrolled into using it (cognitive participation), the method has to be enacted (collective action) and work has to be done to organise, collect and interpret data about the method's effects (reflexive monitoring) (5).

The four main constructs are characterised by specific types of 'investments' required from implementers, without which successful normalisation becomes highly unlikely. Each construct is further divided into four sub-constructs or 'components', which allows the specific nature of the work to be described in more detail. The components have been referred to as 'generative mechanisms' because aggregating their different and specific 'work'/tasks produces the outcomes from implementing an intervention (5).

Although the constructs and components describe different types of 'work', they are highly synergistic, fluid, dynamic and often exist concurrently (5). In practical terms, this means that constructs and components constantly 'interact' with the potential to influence and change each other. The relative importance of each of the constructs and components fluctuate over the implementation period but also between particular empirical contexts. A simple analogy of NPT would be that 'components' are atoms and 'constructs' are molecules. Just as atoms and molecules may potentially interact, influence each other to varying degrees and change over time, the constructs and components affect each other and are also affected by external factors. An example would be the work of 'enrolment' of staff and 'initiation' of a project which are clearly related in a practical and temporal sense. These tasks are particularly important during the early phases of implementing a new initiative.

The NPT constructs and all of their components are described next in more detail (5, 243, 264). The NPT coding framework that was used in this study are summarised in Table 4.1.

Table 4.1. The NPT Framework

NPT constructs and components	Description
Coherence	The work participants have to do to understand the
	TRM
Differentiation	The work participants do to understand the
	differences and similarities between the TRM and
	their existing methods, tools and practices.
Communal Specification	The work required by all those involved to understand
	the purpose, aims and potential benefits of the TRM.
Individual Specification	The work of considering and quantifying the expected
	'effort', time and resources that will be required by
	the individuals involved to successfully implement the
	TRM. This work helps to determine whether the TRM is
	perceived as feasible.
 Internalization 	The work participants do to understand and interpret
	the TRM in relation to their own principles and beliefs
	and also the prevailing culture in their team or
	organisation. This work helps to determine whether
	the TRM is perceived as acceptable.
Cognitive Participation	The relational work required to build and sustain a
	community of practice around TRM.
Initiation	The work of preparing for the implementation of the
	TRM. This includes identifying and involving key
	participants willing to 'drive forward' the
	identification of the TRM.
Enrolment	The work of recruiting participants who will
	implement the intervention and keeping them
	engaged in the process.
Activation	The continuing support work that is necessary to
	sustain the use of the TRM
Legitimation	The work required by everyone involved to legitimize
	and justify their involvement with the TRM to
	themselves and each other
Collective Action	The operational work required to enact the TRM and
	requires participants to invest effort
Interactional workability	The work of applying the TRM in practice (e.g. what
	respondents actually did) and the time and effort this
	required (as opposed to what they perceived it would
	be). What effect (if any) did implementing the TRM

	have on their routine work?
Relational integration	The work of building and maintaining confidence,
	trust and accountability in the TRM and in each
	other's ability to successfully implement it. To be
	effective, this work requires formal and informal
	sharing of trigger review findings
• Skill-set workability	The work of dividing tasks and allocating resources so
	that the TRM can successfully be implemented in
	practice
Contextual integration	The work of integrating the TRM into existing
	structures, contexts and policies; the level of
	organisational support and resources at local, regional
	and national level.
Reflexive Monitoring	The work of assessing and appraising the individual
	and communal worth of the TRM
Systemisation	The work of collecting and organizing adequate and
	reliable data about the TRM to enable evaluation and
	to identify the benefits or problems of the TRM
Individual appraisal	The work participants do to evaluate the TRM's worth
	for them
Communal appraisal	The work of participants to collectively evaluate the
	TRM's worth, including to others
Reconfiguration	TRM's worth, including to others The work participants do to modify the TRM,

4.3.1.1. Construct: Coherence

Coherence (CO) is the *work* individuals and teams have to do in order for them to make sense *of* an intervention - in this case, the TRM. This includes the work required to understand an intervention as something new and worthwhile. In other words, they need to create 'a cognitive and behavioural ensemble' of the TRM (5). In practical terms this means working to package the TRM so it becomes a unique and recognizable entity that can 'stick' within specific contexts. Participants' understanding typically increases and evolves over time and requires them to invest meaning in the TRM. Coherence has four components: differentiation; individual specification; communal specification; and internalization.

Differentiation

Differentiation is the work participants do to understand the differences and similarities between the proposed method, tool or intervention and their existing methods, tools and practices. In the case of the TRM participants may have to do work to compare and contrast the TRM with formal and informal review processes of medical records they currently undertake and with existing QI methods such as SEA and clinical audit they already routinely applied. They also need to understand how the 'steps' of the TRM relate to each other and how a first and subsequent reviews may be different.

Communal specification

Communal specification is the required work to understand the *purpose* of the intervention (the TRM). What is the TRM's likely value for them collectively, for the practice and for their patients? In other words, do they think the TRM is useful? Participants' understanding of the aims and benefits may be different from the intervention's intended aims and benefits. If this difference is small, there is a 'high degree' of 'communal specification', which is desirable as it helps to facilitate normalisation.

Individual specification

Individual specification (IS) is the work of considering and quantifying the expected 'effort', time and resources that will be required by the individuals involved to successfully implement the intervention, and how this could best be done. In other words, how feasible would it be to implement the TRM? While a sound understanding of the task requirements are important, it does not necessarily mean that the resources or appropriately skilled, trained and experienced staff are available or willing to participate. Allocating adequate resources and effectively recruiting (the work of 'enrolment') and engaging clinicians in the project require the additional work of 'skill set workability' (discussed later).

Internalisation

Internalisation describes the work participants do to understand and interpret the TRM in relation to their own principles and beliefs as well as the prevailing culture in their team or organisation. In other words, is the TRM acceptable to them and do they see the benefits of using it? Internalisation also includes the work they have to do to interpret the findings from the trigger reviews in relation to their own contexts.

4.3.1.2. Construct: Cognitive participation

The cognitive participation (CP) construct describes the relational work that is required to *build and sustain a community of practice* around an intervention and requires participants to invest commitment. This involves identifying *who* should be involved with the TRM, *recruiting* and organizing them and keeping them *engaged* throughout the implementation process. CP's four components are: initiation, enrolment, activation and legitimation.

Initiation

Initiation (IN) is the initial work of successfully 'bringing forth' an intervention and requires key participants who are capable of 'driving forward' an intervention. Examples of initiation work include promoting and raising awareness of the TRM and planning and delivering educational events.

Enrolment

Enrolment (EN) is the work of recruiting participants who will implement the intervention and keeping them engaged in the process. In other words, EN describes *who was involved* with the TRM and when, in what capacity and to what degree they contributed. In particular, it addresses the degree of 'buy-in' that individuals show towards the intervention. However, the work of determining *who* should be involved in implementing the TRM is described by the 'skill set workability' component.

Activation

The activation (AC) component describes the *continuing support* work that is necessary to sustain the use of an intervention. From a TRM perspective, 'activation' work includes the different improvement actions reviewers and teams considered or undertook after trigger reviews. For this work to be successful, participants needed to remain actively involved in the process. This, in turn, depends on whether they feel 'empowered' to enact change and they consider ongoing participation as 'right' for them (the work of Legitimation - see below).

Legitimation

Legitimation (LE) is the work clinicians and general practice staff have to do, but also the work of policy makers and professional organisations, in order to legitimise and justify their involvement with the TRM to themselves and each other. In other words, it is the work that is necessary so participants believe implementing the TRM is the 'right' thing to do.

4.3.1.3. Construct: Collective action

Collective action (CA) is the operational work required to *enact* the TRM and requires participants to invest effort. The CA construct describes the organisational, external, immediate and internal factors that may hinder or facilitate the implementation process. The four components of CA are: interactional workability, relational integration, skill-set workability and contextual integration.

Interactional workability

Interactional workability (IW) is the work of applying the intervention and, in particular, whether the intervention makes the 'day job' easier or harder for individuals. In the case of the TRM it therefore involves screening samples of electronic patient records for PSIs. It is important to consider whether informal work-place rules may be affecting this work, and if the intervention helps or hinders clinicians' 'normal' work. The IW component includes the actual time and effort the work required, whereas participants' perceptions of the time and effort they invested is described by the 'individual specification' component (part of the coherence construct).

Relational integration

Relational integration (RI) is the work of building and maintaining confidence, trust and accountability in an intervention and in each other's ability to successfully implement it. In other words, RI is the work of incorporating the TRM within existing relationships. This component should be differentiated from the work of 'legitimation', which is the work participants and external parties do in order to agree that they should be enacting the intervention. Are TRM findings being shared and, if yes, how and with who?

Skill-set workability

The Skill-set workability (SW) component describes the work of dividing and allocating resources so that an intervention can successfully be implemented.

This requires division of labour, e.g. 'who does what?' In the case of the TRM, records may only be screened by trained clinicians. The SW component also includes the work of deciding who has the power to make the decisions about resource allocation and work delegation. To be effective, skill-set workability requires consideration of the knowledge, skills, attitudes and capacity of staff.

Contextual integration

Contextual integration (CI) is the work of integrating the TRM into existing structures, contexts and policies: the Quality and Outcomes Framework (QOF); Local Enhanced Service (LES); appraisal and revalidation; GP Specialist Training (GPST) curriculum; regional and national Quality improvement initiatives such as SIPC and SPSP-PC. In addition, there is work to incorporate the TRM within the prevailing safety culture (work that is shared with the 'initiation' component) and professional roles (work that is shared with the relational integration component). The work of CI depends on the availability of adequate and appropriate resources and therefore also includes: provision of new resources or re-allocating existing resources; senior leadership support; restructuring current policies and infrastructure to include, promote and support the TRM.

4.3.1.4. Construct: Reflexive monitoring

Reflexive monitoring (RM) is the work of assessing and appraising the individual and communal worth of the TRM. It has been defined as: '*the work of understanding and evaluating a complex intervention in practice*' (269) and requires participants to invest comprehension. This work requires adequate time, reliable metrics and sharing of results. The four components of RM are: systematisation, individual appraisal, communal appraisal and reconfiguration.

Systematisation

Systematization (SY) is the work of collecting and organizing adequate and reliable data about the TRM to enable evaluation and to identify the benefits or problems of the TRM. This work is normally undertaken through a combination of formal and informal methods. This component is therefore also concerned with the methodological formality with which implementers derive value-judgements about the TRM's usefulness.

Individual appraisal

The individual appraisal (IA) component describes the work participants do to evaluate the TRM's worth for them, e.g. the clinician reviewer, the reviewer's practice team and their patients. IA is typically informed by data derived from informal methods and relies on 'experiential and unsystematic practices of judging the value and outcomes of practice'.

Communal appraisal

This component describes the work of participants to collectively evaluate the TRM's worth, including to others, *e.g.* clinicians, practices and specific professional and patient groups other than their own. Communal appraisal is typically informed by data derived from formal methods (at least in comparison with IA).

Reconfiguration

Reconfiguration (RE) is the work participants do to modify the TRM, themselves (e.g. their attitudes, skills, knowledge, tasks) or their contexts (practice procedures, policies and infrastructure).

4.3.2. The rationale for selecting NPT for this study

NPT has at least three important 'strengths' which help justify its selection to inform the study design and analysis in this work. First, it was developed in the UK primary care setting with methodological rigour, its design process was transparent and it was subsequently validated through practical application in 'real life' settings. Second, because NPT is about 'workability in practice' (269) it can be applied iteratively to study temporal changes in perceptions, actions and outcomes. Third, it is transferable to different healthcare settings and can be applied to a wide range of innovations, methods, tools and interventions. Even though NPT is relatively young in research terms - it was developed only a decade ago - it has already been successfully used in multiple studies and settings.

For example, in the UK, NPT has been used to study the implementation of nutrition guidelines for elderly patients in residential care homes (270), to evaluate the integration of Telecare management of chronic diseases in the community (268, 271), as a framework for generating and analysing data relating to the management of early chronic kidney disease in primary care (272) and to assess the treatment burden among patients with chronic heart failure in general practice (273). In Australia, the barriers and enablers to initiating insulin in primary care were identified (274) and a conceptual NPT framework was used to design a new model of depression care in general practice and study its effective implementation (275). A similar approach was adopted in the Netherlands to implement a stepped-care approach to the management of depression in primary care (276). Finally, NPM was used in South Africa to analyse provider experiences of a new tuberculosis treatment programme (277) and a provider-initiated HIV testing and counselling intervention (278).

This small selection of studies help to demonstrate some of NPT's multiple potential applications. Its main use so far has been as a validated, conceptual framework for *describing*, *understanding* and *evaluating* complex health care interventions. However, it can and has been used to help design, develop and test complex interventions and to optimize trial parameters (269, 279). In this way, NPT allows implementers to focus through a socio-technical lens on areas likely to be problematic and direct their efforts accordingly (243). Another important application is to help 'bridge the translational gap' between research and innovation and their practical implementation by identifying possible barriers and facilitators to these processes (245).

While post-hoc explanations and interpretations of events are important, an 'ideal' theory overcomes the significant methodological and theoretical challenge inherent in predicting future outcomes with at least a modicum of reliability and success. This holds true in the field of implementation science because, no matter how complex and emergent implementation processes may appear, they are subject to normative and structural constraints that are usually not arbitrary. Consequently, the 'trajectory of a practice can be anticipated within certain limits' which in turn enables 'prospectively assessing the potential of practices to normalise' an intervention (5). In other words, NPT can be applied to determine the more likely outcomes of an intervention, e.g. whether it will be successfully normalised or not. At the very least, NPT can be applied to judge the implementation potential of an intervention. The practical implication is that NPT may therefore have utility as a potential 'trial killer' (243, 269). Applying the NPT framework to a study proposal would help to determine the likelihood of that intervention being successfully normalised, and hence whether it is worthwhile to proceed with the study or not (245).

4.4. Study aims

This thesis describes the study of the implementation of the TRM in the general practice setting in Scotland. There are four main study aims.

- To describe the perceptions and understanding of general practice clinicians and staff about the concept of 'patient safety' prior to the TRM being implemented. Perceptions powerfully influence and create the unique contexts within which health care interventions are implemented. The current consensus is that the successful implementation of interventions is strongly associated with the degree to which the planners consider and take into account the contextual factors and the setting into which they transfer their interventions (243, 250, 251, 280). Becoming aware of and understanding preexisting perceptions and contexts are therefore important to understand the outcomes from implementing the TRM, and are described in Chapter 6. Considering the perceptions of the participants in the design and planning stages of the study helped to increase the likelihood of the TRM being understood as acceptable, feasible and potentially useful.
- To evaluate the usefulness of implementing the TRM by describing its main outcomes. Chapter 7 considers the questions related to this aim: how did the general practice staff implement the TRM, what did they find and what did they do next? More specifically, how many trigger reviews were undertaken,

what number and types of patient safety incidents were detected, which improvement actions were considered or undertaken and how much effort did this require? In addition, the degree to which the study participants perceived the TRM as feasible, acceptable and useful is also described. The qualitative and quantitative results that are presented in Chapter 7 provide evidence of the different aspects of the usefulness of the TRM. The study findings therefore help to answer an important question which is raised in Chapter 10: should the TRM be normalised?

- To explain how the TRM works. Chapter 8 explains how the TRM produced the results that were observed - whether positive or negative. It also describes the related issue of the learning that occurred as a result of the trigger reviews.
- To identify and describe the main factors that facilitated or hindered the implementation of the TRM in general practice. Understanding what the main factors are that determine whether interventions are successfully implemented and normalised extends the potential impact of the study from a single intervention in a single setting. However, for this potential to be realized, a structured assessment of potential factors that may facilitate or hinder these processes are required, which is provided in Chapter 9. The study findings also help to inform recommendations about future improvement interventions and application of the TRM; whether the TRM should be normalised or not and how the process may best be facilitated; and which areas of research should be prioritised (Chapter 10).

Before the main findings are presented, the different methods that were used to achieve the four main aims are first described and discussed in the next chapter.

Chapter 5. Methods

This chapter begins with an overview of the study design and a description of the sampling and recruitment strategies. The two main data sources are described: semi-structured interviews with practice managers (PM), practice nurses (PN) and GPs; and (iii) Trigger Review Summary Sheets (SS). In particular, the data collection strategies and the methods that were used to analyse the qualitative and quantitative data are described. The chapter concludes by considering data protection issues and ethical requirements.

5.1. Study design

The study design was a mixed-methods approach comprising semi-structured interviews with a range of general practice clinicians and staff at two different points in time and a cross-sectional review of electronic patient records by trained clinicians who applied the Trigger Review Method (TRM).

Some of the early studies about patient safety in general practice have been likened to '*picking low hanging fruit*' because they were opportunistic, the results could not be generalised from the original to other settings and they often provided only a narrow perspective of certain types of harm and error (177). In reality, patient safety incidents (PSIs) have complex and diverse contributing causes that can be understood from different perspectives (281). Four specific perspectives of the contributing causes of PSIs will be described in Chapter 6, page 163. For now, the important implication is that different improvement approaches and interventions only provide information about specific aspects of patient safety. Similarly, different research methods can only provide patient safety-related data from specific perspectives.

One of the key benefits of a mixed-methods study design is therefore that the different, complementary perspectives about PSIs that they provide can, in combination, enhance our overall understanding of patient safety. A second benefit of this approach is that, by combining different methods, their respective limitations are compensated, at least to some degree, while the veracity of the main findings are enhanced (282). In this study quantitative
methods were used to collect data about the incidence and nature of PSIs and the range of improvement actions considered or undertaken during and after the trigger reviews. Simultaneously, qualitative methods were used to explore and describe the perceptions of general practice staff and clinicians about the TRM, the contexts within which it was being applied and the related implementation processes. Combining the results from the different sources of data provided more complete and reliable answers to the questions posed in Chapter 10 about the potential usefulness of the study and how generalizable its findings were than would have been the case if only a single source of data had been available.

5.1.1. Setting and sample

The study was undertaken in the West of Scotland region in two NHS Health Boards (henceforth BOARDS): Greater Glasgow and Clyde (GGC) and Ayrshire and Arran (A&A). The BOARDS contain 262 and 57 general practices respectively and cover a wide range of settings from severe socioeconomic deprivation to areas of affluence, with practices located in urban, suburban and rural areas.

A range of different types of general practice staff was included in the study to allow exploration and comparison of clinical and non-clinical perceptions of patient safety in general and the trigger review method in particular. The decision to include multiple staff groups was also motivated by the multidisciplinary nature of general practice and its strong ethos of working in teams.

The study sample was obtained from two main sources. The first source was a convenience sample of 12 GP practices from GGC (n=10) and A&A (n=2) BOARDS. In each surgery, at least one GP and the practice manager actively participated, while participation by a practice nurse was strongly encouraged. The second source was a convenience sample (n=25) of GP specialist trainees (GPSTs) from the West of Scotland.

The pragmatic choice of a convenience (or non-probabilistic) sampling strategy was motivated by time and resource constraints and ready access to an established network of general practices with an interest in quality improvement and research. The sample size was informed by the need to collect sufficient qualitative and trigger review data, to allow for a proportion of practice teams potentially withdrawing during the study period and the budget that had been allocated to this specific research project.

The GP teams and GPSTs were not involved in any other research projects during the study period. However, some of the practices were participating in two Local Enhanced Service (LES) Quality Improvement projects. The projects were medication reconciliation; and improving the reliability of care delivered for patients prescribed disease-modifying anti-rheumatic drugs (DMARDs) by applying a Care Bundle approach.

5.1.1.1. The 'QOF study' sample

The inclusions of the TRM in the Scottish QOF and as an integral component of the SPSP-PC were described in Chapter 2. These events occurred shortly after the data collection phase of this study had been concluded and before the analysis had been completed. A timeline of the events were provided in Figure 1.1, page 27. The implication was that virtually all general practices in Scotland implemented the TRM from April 2013 onwards, using essentially an identical approach as in this study. Consequently, thousands of patient records were reviewed in Scotland, and more than a thousand Trigger Review Summary Sheets (SS) were submitted to the BOARDS in the 2013-14 financial year.

CdW was commissioned to evaluate the implementation of the TRM as part of QOF by NHS Education for Scotland (NES) on behalf of the BOARDS. The final, approved evaluation reports were subsequently rewritten, submitted for peerreview and published (48). CdW also provided informal support and advice to a third BOARD, NHS Lothian. Direct access to the aggregated trigger review data from the three BOARDS (with their express permission) provided a unique opportunity for comparative analysis with the data collected for this study. In Chapter 7, when the quantitative results of this study are presented, the aggregated data from the BOARDS will also be presented for comparison and context. The results from the BOARDS will henceforth be referred to as the 'QOF study' to differentiate it from this study.

5.1.2. Recruitment

Potential participants for this study were identified from two different NHS Education for Scotland (NES) 'lists'. The first list consisted of the contact details of approximately 21 practices that had previously supported NES research projects, including the implementation of other quality improvement interventions, and who had provided invaluable feedback about their perceived utility.

The second list was the e-mail addresses of all of the active general practice educational supervisors in the West of Scotland. NES maintains this database to help support its core business, which is postgraduate medical education, including the delivery of GP Specialist Training in Scotland. The strategies that were used to recruit general practices and GPSTs for this study are described below.

5.1.2.1. Recruitment of practices

Practice managers (n=21) were sent written information about the proposed study, its rationale and an invitation to participate via e-mail in April 2012. The invitation letter is included as Appendix 2, page 293. There were three main requirements:

- The designated clinician reviewers and the practice manager would undergo TRM training;
- (ii) The general practitioner, practice nurse and practice manager would each participate in *two* individual, semi-structured interviews. The first interview would be scheduled before the training event and trigger reviews. The second interview would be conducted after all of the SS had been submitted;
- (iii) The practice would submit four SS, with at least two of the four prepared by a GP reviewer.

110

The first 12 general practices who responded affirmatively to the invitation were selected. Each practice nominated at least two clinicians, one of which had to be a GP, to receive TRM training. Each participating practice received an equal professional fee paid once they had agreed to participate in order to help recompense them for the time spent on the project. The amount was equivalent to that of one practice manager, three GP and three practice nurse sessions and was calculated from the sessional payment rates NES agreed at that time.

5.1.2.2. Recruitment of GPSTs

In April 2012, all GP Educational Supervisors in the West of Scotland Deanery were e-mailed (n=322) with details of the study and were asked to discuss this with their trainees. The invitation letter is included as Appendix 3, page 295. There were two main requirements of participants:

- (i) They would undergo TRM training;
- (ii) Each participant would submit one SS;

The number of participants was restricted to the first 25 GP trainees to indicate a willingness to participate via email response. The GPSTs did not receive a professional fee but could claim their travel costs incurred in relation to the TRM training event.

5.1.3. TRM training

The TRM training sessions had a number of key components, which are shown in Box 5.1 These components were developed and refined over several years prior to this study through practical experience and participant feedback during initial TRM pilot work and the SIPC programme. The components are flexible and can be tailored to the size of the group of learners, their specific learning needs and the available time.

The aim of a TRM learning session is to help prepare clinicians to successfully apply the TRM. The two main objectives are that, as a result of the training sessions, reviewers will be able to: (1) review their own clinical records in a rapid, structured and focused manner in order to detect PSIs and potential

- A *patient safety quiz* with feedback to introduce the evidence base for the epidemiology of error and harm in primary care.
- A group work exercise on matching a range of patient safety terms to their definitions in order to help increase participants' shared understanding of key terms such as 'patient safety incident'.
- A short PowerPoint *presentation* about the Trigger Review Method (TRM)
- Provision of a TRM *educational support package* consisting of: (i) a step-bystep implementation guide; (ii) simulated patient records with 'worked out' solutions; (iii) the Trigger Review SS template; and (iv) practical examples of how to rate PSI severity and preventability.
- Opportunity for individual participants to *practice* performing 'trigger reviews' of simulated patient records, followed by discussions in small groups and then feedback in an open forum. The objective is for reviewers to identify at least four of the PSIs contained in the examples within 20 minutes, and to make judgements on the severity and preventability of these incidents.
- *Clarification* of the study's expectations, including informing participants about the 'high risk' patient groups from which they were to select their sample of clinical records for trigger reviews and where to send their completed SS documentation.

safety-critical issues and; (2) learn from and act on the findings in order to improve the quality and safety of the care they deliver.

5.1.3.1. Training of practice teams

The nominated clinicians from each participating practice were trained to use the TRM by CdW at a time of their choice. Any number of additional team members was able to attend. Training lasted up to three hours and was delivered on the practice premises. All of the training components were delivered, but the duration of each component was adjusted according to the needs of each practice team.

5.1.3.2. Training of GPSTs

A dedicated three-hour training session was offered exclusively to participating GPSTs. They were offered two possible dates in May 2012 in a central Glasgow location to give some attendance flexibility to the volunteer participants. The two training sessions were delivered by the same facilitators (PB and JM).

5.2. Data collection

The data in the study were collected with two different methods, semistructured interviews and Trigger Review SS templates, and are described below.

5.2.1.Semi-structured interviews

Two different interview schedules were developed - one for the first round interviews scheduled before the TRM training session - and the other for the second round interviews undertaken after the practice teams had concluded their reviews and submitted their SS. The schedules were derived from patient safety literature, previous experience of research in this discipline and through discussion with PB and COD (283). In addition, the second interview schedule was based on the NPT framework. The interview schedules are included in Appendix 6 and 7, pages 299 and 300.

The first interview with each participant was conducted *before* the training session and aimed to explore in confidence participants' awareness, experiences and perceptions of the state of patient safety in primary care and how safety may be addressed. The second interview was conducted *after* all SS had been submitted. The second interview aimed to explore the potential acceptability, feasibility and usefulness of the trigger review method.

The interviews were conducted in the participating GP practice premises at a time convenient to the participants. Informed consent was obtained before commencing interviews. The practice and individual consent forms are included as Appendix 4 and 5, pages 297 and 298. Following written consent, interviews were recorded using a digital recorder and supporting field notes were made. The number of interviews, their duration and the types of individual participants are summarized in Table 5.1.

Only one participant was unable to commit to a time for the first round of interviews (GP12). A minority of participants were unable to participate in the second round of interviews. This was because two PMs had developed significant health problems, while PN07 had left the practice and her position had not been filled by the end of the study period.

5.2.2. Trigger reviews

Each practice was asked to conduct four trigger reviews in total during the period of time from June 2012 to July 2013, with at least two of these performed by a GP. GPSTs were given a four-week period after their training session to undertake the TRM and submit one completed SS each. Each Trigger Review was conducted on a random sample of medical records (n=25).

After the data collection phase of the GPST component of the study had concluded, the trigger review process was slightly adapted. Reviewers in the GP component of the study were advised to stop searching for PSIs once they had detected five. Instead, they were encouraged to use the remaining time to consider how to act on their findings. The reason for 'capping' reviews at five

חו	GP interviews			PN interviews			PM interviews			A []
. שו	1st	2 nd	Both	1st	2 nd	Both	1 st	2nd	Both	All
-										
1	45	*	45	58	*	58	NA	NA	-	103
2	66	52	118	57	45	102	46	19	65	285
3	56	57	113	58	27	85	54	22	76	274
4	59	54	113	49	30	79	58	x	58	250
5	54	62	116	49	28	77	45	x	45	238
6	53	49	102	55	30	85	60	35	95	282
7	57	50	107	38	Х	38	60	x	60	205
8	48	62	110	73	70	143	76	х	76	329
9	54	15	69	56	41	97	64	41	105	271
10	57	44	101	58	54	112	52	28	80	293
11	58	54	112	44	22	66	43	х	43	221
12	х	41	41	49	21	70	59	x	59	170
All	607	540	1147	644	368	1012	617	145	762	2921

Table 5.1. Summary of the number and duration (in minutes) of individual interviews

*GP01 and PN01 requested a group interview in lieu of a second, individual interview. Its duration was approximately 90 minutes and is not included in Table 5.1.

NA = PN01 was also PM01. She indicated her preference to be assigned as PN rather than PM for this study.

X = participant unavailable for an interview

PSIs was pragmatic as it is possible that GP teams would not be able to feasibly deal with all incidents detected during a review, given time and resource constraints.

Reviewers were asked to complete a trigger review SS after reviewing each sample of records (n=25). The SS template is described in more detail in the next section and is included as Appendix 8, page 301. Reviewers were also strongly encouraged to 'fix' any obvious problems that could be achieved quickly and without much effort while conducting the reviews, e.g. updating a patient's allergy coding status. Participants were advised that the total review process (collecting and reviewing a sample of 25 records and completing the SS) should take approximately two to four hours and that no more than 20 minutes should be taken for any single record.

The recommended number of reviews, the study period and recommendation of a maximum number of PSIs per review were informed by feasibility concerns and reflected how the TRM could practically be implemented as part of the QOF. It had been the intention all along to propose that the TRM be included in the QOF if the findings indicated that it was an acceptable, feasible and useful intervention. As it turned out, the TRM was included in the QOF before the study was concluded.

5.2.2.1. Selection of patient record samples for trigger reviews

The recommendation to clinicians was that they should select their random samples of patient records to review from a specific high-risk patient group patients who were older than 75 years *and* had a record of confirmed cardiovascular disease. This 'high-risk' patient group is particularly prone to error and harm (as described in Chapter 3) and therefore increases the likelihood of detecting PSIs. However, reviewers were informed that they had the final say in deciding which patient group to select. The decision to allow reviewers discretion in relation to their choice of patient population was pragmatic, to account for the professional requirements and attitudes of some reviewers and because, based on previous experience with the TRM, increasing the flexibility of the TRM had improved its acceptability. The reviewers were therefore also provided with a pre-selected list of alternative 'high-risk' patient populations group (Box 5.2).

5.2.2.2. The 'review period' for records

The 'review period' was defined as any three consecutive calendar months in each patient record before the review date, as this period of time arguably offered the most efficient compromise between yield of triggers and time spent by the healthcare professional examining the notes [1]. It was left to each practice to decide which three-month period clinicians would review.

5.2.2.3. The Trigger Review Summary Sheet (SS)

The trigger review SS is a structured, two page template guiding reviewers through the three steps of the TRM, which are: (1) planning and preparation; (2) reviewing a sample of patient records and (3) reflection and action. The SS were used to collect and summarise anonymised data. It contained a number of tickboxes as well as free-text data fields for clinicians, including those in the GPST group, to: record triggers and details of detected PSIs, including their perceived severity and preventability; describe the improvement actions and intended actions undertaken during and after the trigger reviews; record their own learning needs and points and those of their practice teams; and reflect on or provide feedback about the trigger review process.

In addition to the eight validated, pre-defined triggers on the SS template, reviewers could add optional triggers, depending on the characteristics of the specific 'high-risk' patient population they selected to review. For example, 'prescription of a NSAID' might have been a useful optional trigger to add to the list of validated triggers when screening the records of patients with cardiovascular disease. The SS continued to evolve and a number of minor edits that were made as a result of and subsequent to this study are described in Chapter 9, page 240.

Box 5.2. Potential high-risk patient groups from which samples of records could be selected for Trigger Reviews

- Patients on DMARD therapy;
- Patients with diagnosis of Left Ventricular Systolic Dysfunction;
- Patients on Warfarin therapy;
- Patients with a higher SPARRA score, e.g. >40;
- Recent admissions with COPD;
- Care home residents;
- Patients on chronic district nursing caseload;
- Patients aged >75 years on 6 or more medications

5.2.2.4. Severity and preventability of detected PSIs

The dual scoring system included on the SS to allow reviewers to judge the perceived severity and preventability of detected PSIs is shown in Table 5.2. The severity classification system was adapted from previously published work to suit the general practice context (284).

5.3. Data analysis

5.3.1. Quantitative data analysis

The quantitative data from the submitted SS were extracted and coded in an Excel spreadsheet and included: (i) the numbers of individual triggers and PSIs identified; (ii) the severity and preventability ratings of Patient Safety Incidents; (iii) actual or proposed improvement actions resulting from the trigger reviews; and (iv) the time required to conduct the reviews.

In general, data can either represent measurements on a continuous scale or information about categorical (discrete) characteristics. For example, age is 'continuous' and gender is 'categorical' data. Some variables may be considered as one or the other. For example, the ratings of PSI severity on a 4-point scale may be considered as either continuous or categorical data. In this study, all ratings were considered to be categorical data. The characteristics of the quantitative data collected during the study and the statistical methods to analyse them are summarized in Table 5.3.

5.3.1.1. Descriptive statistics

The clinician reviewers were classified into one of *three groups* according to their professional roles. The three groups were: 'GP', 'nursing' and 'GPST'. The nursing group included practice nurses and nurse practitioners.

Severity	Scale	Preventability
Any incident with the potential to cause harm	1	Not preventable and originated in secondary care
Mild harm: inconvenience, further follow-up or investigation to ensure no harm occurred.	2	Preventable and originated in secondary care OR not preventable and originated in primary care
Moderate harm: required intervention or duration for longer than a day	3	Potentially preventable and originated in primary care
Prolonged, substantial or permanent harm, including hospitalization	4	Preventable and originated in primary care

Table 5.2. Severity and preventability rating scales for detected PSIs (39)

Description	Type of data	Descriptive statistics	Statistical tests
Reviewer characteristics	Categorical, nominal	Counts, percentages	
Triggers	Continuous, ratio	Counts, percentages, means, SD, range	Comparison of means and proportions: two- sample t-tests, ANOVA, Chi-square
Number of PSIs	Continuous, ratio	Counts, percentages, means, SD, range	Comparison of means: two-sample t-tests, ANOVA
PSI Severity / Preventability ratings	Categorical, ordinal	Counts, percentages	Comparison of proportions: Chi-square
Classification of PSI types and commonly implicated medications	Categorical, ordinal	Counts, percentages	Comparison of proportions: Chi-square
Number and types of actions or intended actions	Continuous, ratio	Counts, percentages, means, SD, range	Comparison of means and proportions: two- sample t-tests, ANOVA, Chi-square
Time taken for reviews	Continuous, ratio	Counts, percentages, means, SD, range	Comparison of means: two-sample t-tests, ANOVA

Table 5.3. Summary of the quantitative data derived from the trigger review summary sheets

Microsoft Excel was used to perform simple descriptive statistical calculations, including: counts, percentages, means, standard deviations and ranges of the relevant study variables. All other statistical tests, e.g. comparison of means (section 5.3.1.2) and proportions (section 5.3.1.3), were performed with MedCalc Version 14.1.

5.3.1.2. Comparison of means

Two statistical methods were used to compare means: two-sample t-tests and ANOVA.

Two-sample T-tests: This test is used to compare the means of two study populations (groups). More specifically, t-tests are used to compare *one continuous variable* between *two* groups. To compare how *two* variables vary together, correlation and regression should be used. If one variable is compared for *more than two groups* ANOVA should be used.

The validity of T-tests is based on two assumptions. The first is that the populations being studied have a normal distribution. Although the variance within each group is unknown it is assumed to be the same. The second assumption is that the samples of data from the two groups are independent.

T-tests were used to compare the aggregated mean scores of this study with the aggregated mean scores of the QOF study for the following data:

- Time required to complete Trigger Reviews and SS;
- Numbers of detected triggers;
- Numbers of detected PSIs;
- Numbers of improvement actions and intended actions undertaken during and after Trigger Reviews.

In other words, t-tests were used to determine whether, for example, significantly more (or fewer) PSIs were detected in this study compared with the QOF study. In each instance when t-tests were used, the following were calculated: t-value, p-value, degrees of freedom (DF), 95% confidence interval (CI) and the standard error of difference ANOVA: This test was used to compare the mean scores of three or more groups. In addition, post-hoc t-tests were performed to further analyse observed intragroup variation. ANOVA tests were applied to the following data of the three reviewer groups (GP, GPST and PN):

- Time required to complete Trigger Reviews and SS;
- Numbers of detected triggers;
- Numbers of detected PSIs;
- Numbers of improvement actions and intended actions undertaken during and after Trigger Reviews.

In each instance when ANOVA was used, the following were calculated: Sum of squares, F-ratio, p-value and degrees of freedom (DF)

5.3.1.3. Comparison of proportions

Chi-square (x^2) *tests:* The Chi-square test is used to 'test' how likely it is that an observed distribution is due to chance. It is also called a 'goodness of fit' statistic because it measures how well an observed distribution of data fits with the expected distribution if study variables were independent. Chi-square tests may only be applied to categorical data. The validity of the test is based on the assumptions that data were randomly and independently gathered from the populations (groups) and that there are at least five study participants in each group that is being compared. These minimum requirements were exceeded in this study.

Chi-square tests were applied to the following data to compare the difference in proportions for:

- Reviewer groups (GP, nursing and GPST) who selected their samples of patient records from the recommended 'high-risk' patient group.
- The severity and preventability ratings applied by the reviewer groups in the study and the BOARDS;
- The different PSI types in the preliminary classification;
- The most commonly implicated medications in detected PSIs;
- Actions and intended improvement actions undertaken during and after

trigger reviews.

In each instance when chi-square tests were used, the following were calculated: the difference in proportion (%), the 95% confidence interval (CI) of the difference, the chi-square value, p-value and degrees of freedom (DF).

5.3.2. Qualitative data analysis

Analysis of the qualitative data commenced shortly after the first interviews had been conducted in 2012 and continued until the second half of 2016. This section describes the different qualitative data analysis processes and strategies that were used in order to achieve the aims of this study. Table 5.4 provides a summary of the main points.

5.3.2.1. Assignment of unique identifiers to practices and participants

All interviews were carefully transcribed verbatim in order to preserve colloquialisms, repetition and other important aspects of communication, e.g. 'pause', 'giggle', 'laugh', 'cough' or 'a knock on the door'. The transcripts were anonymised by removing all names from them. Each of the twelve participating practices was assigned a unique, double digit identifier, commencing with '01' through to '12'. The same unique identifier was applied to every participant within a given practice. Participants from the same practice were differentiated by adding a further, unique identifier as a prefix, derived from their professional role: general practitioner - GP; practice nurse - PN; and practice manager - PM.

The results chapters (chapters 6 through 9) include a large selection of verbatim quotes from the interview data that are presented in a similar way to the following example:

'I like nothing more than going back over notes, and reviewing and researching what we have or haven't done' (GP06)

The unique identifier at the end of the quote in the example, e.g. 'GP06', indicates that this exact phrase was spoken by a general practitioner (GP) from

Processes	Analysis strategies						
	Thematic analysis	NPT framework application					
Sources of data	• All of the data from the	• The majority of the data					
	first round of semi-	from the second round of					
	structured interviews with	semi-structured interviews					
	PNs, GPs and PMs						
	 Free-text entries by 						
₽	clinician reviewers on the						
	Trigger Review SS						
	• Some of the data from the						
	second round of semi-						
	structured interviews						
First cycle coding	The data were coded using:	The data were coded					
	 Grammatical methods 	according to the main NPT					
	(attribute and simultaneous	constructs					
	coding); and						
₽	 Elemental methods 						
	(descriptive and structural						
	coding)						
Second cycle coding	The initial codes and data	The data were sub-coded to					
	were analysed further	the NPT components					
	through pattern and focused						
Ţ	coding						
Derive categories,	The findings relate to:	The findings relate to:					
themes and main study	• Perceptions of patient	• Factors that facilitated or					
findings	safety (Chapter 6)	hindered implementation of					
	 Classification of PSIs 	the TRM (Chapter 9)					
	(Chapter 7)						
	• How the TRM works						
	(Chapter 8)						

Table 5.4. Summary of the study's qualitative data analysis strategies and processes (285)

practice six of the twelve participating practices. In order to reflect the reality of modern general practice, which is that it is delivered by teams, the qualitative data included 'matched' interviews within each practice, e.g. a practice nurse, GP and practice manager, who were interviewed (where available) using the same schedules. In this example, data from the practice nurse and manager from the same practice as GP06 would be indicated by the unique identifiers 'PN06' and 'PM06' respectively.

5.3.2.2. Thematic analysis

The first round interview transcripts were read and re-read to provide an overview of the contents and as a primer for the 'first cycle' of coding (285). The two main tasks during the first cycle were to: (i) develop and apply codes to the data and; (ii) to write reflective memos about the codes and initial impressions of the data. An exemplar reflective memo is provided in its original format in Box 5.3 for interest. The memo was written while analysing the first interview with PM08.

For the purposes of this study, a 'code' was understood to mean 'a word or short phrase that symbolically assigns a summative, salient, essence-capturing and/or evocative attribute for a portion of language-based data' (285). The majority of the first cycle coding was performed using pen and paper, as this was found to be easier and also more conducive to reflection than using Computer-assisted qualitative data analysis software (CAQDAS).

After initial experimentation with various coding methods, I found the grammatical (e.g. attribute and simultaneous coding) and elemental (e.g. structural and descriptive coding) methods to be the most useful and intuitive for application to the first round interview data (285). A brief description of these methods is provided in Box 5.4.

Once first cycle coding had been completed for the majority of transcripts, the second cycle of coding were commenced. During 'second cycle' coding, the memos and codes from the first cycle were searched for patterns and categories and were combined, re-combined and reconfigured. The codes were then

12/01/2013; 10:00 am

'I selected PM08 to code, as I hoped it would be a little 'easier' than another GP/PN. So far, quite a few 'memorable quotes'. She is clearly a very enthusiastic person, and just about the only one so far enthusing about QOF. Not surprising, as it gave her a number of opportunities, including becoming a QOF practice inspector. However, after a few years the health board 'didn't need us anymore'. I really like her philosophical, no-nonsense response to being fired: '...anyway that was that [chuckle] and this is now...' No wasted self-pity here! Reminded me about the well-known poem 'if' ...' if you can meet success and sorrow the same...' (I paraphraze)

11:40: Still coding PM08: The longest passage is when I ask her about 'culture' (ethos) - more unstructured than semi-structured interview? One of the reasons she left her previous practice was because she was 'frustrated' and it was the only way she could implement change. What a loss to the previous practice! Interesting, that the PM from the previous practice retired?? The 'new' PM came to her and has now adopted some of the systems (access/triage) that she put into place! Excellent example of cross-practice relationships and learning. How can this be captured better, and disseminated?

I've come away very impressed by the 'flat' hierarchy in this practice. They don't 'vote', just reach consensus. Lots of different, regular meetings, with standing items on their agenda. All meeting minutes are disseminated to 'everyone'. Attached staff are involved in meetings and considered part of the team. Having said that, the GPs still seem to have all of the 'power'. As far as the 50:50 split of types of PMs, I think she slightly misunderstood what I asked and insinuated. For all that the PM has autonomy, she very much seems to leave decisions to GPs. It was clear that she doesn't censure information for them.

Of all the PM transcripts I've read so far, I'm most impressed by her - very positive, practical attitude, good systems in place - in one word, I'll describe her as a 'fixer'.

Box	5.4.	Summary	of	selected	coding	methods	(285))
-----	------	---------	----	----------	--------	---------	-------	---

Qualitative data	Description
coding methods	
Grammatical methods:	Grammatical methods are used to enhance the
	organisation and texture of qualitative data. The word
	'grammatical' refers to the principles of the specific
	coding techniques and not to the grammar of
	language.
• Attribute coding:	Attribute codes are used for basic, descriptive
	purposes. They typically relate to a study's setting and
	the demographics and characteristics of participants.
	Examples from this study include: 'teaching practice'
	and 'salaried GP'
 Simultaneous coding: 	Two or more different codes are applied to describe a
	single piece of text. For example, a participant's
	description in this study of sharing a mistake with his
	colleagues during a dedicated SEA meeting was
	simultaneously coded as: 'positive safety culture',
	'SEA' and 'practice meeting'
Elemental methods	Elemental methods are the 'foundation' approaches
	and focused filters for coding qualitative data
• Structural (or	Structural codes are used to categorise data and help
utilitarian) coding:	to identify their commonalities, differences and
	relationships. Examples from this study include the
	codes 'assumptions', 'distractions', 'lack of
	knowledge' and 'fatigue' to describe some of the
	reasons clinicians gave for making errors.
• Descriptive (or topic)	Descriptive codes summarise the topic (not content) of
coding:	text passages in a single word or short phrase. An
	example from this study would be the descriptions
	participants provided of patient harm, that were
	descriptively coded as 'patient safety incident'

gathered into related categories and themes, which were mapped and displayed with NVivo version 9.2.81.0. The codes were therefore not simply labels for the data but the 'critical links' between the interview data and the explanations of what the data were understood to mean (286). This is why, although coding and analysis may not be synonymous, 'coding is a crucial aspect of analysis' (287). The number and nature of the themes are described in detail in Chapters 6 and 8. However, overall, the analysis identified 21 broad themes relating to the different study aims.

Two specific, related strategies were used to reduce bias: double coding and coding clinics (288). The double coding was performed at the beginning of the first cycle by CdW and JF, who independently analysed the same interview transcripts of GP05, PN05, PM05 and GP06. During a series of coding clinics, they then compared their initial codes, preliminary understanding of the data and any differences in interpretation. However, because 'all coding is a judgement call' and dependent on the researchers' 'subjectivities, personalities, predispositions and quirks' (289) any unresolved disagreements about coding were subsequently discussed with either COD or PB. In this way, the different perspectives were triangulated to inform the generation and application of suitable codes.

5.3.2.3. Application of the NPT framework

In thematic analysis, as described in Section 5.3.2.2, codes are constructs generated by the researchers themselves, in order to symbolise and summarise attributed meaning to data. In contrast, theoretical frameworks provide a-priori codes for researchers to apply to their data. In this study, the majority of the qualitative data from the second round of interviews were coded using the constructs and components of the NPT framework. The four main constructs are: coherence (sense-making work); collective action (operational work); cognitive participation (relational work); and reflexive monitoring (appraisal work). The NPT framework has previously been described in detail in Chapter 4 and is summarised as Table 4.1, page 96.

The sequential processes that were involved in analysing the data using the NPT framework are described next, and illustrated through the practical example of

the following verbatim quote (a datum) from GP05: 'I think it's useful as a learning tool to learn about your own systems and a way of trying to improve those systems and a way of learning as a team with the results' The quote is also shown in Box 5.5 with the original highlights, first and second cycle codes and written notes that were made while analysing the data.

Preliminary, first cycle coding was performed independently by CdW and JF, who used coloured pens and paper copies of selected interview transcripts (GP05, GP06, GP02, PN02) for this purpose. They held data clinics to compare and evaluate the consistency of their coding, to discuss their interpretations of the data and understanding of the NPT coding framework and to identify exemplar quotes for each construct and component. They discussed any differences or uncertainties with COD in order to reach a shared understanding of the meaning of the NPT constructs and components. Once this has been achieved, all further coding were performed by CdW in NVivo version 9.2.81.0.

During first cycle coding, most data were coded using one of the four main NPT constructs. However, some data clearly related to a more specific component, and could therefore immediately be coded accordingly. In the example of GP05, the participant's words, '*1 think*', was interpreted as him doing the work of trying to understand the potential value of the TRM. The quote was therefore coded as 'coherence', which is one of the four main NPT constructs. The validity of this code was examined by CdW in a related memo. If the words '*it*'s useful', rather than '*1 think*', had been considered in isolation, they could have been coded as 'reflexive monitoring', one of the other four main NPT constructs. However, when the rest of the quote is considered, it is clear that the participant was doing 'sense-making', rather than 'appraisal' work, which is why this piece of text was coded as 'coherence'.

Four informal, aide-memoires questions were used to help differentiate between the main NPT constructs. The questions were:

- Coherence what do participants think the work is?
- Cognitive participation do people join in to do the work?
- Collective action how do people do the work?
- Reflexive monitoring how do we know the work is happening?

Box 5.5. An example of a datum after the first and second coding cycles



During second cycle coding, data were coded in more detail by using the NPT components. In the example above, the participant's words '*trying to improve those systems*' and '*learning as a team*' were interpreted as him doing a specific kind of coherence work, which was to try and understand the potential value of the TRM. This kind of coherence work is described by either the 'individual specification' or 'communal specification' components. Because GP05 described his understanding of the potential value of the TRM in terms of a practice team, rather than in relation to himself or another individual, the datum was coded as 'communal specification'. The other kinds of coherence work GP05 might have done were: comparing the TRM with improvement methods he already used (the 'differentiation' component of coherence); or considering how the TRM would fit within the existing culture of his practice (the 'internalisation' component of coherence).

The first and second cycle codes, in combination with their related, reflective memos, were then analysed further in order to extract and summarise their meaning, which produced the results that are reported in Chapter 9. In the example of GP05, CdW considered the original quote, its first cycle code (coherence) and second cycle code (communal specification) before describing the result as: 'one GP participant thinks the TRM is useful because of its potential for helping practice teams learn'. Then, through a process of constant comparison, this result was compared with the rest of that participant's data and with the related results of all of the other participants, e.g. the data relating to the 'communal specification' component of the NPT framework. In many instances the results from the other study participants provided additional insights and allowed more nuanced interpretations that helped to enrich the preliminary results and refined them until they became the study's main findings.

In the example of GP05, many other participants shared his perception about the usefulness of the TRM, and no one fundamentally disagreed with this viewpoint. When all the different perceptions were combined, the preliminary result was expanded and refined into a main study finding, which is that the majority of participants perceived the TRM as useful, because it helped them identify

specific learning needs and points relevant to them and their practice teams. This finding will be discussed in more detail in Chapter 9.

5.3.2.4. Development of a preliminary classification of PSIs

The PSIs that had been detected, recorded and submitted in this study, as well as the QOF study, were coded and thematically analysed by CdW, SL and CB. Initially, CdW, SL and CB independently coded the same 100 PSIs. They met to compare their codes and agree which categories of types of PSIs were emerging from the data. Next, CdW, SL and CB coded 100 additional PSIs each, before meeting for a second time to clarify any remaining discrepancies and to further refine the categories describing the different types of PSIs. CdW then double coded two samples, different from the aforementioned 200 PSIs, that were also coded by SL (n=50 PSIs) and CB (n=50 PSIs) respectively. No further discrepancies or new types of PSIs were identified. Consequently, the remaining PSIs were coded individually by either SL, CdW or CB.

The nine categories of different types of PSIs that were identified through this analytical process provided a preliminary classification of the types of PSIs that were detected with the TRM in this study and the QOF study. The classification and the proportions of PSIs in each category are reported in Chapter 7, page 187.

Classification systems are valuable tools in patient safety research. They provide a systematic and potentially transparent approach for describing health care processes and systems; support reliable analyses of PSIs and clinical outcomes; and allow integration of data from heterogeneous data sources. Consequently, there is considerable interest in this concept and a wide range of patient safety classification systems and taxonomies have been proposed. However, the vast majority of these were considered unsuitable for direct application in this study. This is because: (i) they were developed for application in secondary care settings; (ii) describe related yet fundamentally different concepts to PSIs such as 'error', 'harm' or 'adverse event'; or (iii) had not been appropriately validated. One of the first taxonomies of medical error in primary care was developed fifteen years ago in the USA (290). It consisted of eight categories: administrative failures, investigation failures, treatment delivery lapses, miscommunication, payment system problems, error in execution of a clinical task, wrong treatment decision and wrong diagnosis. Shortly after, an international taxonomy of general practice errors was proposed. It was developed from voluntary incident report data and was more detailed with five different levels and 171 different types of error (291). A few years later taxonomies of medical error in general practice were developed in Canada (six error categories: administrative; communication; diagnostic; documentation; medication; and surgical/procedural) (292); New Zealand (three levels: 'patient', 'clinician' and systems' with 70 types of errors) (293); and the UK (six error categories: prescriptions, communication, appointments, equipment, clinical care and 'other') (72).

A comparison of general practice 'error' taxonomies suggests that the main source of their differences may be attributable to the development approach the researchers selected rather than their content. There are two main approaches when developing taxonomies - 'lumping' or 'splitting' categories, referring to the predilection of researchers to merge or subdivide categories (292). Therefore, while some taxonomies have now been superseded by more recent, validated instruments, the categories they describe remain relevant.

One example of a new and validated classification system that will be of particular importance to future primary care patient safety research is the LINNEAUS collaboration's recent, comprehensive and multi-dimensional patient safety incident classification-system which they developed specifically for this purpose (294). The system provides definitions of dimensions and classes of PSIs, and is independent of data sources and accounts for the different organisations and professions involved in care episodes.

If the LINNEAUS system of classifying PSIs had been available at the time this study was designed or when the data were analysed, it would have been applied in preference to the method that was used. The classification of PSIs in this study is therefore relatively simple in comparison to the LINNEAUS framework, but the overall approach, content and PSI classes are still compatible with the newer, validated system.

5.4. Ethical approval and considerations

The study was submitted to and approved by the Glasgow University's College of Medicine, Veterinary & Life Science's Ethical Committee, reference number 2012054. The ethical approval letter is included in Appendix 1.

5.4.1. Consent

The practice team agreed to participate in writing. They gave consent to allow the researchers supervised access of the premises and medical records at arranged times. Every individual participant (GP, PN, PM, GPST or any other) was asked to indicate their consent to participate in the study in writing separately to the practice - to actively participate in the training, reviewing of medical records and before any research-related activity commenced. The surgeries, individual team members and GPSTs retained the right to withdraw from the study at any time. The consent forms are included in Appendices 4-6.

5.4.2. Analysis of risk

Prior to commencement of the study, a risk-analysis was performed. The potential risks associated with the study were considered to be very low. However, a separate concern was whether and what risk there may be from implementing the TRM and clinician reviewers finding previously undetected PSIs. Practical experience with the trigger review method in Scotland during the previous four years suggested that the vast majority of detected patient safety incidents would be of low or moderate severity. Only in exceptional cases would an incident of severe patient harm have gone undetected. However, this possibility was considered and reviewers were encouraged to deal with it in an appropriate manner through disclosure to the patient and family, reporting the incident (if relevant) and significant event analysis. Reviewers who may have experienced emotional difficulties as a result of the trigger review method would have been provided with contact details of a health care professional to

discuss their concerns in private but this eventually did not arise during the study period.

5.4.3. Data storage, access and handling

All data were stored securely on NHS Education for Scotland IT systems in accordance with the Data Protection Act of 1998. CdW, PB and HB had access to all of the data. COD could request access to any data considered relevant and had access to selected transcripts for analysis purposes. Data were anonymised and stored on an internal network for the duration of the study and will be stored for five years after its completion.

Conclusion

This chapter described the study design and the methods that were used to collect and analyse the data. The two main sources of data were:

- 1. Two rounds of *semi-structured interviews* with practice managers, practice nurses and GPs. The first round of interviews were analysed thematically and the results are provided in chapter 6. The majority of data from the second round of interviews were coded to the NPT framework, although some data were also analysed thematically. The results are presented in chapters 7 to 9.
- 2. *Trigger Review summary sheets*, containing both numerical and free-text data. The quantitative data were counted and percentages were calculated. Means, standard deviations and ranges were calculated for continuous and ratio-type data. Comparisons of means were performed with t-tests (two variables) and ANOVA (three variables) and comparison of proportions with Chi-square tests. The qualitative data were analysed thematically. The results of the quantitative analysis are reported in Chapter 7 and the qualitative analysis results in Chapters 7 and 8.

The specific strengths and limitations of the study design and methods are described in Chapter 10, from page 269.

Chapter 6. Participants' perceptions of patient safety

This chapter describes the results that relate to the study aim of describing the perceptions and understanding of general practice clinicians and staff of patient safety. The main source of data for this chapter was the first round of interviews with practice managers, general practitioners and practice nurses. The data were thematically analysed.

Introduction

Chapter 2 described how patient safety incidents (PSIs) commonly occur, and that a substantial minority result in preventable, iatrogenic harm to patients. From a patient safety perspective, the main priorities in modern health care systems are therefore to improve standards of care, reduce the risk of avoidable harm and remove (or contain) latent safety threats from systems where possible. Consequently, a number of improvement initiatives ranging from small-scale, informal actions in single units or teams to formal collaborative-type programmes at regional and national levels have been implemented in the UK, including in general practice. Chapter 3 described five main types of improvement methods - including the TRM - that were specifically developed or adapted for use in patient safety research and in these improvement programmes.

However, despite more patient safety initiatives interventions and research than ever before, there is still very little evidence that health care safety standards, including in general practice, have been substantially improved as a result (35). There are many potential reasons for this, including some that are still unknown at this time. Arguably, the most pressing question right now is therefore whether the safety improvement methods available to us are actually useful? It is challenging, if not impossible, to provide a short and clear answer to this deceptively simple question. The reality is that the utility of a method or intervention is determined by a range of emergent, complex and interlinked factors that are typically absent when its performance is evaluated during testing or pilot studies. Three of these factors that are particularly important are described to help illustrate this point further. The first factor that helps to determine the practical usefulness of improvement methods is the reciprocal effect of the diverse, dynamic and complex health care contexts within which interventions are implemented (295). The same method produces significantly different results depending on when, where and how it is implemented. The second factor is the perceptions of the intended users. If health care staff perceive methods as acceptable, feasible and useful they are more likely to implement and use them. However, positive perceptions about the utility of a method do not necessarily imply a willingness or ability to use it. The third factor is therefore whether the knowledge, skills and attitudes of the intended users are sufficient to enable effective use of the method.

Before health care policy makers and researchers attempt to implement complex health care interventions, it is therefore important, and possibly essential, that they first consider and understand whether and how frontline clinicians and staff perceive related problems and their potential solutions. This is because perceptions about the nature and scale of the problem, and the need for, and characteristics of, interventions will largely determine how they are received, interpreted and applied in practice. In fact, this may be more important than the potential or actual 'technical' value of the intervention. However, the perceptions of general practice staff about patient safety are currently largely unknown.

The aims of this chapter are therefore fourfold:

- To explore the perceptions and experiences of a range of general practice staff about the concept of 'patient safety';
- To list and describe the factors they perceive as most important in contributing to patient safety incidents (PSI);
- To identify which improvement actions, methods and tools are known or currently being used by practices, which areas they perceive as important for further improvement and the requirements to do so; and
- To consider the safety culture that was prevailing in the general practice teams when the TRM was implemented.

Throughout this chapter, the main findings are discussed when they are reported to raise a number of directly relevant issues, and to compare them with the patient safety evidence-base where applicable. The chapter concludes with a summary description of the four main perspectives of the contributing factors to PSIs, and explain how each perspective helps to inform the selection of different approaches to health care improvement (296).

6.1. Perceptions and experiences of 'patient safety'

6.1.1. Patient safety definitions and perspectives

Only a few participants were able to provide a definition of patient safety. Others felt that a formal definition of patient safety had little value for them and could, paradoxically, even distract them in their efforts to provide high quality care. One practice nurse said: 'sometimes it can feel in health care like we just define things that we are already doing and you're not sure why. Who is doing the defining? It would be people higher up... sometimes it can feel like it's quite divorced from where we are.' (PNO2) However, all participants were able to clearly articulate their understanding of patient safety through practical examples and rich descriptions.

Despite the seemingly disparate nature of the patient safety narratives, three themes clearly emerged during analysis, based on the different perspectives from which participants initially understood and explained patient safety. The perspectives were related to: (i) patients; (ii) clinicians and health care staff; and (iii) systems and processes. The different perspectives are summarised in Table 6.1 and illustrated with a selection of verbatim quotes.

Those participants who interpreted safety from a 'patient' perspective, typically talked about 'harm' or 'journeys' and thought patients could actively contribute to their own safety, but also to PSIs. Other participants had a predominantly 'clinician and staff' perspective of safety. They were more likely to describe patient safety in terms of 'responsibilities' and 'duty of care' and considered the actions of health care workers, whether intentional or not, as particularly important determinants of PSIs. The remaining participants had a third

 Table 6.1. Participants' patient safety narrative perspectives

Perspective Verbatim quotes

Patients	I always put myself in the patient's point if that's the way I was treated how would I feel about that? I suppose I always do that and it's just natural for me (PM07) I don't think it [patient safety] is as simple as making sure you are prescribing the right drug - it is looking at the patient as a whole and thinking outside the box (PN06)
Clinicians	It's about taking personal and professional responsibility for
and staff	that patient not passing the buck, taking responsibility for each patient (PN02)
	As practice manager I have to make sure that when the
	patients come in to the health centre they are protected
Systems and	Double checking, checking, double checking. I think that keeps
processes	everybody safe. That's how I view it [patient safety] (PN11)
	It just makes me think about clinical safety - I think to make
	sure that prescriptions are correct, and referrals are dealt with
	properly and that patients are followed up (PM11)

perspective of patient safety, namely that it is about 'systems and procedures'. Their narratives typically included terms such as 'protocols' and 'guidelines' and they perceived organisational factors, systems and processes to be the main drivers of patient safety.

The three different perspectives from which participants shared their patient safety narratives were not associated with their professional roles. Being in the same practice team did not increase the likelihood of participants having the same perspective either. Rather, the specific perspectives from which participants understood patient safety in this study seemed to be mainly derived from their own, unique and individual experiences.

These study findings, i.e. that the vast majority of participants were able to explain patient safety in practical terms, even though they were unable to provide formal definitions of the concept, and that their patient safety narratives were delivered from different perspectives, are similar to the international literature. In a qualitative study of GPs (n=22) and practice nurses (n=7) in the Netherlands, none of the respondents provided a definition of patient safety, but they were also able to offer a wide range of descriptions and perceptions of this concept when asked (297).

One implication of this finding is that the way in which researchers and policy makers define patient safety may not have any practical meaning for frontline staff. In order for improvement interventions to be successfully implemented, it may therefore be necessary to first align how patient safety is understood by *all* stakeholders, and especially their intended users, to ensure overall consistency of purpose and application. From an NPT perspective, the work to achieve this goal is described by the 'coherence' construct. This will be considered in more detail in Chapters 9 and 10.

6.1.2. Patient safety perceptions and experiences

The experiences and perceptions participants had of patient safety could be summarised as four main themes. They are, that patient safety is: (i) *important*; (ii) *integral* to care; (iii) characterized by *impermanence*; and (iv) *imperfect*, but amenable to improvement. The themes are discussed below and summarized in Table 6.2 with a selection of verbatim quotes.

6.1.2.1. Patient safety is important

All participants agreed that patient safety is important, and for some of them, it was *the* most important characteristic of the care they aim to deliver. However, the majority acknowledged that there are a number of equally important

priorities competing for increasingly limited time and resources. As a result, the relative importance they placed on patient safety could change in response to internal and external factors. Two examples of the relative importance of patient safety increasing would be after detection of a significant PSI in the practice, and in response to the promotion of a safety improvement initiative by a local Health Board.

6.1.2.2. Patient safety is integral to care

The vast majority of participants expressed the opinion that patient safety had been an integral part of their practice for many years, even if this had not been explicitly acknowledged. According to them, the significant change had been that external agencies had started taking an interest in it too. As a result, many felt patient safety had become '*fashionable*' and '*sexy*' (GP04) and also: '...anything that's got a government initiative becomes a buzz word' (GP08). This perception was re-enforced by the fact that patient safety had been formally prioritised and improvement programmes were promoted by policy makers at around the time of the study. Arguably the best example of this was the launch of the national Scottish Patient Safety Programme (SPSP) in 2008 and the subsequent expansion to include general practice in 2013 (42). Participants recognised the importance of this aspect of the care they deliver being newly labelled as 'patient safety' by external agencies. In NPT terms this legitimised their involvement in the improvement initiatives - a factor that helped to facilitate the TRM. This issue is discussed further in Chapter 9.

Theme and description	Verbatim quotes
<i>Important</i> : Patient safety is important, but one of several priorities competing	• [Patient safety] is almost the 'be all and end all'. Whatever you do you have to make sure that patient safety is your highest priority (GP11)
for limited time and resources	• When things come along we should be grabbing them and taking the opportunity to look at things and how you do things and how you're going to change things for the better it's just I suppose for most practices, it's just something else to do in your already crammed up day (PM10)
<i>Integral</i> : Patient safety is and has been integral to care delivery for many	• We've been doing this same thing under different names for years. It has been going on for years. It's just been called different things (PM04)
years, but was only recently formally prioritised	 I think [patient safety is] the next big thing really, and somebody in government's decided to throw money at it, so that's what we're doing now. That's not to say it's a bad thing or that it shouldn't have been done years ago, but I think it is [pause] today's big thing (PM11)
Impermanence: Patient safety is the dynamic and emergent product of the many health care processes that shape patient journeys	 I suppose [GP X] has changed my view of patient safety from things that happen from a sort of medication or a medical point of view, wrong diagnosis, error in prescribing that sort of thing. So now I look at patient safety from that point of view, as opposed to the view that I had (PM06)

Table 6.2. The four main patient safety themes participants described
It's a whole kind of journey, and we're involved in so many aspects of it. There's safety in the physical viewing, there's safety in the medical assessment, medical administration and then the ongoing care, and I suppose secondary care. You know, you can talk about the journey into it secondary care as well. So yes, it [patient safety] covers everything really (GP07)

Imperfect: the quality of safety can - and should - be improved further. However, it is unlikely that it will ever be 'perfect'

- Nothing can be 100% safe. You can always aim for that as an ideal but, for anybody that thinks they're going to be 100% safe all the time - they are fooling themselves and probably missing what they are doing wrong... When I say inevitable, I didn't mean it in the sense of 'well it's inevitable so just live with it' I mean it's inevitable so don't accept that it [PSIs] doesn't happen (GP01)
 - Total patient safety is [pause] will I say unachievable? I don't think it's ever going to be achievable, but I think there's a lot of things can be done to change things, but I don't think you'll ever get 100% perfect - you've got too many ingredients for that, too many ingredients (PN08)

6.1.2.3. Patient safety is characterized by impermanence

The majority of participants explained that they understood patient safety not only as the state of care at important moments, such as when PSIs occur, but as the dynamic and emergent product of many different and variable processes in health care, all in temporal (and sometimes contradictory) relationships with each other. They were aware that, just as levels of relative safety risk fluctuates, their own understanding and experiences of the prevailing levels of safety changed too. There were many potential reasons for this, such as: direct or indirect involvement in PSIs - not only as clinicians but also as patients; personal reflection; feedback from patients and colleagues; and the degree to which national policy prioritises safety. A practical example of how a participant's recent, personal experiences as a patient informed her perceptions of care safety is provided in Box 6.1. Many participants explained the impermanence of patient safety by referring to patient journeys, both within primary care but also interfacing with hospital care, and how they are often unpredictable with many twists and turns and may include a range of health care staff able to influence the different potential endpoints.

6.1.2.4. Patient safety is imperfect but can be improved

All but one of the participants thought that PSIs were the inevitable consequence of clinical care and even if infinite resources were theoretically available to try and prevent them, they could never be completely prevented. This perception is widely shared by other clinicians worldwide and fits with what we know about the inevitability of failure at some point in highly complex sociotechnical systems (281, 298, 299). Some participants therefore described the possibility of 'perfect' patient safety as a 'pipedream' or a 'wish'. Despite this perception, all participants agreed that this was not an excuse for not attempting to reduce PSIs and that much could still be done in this regard. They also all agreed that improving patient safety was an ethical and professional responsibility for every health care worker, for general practice teams and also organisations. However, according to them, the expectations about the results of these efforts should be tempered by the acknowledgement that clinicians are 'human' and therefore imperfect and prone to err.

Box 6.1. Example of a participant's personal experience as a patient

I was an inpatient just for, well, for two nights. I had a thunderclap headache and I had to have an ehm CT and an LP. Anyway, it happened on a Friday, which was unfortunate ehm and I had to go to A&E was so sick, got reviewed, bloods, CT, all within an hour, so impressive. Sent over to medical assessment unit, waited about five hours to be seen then was seen by the consultant who said: 'okay, your CT's fine, we need to do an LP. We'll get that done, we'll get the results tomorrow and if it all is okay, you can get away.' Well, so I was like 'oh, that's fine'.

So then nothing happened the next day. It was so loud overnight - there was an old demented lady who was so agitated and they were speaking quite [firm voice] 'Shh, come on shhh' in the middle of the night. It was ehm quite upsetting. Anyway, she got moved to another ward in the middle of the night. I think probably cause she was quite hard work. There was another lady who was really wheezy who they also moved in the middle of the night and I was going to go and say I think she needs a nebuliser, I think she needs reviewed. None of this was done on the acute medical ward ehm and anyway, they did the LP too late, they sent the samples off, they told me they hadn't arrived. Turns out they don't process at the labs after twelve o'clock on a Saturday. Had to wait til the Sunday. They wouldn't let me go home even though it was less than a mile from the hospital, so I had to wait another twenty four hours but they kept me because I was independent and self caring...

I was not impressed at the level of nursing and care... there's not enough time nursing anymore I don't think, and I think we need to get back to good caring, nursing, that's my rant [chuckle]. I was, I was very saddened by my stay, and it actually makes me think I don't want to admit people to the [hospital X]. I think patients might be better dealt with at home. So nursing, more nursing time, more nurses on the wards so they can do their job fantastically but also care for the patients...if they don't increase the number of nurses you're going to have people having acute LVF [heart failure] or people going off with COPD, people arresting and no-one else noticing...we've lost the human aspect of it and I think that's a big loss... you're not meant to move old delirious, demented patients in the middle of the night cause it increases delirium, it increases that [whispering] we all know that. They should have moved me but they moved the one who was hard work, short of breath, not being reviewed. I mean, I, was just about to go and say 'look, can you get this person', then they wheel her, but I was 'oh'? It was quite bad from the other side (GP07). There are at least three practical reasons that support the perceptions of the participants that patient safety will always be imperfect. The first reason is the variable quality and timeliness of information. Clinicians often have insufficient information or the wrong information and increasingly also have to contend with information overload and information 'chaos' (300). Even when they are familiar with all the latest evidence-based information and have access to reliable investigations they may still not understand or know all the relevant contexts and patient-specific issues. For example, some medication side effects and serious drug interactions may be known but cannot reliably be predicted for every patient; and rare and occult diseases may go undetected or unrecognised. The second reason is that clinicians are in many instances only one contributing factor to PSIs. Technology can fail without human involvement and patients contribute to some PSIs too. The third reason is that, even if clinicians worked with 'perfect' patients in 'perfect' systems, their own psychological and physiological limitations predispose them to error. These factors will be considered in more detail next.

6.2. The main contributing factors to patient safety incidents (PSI)

Participants felt PSIs were most likely to occur in medication and medicationrelated processes, such as prescribing, dispensing, administration and monitoring: 'our biggest area where things can go wrong is generally through prescribing, and it's where you can do the most harm as well' (GP03). Three specific examples of medication-related risks participants identified were: patients intentionally and unintentionally varying from prescribers' intended use of their medication; a lack of reconciliation of medication items between different health care settings; and the sheer volume of prescription requests that made effective monitoring a challenge. One GP explained: 'the way that the service is set up, we as prescribers aren't the ones that are doing the monitoring' (GP10).

Another important practice system participants identified as high-risk for PSIs was ordering investigations and managing results, e.g. '*I'm really worried I put the wrong label on things - that's awful easy done' (PN11)*. Many clinicians recounted personal experiences of specimens, requests and reports being

mislabelled, lost or not followed-up appropriately. Despite this, the vast majority thought the volume of data made it unfeasible to implement a monitoring system, e.g. 'We don't have any processes [to monitor investigation requests] and a lot of that is about capacity you know. The practice just doesn't have capacity to manage everything' (GP08). Participants also identified the processes of sharing, storing and accessing information as important safety risks: 'the ability for error is huge. I mean I scan something and then you drop and drag it into a patient's records. How do you know you've put it into the correct patient?' (PM02).

Elderly and housebound patients and nursing home residents in particular were considered to be at increased risk of PSIs and more likely to suffer harm. Participants explained this risk was because of relatively higher levels of multimorbidity, polypharmacy and reduced physiological reserves compared with other patient populations, but also because elderly patients often struggle with access to appointments and clinicians have insufficient time for pro-active management of physical and social problems.

The specific patient characteristics and practice processes the study participants identified as being particularly 'high-risk' for PSIs are almost identical to those identified through years of patient safety research and which was summarised in chapter 2. A further, specific example from the international literature is a webbased survey of GPs in the Netherlands. The two risk factors they perceived as most frequently constituting safety threats were, similar to this study, prescribing and monitoring of medication and patient age over 75 years. In addition, they identified polypharmacy, poor doctor-patient relationships and insufficient continuing education by GPs as important (301). The recommendation to practice teams in this study was to select the records of high-risk, elderly patients for trigger reviews. One of the pre-defined triggers required reviewers to screen for medication changes, while several others were related to specific laboratory results. In terms of NPT, the high degree of congruence between the perceptions of the study participants, the international literature and the TRM helped to facilitate its implementation through the work of coherence.

All participants were able to identify a range of contributing factors to PSIs. The factors were related to five main themes: (i) inadequate time and resources to deal with increasing workloads; (ii) lack of care continuity; (iii) patient-related factors; and (iv) clinician-related factors; and (v) chance. The participants explained that all of the contributing factors were potentially relevant to *all* types of PSIs and could affect any practice system, process or patient. The implication is therefore that, in order to improve the safety of care, it is necessary to adopt a holistic, systems-perspective rather than simply attempting to 'fix' each factor in turn. These themes are discussed below and summarized in Table 6.3 with a selection of verbatim quotes.

6.2.1. Inadequate time and resources to deal with increasing workloads

All of the participants described how they, and the rest of their practice teams, were struggling to safely manage their existing workloads and that their workloads continued to increase. They responded to the increasing workloads with a range of formal and informal adaptive behaviours. Four practical examples of adaptive behaviours provided by the participants were: intentionally deviating from policies and procedures (e.g. violations); working additional, unpaid hours; choosing to forego breaks and meals; and reprioritising the urgency of competing tasks and making changes to their appointment systems.

Despite these behaviours - in a sense their best efforts to adapt - they remained aware of potential safety threats in the practice which, in some instances, had even increased. For example, some patients may have been inappropriately triaged, prescriptions were often signed without being reviewed and reception staff sometimes offered patients appointments with team members who were not clinically appropriate to their needs. One nurse explained: 'we can't see these types of patients' but because 'there's no appointments [reception staff is] putting it in for us just to get them really off the phone' (PN09).

Participants clearly understood the potential risk of increasing workloads and relative lack of time, but were unable to find an alternative solution that was acceptable and feasible in practice. Procuring additional time (whether through

Main factors	Specific examples	Verbatim quotes								
Inadequate resources	Excessive workload	It was almost getting really too dangerous - we're talking sixty people in that hour and a half every morning (GP02)								
		The pressure that is on practices to churn out patients and churn out facts, figures, returns - it's phenomenal (PM02)								
	Time constraints	At my lunch break I'm putting information on the computer (PN03)								
Clinician factors	Distractions	If you're doing that while you're trying to do six million other jobs, your phone's ringing and everyth else then you could miss it. Discharges are particularly bad - we've had it one or two incidents with discharges [deep breath] (PM04)								
	Assumptions	I don't really remember what I was doing at the time. I do remember seeing it, but I made an assumption. Assumptions aren't good for safety (GP02)								
	Difficult 'personalities' in the team	[There is] huge variance from practice to practice and I guess it's all about the GPs that you work with (PM08)								
	Lack of knowledge or skills	I think the biggest threat to patient safety is the fact that loads of people do things that they don't actually know how to do in nursing (PN02)								
	Stress	You get stressed; you're kinda rushing against the clock all the time. So because you're rushing it's probably more stresswhich in itself isn't a great thing either from a safety point of view (PN03)								

Table 6.3. Factors perceived as contributing to patient safety incidents (PSIs)

Main factors	Specific examples	Verbatim quotes
Clinician factors continues	Being unaware of personal limitations	Nobody is perfect ever, so you have to be aware of your limitations (PN09)
	Violations	The systems are there. People either don't follow the systems or don't have time so try to cut corners with the systems, and that's why things fail (PM07)
Loss of care continuity		Sometimes we've had issues over the last few years and I think really it's because people are darting in and out and don't really know what's going on (PM12)
Patient-related factors	Patient expectations	I think they [patients] have some real unrealistic expectations of what doctors can and can't do (PM03)
	Compliance	If you sat down and you've agreed that's what you're going to do, it's then the patient's responsibility to keep their part of the deal in inverted commas (GP11)
	Diversity of presenting problems	We have such a big remit as practice nurses that we're expected to be, to at least a degree, experts in, that you can sometimes feel like you're drowning a wee bit (PN02)
	Multimorbidity	People come in with a shopping list you know (PN08)
Chance		It's really just luck when bad things don't happen (GP08)

employing new staff or increasing the hours of existing team members) reduces the income and livelihood of the partners and would in some instances make the business non-viable. One GP explained how they: 'obviously have a set amount of money. If you employ too many doctors you're not going to have a practice because it won't be financially viable and it will close' The challenge is 'getting the right balance of what you can afford and what you can do' (GP03). This important issue will be considered in detail in Chapter 10.

Inadequate resources and high workload were not only perceived as important safety threats by the study participants, but were also understood to have a negative impact on the performance and wellbeing of clinicians and staff. These key findings are comparable with the international literature (302). In a focus-group study of primary care physicians (n=32) in the USA, the lack of resources and time pressures were perceived as particularly important impediments to care quality. In fact, inadequate resources 'often force physicians to compromise standards of care' (303). Other important factors that were identified in that study included: lack of control over work environments and inflexible and unclear policies and procedures.

Insufficient time to provide all of the necessary care patients require is a wellrecognised and important contributing factor to PSIs. A framework identifying specific types of 'time problems' or 'tempos' in general practice was recently proposed (304, 305). One of the five components of the framework is the 'office tempo' and quantifies the amount of time clinicians have available to provide care for patients. All participants identified this tempo as a particularly important risk factor for PSIs. The other four tempos relate to: (i) 'diseases' (estimated evolution and response to treatment times); (ii) 'patients' (wide variability in compliance with management plans and follow-up); (iii) 'out-ofoffice coordination' (obtaining investigation results and access to specialists); and (iv) 'knowledge' (misinterpretation of initial symptoms or having to acquire new information). However, the study participants were less likely to identify these four tempos when discussing the importance of time.

6.2.2 Lack of care continuity

Participants explained continuity meant ensuring that relevant, correct information accompanied the patient throughout their journeys and that named clinicians were involved in their care. Continuity was generally perceived as an important contributing factor to safe care. Conversely, a lack of continuity was perceived to negatively impact on patient safety, particularly during care transitions between health care providers and at the interface between different organizations. One participant explained: 'I refer them [patients] to hospital. They see two different consultants [pause] it's a system it's a patient pathway but it doesn't necessarily mesh like that. It could butt up against each other like that and that's where the problems lie. They don't know what I'm doing and I don't know what they're doing' (GP04).

Many participants were also concerned that care continuity was being eroded at a practice level, which they attributed to increasing workloads; patient expectations of same-day consultations; and use of locum staff. However, one participant felt continuity paradoxically decreased patient safety in some instances. She explained that: 'sometimes it can be the other way. It can put a lot of demand on you because they know you' (PN05). In response, clinicians might be more inclined to accede to the requests of the patients who are wellknown to them or try and fit in additional issues within a single consultation.

6.2.3. Clinician-related factors

Participants described a wide range of 'clinician' factors that may contribute to PSIs (Table 6.3). However, they recognised that some behaviours intentionally (for example violations) or unintentionally (for example assumptions) ensured patients received an effective and ongoing service despite distractions, competing priorities and having to deal with large workloads. What the participants were describing are clear examples of 'efficiency thoroughness trade-offs' (ETTOs), although they did not use this term. ETTOs and the ETTO principle are therefore important in relation to this study and will be discussed on page 166.

One of the more insidious clinician factors that was identified by some participants was 'personalities' in their teams. They used the word 'personality' euphemistically to explain how specific characteristics of some clinicians increased the risk of PSIs occurring in the practice. These characteristics included clinicians being afraid to ask for help; lacking insight about their own knowledge and skill deficiencies; interpersonal relationships and communications skills that made it challenging for others to raise concerns about potential safety threats with them.

6.2.4. Patient-related factors

The majority of participants - and especially the practice managers - felt the expectations of patients about their care, of GP practices and the NHS had increased to the point where it was difficult or impossible to meet. This feeling was compounded by the perception that some patients take very little or no responsibility for their own health, but also by the increasing clinical complexity and prevalence of multimorbidity. One practice manager described her perspective: 'we are becoming an older generation. I think there's more and more, three or four things that people are wanting to discuss, where you just can't do that in ten minutes' and then there is the additional workload of 'all the paperwork that comes behind that as well' (PM02).

The clinicians reported struggling and often failing to effectively and safely manage the 'shopping lists' (PN08) of their patients in the ten minutes allowed for consultations. However, while some patients were perceived as 'demanding', participants acknowledged that the majority of patient requests and expectations were appropriate and that clinicians and practices were responsible for meeting them.

The first four groups of contributing factors to PSIs the participants in this study identified (sections 6.3.1 through to 6.3.4) are comparable to those reported by clinicians in other health care settings and countries. Three of many possible examples are provided as evidence. The first example is a qualitative study conducted in the USA more than twenty years ago. The authors explored the recollections of family physicians (n=53) of their most memorable errors and the

perceived causes through in-depth interviews. Similar to this study, many different possible contributing factors (n=34) were considered, which the authors categorized into three groups: physician stressors and characteristics; process-of-care factors; and patient-related factors. Some of the more important, common and specific factors within these categories included distractions, 'hurry', lack of knowledge and premature closure of the diagnostic process (306).

The second example is a survey of clinicians (n=848) working in outpatient settings in the USA that identified many cognitive and systems factors felt to be related to diagnostic errors. Some of the more common factors reported by clinicians were: inadequate knowledge, detection or perception of clinical problems; excessive workload; issues in relation to investigation and information systems; and patients' lack of adherence to physician recommendations (110).

The third example is the qualitative study by Slight et al about the perceived causes of prescribing and monitoring errors in English general practice. They identified seven high-level categories, each containing a number of error-producing conditions. The categories are: the prescriber; the patient; the team; the working environment; the task; the computer system; and the primary-secondary care interface (307).

In addition to these well-known contributing factors to PSIs, the participants in this study also identified a fifth group, 'chance'.

6.2.5. Chance

The vast majority of participants considered 'chance' or 'luck' to be the most important contributing factor to safe care or, alternatively, whether PSIs occurred and the resultant harm severity. One GP explained how 'things that could have gone wrong but didn't' was because of 'the grace of God and good luck' (GP10). Consequently, many participants expressed feeling helpless and unable to prevent these types of PSIs in the future because, as another GP explained, 'if she were to have that same day again, probably the same thing would happen again. It's just a set of circumstances... you might be lucky, you *might be unlucky*' (GP02). However, all participants acknowledged that chance was not always implicated or the main contributing factor for every PSI.

This perspective of PSIs may seem negative, pessimistic or fatalistic to some. However, given the vast majority of participants share this understanding about at least some PSIs, their perception should at least be acknowledged, and its veracity examined. If we accept for a moment that some PSIs truly occur by random chance, what type of method or improvement initiative could we propose that would be acceptable for clinicians? How do we best safeguard clinicians and patients alike? This issue will be considered further in section 6.5, from page 166.

6.3. Existing improvement actions and potential for future interventions

6.3.1. Existing improvement actions

Participants were aware of many different formal and informal methods that could be used to help improve the standards of care they deliver. Table 6.4 lists specific examples of methods and verbatim quotes. The methods ranged in scope from small changes for a single patient to system-wide re-organisation of practice systems. For example, a practice nurse (PN04) was able to recollect three recent improvement actions she had undertaken: (i) covering the light bulbs and fixing in place the examination lamps to prevent them falling or shattering on patients during cervical screening tests; (ii) convincing her practice to procure a hydraulic bed to facilitate the physical examinations of disabled patients; and (iii) she had designed and implemented a protocol for travel vaccinations to standardize management. This finding, that many clinicians develop informal, unique and successful solutions to safety risks in their own working environments, are consistent with the literature (308).

A significant minority of participants, and many from the practice nurse group in particular, perceived the active involvement of patients as an important approach to deliver high-quality, safe care. For them, patient involvement meant involving patients in their own management plans through shared decision-making, providing patients with clear information about their conditions

Action or method	Selected verbatim quotes
'Formal' actions	
Significant event analysis (SEA)	We do significant events regularly we will meet to discuss it (PM06)
Clinical audit	We do lots of audits around [access] and check that it's still as good as we think it is, and we occasionally have to tweak the amount of triage (PM08)
Protocols	Over the last few years with being a training practice we have tried to put a lot of protocols and systems in place to protect it (PN05)
CPD, appraisal and revalidation	Individually you are doing the best for the patient that you have and that is your responsibility, so there is a bit about professional development, CPD and maintaining your knowledge and recognising your weaknesses (GP08)

Table 6.4. Examples of improvement methods participants already used

'Formal' and informal actions

Involving patients	We're calling it 'complaints, comments and compliments						
	and what we're asking, we'll go out regularly and speak						
	to the patients and say 'how do you feel about how we're						
	doing? Is there anything we can improve on?' How do we						
	know we're completely safe? I think this is maybe a way						
	of us checking are we doing enough (PM02)						

Informal actions

Raising awareness of	People are making others aware of what has happened
safety critical issues	and that is the way forward and we will just continue to
	do that, and hopefully we will get better and better at it
	(PM06)

Sharing information / I think being able to discuss things with my nursing peer feedback colleague - on a Wednesday I start at one, we have an hour's handover - I find that really useful (PN02)

Mitigation, esp. I think there are lots of sources that stop us from falling pharmacists and patients short more of the time, to be honest (GP03)

and treatments, pro-actively planning for future events (e.g. exacerbations of their chronic conditions) and eliciting patients' feedback about the standard of care they received. One practice nurse explained that when she's consulting: 'you're hoping that they're [patients] going to give you a wee bit of feedback as well... they're the first ones to say: 'hey wait a minute!' (PN07)

Many participants recognised or referred to instances where PSIs or their impact had been mitigated. This happened through the intervention of a wide range of potential factors, but the most important were other practice team members, pharmacists and patients. In other words, 'people' were responsible for creating patient safety. Specific instances of mitigation were considered to occur mainly through chance, but the overall likelihood of mitigation increased when there was effective teamwork and communication in the practice. One participant described how: 'there is a whole team, you know - GP, staff, pharmacists. If everybody works together as a team then these things can be resolved before they get to the critical stage' (PM06). This issue will be considered again in Chapter 10.

6.3.2. Potential for future improvement initiatives

When asked to suggest high-priority patient safety areas for future intervention, the majority of participants identified two specific issues. The first issue was medication and medication-related processes. Participants were concerned about the very large volume of repeat prescriptions generated on a daily basis, usually by administrative team members, which were signed without review by GPs: 'If you see the amount of repeat prescriptions that wait to be signed, it can be two inches thick [this] puts a lot of stress on the doctors, because then it becomes a machine. So all you're doing is you're signing through the prescriptions and not actually looking at the prescriptions. How safe is that?' (PM02). For them, the implication was that more, and more thorough, medication reviews would be useful to help prevent PSIs. The perceptions of the study participants are consistent with the international literature that have also identified the management of repeat prescriptions as an 'important source of risk' (309).

The second issue was ensuring housebound patients receive the care they require. One participant described how 'housebound chronic diseases get neglected' because 'you can tick a box to say they're housebound so therefore you don't have to achieve targets for them' (PN08). A substantial minority of participants therefore thought it might be helpful to create nursing roles specifically aimed at providing care to these patients, similar to how public health nurses care for children or incentivizing practices to provide this service. A practice nurse explained: 'if the funds were there, that would be a good idea' (PN12).

In addition to these two issues, a substantial minority of participants also suggested: improving the reliability of their clinical coding and computer software usability; reducing the number of interruptions during specific tasks by allocating protected time for staff; and collectively learning from PSIs.

Despite recognising potential areas for improvement, the study participants unanimously agreed that they had no time, resources or spare capacity to consider implementing these suggestions, or *any* other new interventions. They also doubted whether any other team in the general practice setting could feasibly undertake any additional, unfunded work. Some participants also expressed concern that quality improvement interventions may increase workload and paradoxically decrease patient safety by reducing time to provide clinical care. One participant explained that this was why *'it can be seen as hassle for some'* because *'it takes us out from the day to day practice when I have still got other patients and normal work carries on'* (GP06).

Perhaps unsurprisingly then, most participants identified externally funded, additional clinician and staff time as the most important determinant of patient safety, and whether they would be able to implement additional improvements. One practice manager described that 'what we need is another body in the practice who'll see another hundred patients a week, which would make a tremendous difference' because 'it means they [admin staff] can get on with the safety aspects of downloading blood results instead of answering a patient's phone call four times when one would do if we had appointments to give them' (PM12). A small minority of participants disagreed, and thought increasing staff levels could actually decrease patient safety. They reasoned that this would be because of loss of care continuity, increased use of locums and the likelihood that patient demand would keep outpacing any additional capacity. They argued that rather than employing additional staff, existing team members could work more efficiently instead: 'what you need is a good dependable well trained staff rather than more bodies on the ground' (PM06).

Participants also recognised a need for additional training, new quality improvement tools and IT support if they were to successfully implement further improvements. However, these factors were relatively less important than workload, time and resources. One practice manager made it clear that 'we don't have the money for the other options [time, staff] so therefore we'll take the tool' (PM11). This key finding is reflective of the wider patient safety literature. In a survey conducted by Sarkar et al in the USA clinicians provided open-ended recommendations to reduce diagnostic errors. The vast majority suggested addressing workload issues and time constraints, improving investigation management systems and strengthening collaboration and communication between different health care providers. A tiny minority (3%) cited 'training opportunities' as a potential improvement strategy (110).

6.4. Participants' perceptions of safety culture

All participants thought culture was an important determinant of patient safety in their practice but also in the wider health care service. They understood culture as a dynamic and evolving construct and were able to provide many examples of how cultures in different parts of the health service interacted to produce positive, but also negative, outcomes for staff and patients (Table 6.5). However, all participants perceived the culture in their own practices and, with a few exceptions, other general practices in Scotland as conducive to safe care. They felt the overall culture in general practice was characterized by effective teamwork, communication and leadership, but that this was not necessarily the case for other parts of the health service. The sentiment of PN02 that 'there's a very open communicative atmosphere [and] we're all fairly aware of the

Table 6.5. Examples of the potential effects (positive or negative) whe	en
different cultures interact in health care	

Providers	Verbatim quote There was a little bit of a stand-off between the receptionists and the district nurses but I don't think, you're not going to resolve that (GP03)								
District nurses									
General Medical Council	At some level there is the big bad witch called GMC that must have some impact actually, if you end up making a lot of errors that is one driver for people to not end up in trouble with GMC (GP06)								
NHS Health Boards	There is always this suspicion as well of Health Board managers. Real suspicion from a lot of GPs that engagement that you can actually work together - they don't believe that that is what is happening (GP08)								
Secondary care referrals	I think a lot of the systems that have been put in place to improve the flow of information haven't really done that well. They've done it on one level but what we used to have in place - the sort of personal relationships that we used to have between primary and secondary care - that's all been broken down, but it's not been replaced by anything so I see that as a real problem (GP09)								
New practice staff	I think it's hard when you come in somewhere new to you. You've worked a certain way before. There is a tendency to want to come in and change everything but you have to obviously respect other people's working practices as well (GP02)								
Accident & Emergency departments	A&E letters at the moment from [x] hospitals gives a disclaimer that drugs are not included in the letter which to me seems like a shocking admission in any kind of a clinical correspondence but we live with it, for the time being (GP01)								
Improvement initiatives (SPSP- PC)	These [GPs] are independent contractor communities. We have to engage with them on that basis. We can't tell them what to do (GP04)								

threats to patient safety' was representative of the perceptions of the study participants.

Participants also recognised that the prevailing culture is an important determining factor of any new initiative's degree of success. One GP explained that an essential prerequisite for the successful implementation of change is to 'get the culture right' so that 'there'll be an appetite for the tools' (GP04). Another participant agreed 'the fundamental thing is culture'. According to him the specific methods were secondary to the overall success of a venture: 'you are just using the tools to try and emphasize and drive and direct people' (GP06). The importance of culture will be considered again in Chapters 9 and 10.

While participants were aware of culture changing over time, few were able to describe how or why these changes occurred. A number of participants attributed a strong, positive safety culture simply to 'luck': 'maybe we're just a lucky practice because we all more or less sing from the same hymn sheet' (PM08). However, one factor they felt did affect their culture (in a positive or negative way) was team members leaving or joining the practice: 'we'll have someone new in the practice and that kind of gives everyone a bit of a buzz really' (PM08).

The characteristics and personality traits of team members also affected the culture. Some team members - usually one of the GPs and/or the practice manager - seemed to have a disproportionally large impact on the prevailing culture. These team members strongly influenced the practice safety culture through their actions, examples and leadership. GP06 was one of many participants who was identified by his team as being influential and important to their culture. The following verbatim description of how he approached PSIs provides evidence of how he helped to build a positive and open culture: '*if 1* haven't done something correct, *1* have always first to be self effacing and fall on the sword and say: 'that was my fault, *1* made that mistake. *1* apologize, how do we make it better from here?' Also being accepting of when other people say 'oh this didn't go quite right' - *1* won't be likely to hit them over the head with a stick about it. *1* would far rather that *1* knew about it and then you can adapt' (GP06). The issue of leadership will be considered again in Chapter 10.

Many participants were aware that the culture in health care had changed over the years and continues to change. While some of the changes were incremental and barely perceptible at first, other changes had been more dramatic and rapid. For example, some participants described being aware of a prevailing 'blame and shame' culture' in health care for a long time, but how this had changed for the better in more recent times. One participant thought: 'the culture within medicine is far more open than it used to be, and everybody knows that they make errors (GP06). Another participant described 'a big culture shift' in the willingness of clinicians to admit their errors and interpret them as 'opportunities to learn and make positive changes' (GP10).

There is corroborating evidence for the perceptions participants had about the prevailing safety culture in general practice in Scotland and England. A safety culture survey of general practices in England with the 'Competing Values Framework' instrument just over a decade ago found the majority of teams had a 'clan' culture type, which is characterised by high levels of teamwork (310). For the last few years perceptions of safety climate in general practice have been measured annually in Scotland with SafeQuest, a validated 30-item questionnaire (170, 311). The quantitative survey data suggest the safety climate in primary care is generally positive, although a small minority of practices has significantly less positive perceptions of safety climate than the main group (41, 42, 170, 311).

6.5. Different perspectives of the causes and contributing factors to PSIs

This section describes four main perspectives about the contributing factors to PSIs in the international literature (296). They are that PSIs result from: (i) individual human errors; (ii) systems and technical failures; (iii) weak safety cultures; and (iv) Efficiency-Thoroughness-Trade-Offs (ETTOs). These perspectives were reflected in the wide range of patient safety experiences and perceptions reported by participants in this study. The practical importance of the different perspectives is that they strongly influence which type of improvement approach individuals and teams are likely to select.

6.5.1. Individual human error

The first perspective of the causes and contributing factors to PSIs is that they are the product of individual human failures and errors. Human error is considered a natural phenomenon and an important mechanism through which we learn. The phrase 'to err is human' is well-known and recognises the existence of finite physiological (e.g. attention span, short-term memory) and psychological (e.g. resilience, motivation) resources that will eventually limit the ability of even the most skilled and talented human being to continue to function effectively and safely (312). For example, an inexperienced, tired and hungry GP specialist trainee (GPST) working in a busy and unsupported practice and who is often interrupted is more likely to make a diagnostic or prescribing error than her well-rested and experienced colleague working in a supportive environment.

The implications from this perspective are that *all* health care workers are susceptible to err; that the likelihood of error increases as the number of 'demands' on finite human abilities increase; and that the frequency and types of clinical errors are largely predictable. From this perspective, the contributing factors to PSIs can be conceptualised as simple linear cause-and-effect models. A practical example is that interrupting a clinician causes a lapse in concentration which results in a diagnostic error (effect). These models implicitly or explicitly implicate health care professionals in PSIs and may inappropriately and sometimes unintentionally be used to justify the 'blame and shame' culture that still prevails in some health care organisations.

There are at least three reasons why individual clinicians may be inappropriately blamed for certain PSIs. The first reason is that, from a legal perspective, it is easier to prosecute an individual than an organisation. The second reason is that disciplining individual employees are more convenient for some health care organisations than analysing and improving their systems. However, the third reason is arguably the most pervasive and important, yet also the most challenging to overcome. As individuals and societies we are psychologically predisposed to attribute blame for unfortunate events to those visibly and directly involved in them. Two examples of psychological predispositions are: 'hindsight bias' and the 'just world' hypothesis. Hindsight bias creates the 'I-knew-it-all-along' effect. This means knowledge of an action's outcome makes warning signs appear more obvious and consequences more foreseeable than they were to those involved in the PSI (313). The 'just world' hypothesis is the (often unconscious) assumption that 'bad' things happen because of 'bad' people (312). Unfortunately, the reality is that it is often the 'best' people who make the worst errors as they are more likely to be performing the most difficult tasks (164).

From the perspective of 'human error', certain improvement initiatives will appear more appropriate than others. Examples of interventions that are strongly associated with this perspective include: (i) continuous professional development (CPD), providing educational resources and promoting (or mandating) specific training activities; (ii) increasing professional regulation and implementing a revalidation system; (iii) taking disciplinary actions against individual clinicians who are involved in PSIs; (iv) incorporating human factors engineering in health care organisations; and (v) clinical decision support systems (109, 314).

6.5.2. Systems and technical failures

The second of the four patient safety perspectives is that health care systems and technical failures contribute to PSIs. Health care organisations consist of many highly complex systems, and even relatively straightforward and common processes have multiple, interacting steps (281). The inherent active and latent safety risks in these systems and processes are often graphically depicted by the admittedly oversimplified metaphor of the 'Swiss-cheese' model (1). In the model the slices of cheese represent the various system defences between potential hazards and accidents. The holes in the cheese represent active and latent errors in the system and the slices of cheese have to be visualized as being in constant motion. The holes in the cheese rarely form a straight line of openings because at least one slice typically blocks the hazard from reaching patients. PSIs occur when the holes in the slices of cheese temporarily align, which allows hazards to reach patients. Systems and technical failures can be represented by linear models in which it is possible to identify simple, complex and cascading causes, contributing factors and outcomes (315). In other words, a PSI is conceived as the product of a series of events which occur in a specific and (retrospectively) recognizable manner. The model therefore allows some knowledge about the future which is conceptualized as a mirror image of previous events and conditions. The implication for patient safety is that some PSIs may be prevented by detecting and eliminating potential threats proactively and by designing, incorporating and strengthening health care system defences (316). From this perspective, examples of system defences may include: guidelines; protocols; automating processes or including visual reminders for clinicians; redundancies (e.g. additional steps or duplication of actions); and 'forcing functions' in software.

6.5.3. Safety culture as a contributing factor to PSIs

The third perspective about the causative and contributing factors of PSIs is that the prevailing safety culture in a health care team or organisation helps to determine whether they occur or not. As previously discussed in Chapter 2, organisations and teams with a positive and strong safety culture are more likely to learn openly and effectively from error and adapt their working practices appropriately when PSIs occur. Safety culture will be discussed in more detail in Chapter 10.

6.5.4. Efficiency-thoroughness trade-offs

The fourth perspective about the causes and contributing factors of PSIs are that they are the inevitable by-products of efficiency-thoroughness trade-offs (ETTOs). ETTOs are the approximate performance adjustments individual clinicians make in order to manage expected and unexpected variability in their environments and tasks (317, 318). The effectiveness and appropriateness of ETTOs can only be evaluated retrospectively, once patient outcomes are known. The implication is that ETTOs are not intrinsically 'right' or 'wrong'. An example of the ETTO principle in general practice is the challenge of GPs to balance their available time between taking complete histories and performing thorough examinations, and offering appointments to all the patients who request them. Taking incomplete histories and conducting partial examinations would allow GPs to see more patients. This approach prioritises efficiency over thoroughness. Alternatively, GPs may decide to take complete histories and conduct full examinations (increased thoroughness) but will then be able to offer fewer appointments or have to work longer hours (decreased efficiency). The benefit of maximum efficiency is that more patients will receive at least some care. On the other hand, it may increase the risk of diagnostic errors and only the most pressing problems may be dealt with. The benefit of maximum thoroughness is that rare or dangerous clinical conditions are more likely to be identified. However, fewer patients will receive care and other, equally important tasks, may be neglected.

The stark reality in present-day UK general practice is that resources and time are at best finite but more typically scarce. The information available to GPs with which to diagnose and plan care is always underspecified. This is because undifferentiated patient populations present with complex, diverse and often atypical symptoms and signs and medical records are often incomplete or unavailable. To provide effective patient care despite the challenge of everchanging demands, clinicians have to make approximate adjustments to their performance. In other words, they have to adjust (consciously or unconsciously) the efficiency in relation to their thoroughness. The adjustments are based on their interpretations of the requirements of their patients and the available resources (of all types) at given points in time. It is 'performance variability' that allow clinicians to balance the demand for resources with what is actually available and possible to do in practice (319). The challenges inherent in this balancing act will be considered in detail in Chapter 10.

From an ETTO perspective, performance variability is therefore not only considered normal, but necessary, and the origin of *both* successful *and* unsuccessful health care outcomes. The ETTO approach to improve patient safety is by learning from positive deviance and increasing the resilience of clinicians and systems (6, 319-321).

Conclusion

This chapter began with a description of the perceptions of a range of general practice staff about the concept 'patient safety' in Section 6.1. All participants considered safety to be an important and integral part of routine care. The vast majority also perceived it as impermanent and imperfect. However, they recognised an ethical and professional duty to continue to improve the standards of care in their practices and felt that there was potential to do so. The TRM should therefore be acceptable to the participants, as the main reason for its development was to provide frontline staff with a practical method to examine their own records in order to detect and learn from PSIs and identify and address latent safety threats in their systems and processes.

Section 6.2 described five groups of factors participants identified as causing or contributing to PSIs. In addition, they identified medication and medication-related processes and elderly, housebound patients as being particularly at risk for PSIs. Most participants perceived a proportion of PSIs as inevitable, and therefore not preventable. However, they unanimously agreed that many contributing factors and high-risk processes and systems are amenable to improvement efforts. The TRM is compatible with these perceptions as it recommends selecting high-risk patient groups and the predefined triggers are consistent with the contributing factors to PSIs.

Section 6.3 listed a range of improvement methods the study participants were aware of or already using. The implementation of the TRM may potentially be facilitated through its association with some of these tools. However, there is also the possibility that implementation may be hindered if participants are unable to differentiate the TRM from their existing tools. This issue will be explored in Chapter 9. The vast majority of participants also indicated that additional resources would be required as a critical prerequisite before they would consider participating in future improvement initiatives. This issue will be considered in Chapters 9 and 10. Section 6.4 described the prevailing safety culture in general practice in Scotland, which was perceived as 'positive'. The NPT work of 'internalization' is facilitated when interventions and practice cultures are compatible. The TRM should therefore be of interest to clinicians and staff. This issue will be considered in more detail in Chapter 9.

The final section described four different perspectives from the international patient safety literature about the contributing factors to PSIs. Each perspective helps to inform the selection of specific strategies to improve the safety of care. For example, the ETTO principle helps to explain the perception of many participants that some PSIs occur by 'chance', and suggest increasing the resilience of clinicians, systems and organisations.

In summary, this chapter provided an overview of the perceptions study participants have about patient safety. More specifically, it described what participants perceived as important patient-safety related problems; whether they believed the issues were amenable to change and their responsibility to deal with; and what they were already doing about potential safety threats. It also considered the prevailing cultures in the practice teams as an important determining factor for successfully implementing complex health care interventions.

The main findings are that the all participants considered patient safety an important and integral part of their 'jobs'. They were aware of many predisposing factors to PSIs and conceded that there are potential improvements that could be made in their systems and procedures. The implications are that the TRM should be: acceptable to the study participants and compatible with their perceptions about patient safety; compatible with the prevailing cultures within the teams; and suitable for contextual integration in the general practice setting. Chapters 7, 8 and 9 will present evidence in support of these assumptions.

Chapter 7. Main outcomes from implementing the TRM

This chapter describes the results that relate to the study aim of determining the usefulness of the TRM by describing the outcomes from its implementation. The main source of data for this chapter was the Trigger Review Summary Sheets (SS). The data were analysed using the statistical methods described in Chapter 5.

Introduction

Chapter 3 introduced the concept of complex health care interventions and their three main components: 'actors', objects' and 'contexts'. The results that were presented in the previous chapter mainly related to the 'actor' and 'context' components. This chapter will focus mainly on the third component - the 'object', which in this study is the TRM - how it was enacted and the outcomes that resulted from this. From a NPT perspective, the main work that will be considered is that of 'collective action' and in particular its component of 'interactional workability'. In other words, how did the clinician reviewers apply the TRM, were they able to detect triggers and patient safety incidents (PSIs) and what actions (if any) did they subsequently take?

This chapter has four main sections. The first section describes the characteristics of the clinician reviewers and participating general practices, the types of electronic patient records that were selected for review and the number of completed Trigger Review Summary Sheets (SS) that were submitted during the study period. The second section lists the number and types of predefined triggers. The third section reports the number of detected patient safety incidents (PSIs) and their severity and preventability ratings. The different types of PSIs are then described and a preliminary classification of the main types is provided. In the fourth and final section of this chapter the actions and intended actions of the reviewers and their teams during and after the trigger reviews are discussed.

Throughout this chapter the main findings are discussed when they are reported in order to raise a number of directly relevant issues and to compare them with the patient safety evidence-base where applicable. In addition, the main findings are compared with the aggregated data from the general practices in three Scottish NHS Health Boards who implemented the TRM subsequent to this study as a Quality and Outcomes Framework (QOF) requirement for the financial year April 2013 to March 2014 (48). This data will henceforth be referred to as the 'QOF study' and provide additional context for interpreting and discussing the main findings of this study.

7.1. Characteristics of clinician reviewers and trigger reviews

7.1.1.Reviewers

A total of 47 clinician reviewers participated in the study. The professional roles of the reviewers were: general practitioners (n=12), practice nurses (n=11), a community pharmacist (n=1), GPSTs (n=22) and a nurse practitioner (n=1). They were recruited through two different strategies, as described in Chapter 5. The first strategy recruited 26 clinician reviewers from ten NHS Greater Glasgow and Clyde (Health Board 'A' - HBA) and two NHS Ayrshire and Arran Health Board (Health Board 'B' - HBB) general practices and exceeded the initial aim of recruiting two reviewers from each of twelve practices. The characteristics of the twelve participating practices, the reviewers and the number of Trigger Review Summary Sheets (SS) they submitted are summarized in Table 7.1.

Through the second strategy, 25 GPSTs from the West of Scotland region were recruited and all of them attended the Trigger Review training sessions. 21 (84%) of the participants subsequently implemented the TRM and submitted SS that were suitable for analysis.

The reviewers were classified into three main groups according to their professional roles. They are: (i) a GP group (n=12); (ii) a GPST group (n=22); and (iii) a nursing group (n=12). The data from the community pharmacist's reviews were included in the overall analyses but not in the group analyses.

7.1.2. Trigger Review Summary Sheets

A total of 67 SS were submitted, comprising 1659 individual electronic patient records. Clinicians from the general practice teams reviewed 1139 individual patient records and submitted 46 SS (see Table 7.1). The West of Scotland GPST group reviewed 520 individual patient records and submitted 21 SS. A three-month period was reviewed in each of 1659 individual electronic patient records.

Data were recorded and summarized by the clinical reviewers along with their actions, intended actions, learning points and needs, reflections and feedback on the trigger review summary sheets (SS) that were provided.

The expected total number of SS for the study had been 73 - 48 from the practice teams (four SS from each of the twelve practices) and 25 from the West of Scotland GPSTs (one SS per GPST). The overall response rates for submitted SS were therefore 91.8% (67/73), with response rates of 96% and 84% from the practice teams and GPST groups respectively.

The evaluation of the subsequent implementation of the TRM as part of the QOF for the financial year from 2013 to 2014 involved analysis of 755 submitted SS. The SS summarized the findings from reviewing a total of 18826 individual patient records from three NHS Health Boards in Scotland. The overall response rate for the general practices from the three Boards could not be calculated from the available data, but the response rate for Health Boards 'A' and 'B' were 66.1% and 88.2% respectively.

The response rate in this study was higher than in the QOF study. There are a number of possible reasons for this finding. The practices and most of the reviewers in this study were essentially a self-selected group who were highly-motivated to implement the TRM. The vast majority of the practices and many of the reviewers had participated in previous research projects and improvement initiatives. The participants invariably described their prevailing culture as proactive and took pride in being 'early adopters'. In addition, participants were offered one-to-one, tailored training in the use of the TRM at a time and place of their choice and had direct access to external 'expert' support. They were also offered a slightly more generous financial remuneration for their time. These facilitating factors will be considered in more detail in chapter 9.

Practice	Patient	GPs in pra	ctice (n)	A	Training	Clinician	SS	
no	list (n)*	Partners	Other	Area	practice	reviewers (n)	(n)	
1 2100		1	-	Suburban	No	1 PN	1	
						1 GP	4	
2	4300	3	1 salaried	Urban	Yes	1 PN	4	
						1 GP	4	
			1 salaried					
3	3200	1	1 long-term	Urban	No		4	
			locum			1 Gr		
4	4100	3	1 Retainer	Urban	Yes	1 PN	4	
						1 GP	4	
5	11000	8	-	Suburban	Yes	1 PN		
						1 GP	4	
						1 Nurse	4	
						practitioner		
6	5900	4	1 Salaried	Urban	Yes	1 PN	з	
						1 GP	J	
7	8200	7	_	Urban	Voc	1 Community		
7	0200	,	_	Orban	163	pharmacist	4	
						1 GP		
8	6800	3	2 Salaried	Urban	Yes	1 GP	4	
						1 PN	7	
9	6400	3	1 Salaried	Urban	No	1 PN	٨	
						1 GP	т	
10	9900	6	1 Retainer	Urban	Yes	1 PN	Л	
						1 GP	-	
11	3000	4	1 Retainer	Inner City	Yes	1 PN		
						1 GP	4	
						1 GPST		
12	7500	6	1 Salaried	Urban	Yes	1 GP	R	
						1 PN	J	
All	72400	49				26	46	

Table 7.1. Characteristics of the participating general practices

*rounded to nearest 100

Abbreviations: GP=General practitioner; PN=Practice nurse; GPST=GP specialty trainee; SS=Summary Sheet

The vast majority of trigger reviews (57/67, 85.1%) were performed by applying the TRM to the records of the specific 'high risk' patient group that had been recommended, i.e. patients aged >75 years and with confirmed cardiovascular disease. Reviewers from the GP, nursing and GPST groups indicated selecting this high-risk patient group on 22/24 (91.7%), 15/19 (78.9%) and 18/22 (81.8%) SS respectively. The differences between the reviewer groups were not statistically significant (chi-square 1.518, DF=2, p=0.151). Examples of alternative patient groups the remaining reviewers selected were: 'patients prescribed warfarin'; 'patients with COPD'; and 'patients seen in consultation in March'. One reviewer expressed an interest in the management of patients with diabetes mellitus and therefore applied the TRM to a selection of records from that patient group.

7.1.4. Review time

The mean reported time reviewers in the study required to conduct a trigger review and complete a SS was 138.3 minutes (min), SD 48.3, range 60 to 240. The mean times of the GPST, nursing and GP groups were 148.4 min/SS (SD 49.6, range 60 to 240), 140.7 min/SS (SD 52.3, range 65 to 240) and 126.3 min/SS (SD 46.7, range 70 to 240) respectively. The difference between the group means was not statistically significant (p=0.351, DF=2, Sum of squares 5159.8 and F-ratio 1.068).

An alternative way to quantify the reported time requirements would be that reviewers required, on average, approximately five minutes to review a single patient record. This is well short of the recommended maximum of 20 minutes per record. It is also considerably less than a trigger review study in ambulatory primary care in the USA in which reviewers required approximately 20 minutes to screen each record (322). However, the different study methods mean the results are not directly comparable.

The mean time to conduct a trigger review and complete a SS for reviewers from the QOF study was 165.7 min (SD 62.1, range 30 to 300, n=656/755 SS). The difference of 27.4 min is significantly longer than reported by reviewers in this

study (p=0.001, T=3.3, df=712, 95% CI -43.8 to -11.0, standard error of difference 8.371). In other words, reviewers in this study spent on average approximately two hours conducting a trigger review and completing the associated SS, while reviewers in the QOF study reported taking about two and a half hours. It is unclear what the practical implication (if any) of this difference may be or the reason for it. One potential explanation may be the difference in training between reviewers in this study (face-to-face) and the QOF study (in groups). The reported difference intuitively seems small, but the issue of time (or lack thereof) was repeatedly and strongly identified as a key barrier to the implementation of the TRM. This issue will be considered again in more detail in Chapter 9.

A small minority of reviewers indicated requiring more than four hours to complete the trigger reviews. It is possible they may have overestimated the actual time required, but this seems unlikely as there were no formal or informal incentives or assessments associated with the duration of reviews. It seems more likely that this finding is reflective of the characteristics of individual reviewers. Alternatively, it may indicate the need for further training to re-emphasize that the TRM should be structured and focused. It is also possible that some trigger reviews, PSIs and further actions are more complex than others and therefore required additional time.

7.2. Triggers

Overall, a total of 1407 triggers were detected, with a mean of 21.0 per Summary Sheet (SD 13.7, range 0 to 66). Reviewers from the QOF study detected a mean of 17.5 triggers/SS (SD 11.9, p=0.023). The GP, nursing and GPST groups detected a mean number of triggers of 24.3 (SD 15.1), 15.4 (SD 9.3) and 21.4 (SD 14.6) per SS respectively. The comparative differences between calculated means were not statistically significant (p=0.104). The most commonly detected trigger was 'repeat medication item discontinued' (n=290, mean 4.3/ SS, SD 4.4, range 0 to 18). The frequencies, proportions, means, standard deviations and comparison of means between groups and between this study and the QOF study are shown in Table 7.2.

	This study														
Trigger	G	GP		Nursing		РST	Comparison of means (ANOVA)	Total*		QOF study		Comparison of means		Comparison of proportions	
	n (%)	Mean (SD)	n (%)	Mean (SD)	n (%)	Mean (SD)	Sum of squares, F-ratio (p-value) and DF=2	n (%)	Mean (SD)	n (%)	Mean (SD)	t-value (p-value) and DF=820	95% Cl (Std error of diff)	Difference in proportion (%), 95% Cl, chi-square (p- value) and DF=1	
	74	3.1	40	2.1	121	5.5	128.1, 64.1	249	3.7	1348	1.9	5.2	1.1 to 2.5	7.5, 5.5 to 9.7, 72,4	
>3 consultations	(12.7)	(3.2)	(13.7)	(2.3)	(25.7)	(6.2)	(0.037)	(17.7)	(4.4)	(10.2)	(2.5)	(<0.001)	(0.345)	(<0.001)	
New high priority	128	5.3	55	2.9	75	3.4	71.3, 3.073	271	4.0	2185	3.1	2.3	0.1 to 1.7	2.7, 0.6 to 5.0, 6.421	
clinical code	(22.0)	(3.9)	(18.8)	(2.5)	(16.0)	(3.5)	(0.053)	(19.3)	(3.5)	(16.6)	(3.1)	(0.025)	(0.400)	(0.011)	
Allergy read code	19	0.8	12	0.6	22	1.0	1.6, 0.461	59	0.8	310	0.5	2.8	0.1 to 0.5	1.9, 0.9 to 3.1, 18.178	
Allergy read code	(3.3)	(1.7)	(4.1)	(1.0)	(4.7)	(1.1)	(0.633)	(4.2)	(1.3)	(2.3)	(0.8)	(0.006)	(0.109)	(<0.001)	
Repeat medication	146	6.1	71	3.7	64	2.9	127.7, 3.429	290	4.3	2458	3.4	2.3	0.2 to 1.7	2.0, -0.2 to 4.3, 3.203	
item discontinued	(25.0)	(6.0)	(24.3)	(3.0)	(13.6)	(2.8)	(0.039)	(20.6)	(4.4)	(18.6)	(3.0)	(0.025)	(0.400)	(0.074)	
OOH / A&E	96	4.0	52	2.7	70	3.2	18.7, 1.817	229	3.4	2335	3.2	0.4	-0.7 to 1.1	1.4, -0.7 to 3.4, 1.626	
attendance	(16.5)	(2.0)	(17.8)	(2.3)	(14.9)	(2.5)	(0.171)	(16.3)	(2.8)	(17.7)	(3.7)	(0.666)	(0.463)	(0.202)	
	75	3.1	34	1.8	66	3.0	21.2, 2.148	178	2.7	2461	3.4	1.5	-1.6 to 0.2	6.0, 4.0 to 7.8, 30.444,	
Hospital admission	(12.9)	(2.0)	(11.6)	(2.0)	(14.0)	(2.6)	(0.125)	(12.7)	(2.2)	(18.7)	(3.7)	(0.128)	(0.459)	(<0.001)	

Table 7.2. Number, type and comparison of triggers detected by reviewers in this study and the QOF study

Hb < 10	8	0.3	4	0.2	19	0.9	6.1, 3.706	32	0.5	365	0.6	1.0	-0.3 to 0.1	0.5, -0.5 to 1.3, 1.008
	(1.4)	(0.5)	(1.4)	(0.5)	(4.0)	(1.4)	(0.030)	(2.3)	(0.9)	(2.8)	(0.8)	(0.332)	(0.103)	(0.315)
Optional trigger(s)	37	1.5	24	1.3	33	1.5	20.5, 0.081	99	1.5	1731	2.3	2.1	-1.6 to -0.0	6.1, 4.5 to 7.5, 42.663
	(6.3)	(1.3)	(8.2)	(1.2)	(7.0)	(2.6)	(0.923)	(7.0)	(1.8)	(13.1)	(3.1)	(0.038)	(0.384)	(<0.001)
Total	583	24.3	202	15.4	470	21.4	855.5, 2.352	1407	21.0	13103	17.5	2.3	0.5 to 6.5	
	203	(15.1)	272	(9.3)	470	(14.6)	(0.104)	1407	(13.7)	3.7)	(11.9)	(0.023)	(1.535)	

*Includes data from a community pharmacist

Abbreviations: OOH=out of hours; A&E=Accident and Emergency

There are at least two explanations why the 'repeat medication item discontinued' trigger was detected the most. The first is the relatively high incidence of polypharmacy and multimorbidity in the high-risk patient populations that were selected. The second reason is that during the study period a number of medication reconciliation initiatives were underway locally and involved the participating general practices. Two examples of medication changes that were being promoted at that time were: (i) decreasing the dose of simvastatin or prescribing an alternative lipid-lowering drug for patients concurrently prescribed amlodipine; and (ii) changing omeprazole to lansoprazole for those patients concurrently prescribed clopidogrel.

The second most common trigger in this study (and the third most common in the QOF study) was 'new high priority clinical code added'. This finding may indicate the increasing prevalence of chronic disease detection; the increasing workload of general practices; and the fact that coding clinical conditions have become a routine part of day-to-day work because of the QOF. From an NPT perspective, it would seem that coding has been 'normalised' through its contextual integration with QOF.

The predefined trigger that was detected least in both studies was 'haemoglobin (Hb)<10' (n=32, mean 0.5/SS, SD 0.9, range 0 to 7). Reviewers were also encouraged during the training sessions to add the trigger 'prescription of a non-steroidal anti-inflammatory drug (NSAID)'. The rationale for adding this trigger was that elderly patients and patients with heart disease are at increased risk of adverse drug events if prescribed NSAIDs. In fact, Morris et al previously found NSAIDs were the drug most commonly associated with adverse drug events in UK general practice (323). However, in this study the NSAID trigger was least detected. This finding may indicate a high level of awareness of this potential patient safety threat amongst the clinical workforce and might provide indirect evidence of improved, safer prescribing.

Selecting the optimal number of pre-defined triggers was discussed in Chapter 3. Despite much research, the choice of triggers essentially remain at the discretion of researchers and reviewers and require reaching a compromise between the greater sensitivity afforded by more triggers and the improved efficiency of fewer triggers. As a result, the numbers of triggers in different studies range from two (324), six (325) through to 72 (326). A study of preventable adverse drug events in primary care found the majority (60%) were detectable by four triggers only (323). In general practice, increasing the numbers of triggers may help to detect additional PSIs, but with variable and diminishing returns which are not commensurate with the increase in research/review effort this requires. Given that the intention of the TRM is to facilitate rapid record reviews, larger numbers of triggers may in fact decrease the effectiveness of this method.

The overall number of triggers that arguably provide the best compromise between feasibility and likelihood of detecting PSIs seem to be approximately ten - at least for ambulatory and primary care settings. The overall number comprises a combination of 'global' (non-specific) and specific triggers. A practical example is Kaafarani and colleagues' systematic development of a trigger tool for detecting adverse events in ambulatory surgery. Their study design used the 'gold standard' for trigger tool development and incorporated literature reviews, end-user preferences and expert opinion to derive ten triggers, half 'global' and half specific (327). A further example is Singh et al's experience with a trigger tool for identifying adverse drug events among older patients in primary care. They found nine of their 39 triggers accounted for approximately 95% of the adverse drug events they detected (322).

7.3. Patient safety incidents (PSI)

7.3.1. Incidence of PSIs

A total of 216 PSIs were detected and recorded by the clinicians in this study, with a mean of 3.2 PSIs per SS (SD 2.0, range 0 to 9). Reviewers in the QOF study recorded a total of 2753 PSIs, with a mean of 3.6 PSI/SS (SD 1.5, range 0 to 5). The difference in means between the two studies is statistically significant (t=2.0, df=840, standard error of difference=0.197, 95% CI -0.8 to -0.0, p=0.042). The GP, nursing and GPST groups respectively recorded: 75 PSIs (mean 3.1 PSIs/SS, SD1.8, range 0 to 5), 46 PSIs (mean 2.4 PSIs/SS, SD1.9, range 0-5) and 85 PSIs (mean 3.9 PSIs/SS, SD2.3, range 0 to 9). The differences between the
means of the three groups were not statistically significant (p=0.065). Selected examples of PSIs detected during the study are included in Tables 7.3 to 7.6.

Five of the 67 Summary Sheets did not have any PSIs recorded (7.5%). The number of SS without a PSI from the GP group (1/24, 4.2%), nursing group (3/19, 15.8%) and GPST group (1/22, 4.5%) was comparable (p=0.289). In the QOF study 32/755 (4.2%) of SS did not have a single PSI recorded. A comparison of the difference in proportions of SS without PSIs in this and the QQOF study were not statistically significant (Chi-square results: difference 3.3%, 95% CI -2.0 to 12.5, chi-square=0.892, DF=1 and p=0.345).

The majority of participants agreed that 'finding nothing' is a potential barrier to the implementation of the TRM. This will be considered in more detail in Chapter 9. For now, it is important to consider two different reasons why some reviewers may have failed to detect a single PSI. The first reason is that there were no PSIs to detect in their random sample of records. This argument is supported by the fact that a similar, small proportion of reviews in both studies failed to produce any PSIs. The other reason is that reviewers did not recognise or report a PSI that was in fact detectable in the record. Without external or at least additional reviews of the same sample of records it is not possible to explore this option. From a purely personal perspective it seems likely that both reasons can be true.

It is possible to express the number of detected PSIs in this study (216) as estimated rates, including: one PSI per 7.7 reviewed records; one PSI per 7.7 patients; one PSI for every 23 months of documented care; or 13.0% of the patients whose records were reviewed. While the estimated incidence of PSIs in this study may help to inform our understanding of the safety of care in general practice, it is important to recognise that *all* rates have limitations. Rates are highly dependent on the numerator. If 'PSIs' had been replaced with alternative numerators such as 'adverse drug events' or 'incidents of avoidable harm' the estimated rates in this study would have been lower. Another limitation is that estimated rates are normally calculated for specific patient populations, in this case 'high-risk, elderly patients'. They should therefore not be generalized to other patient groups and are not suitable for direct comparison with the estimated rates of error and harm in other studies. The potential value and limitations of estimated harm rates are considered in detail in Chapter 10.

7.3.2. Severity of PSIs

The proportions of PSIs - as rated by reviewers in this and the QOF study as being of severity '1' through '4' - are shown and compared in Table 7.3. The severity of the majority of PSIs in this study were rated as '1', i.e. having the 'potential for harm' (n=78 PSIs, 36.1%) or '2', i.e. leading to 'mild harm' (n=75 PSIs, 34.7%). The minority of PSIs were rated '3' or '4', i.e. judged to have caused 'moderate' or 'severe harm' (n=63 PSIs, 29.2%).

There was no significant difference in proportions of the ratings of PSI severity between this study and the QOF reviewers (chi-square 3.924, DF=3, p=0.2698). However, the severity ratings of PSIs between the three different reviewer groups in this study were significantly different (chi-square 21.259, DF=6, p=0.002). The GPST group rated fewer PSIs as having a severity of '1' (potential for harm) and were more likely to rate PSIs as '3' (moderate harm) or '4' (substantial harm) compared with the GP and nursing groups.

The key finding is therefore that the majority of detected incidents have low-tomoderate severity or are 'near misses'. This finding is consistent with the international general practice patient safety literature and independent of the research method (33-35). Two further practical examples are provided. The first example is a study of medication errors reported by family practice teams in the USA. A minority (16%) of errors resulted in adverse drug events. The authors used a 5-point scale to rate the severity of adverse events: (i) 'did not reach the patient' - 41%; (ii) 'reached patients but did not require monitoring' - 35%; (iii) 'required monitoring' - 8%; (iv) 'required intervention' - 13%; and (v) 'resulted in hospitalization' - 3% (76).

le			This study				Comparison of proportions		
Rating sca	Scale item description	Selected examples of PSIs from study	GP n (%)	Nursing n (%)	GPST n (%)	All* n (%)	study n (%)	Difference in proportion (%), 95% CI, chi-square (p-value) and DF=1	
	Severity								
1	Any incident with the potential to cause harm	Diagnostic read codes from hospital discharge (aortic stenosis, postural hypotension and osteoarthritis) were added to record, but with low priority, so not present on summary	35 (46.7)	21 (45.7)	19 (22.4)	78 (36.1)	1090 (40.4)	4.3, -2.7 to 10.9, 1.366 (0.243)	
2	Mild harm, inconvenience,	Verapamil stopped as could have been							
	further follow-up or	causing constipation, exacerbating	25	17	29	75	788	5.5, -1.1 to 12.5, 2.645	
	investigation to ensure no harm occurred	underlying gastro-intestinal problems	(33.3)	(37.0)	(34.1)	(34.7)	(29.2)	(0.104)	
3	Moderate harm: required	A delay in monitoring after an increased							
	intervention or duration for	dosage of nephrotoxic medicine leading to	8	1	19	31	454	2.4, -3.2 to 7.0, 0.667	
	longer than a day	a significant decrease in renal function with increased monitoring requirements	(10.7)	(2.3)	(22.4)	(14.4)	(16.8)	(0.414)	
4	Prolonged, substantial or	Delayed diagnosis of ischaemic heart	7	7	10	22	267	1 2 2 5 + 5 4 9 0 152	
	permanent harm, including hospitalisation	disease led to avoidable admission	(9.3)	, (15.2)	(21.2)	32 (14.8)	(13.6)	(0.696)	

Table 7.3. PSI (n) severity and preventability ratings recorded by reviewers

Preventability

		77 year old with IHD and paroxysmal AF admitted to hospital for elective						
1	Not preventable and originated in secondary care	pacemaker insertion. Post procedure, required two courses of antibiotics and repeated contacts with GP surgery due to medication side effects and slow recovery	6 (8.1)	7 (16.7)	5 (5.9)	20 (9.6)	379 (14.2)	4.6, -0.4 to 8.5, 3.044 (0.081)
2	Preventable and originated in secondary care OR not preventable and originated in primary care	Elderly patient did not have her vitamin B12 injection in over 6 months. The receptionist contacted the patient's daughter and asked her to make an appointment, but this was never done	23 (31.1)	7 (16.7)	41 (48.2)	74 (35.6)	995 (37.3)	1.7, -5.5 to 8.5, 0.172 (0.679)
3	Potentially preventable and originated in primary care	Failure to initiate prophylactic treatment for or follow up a patient with gout resulted in a hospital admission	20 (27.0)	15 (35.7)	22 (25.9)	59 (28.4)	732 (27.4)	1.0, -5.3 to 7.9, 0.0532 (0.818)
4	Preventable and originate in primary care	Patient given inappropriate dosages of anti-diabetic medication with resultant renal injury	25 (33.8)	13 (31.0)	17 (20.0)	55 (26.4)	562 (21.1)	5.3, -0.8 to 12.0, 2.907 (0.088)

*Includes a community pharmacist

The second example is a mixed-method study of adverse events in general practice in the Netherlands. Adverse events were detected prospectively by GPs and through a retrospective record review. Approximately half of the events were not associated with any harm, which is comparable with a severity rating of '1' in this study. Approximately a third of the detected adverse events led to worsening of symptoms (severity ratings '2' and '3') while a few resulted in hospital admissions (severity rating '4'). However, the majority of events were associated with the risk of harm (328). The implication of these findings is therefore that improvement efforts in general practice should focus on reducing the risk from avoidable PSIs irrespective of the severity of harm associated with them.

There are at least three reasons why the TRM is more likely to detect PSIs associated with no or mild harm rather than with moderate or severe harm in general practice. The most obvious reason is that there may simply be fewer incidents associated with moderate to severe harm to detect. The second reason is that trigger reviews are conducted on relatively small samples of records and focus on specific periods of care only. The potential windows through which PSIs may be detected are therefore relatively narrow. The third reason is that more serious incidents are less likely to go undetected. Depending on the severity of harm patients experience, they may re-present to the practice, attend accident and emergency, be admitted to hospital or make formal complaints, thereby alerting clinicians of a potential problem.

Even in those few instances when PSIs with moderate to high severity are detected in general practice, they often originate in secondary care. Three clinical examples would be: patients presenting to their GP with urinary sepsis after being discharged from hospital with an indwelling catheters; patients presenting to their GP with phlebitis after intravenous cannulation during their hospital admissions; and patients with post-operative wound infections presenting to primary care clinicians. This does not necessarily imply that the majority of hospital-related PSIs have serious consequences for patients. The proportions of detected PSIs associated with moderate to severe harm may be relatively higher in hospitals than in general practice when the results of individual trigger reviews are compared. However, the majority of hospital PSIs is still associated with low to moderate harm. A practical example is Forster et al's study to detect adverse events in an obstetric service through trigger-based clinical surveillance. They found serious adverse events occurred 'infrequently' and there were no instances of permanent harm or death. However, they commonly found 'quality problems' which they felt could (and should) be addressed (326).

In other words, the perceived severity of PSIs should not (and do not) automatically correlate with their usefulness in relation to informing subsequent improvement actions. Despite the relative low impact of the majority of PSIs, they still provide many opportunities to tackle issues that can be addressed by the practice team or a clinician reviewer (326, 329). However, many participants associated more severe PSIs with relatively higher importance and usefulness. From an NPT perspective this issue has to be addressed by doing the work of coherence in order for clinicians to understand that the value of PSIs should be measured by their potential to inform subsequent improvement actions rather than the severity of harm associated with them. This work was performed to some degree through the TRM training sessions. Still, participants' perceptions of the importance of the severity of PSIs strongly influenced the implementation of the TRM and will be discussed in more detail in chapter 9.

7.3.4. Preventability of PSIs

The potential preventability of detected PSIs ranging from '1' through '4' - as rated by the reviewers in this and the QOF study - are shown and compared in Table 7.3. The preventability of 8 (3.7%) PSIs were not rated. 114 PSIs (54.8%) were rated by the reviewers in this study as being preventable or potentially preventable, compared with 1294 PSIs (48.5%) in the QOF study. There was no significant difference in the proportions of preventability between the studies (chi-square 5.745, DF=3, p=0.125). However, the ratings of PSI preventability of the three different reviewer groups in this study were significantly different (chi-square 16.300, DF=6, p=0.012). The GPST group were less likely to consider PSIs as being preventable or potentially preventable compared with the GP and nursing groups. The key finding from both studies is therefore that

approximately half of all detected PSIs may be preventable. This seems to be a consistent finding in the international literature in different health care settings (33-35).

The importance of the concept of 'preventable' harm was previously discussed in Chapter 2. This is a critical and often overlooked issue in the patient safety literature: unfortunately but inevitably some patients will be unavoidably harmed as a result of their interactions with healthcare for a range of highly complex reasons. The key focus from the perspective of patients and clinicians should therefore be detecting and learning from those incidents which are judged to be preventable.

A literature review published after this study had commenced found there is still insufficient empirical evidence to support a specific definition of 'preventable harm'. However, the most common definition is the 'presence of an *identifiable, modifiable cause of harm*' (55). The authors suggest three ordinal categories for the degree of preventability: 'definite', 'probable' and 'plausible'. These categories seem intuitively sensible and may prove to be more sensitive and useful compared with the two categories ('preventable' and 'potentially preventable') that were provided in the trigger review template in this study. However, the intrinsic challenges of subjectivity and inter-rater reliability are arguably more important than the type of rating scale reviewers use (330).

The basic preventability rating scale used in this study was specifically developed to assist participants in making professional judgements, e.g. whether identified PSIs had modifiable causes. The rating scale also encouraged them to consider whether it was feasible for them to address modifiable causes, hence the distinction between 'originate in secondary care' and 'originate in primary care'. Consequently, PSIs rated as '3' or a '4' were more likely to be amenable to corrective actions.

It is unclear whether the relatively small differences in PSI severity and preventability ratings between reviewer groups in this study had a discernible impact on their subsequent actions. The difference between the preventability ratings of the GPST group and the other two clinical groups in this study is consistent with their severity ratings and may therefore be a consequence of applying the rating systems to different PSIs, rather than indicative of disparate interpretations of key concepts. In general, more severe PSIs (as detected and rated by the GPST reviewers) tend to be associated with secondary care. From a general practice perspective, the corresponding preventability rating of '2' signals that, even if modifiable causes could be identified, it is unlikely for primary care to feasibly address them. However, the potential importance of reviewer characteristics, e.g. number of years of clinical experience, and differences in TRM training on the reliable application of the rating scales should also be considered and may be a focus for future research.

7.3.5. Classification of PSI types

The iterative development of a preliminary classification system for the different types of detected PSIs was described in Chapter 5. The nine types of PSIs are: (1) medication and medication-related activities; (2) communication and correspondence; (3) record keeping and coding; (4) care monitoring, access and continuity; (5) diagnoses; (6) medical equipment including IT; (7) investigations; (8) healthcare acquired infections (HAI); and (9) insufficient information to classify. The proportions of each of the different types of PSI are shown and compared in Table 7.4. There were significant differences between the proportions of PSIs detected by the reviewers from this and the QOF study for the following types: care monitoring, access and continuity (p=0.004), diagnoses (p<0.001) and 'unclear' (p<0.001).

PSI types	Selected examples of PSIs from this study, taken verbatim from the submitted Trigger review summary sheets	This study n (%)	QOF study (2 Boards) n (%)	Comparison of proportions Difference in proportion (%), 95% CI, chi-square (p- value) and DF=1
Medication	Two different strengths of co-codamol prescribed for patient. [Also]	88	650	6.0, -1.0 to 13.2, 2.795
	on repeat NSAID - elderly patient with renal impairment.	(40.7)	(34.7)	(0.095)
Communication	Bisoprolol increased from 5mg to 10mg at admission. Medication not	12	108	0.2, -3.9 to 3.1, 0.0012
	altered on repeat script. Increase NOT highlighted by secondary care (and CCU and Ward doctors different [doses])	(5.6)	(5.8)	(0.972)
Record keeping and coding	Same diagnosis recorded twice, discharge letter filed 3 times with	18	145	0.6, -3.0 to 5.3, 0.0316
	different dates - potential for confusion regarding dates	(8.3)	(7.7)	(0.859)
Care monitoring, access and	Elderly lady with heart failure on ramipril. Bloods not done for 2	55	324	8.2, 2.3 to 14.8, 8.223
continuity	years. Letters sent but not attended	(25.5)	(17.3)	(0.004)
Diagnoses	Chest x-ray 'non-specific opacity' - repeat in two months. Patient presented one week later with nausea, then dysphagia. Oesophageal CA. Died	16 (7.4)	46 (2.5)	4.9, 1.7 to 9.3, 14.307 (<0.001)

Table 7.4. Preliminary classification of types of PSIs (n)

PSI types	Selected examples of PSIs from this study, taken verbatim from the submitted Trigger review summary sheets	This study n (%)	QOF study (2 Boards) n (%)	Comparison of proportions Difference in proportion (%), 95% CI, chi-square (p- value) and DF=1
Investigations	Unlabelled specimen bottle sent for FBC check	6 (2.8)	54 (2.9)	0.1, -3.2 to 2.1, 0.0173 (0.895)
Medical equipment including IT	Female patient age 88, terminal Ca. Problems with syringe driver. No number for district nurses. Pump alarming - unable to fix	2 (0.9)	15 (0.8)	0.1, -0.9 to 2.5, 0.0603 (0.806)
Insufficient information to classify PSI	Anaemic - required transfusion	17 (7.9)	512 (27.4)	19.5, 14.6 to 23.3, 37.857 (<0.001)
Healthcare acquired infection	83 year old developed chest infection and cellulitis after discharge from [hospital] post hip dislocation	2 (0.9)	18 (1.0)	0.1, -2.3 to 1.1, 0.0495 (0.824)

The majority of detected PSIs were of the 'medication' type in both this (n=88 PSIs, 40.7%) and the QOF study (n=650 PSIs, 34.7%). A selection of the ten most common types of medications implicated in PSIs and their relative frequencies are shown in Table 7.5. The most common medication involved in this and the QOF study was Warfarin and seven of the top ten medications were the same in both studies. The key findings of this study are consistent with the international literature: the most common types of PSIs in general practice are related to medication and medication-related processes; and a small number of medications are associated with the majority of medication-related adverse events (77, 78, 83, 87, 88). A systematic review of the international literature (n=19 studies) was recently conducted to study the incidence of inappropriate medication prescribing (IMP) to elderly patients in primary care (331). The authors found a median rate of IMP among the elderly of 20%. The four most commonly prescribed inappropriate medications were from the following groups: analgesia (propoxyphene, a type of NSAID); antihypertensives (doxazosin); antihistamines (diphenhydramine); and anti-depressants (amitriptiline). The availability of these drugs differs widely between countries and the findings can therefore not be directly generalized to Scotland. However, it is striking that six of the ten types of medications identified as high-risk in this study also belong to the analgesia and antihypertensive groups.

7.4. Actions and intended actions of the reviewers and practice teams

This section describes the actual and intended actions of reviewers and their practice teams as a result of the trigger review process. These actions can be divided into two distinct groups, depending on their temporal relation to the trigger reviews. The first group includes any action that was undertaken while the trigger reviews were being conducted. These types of actions do not necessarily require detection of PSIs. For example, a clinician reviewer may have updated the clinical codes for a patient while reviewing the record or clarified a clinical entry. The second group includes actions intended or undertaken after the trigger reviews were concluded. These actions typically required detection of one or more PSIs.

Medications	Selected examples of PSIs from this study, taken verbatim from the submitted Trigger review summary sheets		This study		study ards)	Comparison of proportions
		n (%)	Rank	n (%)	Rank	Difference in proportion (%), 95% CI, chi-square (p-value) and DF=1
'Top 10' medications, This st	udy					
Warfarin	Patient's INR > 5 after Warfarin dose changed. Patient unsure of what dose he was meant to take and didn't know what he had been taking	10 (11.2)	1	139 (19.5)	1	8.3, -0.6 to 14.8, 3.075 (0.080)
Calcium Channel Blockers	Ankle swelling secondary to amlodipine. No allergy coded	9 (10.1)	2	13 (1.8)	13	8.3, 2.8 to 16.6, 17.555 (<0.001)
ACE / ARB	Trial of ramipril 1.25mg started. Patient already on ramipril 10mg which had been withheld but not discontinued on repeats. Nurse at an appointment printed 1.25mg from 'acutes' and also all repeats, including Ramipril 10mg. Patient noticed and called for advice	8 (9.0)	3	45 (6.3)	7	2.7, -2.7 to 10.9, 0.549 (0.459)
Diuretics	Diuretics discontinued completely instead of dose reduced	7 (7.9)	4	47 (6.6)	5	1.3, -3.8 to 9.2, 0.0561 (0.813)
NSAIDs, including aspirin	Chronic cardiac failure admission. Had been getting as required diclofenac. Apparently told that prn ok by cardiology. May have exacerbated underlying heart failure	6 (6.7)	J5	52 (7.3)	4	0.6, -7.0 to 5.3, 0.000122 (0.991)
B-blockers	Given atenolol 100mg by chemist instead of allopurinol 100mg. Already on metoprolol - felt unwell. Checked by GP stopped himself	6 (6.7)	J5	15 (2.1)	12	4.6, 0.2 to 12, 4.896 (0.027)

Table 7.5. The 10 most common medications associated with PSIs

Opiates, e.g. codeine, morphine, tramadol	Injured back - was given co-codamol. Co- codamol made her sick.	5 (5.6)	J7	27 (3.8)	8L	1.8, -2.3 to 8.9, 0.281 (0.596)
DMARDs including methotrexate	Female patient age 76. GP discovered her medication had not been collected for 5 months. This was missed when patient attended for routine monitoring bloods. Resulted in flare up of arthritis. Pt stopped due to side effects which could have been easily treated with an additional tab.	5 (5.6)	J7	78 (11.0)	2	5.4, -1.9 to 9.9, 1.930 (0.165)
Antidepressants and antipsychotics	Prescribed citalopram when already on sertraline. GP did not check. Patient did not start and phoned her usual GP	5 (5.6)	J7	20 (2.8)	10	2.8, -1.3 to 9.9, 1.233 (0.267)
HMG-CoA reductase inhibitors (statins)	Patient with GI upset following prescription of simvastatin. Adverse reaction to simvastatin not coded and simvastatin restarted 18 months later as patient's cholesterol noted to be elevated at CHD annual review.	4 (4.5)	10	9 (1.3)	J15	3.2, -0.3 to 9.9, 3.203 (0.074)
'Top 10' medications, QOF st	udy					
Unspecified	Ordering medication every month and getting 2 months' supply. Why has no-one noticed this, and why has the pharmacist not noticed this?	2 (2.2)	J12	56 (7.9)	3	5.7, -0.2 to 8.7, 3.015 (0.083)
Antibiotics	Diarrhoea following antibiotic treatment for chest infection. This worsened haemorrhoids. Had further 2 courses in Nov & Dec. Referred to surgeons for haemorrhoids	2 (2.2)	J12	46 (6.5)	6	4.3, -1.6 to 7.1, 1.880 (0.170)
Oral anti-diabetic drugs	Proteinuria as code, taking incorrect glicazide and metformin dose.	2 (2.2)	J12	27 (3.8)	8L	1.6, -4.2 to 4.2, 0.212 (0.645)

*Abbreviations: INR=international normalised ratio; ACE=Angiotensin-converting enzyme (ACE) inhibitors ; ARB=Angiotensin II receptor blockers ; NSAID=Non-steroidal anti-inflammatory drugs ; DMARD=Disease-modifying anti-rheumatic drugs ; J=Joint.

7.4.1. Actions undertaken during trigger reviews

Clinicians in this study indicated undertaking specific actions relating to patient care and practice processes in 44/63 SS (69.8%). Reviewers in the GP, GPST and nursing groups indicated performing one or more actions while conducting reviews in 21/24 (87.5%), 11/20 (55.0%) and 12/19 (63.2%) SS respectively. The differences are statistically significant (Chi-square statistic 13.431, DF=2, p=0.001).

Reviewers in the QOF study recorded undertaking one or more actions during the trigger reviews in 654/755 (86.6%) of the SS. This is comparable to the GP group in this study but significantly more than the nursing and GPST groups (difference in proportions=16.8%, 95% CI 5.6 to 29.9%, Chi-squared 11.787, DF=1 and p<0.001). Essentially all reviewers in the QOF study were GPs, which is probably why their proportions of actions undertaken during reviews were comparable with those of the GP group in this study.

The implication of this finding is that GP reviewers are more likely than other clinical groups to take some kind of action relating to patient care during trigger reviews. This is understandable as GPs would typically have more clinical autonomy and a wider range of responsibilities for care delivery compared with GPSTs and practice nurses. It should be recognised though that GPST and nurse reviewers are able to make meaningful contributions to improvement efforts as borne out by the fact that more than half of them reported taking some patient care related action while conducting their reviews. However, at the very least this finding raises the question of who ideally should be performing trigger reviews in general practice to help maximize its potential for improving care. This question will be considered again in Chapters 9 and 10.

The actions reviewers undertake during trigger reviews can be classified into one of four main types, depending on whether they relate to: (i) record keeping and coding; (ii) medication; (iii) communication, e.g. with the patient, carers or other health care providers and could be telephonic, face-to-face or via correspondence; and (iv) providing or arranging additional care such as

193

investigations or consultations. Each of these has a number of more specific actions that are summarized and illustrated with selected quotes in Table 7.6.

7.4.2. Actions and intended actions after trigger reviews

A total of 203 specific actions were taken or intended by reviewers and their practice teams after the trigger reviews were completed. These actions were typically prompted by the PSIs they detected. A mean of 3.1 actions or intended actions were taken or intended after completion of each trigger review in this study (SD 4.2, n=65/67, range 0 to 27). The means for the GP, nursing and GPST groups were 2.7 (SD 3.1, n=24/24, range 0 to 12), 1.8 (SD 1.8, n=19/19, range 0 to 7) and 5.2 (SD 6.0, n=20/22, range 0 to 27). The differences in group means were statistically significant (DF=2, sum of squares 123.2, mean squares = 61.6, F-ratio = 3.837 and p=0.027s) with post-hoc tests finding significant differences between the GPST and GP (p=0.046) and GP and nursing (0.012) groups. The specific types of actions undertaken after the trigger reviews are summarized and compared in Table 7.7.

Reviewers and practice teams from the QOF study indicated taking or intending to take a total of 3124 actions after completing their trigger reviews, with a mean of 4.1 actions or intended actions per trigger review (SD 3.4, n=755/755, range 0 to 21). This was significantly more than in this study (difference of means=1.0, standard error 0.45, 95% CI 0.1 to 1.9, t-value 2.230, DF=818, p=0.026).

However, there were no statistically significant differences when each specific action is compared, apart from 'discuss with educational supervisor', which is expected, given this study had a much larger proportion of GPST reviewers than the QOF study. The difference in reported actions between the two studies can be partly explained by small but potentially significant adaptations that were made to the SS between this study and the subsequent QOF study. The additional option of 'other' was added in 'step two' of the SS and the 'discuss with educational supervisor' was changed to 'formal submission of incident' in at least one Health Board. In other words, reviewers in the QOF study had more options to record their actions or intended actions. However, it is unclear

Main type of action	Specific examples of actions	Selected quotes from Trigger Review SS
Record keeping and coding	Update coding (disease registers, QOF, medical history of clinically important procedures and diagnoses)	Low priority codes from recent discharge were upgraded to 'active' so they are present on the patient summary (GP02) Add code 'deteriorating renal function' to patient (with declining eGFR) record (PN01)
	Correct coding / records	Duplicate code removed (GP11)
	Add a clinical entry, allergy or set a task 'reminder'	Coded [that] ACE-inhibitor not tolerated (GP05)
Actions related to medication	Changes to medication items (commence new drugs, change or discontinue current drugs)	A handwritten note from secondary care outpatient clinic was overlooked. Request to start vitamin D as adcal D3 not tolerated. Contacted pharmacy to see what cholecalciferol preparations are available and commenced [patient] on it (GP02) I changed a patient's antidepressant and added it to his repeat script. Chemist and patient notified to avoid confusion (GP03)
	Perform medication reviews	Medication reviews carried out on all patients reviewed (GP08) Two medication reviews updated (GP04) Medication reviews done / medication adjustments made (GPST9)

Table 7.6. The types of actions reviewers undertook during trigger reviews

Communication	Clarify management plan and responsibilities with	Patient contacted about INR - district nurses requested to do [patient's] fasting
(monitoring,	other health care workers, including secondary care	glucose (PN02)
clarification, review,		
education)	Clarify patient understanding of management plan	NSAID on repeat [despite] myocardial infarction - looks like patient taking it, but not
		actually, so stopped: confirmed with phone call (GP07)
	Provision of information / education to patients	Telephoned patient about repeat blood tests, including inflammatory markers (GP02)
	Confirming that intended actions (e.g. scheduled	Follow up abnormal chest x-ray (patient had been seen at respiratory clinic) (GP11)
	monitoring) took place	Totow up ubnormal enest x ray (patient had been seen at respiratory enine) (of rry
Additional care	Arrange an appointment with the reviewer or	Noted antihypertensive compliance poor. Last BP nine months ago was high. Not
(Investigations and	another practice team member	ordered calcium channel blocker for four months. Called [patient] in early for home
follow-up)		blood pressure review diary and entry in case notes to query at review (GP02)
		Called patient overdue annual review (GP12)
	Refer to another health care provider	[I send a] letter to cardiology and to a patient (GP10)
	Arrange repeat or further investigation for a	>14 point drop in eGFR not noted three weeks ago. Distracted by the increased glucose
	patient	and new diagnosis of type two diabetes mellitus. [Patient] 92 [years old] with
		dementia and frail. Arranged bloods [to be] rechecked (GP02)

	This study					Comparison of proportions	
Description of action	GP	Nursing	GPST	All ^{\$}	QOF study	Difference in proportion (%), 95% Cl,	
	n (%)	n (%)	n (%)	n (%)	n (%)	chi-square (p-value) and DF=1	
Significant event analysis	10 (15.4)	3 (8.8)	12 (11.5)	25 (12.3)	372 (11.9)	0.4, -4.0 to 5.8, 000344 (0.953)	
Clinical audit	5 (7.7)	2 (5.9)	5 (4.8)	12 (5.9)	173 (5.5)	0.4, -2.5 to 4.7, 0.00692 (0.934)	
Make a specific improvement	5 (7.7)	2 (5.9)	7 (6.7)	14 (6.9)	352 (11.3)	4.4, -0.1 to 7.7, 3.325 (0.068)	
PDSA cycle	1 (1.5)	0 (0.0)	1 (1.0)	2 (1.0)	61 (2.0)	1.0, -1.6 to 2.0, 0.546 (0.460)	
Feedback to colleagues	29 (44.6)	14 (41.2)	28 (26.9)	71 (35.0)	1265 (40.5)	5.5, -1.7 to 12.3, 2.176 (0.140)	
Discuss with Educational Supervisor	0 (0.0)	3 (8.8)	27 (26.0)	30 (14.8)	2 (0.1)	14.7, 10.1 to 20.4, 402.564 (<0.001)	
Add to appraisal documentation	8 (12.3)	6 (17.7)	12 (11.5)	26 (12.8)	496 (15.9)	3.1, -2.4 to 7.6, 1.159 (0.282)	
Protocol update	7 (10.8)	4 (11.8)	12 (11.5)	23 (11.3)	297 (9.5)	1.8, -2.4 to 7.0, 0.519 (0.471)	
Other**	N/A	N/A	N/A	N/A	106 (3.4)		

Table 7.7. Specific actions relating to patient care and practice processes that were taken or intended after completion of trigger reviews*

*More than one option could be selected

^{\$}Includes a community pharmacist

**The 'other' option was added to the SS after this study

whether this relative increase in reported actions in the QOF study had any additional practical value for patients compared with this study.

The most common action taken by reviewers and their practice teams in this and the QOF study was 'feedback to colleagues' and this accounted for, respectively, 35.0% and 40.5% of all indicated actions. They were least likely to apply the 'Plan-Do-Study-Act (PDSA) method'. This may reflect a lack of knowledge of this specific QI method (and indeed other QI methods such as root cause analysis) since there is no formal training on these methods within the current GP curriculum, but would not explain the relatively low numbers indicating their intention to undertake audits and SEA which are formally taught. From a systems perspective of safety improvement this may indicate a training need to help link PSIs to particular improvement methods and tools.

Conclusion

This chapter described the practical implementation of the TRM in general practice, the main findings from conducting trigger reviews and the actions and intended actions undertaken by clinician reviewers and their practice teams. The main results are summarized below:

- A range of primary care clinicians (n=47) were recruited and trained before applying the TRM to review a selection of 1659 high-risk patients' records. They recorded their main findings, actions and reflections on Summary Sheets (SS) and submitted 67 of these for analysis.
- The mean reported time required to conduct a trigger review (25 individual records) and complete a SS in the study was 138.3 minutes (SD 48.3).
- The reviewers found 1407 triggers, with a mean of 21.0/SS (SD13.7). The most common trigger was 'repeat medication item discontinued'.
- 216 PSIs were detected, with a mean of 3.2 PSIs/SS (SD 2.0). A substantial minority of these were considered to have led to moderate or more substantial harm (n=63 PSIs, 29.2%) while the majority (114 PSIs, 54.8%) were rated as being preventable or potentially preventable. However, a small minority of reviewers did not record any PSIs.

- The PSIs were classified by type using a preliminary classification. The most common type of PSI related to 'medication' (n=88 PSIs, 40.7%). The most commonly implicated drug was Warfarin.
- Reviewers undertook specific, patient safety-related actions *during* approximately two thirds of trigger reviews. The types of actions related to improving (i) record keeping and coding; (ii) medication-related processes; (iii) communication; and (iv) providing or arranging additional care for patients.
- A mean of 3.1 actions were taken by reviewers and their practice teams *after* completing their trigger reviews. The most common action was 'feedback to colleagues' but they also conducted SEAs, performed clinical audits, designed protocols, added findings to their appraisal documentation or made specific improvements to their practices.

Chapter 8. How the TRM works

This chapter describes the results that relate to the study aim of explaining how the TRM worked in practice, how it influenced study participants and helped to bring about the outcomes reported in Chapter 7. The main source of data for this chapter was the second round of interviews with the practice managers, general practitioners and practice managers. The data were thematically analysed. A summary of this chapter and its three sections are provided in Figure 8.1.

The first section describes how the TRM enabled the detection of PSIs by providing the study participants with: (i) the necessary 'knowledge' to ensure they were ready to use the method with a patient safety 'mindset'; (ii) opportunities to perform trigger reviews, and to reflect on the findings and implement changes; and (iii) a *structure* of the cognitive and practical steps required for effective implementation of the TRM.

The terms 'knowledge' and 'readiness' are commonly used and require further clarification. For the purposes of this study, knowledge is understood to 'encompass everything that individuals might know, be able to do and value'. 'Readiness' is defined as 'an individual's ability to learn from what they know, can do [their skills] and value [their attitudes]'. In other words, different yet interdependent types of knowledge (knowing; doing; and valuing) determine the 'readiness' of individuals to engage with, and learn from, their experiences in any given situation (332).

When all three of these components are present (knowledge/mindset; opportunity; and structure) the likelihood of detecting PSIs with the TRM is greatly increased. The importance of detecting PSIs in this study is that it creates potential 'learning moments' (49). Learning moments are by their very nature emergent and temporal and their value is determined by clinicians consciously or unconsciously choosing to either accept 'ownership' of PSIs or not. This pivotal decision is a key determinant of the nature and degree of impact the PSI will have and is therefore crucial to the eventual outcomes of the TRM, e.g. whether it will result in reflection, learning and improvement-related



Figure 8.1. How the TRM works and the related chapter sections

actions or not. The second section of this chapter therefore explains the importance of ownership of PSIs in more detail as well as how this concept influenced clinicians and their teams.

The third and final section describes the potential outcomes from detecting PSIs and accepting ownership for them, ranging from negative emotional and psychological experiences, through none or neutral to very positive effects. Practical examples include: clinicians and teams taking specific actions to help reduce similar PSIs in the future; increased vigilance and awareness of potential safety threats; and identifying learning needs and points.

8.1. The TRM enables the detection of PSIs

8.1.1. The TRM helps to create a patient safety 'mindset'

One of the participants explained how certain tasks created a specific mindset for her, and presumably other health care workers. As a practical example she described how, when she began the daily task of signing bundles of repeat medication prescriptions, she unconsciously went '*into prescriber mode*' (GP03). In the same way, while she was performing trigger reviews, she noticed adopting a 'trigger tool mindset'. Many other participants reported a similar experience of (unconsciously) forming a specific 'mindset' during application of the TRM which made them more vigilant and aware of potential safety-related issues. So, while the patient records were the same ones they had interacted with on a regular basis while providing routine care, when applying the TRM to them this 'mindset' made them feel and appear 'new'.

It is important to note that all reviewers who were aware of this mindset, irrespective of their clinical roles, reported that it developed gradually, over a period of time. The first time they applied the TRM the mindset developed relatively slowly. As they reviewed the consecutive sections in individual records, the 'mindset' started forming until: 'you're looking at the notes from a different view, from a different perspective' (GP02). As reviewers became more familiar with the method, a focused, patient safety mindset developed quicker: 'by the time you're on the third one [record review] it's just happening *automatically' (GP04)*. However, even with experience, reviewers still required a few minutes for the TRM mindset to form.

Once this new perspective (mindset) had been established, participants felt they were more likely to detect PSIs compared with the beginning of the review, or when they were interacting with the record for other reasons during routine care. This mindset was 'built' by reviewing sections in consecutive records in a systematic and focused manner, but could be lost or changed by distractions and interruptions. This is one of the main reasons why implementing the TRM was facilitated by reviewers being provided with sufficient, uninterrupted (protected) time to complete a trigger review as a single activity.

Participants felt these 'mindsets' were beneficial and increased their efficiency as all of their attention could be focused on the tasks in hand: '*it*'s much easier to achieve one specific aim if you're focused to doing it rather than trying to achieve a hundred different things all at the same time' (GP11). Unfortunately, while focusing on completing a specific task such as signing a bundle of prescriptions, potential safety threats and opportunities for interventions or additional care probably go unrecognised in many instances. Box 8.1 provides a practical example of a PSI that was detected during a trigger review in this study and describes its impact.

The clinician reviewer who reported the PSI subsequently reflected on the potential reasons why it may have gone undetected. She concluded that, because she had focused all of her available attention on certain clinical tasks, the unintended consequence had been that other opportunities for proactively offering care to the patient had been missed. She explained: *'I had seen that patient and I hadn't mentioned an ACE inhibitor at all and you also think well why did I never mention that? I'd have just completely excluded it, or was my mind thinking along the track of something else?' (GP03)*

It was unclear to the participants exactly how the 'mindset' formed, but they suspected an association with repetitive and routine clinical tasks: 'we're so used to doing things I think in a certain manner' (GP03). While this explanation seems plausible, an alternative or additional reason may be that participants'

Box 8.1. Example of a detected PSI and the importance of 'ownership'

A GP reviewer (GP03) identified an elderly patient with established chronic kidney disease (CKD) stage three who had not been added to the practice register and had not been offered treatment with a suitable ACE/ARB. In other words, the PSI was suboptimal treatment of a patient with CKD. This was rated as a PSI with low severity and high preventability. The reviewer was surprised at detecting this PSI because the patient had consulted with her on several previous occasions in the preceding months. Her first actions were to add him to the relevant chronic disease register, request a repeat eGFR blood test to check his renal function and arrange a review appointment to monitor his blood pressure and discuss potential further treatment. She identified a professional learning need about the management of CKD while reflecting on this incident and addressed it. The incident was discussed during a practice meeting and the team decided to update the practice protocol for the management of CKD and to perform a clinical audit of the management of their patients with CKD.

'readiness' for the TRM - as determined by their knowledge of it - helped to create the mindset. The TRM knowledge of most participants would have increased as a result of training, increasing practical experience from conducting trigger reviews and evaluating their findings. From an NPT perspective, participants' knowledge - and therefore readiness to learn from PSIs - increased through work relating to coherence and especially its 'internalization' component ('value' or 'attitude' knowledge); collective action ('doing' or 'skills' knowledge); and reflexive monitoring ('knowing' or information knowledge).

8.1.2. The TRM provides opportunities for its application and reflection

Many participants felt that one of the important characteristics of the TRM that made it useful to them was that it not only encouraged reflection and further action, but also provided opportunities to do so. The TRM initiative essentially offered general practice teams a method with potential benefits for them and their patients, an estimate of the time and resources that implementing it would require and an offer of some financial and 'expert' support. During the recruitment stage of this study, participants therefore had the opportunity to decide before the TRM was implemented whether it would be appropriate for them to be involved (or not). From an NPT perspective they had the opportunity to do the necessary work of: initiation, i.e. the preliminary work of deciding they should be involved in its implementation; internalization, i.e. understanding and agreeing whether the proposed intervention (the TRM) was compatible with their prevailing culture; and legitimisation, i.e. feeling justified in allocating and spending time and resources on TRM-related activities.

I really did think [the TRM] was valuable, cause it really does just give you that time out to review patients, just review care and systems, and you know it's something that we often don't have. (GP08)

Chapter 6 described how participants understood patient safety as an important and integral part of the care they provide, and how they perceived a professional responsibility for their patients' safety. It also described how they were able to identify a large number of potential safety threats and contributing factors to PSIs and their willingness to improve their standards of care, provided they had adequate resources to do so. A key element in the way the TRM works is therefore by providing teams with an improvement method (and supporting evidence of its potential value) *along with* some financial compensation for the required time for implementation. This in turn creates the opportunities (and legitimacy) for clinician reviewers to engage in quality improvement work. The implication from an NPT perspective is therefore that, for an intervention to be successfully implemented in general practices, it is important to ensure that the work of contextual integration is supported, i.e. providing teams with appropriate resources and adequate time. This will be considered again in Chapter 10.

The majority of participants reportedly used the opportunities for reflection to some degree, although it is difficult to quantify the true extent or measure the quality of the reflection. However, the descriptions of some clinicians of how they were able to identify learning needs or points while reflecting on detected PSIs provide some evidence of its usefulness. This is an important outcome of the TRM study and will be described in more detail in section 3 of this chapter. Other participants described how reflection helped them understand the latent risks associated with detected PSIs, how these may affect other patients, and how they may be able to proactively manage risk. A practical example is provided in Box 8.1.

The importance of clinicians being able to reflect on clinical care and their own performances is increasingly being recognised. Reflection is considered a competency in its own right and is promoted as a core professional quality in medical education (333). It is a collective term that includes 'those intellectual and affective activities in which individuals engage to explore their experiences in order to lead to a new understanding and appreciation' (334). It should be recognised that, while the TRM strongly promoted and provided opportunities for reflection, it cannot make people reflect or more reflective per se. This issue will be considered in more detail in chapter 9. The implication of this finding from an NPT perspective is therefore the importance of identifying and recruiting those clinicians who are naturally more reflective to perform trigger reviews, e.g. effectively performing the work of skill set-workability.

Another way in which the general practice TRM approach created opportunities for learning and reflection in this study was by encouraging clinicians to consider *all* PSIs as potentially important, irrespective of whether they considered them 'acts of omission' or 'acts of commission'. This is a key difference compared with the Institute for Healthcare Improvement's secondary and ambulatory Global Trigger Tools (GTT) that only record and measure 'acts of commission'. However, the authors of a recent systematic review of the GTT challenged this approach, and recommended inclusion of 'acts of omission' in all settings (242).

The decision to purposefully exclude acts of omission may be justified and considered reasonable for many study designs, as it reduces the subjectivity of the external reviewers' findings, improves inter-rater reliability and therefore also improves the reliability of harm estimates. This is less of an issue when reviews are conducted by internal reviewers where the intention is to improve their own care standard, rather than simply measure clinical performance. This issue, and the other potential strengths and limitations of the TRM will be considered in detail in Chapter 10.

In this study, a relatively large proportion of detected PSIs could be attributable to omissions of care. A practical example would be the suboptimal treatment of a patient with confirmed chronic kidney disease (CKD) who was not prescribed an ACE/ARB even though this is recommended by current evidence-based guidelines (see Box 8.1). The implication is that there are potentially many minor safety threats or quality issues detectable in general practices that either go unrecognised or are recognised but not prioritised or acted on while delivering routine care. If so, it is likely the result of significant time constraints and the need to prioritise more urgent medical concerns and the patient's agenda during consultations that are typically scheduled for a maximum of ten minutes. One clinician explained that 'unless it's something glaringly dangerous [we] might just sort of gloss over it and move on cause you don't have time' (GP02). A practical example would be an elderly patient presenting with symptoms and signs of an acute and significant respiratory tract infection. In a typical ten-minute consultation agreeing and implementing a suitable management plan is prioritised over medication reconciliation or coding. In effect, the clinician is making an ETTO - which was described in Chapter 6.

Unfortunately, these minor issues remain as latent risks in practice systems and could manifest later, potentially as serious incidents (1).

Perhaps unsurprisingly then, many of the detected patient safety incidents in this study were the result of latent safety risks or 'hazards' in the wider practice systems that were already known to many of the reviewers or their teams, but had not been addressed before because of lack of opportunity or perceived urgency. This may also be one of the reasons why the vast majority of study participants indicated they were willing to consider, search for and report PSIs (Chapter 6). In this context, the TRM simply provided the final or necessary impetus (by providing practical examples through detected PSIs of breaches of system defences) for reviewers and teams to implement improvement actions and a little protected time and opportunity to do so. Conceptually, how the TRM worked during its implementation and subsequent application can be likened to a catalyst: it enabled and accelerated existing processes.

8.1.3. The TRM provides structure to the review and improvement efforts

Chapter 3 described the three consecutive steps of the TRM In detail, which are: planning; reviewing records; and action and reflection. Each step has a number of clear and specific recommended tasks which reviewers should perform or consider. The TRM therefore provides a structure for reviewers to follow and apply when reviewing their patient records and taking or considering subsequent improvement actions. Some participants described how the structure and steps of the TRM helped to focus their attention and efforts and they felt this increased their likelihood of recognising PSIs in the records they may otherwise have overlooked, or may have overlooked while providing routine care: e.g. 'the same patient I'd seen regularly, well I didn't make that jump at all [to detect a PSI] until I applied the structure to it [the record]' (GP03). Another clinician agreed that 'the structure helps [reviewers] to process it [PSIs] if you like, and make it a bit more tangible' (GP12). This finding is consistent with international evidence that unstructured reviews are inefficient, inaccurate and highly variable compared with structure approaches (233, 335).

The vast majority of reviewers indicated that they preferred the structured approach of the TRM over the alternative, which would have been to simply conduct an unguided review of the same sample of records: 'I think for people just looking through records, they often don't know what to look for and I think the trigger tool at least gives you a kind of core information' (GP08). The structure of the TRM also required clinicians to focus on performing tasks consecutively. One participant understood the value of the TRM in the following way: 'I think the benefit of doing it as a bundle if you like or as a as a trigger tool exercise is that you're much more likely to be focussed on looking for it [triggers]' (GP12). The structure and focus of the TRM helped a few of the participants who recognised that they were easily 'side tracked' or 'bogged down' when reviewing clinical information such as, for example, correspondence and investigation results. This is why one participant recommended the TRM to her colleagues and advised them: 'well, that's why you might find the structure of this [the TRM] more helpful than just randomly going through it all' (GP10).

These study findings illustrate the work of at least four different NPT components: 'differentiation' - the participants performed work to increase their understanding of how the TRM is different from alternative clinical record review approaches; 'interactional workability' - they had to enact and interact with the method to produce results; 'relational integration' - the work they did to build trust and confidence in the method and in their ability to apply it correctly; and 'individual and communal specification' - the work they did to increase their understanding through practical experience of how to apply the method more effectively, thereby increasing their perceptions of its feasibility and acceptability.

On the other hand, some reviewers detected PSIs that were not directly associated with triggers. These PSIs were still detected in the proximity of triggers, or while searching for triggers in specific sections of records. A practical example would be the detection of the PSI of an elderly patient who suffered a preventable adverse drug reaction after being prescribed different opiate analgesia concurrently. The reviewer detected the PSI while searching the medication section for the presence of the trigger 'medication discontinued'. One of the participants explained the value of the TRM's structured approach in the following way: 'even if there isn't a problem in the crux area, 'cause you've looked at that area there might be something juxtaposed to that that you see because you've actually taken the time to look at it [a specific part of the record] more carefully' (GP06).

8.2. The importance of 'ownership' of PSIs

A crucial difference between the IHI trigger tool method and the TRM approach in this study is 'ownership' of PSIs. In secondary care, external reviewers conduct trigger reviews on samples of records from patients who are unknown to them with very little or no chance of clinician and patient ever meeting in the future. The main findings are estimated harm rates and these are typically 'owned' by researchers and managers. Even when findings are shared with clinical teams they tend to be in aggregated data and chart format. In contrast, primary care clinicians in this study performed trigger reviews on records of patients that, in most instances, were very well known to them. They had usually provided at least some of the care during the periods of review and there was a very high probability that they would continue to do so in the future. Consequently, participants from all clinical groups felt they 'owned' the PSIs they detected.

However, the difference in ownership should not be considered a proxy of reviewer motivation, ability and thoroughness. External reviewers are typically highly-experienced clinicians with additional training in the trigger review methodology. Many of them may also have a personal interest in patient safety and could be expected to do the best job they can. Conversely, some internal reviewers may lack the necessary experience, training or attitude to effectively apply the TRM. Notwithstanding potential differences in reviewer characteristics, ownership of PSIs (or not) was an important determinant of the outcomes in this study as will be explained below, with the implication that it may also be an important factor to consider in other trigger review studies.

How the TRM seemed to work in general practice was that, once potential safety threats became apparent in the practice through the detection of PSIs, the personal relationships with and knowledge of the patients made them difficult to ignore. In other words, the PSIs detected in general practice seemed to evoke a powerful moral, professional and personal imperative to take further remedial actions that were reportedly felt by the vast majority of participants. The importance, emotional dimension and impact of PSI ownership was graphically described by a participant who thought implementing the TRM had 'certainly started the hares running' in his practice. He explained this was because detecting PSIs had felt like 'you've been pricked' (GP04).

The initial actions and intended actions were therefore typically, and unsurprisingly, targeted at individual patients and problems. Once the first few, small actions were taken, this sometimes led to larger and more significant changes or at the very least increased the probability of further improvements to other systems or reducing latent safety threats at practice level. In this way a single incident affecting one patient potentially became the starting point and spark for further assessment and actions at the wider patient population or practice levels. A practical example is provided in Box 8.1 in order to further illustrate the key findings of how the TRM works. The example describes the responses of the reviewer and her practice team to the detection of a single PSI.

In this example, applying the TRM had had the benefit of improving the management of patients with CKD of the practice because of four different sets of actions: (i) immediate actions during the trigger review relating to the specific patient with the PSI; (ii) the team performed a clinical audit to measure their current CKD management performance and to assess the likelihood of a similar incident occurring in the future; (iii) the reviewer undertook self-directed learning about the management of CKD as part of her continuous professional development; and (iv) the team updated the practice's CKD management protocol.

This example helps to illustrate key findings from chapter 7, namely that the degree of severity and preventability of PSIs are not associated with their usefulness. It also illustrates a key finding from this chapter, which is that reviewer characteristics are important determinants of the TRM's outcomes. In this instance the ability and willingness of the clinician to reflect on her findings determined the outcome and value of the TRM.

From an NPT perspective, this example illustrates the work of 'enrolment' and 'activation' - the reviewer had to recruit and engage additional team members (enrolment) to help her implement the desired actions, including the audit and updating the CKD protocol (activation). It also demonstrates the work of reflexive monitoring, and more specifically 'individual and communal appraisal' - the reviewer and practice team had to evaluate the importance of the findings and the potential value (or not) of taking subsequent actions.

Another way ownership of PSIs (and therefore the TRM) works is by generating findings that have personal importance and relevance to the clinicians and teams who apply the method because '*no one else is in the same boat*' (GP04). The value of the TRM is therefore not necessarily only in finding undetected or new types of PSIs but simply the fact that a PSI - or PSIs - was uncovered in the practice at all, and that it occurred to a patient known to them. Once PSIs are detected, the TRM essentially becomes open-ended in the sense that the type and degree of learning and whether and what type of improvement actions are undertaken are left completely to the discretion of the individual clinician reviewer. GP reviewers in particular valued this approach: '*looking for the problem and sorting it myself I felt a degree of empowerment*' (*GP06*). The same participant went on to explain his understanding of the TRM and its potential value (e.g. the NPT work of coherence) as: '*allowing you to take some control and ownership over stuff again and make a difference*.'

Some of the key differences between the TRM and many other quality improvement methods are, therefore, that it provides participants with highly personal, practical and patient-specific information and can enhance clinician autonomy.

Up until now, the discussion about ownership was from the perspective of a professional relationship between patients and clinicians or patients and other general practice staff. There is a second form of ownership of PSIs that were recognised by many participants, but especially the GP group and practice managers. This type of ownership is inferred from the general practice business model and relates to the responsibility of an organisation and organisational

leadership to assure patient safety and compliance with clinical governance obligations. The (abridged) narrative of PM11 in Box 8.2 provides a representative example of the perceived importance of ownership.

The ubiquitous business model of general practice in Scotland is a contractual agreement between the NHS and independent contractors to provide defined health care services on their behalf. GPs and some non-GP partners typically own the business concern and, in many instances, the care facilities. Participants understood the implication, namely that business partners had overall responsibility for the practice, the care it delivered and the repercussions from actions - whether positive (e.g. profit sharing, accumulating assets) or potentially negative (e.g. patient complaints, PSIs). Consequently, many GPs and some PMs were strongly motivated not only from a personal and professional, but also from a business perspective to provide high-quality, safe care. One of the participants explained: '*I sit here in this practice and think, what's happening in this practice? Because the buck stops with me, or with [GP partner X] or with [GP partner Y]' (GP04).*

This is a key difference between general practice and NHS acute care settings, as clinicians working in hospitals arguably have less of a vested financial interest in preventing PSIs. Even when hospital clinicians and staff are directly responsible for managing department budgets, their decisions rarely if ever directly affect their own personal incomes in the way it does general practice.

Box 8.2. Example of the perceived importance of 'ownership' to participants

I've been here seventeen years. I've been a partner for seven or eight of those now and basically I run the business of the practice...I manage everything from buying the envelopes to [pause] to agreeing the contract with the health board. Ehm well at the end of the day if anything goes wrong it's the partners that are going to to [pause] take the can for it... I never felt outside the partnership anyway. I was very taken aback to be offered partnership. It'd never even crossed my mind. I think it crosses more people's minds nowadays, but then it didn't ehm and it was very flattering. I think the only thing that's changed for me is [deep breath] you definitely do feel a much bigger weight of responsibility... I took a year to accept... because I wasn't sure it was really what I wanted, and I was giving up all my employment rights and that was scary, and I was becoming self employed, which was scary. I mean it's my practice, but then I probably felt that way as a manager as well (PM11).

8.3. Outcomes from PSI 'ownership'

There are three main types of outcomes from detecting and assuming ownership of PSIs (Figure 8.1). They are: (i) no impact or change; (ii) positive; and (iii) negative outcomes. Examples of positive outcomes include the different actions and intended actions undertaken by clinician reviewers and theirs teams during and after the trigger reviews as a result of the PSIs they detected and were described in detail in chapter 7. In addition, a practical example illustrating some of the potential and different positive effects of ownership of a single PSI was provided in this chapter (Box 8.1).

This section describes three further potential outcomes from PSI ownership: increased vigilance and awareness of patient-safety related issues; identifying learning needs and points; and their emotional impact on clinicians and their teams.

8.3.1. Increased patient safety awareness

Many participants thought that applying the TRM had increased their awareness of safety-critical processes at the time and a minority felt that this effect was sustained for at least several months after each trigger review. One GP said: *'having done the trigger reviews [we are] more aware of looking for things'* (GP02). Some clinicians also reported that their confidence in their wider practice systems and the standards of the quality of care they delivered had increased after conducting the reviews. From an NPT perspective, this reflects the work of relational integration (increased confidence) but also reflexive monitoring (evaluating their levels of safety) being performed successfully.

It is unclear exactly why the TRM increased awareness of patient-safety related issues and why this occurred to varying degrees for different reviewers. However, a similar effect has been reported in the international literature as far back as 1991, when Wu et al found many junior doctors' vigilance increased as a result of their (known) mistakes (336). Some of this study's participants thought there was an association between the TRM's structured approach, specific triggers and ongoing safety awareness. This perception is supported by the fact
that a few months after the trigger reviews had been completed, a minority of reviewers reported still approaching patient records section-by-section and screening for triggers in a structured manner even during their routine consultations. This was particularly true for the group of PN reviewers, who were initially less confident than the GP group about using the TRM, but found the structured approach and predefined triggers particularly useful. A nurse practitioner explained: 'personally, the trigger review of patient notes did change my mind-set and I look for certain triggers now as part of my practice. You tend to memorise certain triggers and look out for them with all patients it tends to become imbedded into practice' (PN0X).

8.3.2. Learning needs and points

The majority of participants thought that the TRM was useful, because it helped them to identify new learning needs and points. Learning needs were identified at the individual and practice team levels: *'there are always things to learn if you're keen to learn'* (PN12). Some reviewers also described how they met or planned to meet these learning needs. At the individual practitioner level, learning needs often related to management of patients with chronic disease. An example would be updating knowledge on the management of atrial fibrillation.

There are three ways general practices and organisations can respond to and learn from PSIs. The first is 'doing the quick fix'. This is the most common and superficially attractive response. The second is 'going into a black hole', e.g. clinicians report a PSI, but 'nothing' seems to change. The third is 'closing off the Swiss-cheese holes', when sustainable changes are made to systems in order to prevent recurrences (175). Options one and three are appropriate. The challenge is determining when to use which.

At the practice level, reviewers frequently identified the need for: improving communication between primary and secondary care, consistent coding of clinical conditions and 'protocols' for the management of specific high risk medication, e.g. monitoring of nephrotoxic drugs. Further examples with selected quotes are provided in Table 8.1.

Educational	Examples recorded on the Trigger review SS			
value of the TRM	Individual (clinician) level	Practice team level		
Identifying learning needs	• 'I need to review CKD 3 management as I could not remember specific contraindications for ACE inhibitors'	• 'Several patients overdue annual reviews - should we check notes and call patients via telephone rather than just letters?'		
	 [I need to] update my diabetic knowledge - online module' 	• 'Liaise with colleagues who do bloods to ensure the correct tests are carried out and to reduce the need for the patient to return unnecessarily'		
	• [What is the] target heart rate for AF? Read editorial from NEJM	• 'Should discuss and agree how to prioritise read codes. In DOCMAN we should probably advise the staff what priority we want given to codes. Confusing for staff as lots of options.'		
	 '[Find out] how to liaise with social services about respite [care]' 	 'Protocol for monitoring potential nephrotoxic [and hepatotoxic] drugs' 		
Identifying learning points	 'Awareness of need to ensure warfarin is stopped pre- operative where appropriate' 	• 'That, whilst as a practice we are fallible, we do have reasonable processes in place. The missed increase in bisoprolol was a concatenation of events between primary and secondary care at the discharge interface. This is a fraught area.'		
	• 'Be aware and proactive if investigation results do not fit with clinical picture'	'Positive learning that disease monitoring systems work well (COPD) [in this practice]'		
	'[I] need to give more attention to out of hours summary sheets'	• As a team we need to be more aware of encourage those patients with chronic disease to attend for review		

Table 8.1. Educational value of the TRM: selected examples of individual and practice teams' learning needs and points

Whether and to what extent learning from PSIs occurs is determined by many factors (337) and will be considered in more detail below.

8.3.3. Emotional impact of PSIs

The potential adverse impact of PSIs on clinicians is increasingly being recognised and studied (186, 338-341). In fact, a study to determine the factors that influence how students and residents learn from medical errors found even imagining themselves committing errors caused the participants 'significant stress' (299). Twenty-five years ago Christensen et al likened their experience of conducting in-depth interviews with community physicians (n=11) about the impact of their perceived mistakes on them as '*equivalent to a descent into the underworld of medicine. It is a journey into a place of shame, fear and isolation*' (298). The authors described the wide range of negative emotions ('dysphoric feelings') the clinicians experienced and how, in many instances, these feelings of frustration, guilt, fear, anger, self-doubt, anxiety, humiliation, shame and embarrassment remained unresolved for months and years. These findings have since been replicated in larger studies, different clinician groups and settings (186, 339, 340, 342).

The different coping strategies clinicians reported using - if indeed they even consciously committed to deal with the effects of PSIs on them - typically belonged in one of two groups, depending on whether the focus of the strategy was on the 'emotion' or 'problem' (298). This is in some ways analogous to the two different tasks clinicians face in response to a PSI or medical error: '*How do I deal with this*?' And '*How do I learn from this*?' (338) Examples of helpful 'emotion-focused' strategies include disclosing errors to peers, patients and family; and re-establishing and re-affirming their professional identities (e.g. self-forgiveness). On the other hand, participants with a 'problem-focused' approach attempts to analyse the event, learn from it and take remedial actions. Consequently, there is hope in the '*heart of darkness*' for '*this landscape of fallibility contains the spirit of wisdom that can guide the rebuilding*' (298). All but one of the participants in Christensen et al's study reported learning from their mistakes and had subsequently taken specific actions to improve their clinical performance in some way.

In some ways little has changed in the quarter century since this study was published: patients still suffer avoidable, iatrogenic harm; and clinicians still err despite their best and ongoing efforts to provide safe care. As a result, we should acknowledge that the emotional cost of PSIs can be high (338). This is why clinicians involved in serious PSIs have been referred to as 'second victims' (185).

One thing that has changed dramatically in health care, and especially in general practice, is the disclosure rates of medical errors. In the study by Christensen et al, only a tiny minority of physicians admitted to disclosing their mistakes, and those that did only confided in close relatives (298). In contrast, all of the participants in this study unanimously agreed that disclosing errors was not only appropriate but that they felt able to do so, and were confident of the support of their colleagues, should this be required. While it is possible that this may be the unique perception of this study group, consider that significant event analysis (SEA) is ubiquitous in UK general practice, the prevailing safety culture tends to be 'open' and positive in this setting (Chapter 6) and patient safety research is now encouraged in many modern health care systems. This may also be the reason why there were no reported or known instances of detected PSIs having significant or enduring *negative* emotional impact on participants in this study.

There is some evidence that patients also prefer disclosure of medical errors (343). A questionnaire study of German GPs' management of serious medical errors and subsequent outcomes found most errors had been disclosed to patients. The majority of patients continued to trust their GPs despite them disclosing serious errors, especially if the GP had discovered the problem and disclosed it themselves, or if other health care providers also contributed to the PSI (344). The implication is therefore that the TRM recommendations to discuss PSIs within the practice team and disclose errors early are acceptable to clinicians and patients alike.

A common perception is that clinicians learn 'better' from their own mistakes, from those errors they accept responsibility for and from PSIs that result in severe harm (299, 337). The findings from this study that participants consider PSIs with higher severity ratings as more important seem to support this perception, and were described in Chapter 7. However, if there is indeed a link between severity of PSIs and subsequent learning, it has not yet conclusively been shown. In fact, current evidence seems to suggest that clinician characteristics are equally or more important determinants of whether and to what degree learning occur (187, 339). Two other factors that are strongly associated with learning from PSIs are: creating opportunities for critical reflection on incidents; and formal discussions of incidents and providing constructive feedback (49, 337, 342). These findings support the emphasis of the TRM in this study on clinician reflection and the recommendation of formal meetings to discuss trigger review findings in a structured manner within teams.

A recent systematic review (n=24 studies) of the effects of medical errors on clinicians found consistent and widespread evidence of negative psychological and emotional effects on them (186). However, a 'disproportionate' amount of this research was conducted in the hospital setting and the authors admit it remains unclear how error outcomes vary between setting and clinical groups. There is, for example, emerging evidence that PSIs have an emotional impact on all general practice staff, and not only clinicians (7, 187). For now, it remains unclear what the implications of this finding may be. Of interest though, is that the review also found some evidence of positive outcomes from medical error. Examples include: increased assertiveness and confidence; improved relationships with colleagues; and corrective and improvement actions taken from the individual patient through to the organisational levels (186). Constructive feedback and effective learning in the aftermath of a serious PSI or error seems to make the difference between clinicians 'thriving' rather than simply 'surviving' (185). This deceptively simple yet powerful observation resonates strongly with the findings of this study. A key priority for future research is therefore to examine the longer-term impact of PSIs and increase our understanding of how to maximize the positive and reduce their negative effects.

Conclusion

This chapter explained how the TRM worked. In essence, trained clinicians performed structured trigger reviews of samples of patient records with a patient safety 'mindset', ideally as a single uninterrupted activity during protected time allocated specifically for this purpose. In combination, these three factors (structure; mindset; and opportunity) greatly increased the likelihood of detecting PSIs compared with unstructured reviews of clinical records or opportunistically identifying them during routine care.

The detected PSIs in this study had personal and professional relevance for the clinicians and the practice teams involved, because they affected or may have affected patients known to them. Many PSIs also provided specific, uncontroversial information about the standards of delivered care.

Consequently, many participants were willing, and felt a responsibility, to accept ownership of the PSIs they detected, thereby seizing the metaphorical 'learning moments' and extracting value from them in the form of subsequent reflection, learning and further action. The different and important positive outcomes from detecting and taking ownership of PSIs in this study were therefore discussed in more detail. In particular, the different types of learning needs and points that participants identified were listed and the increase in some participants' awareness of safety-related issues were described. Although negative outcomes from PSIs were not an issue in this study, they are potentially important and were also discussed.

The key study findings are therefore that the TRM worked because of specific factors (mindset; structure; opportunity; and ownership of PSIs) and that these factors helped to determine the TRM's positive outcomes (detecting PSIs; increased safety awareness; identifying learning needs and points; taking specific improvement-related actions).

The findings may seem simple or self-evident, but they have potential important implications for other quality and safety improvement efforts. Each factor is related to a more generic and essential requirement for successfully implementing complex health care interventions. They are, in no particular order:

- *'Mindset'* frontline staff must have the necessary *knowledge* to implement the intervention;
- '*Opportunity*' participants require adequate and appropriate resources and time to implement interventions;
- 'Structure' the aims, applications and potential benefits of interventions have to be clearly understood by participants;
- 'Ownership' -interventions and implementation require ongoing clinician engagement.

From an NPT perspective, each of these factors is associated with work relating to specific constructs and components of the framework. The likelihood of an intervention being successfully implemented and eventually becoming normalised is dependent on how much of this necessary work is effectively performed in practice. For example, the work of identifying and agreeing the resource requirements, and which team members have the necessary knowledge to implement the intervention, is described by the 'skill-set workability' component of the NPT framework. The work of providing the resources, recruiting the staff and keeping them engaged are described by the 'contextual integration', 'enrolment' and 'activation' components respectively. Similarly, each of the other factors described in this chapter can be described by applying the NPT framework.

The next chapter will describe these factors in more detail and consider whether they facilitated or hindered the implementation of the TRM. The additional factors that were identified as important facilitators or barriers in this study through application of an NPT framework will also be discussed.

Chapter 9. Factors that facilitated or hindered the implementation of the TRM

This chapter describes the results that relate to the study aim of identifying and describing the main factors that facilitated or hindered the implementation of the TRM in general practice. The main source of data for this chapter was the second round of interviews with practice managers, general practitioners and practice nurses. The data were coded and analysed according to the Normalisation Process Theory (NPT) framework.

The chapter sections correspond to the four main constructs of the NPT framework, which are: coherence; collective action; cognitive participation; and reflexive monitoring. Each of the four sections describes the factors relating to that specific construct, and how they are associated with the other elements of the NPT framework. The fifth and final section considers the main study findings and the potential implications of the identified barriers and facilitators, comparing and contrasting them where relevant with the international literature.

9.1. Coherence - The sense-making work individuals and teams did to understand the TRM

The main findings are summarised in Table 9.1 and illustrated with selected quotes:

- Implementation was facilitated when participants understood that the TRM was a new QI method, but intended to be complementary to existing methods such as SEA;
- Participants were concerned that they would have insufficient time and resources to implement the TRM. However, the vast majority found the method was feasible;
- The vast majority of participants perceived the TRM as acceptable and compatible with the prevailing safety culture in general practice.

NPT component, with a description relating it to the TRM	Success factors and their effect (facilitating or hindering) on implementation of the TRM	Selected quotes	
<i>Differentiation:</i> The work of understanding the differences and similarities between the TRM and existing practices (informal record review) and QI methods (SEA, clinical audit).	Implementation was facilitated when participants understood the TRM as a new QI approach, but complementary to existing methods such as SEA and clinical audit.	I mean essentially what [the TRM] does is a kind of retrospective audit of a cohort of patients looking for any potential areas of harm (GP08) [The TRM] is essentially looking to pick up an SEA I suppose. That's the way that you could look at it - if you need an SEA that's a good way to find one (GP07)	
<i>Communal specification:</i> Understanding the intention and potential benefits of the TRM.	When participants understood the TRM's intended aims, potential benefits and limitations they were more likely to use it, use it as intended and report positive outcomes.	I think it's useful as a learning tool to learn about your own systems and a way of trying to improve those systems and a way of learning as a team with the results (GP05)	
Individual specification: Understanding the required effort and tasks of individual team members to implement the TRM, e.g. is the TRM considered to be feasible and a priority?	All participants were concerned that the available time and resources would be insufficient to implement any new intervention, including the TRM. However, the vast majority found the TRM to be feasible, which then facilitated its further use.	I think the first time doing the first couple of patients was a bit slow and because it's different and you're not quite sure where you're at. So it took a wee while, a couple of patients really to get into the swing of it. I did it again just last week and found it very quick and very easy to go through (GP02)	
Internalization: the work individuals and teams have to do to understand how the TRM 'fits in' with the prevailing culture and processes in the practice, e.g. whether it is acceptable.	The vast majority of participants perceived the TRM as acceptable and compatible with the prevailing safety culture in general practice. The closer the TRM fits with the practice culture and the team's existing tasks, the easier its implementation.	You have to have systems in place that make a safe journey for the patient. So I guess that's why we think we should be doing [the TRM], whether it's a project or an incentive or not, because that's what we're all about really, bottom line (PM08)	

Table 9.1. A description of the four components of the 'Coherence' construct with selected quotes

In the UK, general practices have been incentivised for years to apply QI methods such as SEA and clinical audit (chapter 3) at regular intervals. As a result, most primary care clinicians will have at least some experience with these tools. All primary care clinicians and many non-clinical staff also regularly interact with patients' medical records. Examples of informal and opportunistic interactions may include checking and comparing investigation results and clinical entries while more formal reviews are required to prepare medical reports. The TRM includes references to known and normalised QI methods and its design was based on existing, formal and informal review processes which were adapted to provide a structured approach and a specific purpose, i.e. to detect PSIs. In fact, the structure of the TRM was identified as one of reasons why it was effective, as was described in Chapter 8.

Consequently, the TRM was recognizable to all of the participants, and many of them explained their understanding of the TRM by comparing it with QI methods that were already known to them, such as clinical audit and SEA. Furthermore, they understood the TRM as being complementary to existing QI methods. For example, detecting PSIs with the TRM often required subsequent application of other methods, such as SEA or audit, to better understand or quantify the extent of the problem. Being able to interpret the 'new intervention' in terms of existing approaches and methods helped to facilitate its implementation overall.

On the other hand, when the TRM was perceived as being too similar to existing practices or other interventions, it potentially created confusion and paradoxically decreased the motivation of a tiny minority of participants to implement it. Implementation was therefore further facilitated by participants being able, despite similarities between the TRM and other QI methods, to also recognise important characteristics that differentiated it as being 'new'. The main difference for them was that the TRM aimed to *proactively* detect and prevent PSIs, whereas the aforementioned QI methods were typically *reactive* in nature. One GP explained how '*mistakes usually are only picked up when they've blown up rather than going looking for them*' which is why he thought the TRM was '*better because actually I was fishing for things*' (GP06).

In terms of NPT, the work of differentiation is an important facilitator of implementation because it helps participants to think through how they define, categorize and divide the tasks associated with implementing the TRM appropriately - which is complementary to the work of 'skill-set workability' and essential for effective 'enrolment'. It also helps participants understand how changing aspects of the TRM may affect its outcomes. In NPT terms, it helps to inform the work of 'reconfiguration' of the TRM. For example, selecting different patient populations will affect the type and numbers of PSIs that are detected, e.g. high-risk groups are predisposed to PSIs so their selection will increase the estimated PSI rates. In addition, if new interventions are 'recognizable' to participants it facilitates their work of coherence (sense-making). Participants are able to use their existing skills and previous experience and potentially gain early confidence when using the method effectively. From an NPT perspective, this would be described as 'relational integration'.

Most participants initially expressed concerns that implementing the TRM would increase their workload and require substantial additional resources and time. This perception was moderated as their understanding of the TRM increased through practical experiences with it, so that the vast majority of clinicians conceded that the actual workload and time requirements were lower than they initially expected. For example, GP11 described getting 'bogged down' during the first trigger review, learning from his experience and applying the method more effectively the second time. However, while reviewers typically experienced their second trigger reviews as being quicker and easier than the first, their findings were not necessarily more important or helpful.

The finding that the perceived feasibility of the TRM changed during the study period is the result of the work of 'individual specification', e.g. reviewers changed their perceptions of the TRM so that they came to understand it as being quicker and easier than they thought at first. This was informed by the work of interacting with and experiencing the TRM (e.g. interactional workability) and evaluating its outcomes (e.g. individual appraisal). The implementation of the TRM was facilitated by practice teams understanding the necessary steps to implement the intervention and perceiving the requirements as being feasible. From their perspective, feasibility meant that the TRM required minimum additional resources, time or increase in workload. The implication is that the degree of perceived or actual 'effort' required to implement an intervention is a powerful determinant of its eventual success (less effort, increased feasibility) or failure (more effort, decreased feasibility). Implementation can be facilitated further if the intervention can be assimilated into existing procedures and practices and participants expect it to reduce their workload.

Implementation of the TRM was also facilitated because its intended users clearly understood its aims, potential benefits and limitations. Initially, their understanding varied widely, but after the training sessions the vast majority understood that the main objectives were to search samples of medical records for previously undetected PSIs and implement improvement actions. For some, their understanding of its benefits expanded during the study period as they identified additional applications, such as using the TRM during one-to-one training sessions with GPSTs.

All participants thought the TRM was compatible with the prevailing culture in general practice and understood how it could be incorporated into their existing safety systems and procedures. This perception facilitated the implementation of the TRM as the participating practice teams reported having a strong, positive safety culture which in large part motivated them to participate in this study in the first place. The implication from this study finding is therefore that the more compatible an intervention is with the prevailing culture and existing work practices of a team or organisation, the more likely that its implementation will be successful and that it will become normalised over time.

The degree to which interventions are perceived as being acceptable and feasible to their intended users largely determine whether they will be implemented successfully in the first instance and subsequently become integrated into routine work. The acceptability and feasibility of interventions are in turn determined by the product of the different types of 'coherence' work. The vast majority of participants perceived the TRM as acceptable and feasible and could: (i) clearly differentiate it from other QI methods; (ii) understand how to apply it; (iii) discern what potential benefits it would have for them and their patients; and (iv) understood how it 'fit' within the prevailing cultures in their teams.

9.2. Collective action - The operational work of enacting the TRM and integrating it with existing practices and contexts

The main findings are summarised in Table 9.2 and illustrated with selected quotes:

- Implementation of the TRM was facilitated when PSIs were detected quickly and the PSIs were unambiguous, serious, preventable and originated in primary care;
- Implementation was facilitated by allocating adequate resources and time;
- Inclusion of the TRM in the QOF was an important facilitating factor.

The majority of reviewers conducted the trigger reviews as a single activity by themselves. However, in a small minority of practices the reviewers worked in pairs - at least initially - and felt this helped to improve their confidence in the method and their own ability to apply it. A small minority of practices decided to divide the trigger reviews among several clinicians, so that each reviewed only five or six records from the sample, but this strategy was perceived as ineffective. Similarly, when clinicians interrupted their trigger reviews by only searching a few medical records opportunistically during the day or days, they also reported less benefit compared with those that performed reviews as a single activity. One of the reasons for these findings was discussed in Chapter 8, namely that the TRM works by creating a patient safety 'mindset' which required time to form.

Many reviewers considered the PSIs with greater severity to be more deserving of and amenable to improvement actions compared with those that had resulted in little or no patient harm. The small minority of reviewers that were unable to detect a single PSI or only detect a few PSIs of low severity therefore typically perceived this as an important barrier to the TRM's use. This is understandable, as the main value of the TRM in general practice is derived from detecting PSIs and taking ownership for them (Chapter 8). However, some reviewers

NPT component, with a description relating it to the TRM	Success factors and their effect (facilitating or hindering) on implementation of the TRM	Selected quotes
Interactional workability: the work of applying the TRM in practice, the results (e.g. whether and what type of PSIs they detected and what improvement actions they took) and the time and effort this required (as opposed to what they perceived it would be). What effect did implementing the TRM have on their routine work?	Implementation of the TRM was facilitated when PSIs were detected quickly and the PSIs were unambiguous, serious, preventable and originated in primary care. A small minority of reviewers found no PSIs, which was a major barrier to future implementation.	There's safe and there's safe. I mean there's life threatening and there's a slight error on certain things (PM03)
<i>Relational integration:</i> The work of building confidence in the TRM, their own and their colleagues' abilities to effectively apply it and that the findings are valid and useful. To be effective, this work requires formal and informal sharing of trigger review findings.	Practice teams in general accepted that the TRM findings were valid. However, none interpreted the results as reflecting unsafe or poor overall quality care. Implementation was hindered by the concerns of a minority of reviewers about the validity of the results from other practices or that their findings may be inappropriately interpreted and used at regional and national levels.	You can do it properly or you can have a quick scamper through it and not find anything (GP04) If we get information back and you see practices where they've identified a range of issues, that's good - they've identified them. But what else is going on in that practice? (GP08)

Table 9.2. A description of the four components of the 'collective action' construct with selected quotes

Skill-set workability: the work of dividing TRM implementation tasks. To be effective, this work requires the practice leadership to consider each team member's current role, their experience, skills and attitude and the available resources and then appropriately match staff with the required work. If necessary, additional resources or training may have to be procured. Implementation is facilitated by the allocation of adequate and appropriate resources, balancing competing demands on team members' available time and considering reviewer characteristics when assigning new tasks. The TRM implementation was hindered in those practices that didn't allocate adequate resources and time, or when time was allocated but not protected. The vast majority of clinician reviewers had the necessary skills and experience to perform trigger reviews, but not all of them participated through choice. Time's the biggest killer. I think every practice could open twenty four hours a day and still not have time. Every single thing that comes out: 'we'll get the practice nurse to do it' but just how thin do you get spread? (PN08)

Coming in in my own time was much better because there was no chance of being interrupted and more time to take. I didn't feel pressure to look at my Docman or look to see what other things were going on in the surgery. So I think probably doing that [the trigger review] complete and everything's closed and you know you're up to date with everything is actually more useful (GP02)

Contextual integration: The work of integrating the TRM within the context of general practice, in particular GPST, appraisal and revalidation. CI work requires the provision of adequate and appropriate resources; visible support from senior leadership and restructuring existing policies or infrastructure to include and support the TRM.

Formal inclusion of the TRM in existing, recognised GP contexts such as QOF, appraisal and GPST is an extremely powerful facilitator. However, some participants felt that including the TRM in QOF may paradoxically decrease its impact, as it may simply become 'another tick box' exercise for some practices.

Will I be putting [the TRM] in my appraisal? Of course (GP01)

What we were going to do was get the GP trainee involved... I plan personally to use it with our trainees now (GP12)

had an alternative interpretation of 'finding nothing', which is that it provided them with 'evidence' that the care they provide was safe and of a high standard. Implementation of the TRM was facilitated when the participants or practice leadership accurately assessed the resources and time that would be required for each phase of the implementation process *and* ensured that these needs were met. They also had to decide who would do which tasks. In NPT terms, the work involved in this is described by the 'skill set-workability' component. They also had to critically evaluate the knowledge and 'readiness' (as defined in Chapter 8, page 200) of the clinician reviewers and whether these were sufficient to enable them to successfully implement the TRM.

Further work was then required to recruit additional team members and keep them actively engaged in the TRM process (enrolment) and to allocate and provide the necessary resources (contextual integration) to enact the trigger review findings. The potential types of resources included: protected time to implement the intervention; access to external 'expert' support; availability of training; senior leadership support; restructuring current policies and infrastructure to include, promote and support the TRM; and discretionary budgets. The work of 'contextual integration' should be the responsibility of the host organisation, which in this study included the general practices. However, the local health authorities, NHS Health Boards and even the national government also had important contributions to make.

Providing adequate and appropriate resources was one of the most important facilitating factors for the successful implementation of the TRM. While, the participants valued the TRM training, educational material and support from external 'experts', according to them the crucial resource was sufficient, uninterrupted time to allow a trigger review to be completed as a single activity.

While most practices allocated at least some protected time for TRM work, it was seldom adequate or uninterrupted. One GP described the intensity of the workload in her practice as a *'rollercoaster'* and how she had to *'physically hide'* (GP02) to perform trigger reviews. As a result some clinicians performed the reviews in their own time, in-between other tasks or divided the sample of

records between several clinicians, which as stated earlier reportedly decreased its effectiveness. Some reviewers reported being interrupted by team members or by other, more urgent clinical tasks. A minority of participants were also distracted by the constant feeling of other work 'piling up' and a compulsion to keep checking how their workload was increasing while they were performing trigger reviews. For these reviewers, the TRM was perceived as just one more task competing for their limited time.

The personal and professional characteristics of reviewers strongly influenced the implementation of the TRM. Experienced, enthusiastic clinicians who were motivated to use the TRM and able to critically reflect on their findings and the implications for their patients and the practice derived the most value from the method. On the other hand, adopting a strict 'tick box' approach reduced the effectiveness of the TRM. The study participants thought that a substantial minority of the overall general practice workforce would adopt this approach with any intervention including the TRM, and especially those interventions associated with the QOF. From their perspective, the TRM may therefore be less useful when implemented more widely.

This is an important concern given the TRM was incorporated into the QOF shortly after this study commenced and subsequently implemented across Scotland by hundreds of general practice teams. However, a comparison between this study and the larger QOF study did not find any significant differences in the main findings. The perception that some practices simply 'tick boxes' may therefore be incorrect or, alternatively, occur in only a small number of practices.

Most participants thought that including the TRM in QOF was the most important facilitating factor for its implementation. A practice manager explained how 'being part of the contract it [the TRM] just becomes something that <u>has</u> to be done' (PM03). But for a minority of reviewers it paradoxically decreased their interest in the TRM. This is because they perceived QOF targets as being achievable without discernible improvements in patient outcomes and concerns that targets become harder to achieve year after year. One GP felt that 'it's quite easy to pay, if I'm honest, lip service to some of those changes for the

paperwork that is required [by QOF]' (GP06). In NPT terms the work of including the TRM in the QOF is described by the 'contextual integration' component. Consequently, participants were able to 'legitimise' their involvement with the TRM and this facilitated its implementation. However, including the TRM in QOF paradoxically also hindered its implementation. Some participants struggled to do the work of 'relational integration' because, as described above, they struggled to build trust in the ability of other practices to effectively apply the method.

Apart from the QOF, the implementation of the TRM was also facilitated by its integration in other general practice contexts, including GP specialty training (GPST) and appraisal. A few participants felt the TRM had potential value in medical education: *I actually used it [the TRM] as a teaching tool' (GP11)*. They introduced the method to the GPSTs in their practices, encouraged them to apply it and subsequently discussed their findings in order to identify learning needs and points. Most GP reviewers also indicated that they included or planned to include the TRM as supporting evidence of QI for appraisal purposes. However, these contexts were much less important for the implementation of the TRM compared with the QOF and were not significant facilitating factors.

The vast majority of participants' confidence in the TRM and their own ability to correctly apply it increased with experience and time. However, some participants described a lack of trust in the validity of the findings of reviewers in their own or other practices and a substantial minority of practice nurses doubted their own ability to apply the method correctly (at least initially), e.g.: 'I was just lacking confidence of what I was doing; I think that's what it was. I kept thinking I should go back and I did... I just had to convince myself I wasn't missing anything. I was annoyed with myself for feeling like that cause I felt I shouldn't. I'm experienced enough and it was just something different (PN10).

Despite this, the vast majority of practice nurses were able to detect PSIs. However, they were less likely than the GP group to formally share their findings with the rest of the team or implement improvements at the practice level. This was at least partly because the majority of nurses were not provided with formal opportunities to report their findings during practice meetings, where limited time meant competing agenda items were given priority, or meetings were scheduled on days when the nurses were not scheduled to work. Even when the practice nurses were able to attend team meetings, they typically did not set or influence the agenda.

'The nurse will do it [trigger reviews] because that's what she's been asked to do, but I'm not sure that it'll then go anywhere, or make anything happen unless someone else is involved with that, and it's brought up... it's going to be quite difficult for me to share those findings with other people, because I'm hardly ever at the practice meeting, because they have it on a day I don't work here, and at the other practice I work in I'm never at their practice meetings, so I'm in an oddly peripheral place, you know, and that can be very difficult. I like to feel involved and like I can, you know, influence things and help a practice to work better, but I'm not in that position' (PN02)

The majority of participants felt confident that the findings of their own trigger reviews were valid and that they had applied the method correctly. They acknowledged that some of the detected PSIs may indicate opportunities to improve specific aspects of future care but no one thought it implied that their overall standards of care were unsafe or of unacceptable quality. The majority of participants informally shared the findings from the trigger reviews with at least some team members. The potential impact and value of the TRM was increased by sharing the findings, particularly when this helped team members to identify and agree on the contributing factors of PSIs and collectively take further action to improve practice systems and processes. However, only a minority shared the results with the whole practice team during dedicated meetings.

A substantial minority of participants had concerns about how trigger review findings would be interpreted and used at a national level. For this group, their concerns, and a general mistrust of more senior NHS management, were a potential barrier to effective implementation of the TRM. One GP felt 'we talk about a no blame culture in NHS [Board X]' yet 'the first thing everyone does is 'right, who's to blame for this' and too often the result is that 'we'll hang someone out to dry in a non-judgemental way' (GP04).

9.3. Cognitive participation - The relational work of engaging participants and establishing a community of practice around the TRM

The main findings are summarised in Table 9.3 and illustrated with selected quotes:

- Providing flexible training facilitated implementation of the TRM;
- The TRM was facilitated through voluntary participation;
- The TRM was facilitated when reviewers had sufficient autonomy and opportunities to enact change.

The period prior to and during the initial implementation of an intervention is crucial, as it helps to determine whether the intended users will accept ownership and continue using it. During this period the work of 'initiation' is an essential part of determining participation, but also influences the degree of 'coherence' (understanding) participants develop about the intervention. Factors that were particularly strong facilitators early in the implementation of the TRM were: (i) identifying and recruiting practices and clinicians with an expressed interest in improving care quality; (ii) providing flexible training sessions; (iii) easy access to 'expert' support and advice; and (iv) detecting PSIs early during the trigger reviews, which helped to increase participants' confidence in the method and their own abilities to apply it. Of these factors, participants considered TRM training as the most important.

Training typically required a minimum of one hour and only rarely exceeded two hours. Characteristics of the training that facilitated implementation were flexibility in delivery and timing, using a range of teaching methods and tailoring the delivery to participants' learning needs. In particular, participants valued the chance to practice the method on simulated records and the educational resources they were provided with were useful reminders during the actual trigger reviews. However, whether clinicians received training or not was a more important factor in determining their success in implementing the TRM than the type of training.

NPT component, with a description relating it to the TRM	Success factors and their effect (facilitating or hindering) on implementation of the TRM	Selected quotes	
<i>Initiation</i> : The work of preparing for the implementation of the TRM, e.g. ensuring that implementers are willing and able to start using it.	Training, formal and informal promotion of the intervention (raising awareness) and access to expert support all facilitated implementation. However, training had to be flexible and use suitable delivery methods.	I've been trying to start the ground level approach of saying 'this is how it should be used', you know, used formatively and using it to look at your systems as well, and things like that (GP05)	
<i>Enrolment</i> : the work of identifying and recruiting the necessary people that are required to implement the TRM at the appropriate times. It also involves identifying others who chose to participate and evaluating the contributions of all participants.	Initial recruitment of volunteers facilitated implementation by ensuring clinician engagement. The TRM was facilitated by the vast majority of GPs participating voluntary. However, the majority of practice nurses and administrators were assigned to the TRM which, in some cases, reduced their motivation.	Sometimes you know that, although they're asking you [pause] it's going to come your way anyway (PN09) We decided because I'd already done this last year that one of the other GPs would do the trigger tools this year (GP11)	
Activation: The work of supporting and sustaining the TRM individually and as a team. This work required TRM findings to be disseminated and that clinician reviewers had opportunities and sufficient influence to enact the findings.	The TRM was facilitated when findings were disseminated and reviewers had sufficient autonomy and opportunity to enact change.	[The TRM is] easy until you get the 'what am I going to do about this' stage (GP04) I held a practice meeting afterwards to highlight that perhaps we aren't always that good (GP06) I wasn't involved at all (PM10)	

Table 9.3. A description of the four components of the 'Cognitive participation' construct with selected quotes

Legitimation: The work individuals and teams do to justify implementing and being involved with the TRM. This component also includes the legitimising work other parties do who are not directly involved in the practical implementation of the TRM.

Implementation of the TRM was facilitated when individuals and practice teams were able to justify investing time and resources in its application. Formal inclusion, endorsement and external validation of the TRM by professional and regulatory bodies were particularly powerful facilitators. I'm not sure if I'd have gone back to [the TRM] if it had disappeared off the horizon. If it was out of QOF, but part of what we do, or it's made an RCGP approved tool or it was something that trainees do, then I could see why people would become lured to the idea that it is a good thing... cause it gives it fruits - there are fruits for your labour. Well, you have to justify the time in order to make it happen (GP06) I feel I always have to justify every single working minute I have in here and I don't take tea breaks because it's just constant (PN10) Implementation of the TRM typically began with a GP volunteering to participate in the study or apply the method. These 'early adopter' or local 'champion' GPs described personal and professional reasons for wanting to participate: curiosity, awareness of patient safety as a practice and political priority, and being proactive in this regard. These GPs then enrolled other team members to assist with the work of implementing the TRM in one of three ways. The first was formally assigning specific responsibilities to team members with little or no opportunity to decline the role. This is the way in which the vast majority of practice nurses, managers and administrators were recruited. The second way was to recruit team members opportunistically or wait for them to volunteer their engagement. This is the way the vast majority of other GPs were enrolled. The third and least common enrolment method was formal invitations at practice meetings and GPST training sessions. A few GPs also enrolled nonpractice participants, including IT staff, pharmacy colleagues and secondary care clinicians. Perhaps unsurprisingly GPs were - at least initially - more motivated to apply the TRM than PNs. However, the manner of enrolment and initial levels of motivation did not seem to be associated with the relative success (or not) of applying the TRM.

The autonomy (or alternatively 'power' or 'influence') of the reviewers was identified as an essential factor in the TRM's implementation - without it, the potential for improvement was greatly reduced. However, clinicians with little autonomy could still detect PSIs, endeavour to make some improvements at the individual patient level and learn from the trigger review process. Conversely, greater levels of autonomy meant trigger review findings were shared with the rest of the practice team and therefore increased the educational and improvement value of the TRM to the practice and even regional. In NPT terms, the work of 'activating' the TRM required participants to remain 'engaged' with the intervention. This was facilitated when they were provided with sufficient opportunities to enact the intervention and perceived their involvement to be appropriate, e.g. through the complementary work of 'legitimisation' and 'relational integration'.

Implementation was therefore also facilitated when the TRM was perceived as a legitimate activity and participants could justify investing their time and

resources in it. Legitimation of the TRM occurred through three main mechanisms. These are, in descending order of their perceived importance: (i) contextual integration with, for example, QOF; (ii) endorsement and practical demonstration by a peer, e.g. a clinician in the practice performing a trigger review; and (iii) verbal or written promotion by health care and professional organisations, e.g. the RCGP and NES. The vast majority of participants were also aware that improving patient safety in primary care had been declared a priority by the Scottish Government.

However, just because an intervention is perceived as 'legitimate' does not necessarily imply that it is useful or acceptable. Alternatively, an intervention may be very effective and its users may have confidence in it, but it may still not be legitimate. Therefore, the 'confidence' of health care workers in a method, tool or intervention does not only depend on its legitimacy, but also on its perceived value to them and others, i.e. the work of reflexive monitoring, which is discussed next.

9.4. Reflexive monitoring - The appraisal work of adapting and evaluating the TRM

The main findings are summarised in Table 9.4 and illustrated with selected quotes:

- The Trigger Review Summary Sheets facilitated implementation;
- The vast majority of participants perceived the TRM as potentially useful in their practice, and also in the wider general practice setting.
- A few clinicians adapted the TRM to better suit their own aims

The TRM was intentionally designed to be flexible. Training sessions introduced generic principles and suggested high-yield strategies, but reviewers were actively encouraged to make minor adaptations if they considered this necessary in relation to their own individual circumstances. Examples of how the TRM was customized included: the type of patient population selected for review; what period of time to review in each record; whether to add extra 'triggers'; and the order in which record sections were screened for the presence of PSIs. Clinicians

then used their own discretion whether to record a detected PSI or not, and whether and how they address it.

The TRM's relative flexibility was identified as an important facilitating factor for its implementation. However, despite the potential to adapt the TRM, only a small minority of reviewers actually did so. For example, in 57/67 (85.1%) trigger reviews the sample of patient records were selected from the recommended 'high risk' patient group, e.g. patients aged >75 years and with confirmed cardiovascular disease. Another example is that only a tiny minority of reviewers recorded optional triggers and even then there was no evidence that they were useful to detect additional PSIs.

The Trigger Review SS was intentionally designed as a one page (double-sided) data collection template with a series of tick boxes and adjustable text boxes to encourage a structured approach to recording the detection of PSIs, improvement actions and reflection and learning. Clinicians generally found the SS quick and straightforward to use. However, a minority of reviewers were at first unable to complete the electronic version because of incompatible IT systems - a problem which was resolved - and a small minority of reviewers indicated that they had struggled to rate the severity and preventability of the PSIs they detected, despite referring to the rating scales on the SS. A small minority of reviewers also suggested minor changes to further improve the usefulness of the SS. Two examples of suggested edits that were subsequently made were: adding a section to record whether findings were shared at a practice meeting and removing the 'priority' ratings of PSIs. Overall, in NPT terms, the work of 'systematization', e.g. collecting, analysing and sharing data with the SS facilitated implementation of the TRM.

The vast majority of participants perceived the TRM as a useful approach to improve the safety of the care they provide. They also recognised its potential for identifying learning needs and points, encouraging reflection and raising awareness of potential safety threats. Overall, they thought the TRM had at least as much value as their existing QI methods. While the TRM's perceived usefulness was identified as an important facilitator of its implementation,

NPT component, with a description relating it to the TRM	Success factors and their effect (facilitating or hindering) on implementation of the TRM	Selected quotes
Systematisation: the work of collecting and analysing information about the TRM, e.g. data about its application, findings and outcomes.	The simple, one-page data collection template facilitated implementation by providing a clear, electronic and structured format. However, some clinicians struggled to rate the characteristics of detected PSIs.	The form's [TRSS] helpful although it's perhaps a reporting tool. It forces you down the route of making you think (GP04) To be honest people will make a judgement for themselves about how important something is and subconsciously will probably use the preventability and the severity in their own head without having to have it written down (GP05)
<i>Reconfiguration</i> : The work of adapting the TRM according to local requirements.	The TRM was intentionally designed to be flexible which facilitated its implementation, with evidence that clinicians modified minor aspects. However, increased flexibility reduces the reliability and validity of inter-practice data comparison.	We used the same list but I don't think we used the same patient's records (GP02)

Table 9.4. A description of the four components of the 'reflexive monitoring' construct with selected quotes

Individual appraisal: The work of evaluating the usefulness of the TRM for the individual clinician implementing it, her/his practice team and patients. The vast majority of participants perceived the TRM as a useful approach to help improve the safety of the care they deliver. They also recognised its potential for identifying learning needs and points and encouraging reflection.

I did find it useful... I quite liked it... I thought helpful, clinically helpful... I thought it was worthwhile (GP02) [We] got some really good outcomes from it: a couple of SEAs and an audit... There's learning for the system in there, so worthwhile, definitely worthwhile (GP04) I like this [the TRM] as a kind of start. Here's something we can do regularly that can actually show us how good we are or how bad we are or areas that we need to work at or where we need to go (PM03)

Communal appraisal: The work of evaluating the usefulness of the TRM for other practices and their patients.

The vast majority of participants perceived the TRM as a useful approach to improve the safety of care in the wider general practice setting.

I think it's more valuable than QOF QP to be honest... You will probably find things that will then benefit more than that one patient in the practice, because you might be able to then look at wider practices within your own GP practice and then beyond (GP08) participants were clear that evidence of its usefulness would not be sufficient to ensure its normalisation. Participants also (implicitly) experienced the usefulness of the TRM for others as less important than the perceived value it had for them. The main facilitating value of 'collective appraisal' work therefore seems to be at the regional and national levels. An example of this is the decision to include the TRM in QOF, which was at least partly informed by evidence of its potential usefulness in previous pilot studies (40, 41).

From an NPT perspective, the perceived usefulness of the TRM was mainly determined through the work of 'reflexive monitoring'. Its implementation was facilitated because the vast majority of participants considered the method as flexible, found it easy to collect the necessary performance data and perceived the TRM as being useful for individual clinicians, practice teams, patients as well as the wider general practice community.

9.5. Discussion of the main study findings in Chapter 9

9.5.1. The relationships between NPT constructs and components

Throughout sections 9.1 to 9.4 of this chapter, the relationships between different factors, constructs and components were highlighted. In reality, these effects were emergent, temporal and fluctuated over the course of the study period. While many examples were provided, any of the constructs and components can reciprocally affect all of the others. This is an important point to consider, as the mechanisms, links and associations between the different factors, constructs and components of NPT have not, to the best of my knowledge, been formally considered before. In particular, how and why do the associations between factors change over time? Which of the links, associations and mechanisms are particularly important for normalisation? Do they vary between settings, contexts and with different interventions and implementers?

In terms of the TRM, the following two examples may help to demonstrate the relative importance of associations, links and relationships between NPT constructs and components. The first example is the highly correlated work of relational integration and legitimation. In both cases, the work seems strongly

dependent on confidence - confidence in the method, the reviewers' ability to correctly apply the method, having the support of the rest of the team and confidence that external agencies support the TRM and will not use the results as an indication of substandard care in their practices. However, the work is also different, because relational integration requires actively involving others and legitimisation work requires justification to become and stay involved with the TRM, which is possible to do even when confidence in the intervention may be low.

The second example of highly-correlated work is that of coherence and reflexive monitoring. Being able to evaluate the potential worth of an intervention (individual and communal appraisal) implicitly implies the need of understanding beforehand what its intended aims and potential value are (the work of individual and communal specification). The work of 'communal specification' therefore began prior to the implementation of the TRM as individuals and teams worked to understand its potential worth for them and others. Their initial perceptions then changed as a result of the work of appraisal. The reviewers then (consciously or unconsciously) searched for practical confirmation of their new understanding as they conducted further trigger reviews. In this way, reflexive monitoring and coherence reciprocally influence and shape each other over time.

9.5.2. Comparison with the literature

A systematic literature review (n=47 studies) in 2010 of the influence of context on quality improvement in healthcare identified the most important factors that determined 'success' and normalisation of interventions. They were (are): senior leadership; organisational culture; information systems; previous experience of quality improvement; clinician engagement and resources (345).

A subsequent, narrative review of the Health Foundation's improvement programmes (n=5) by Dixon-Woods et al in 2012 identified ten common challenges when any attempts are made to improve the quality and safety of health care. The ten challenges were related to three broad themes: design and planning of improvement interventions; organisational and institutional contexts; and challenges related to specific professional groups (346).

More recently, a systematic review (n=57 studies) by Braithwaite et al identified eight common factors associated with successful efforts to improve standards of care: preparing for change; capacity for implementation - setting; capacity for implementation - people; types of implementation; resources; leverage; sustainability; and desirable implementation enabling features (45).

The main findings from the two most recent reviews are summarised and compared in Table 9.5. In addition, the findings from the reviews are 'mapped' to the NPT constructs and components that best describe them. There are three important inferences that can be made from Table 9.5. The first is that a discrete number of factors determine whether complex health care interventions are successfully implemented and eventually become normalised.

Second, the same core set of factors seem to determine whether implementation and normalisation are successful (or not), and these factors can be detected irrespective of the differences in terminology, taxonomies and methodologies of researchers.

The third and final inference is that application of a theoretical framework can facilitate meaningful comparisons of data. More specifically, applying the NPT framework in this study and to the related, international literature, clearly identified the similarities between the key factors that were perceived to strongly influence implementation and normalisation processes. Consequently, it can confidently be asserted that the main study findings reported in this chapter are comparable with the international literature.

Consider, as a practical example, the relative importance of providing adequate resources during implementation. The 'resources' factor is a fairly ubiquitous finding in the international literature, and seems to be independent of geographical location; medical specialty; type of intervention; or scale of improvement initiative (270, 276, 347). In fact, one of the key findings from the review by Braithwaite et al is that 'resources' were the most frequently

Implementation success factors (n=8) identified by Braithwaite et al (45)		llenges in quality improvement (n=10) identified Dixon-Woods et al (346)	Equivalent NPT framework constructs (components)	
1 <i>Preparing for change</i> : the organisation, team and individuals have planned for the intervention			Cognitive participation (initiation, activation)	
	1-3	<i>Design and planning:</i> Convince people there is a problem and a solution; match goals and ambitions to what is feasible	Coherence (differentiation; individual and communal specification)	
2 Capacity for implementation - setting: contextual factors	4	Organisational contexts, culture and capacities	Collective action (skill-set workability, contextual integration)	
3 <i>Capacity for implementation - people</i> : There are enough people with the necessary skills, knowledge and willingness to implement the intervention	5	Leadership	Cognitive participation (enrolment); Collective action (skill-set workability)	
4 <i>Types of implementation</i> : the intervention meets the needs of the organisation			Coherence (communal specification)	
5 <i>Resources</i> : adequate and appropriate resources are available, including discretionary budgets, manageria support, Infrastructure, technology, time and staff	l		Collective action (skill-set workability)	

Table 9.5. Implementation success factors, quality improvement challenges and the NPT framework

6	<i>Leverage</i> : there is support and momentum throughout the implementation process; clinical champions			Legitimation
7	<i>Sustainability</i> : this requires planning and ongoing effort	6-7	Securing sustainability and remaining vigilant for unwanted consequences	Sustainability is the product if the work of all four constructs are done effectively
8	Desirable implementation enabling features:			
	Communication and collaboration			Collective action (Relational integration)
	Clear implementation strategy			Coherence (individual specification)
	• Teamwork	8	Tribalism and staff engagement	Collective action (Relational integration, skill- set workability)
	• Monitoring, evaluation and feedback	9	Establish data collection and monitoring systems	Reflexive monitoring (systematization, individual and communal appraisal)
	• Incentives	10	Incentivise participation	Cognitive participation (legitimation)
	• Flexibility, tailoring implementations to the local context			Reflexive monitoring (reconfiguration); Coherence (internalization)

identified factor that were considered to be a determinant of successful implementation (45). Similarly, in this study, the vast majority of participants also identified adequate resources as the single most important factor determining whether an intervention will become normalised (or not).

Three additional examples will be provided next to further support the critical inferences from Table 9.5, and the assertion that the main findings from this study are comparable with the international literature. The study designs of all three were informed by NPT, but they were also selected because they are exemplars of: (i) implementing an intervention in the UK, but in a *non-GP* health care setting (270); (ii) implementing *multiple interventions at scale* (347); and (iii) implementing an intervention in general practice, but in a *non-UK* country (276).

The first example is Bamford et al who used NPT and qualitative methods to determine the barriers and facilitators of nutrition guideline implementation in UK residential care homes. They report that '*improving nutrition only appeared to be a priority if it could be achieved within existing resources*' (270). They also found further significant barriers, including: lack of leadership and managerial support and staff perceiving the implementation as threats to their autonomy and expertise (e.g. the work of cognitive participation malfunctioning); lack of feedback from clients to cooks, or feedback mediated by care staff with biased view (e.g. no reflexive monitoring work was or could be done); scepticism over the value of the guidelines which were perceived to be incompatible with existing practices (e.g. much coherence work was required); and the lack of nutritional knowledge or engagement of the cooks (e.g. the work of collective action could not properly be done). Unsurprisingly, the implementation of nutrition guidelines was 'challenging', although some changes did become embedded in routine practice.

The second example is the Delivering Assisted Living Lifestyles at Scale (DALLAS) programme. This was an ambitious programme attempting change at a much larger scale than this study or previous studies underpinned by NPT. DALLAS was a national, pan-UK programme aiming to promote health and well-being through the delivery of a broad range of digital services and products and was evaluated

using a rigorous, mixed methods and NPT approach (347). Some of the key challenges to normalisation in the DALLAS evaluation are therefore less relevant to this study, such as 'branding and marketing issues' or establishing and maintaining 'multi-agency partnerships'. However, other factors such as the 'need for resilience' and 'information governance' are instantly recognizable and equally applicable.

The third example is Franx et al's introduction and evaluation of a stepped-care model for managing depression in primary care in the Netherlands (276). The model had three components: patient differentiation (reliable identification of patients suitable for treatment); stepped-care treatment (being aware of and accessing a range of available treatments); and monitoring patient outcomes. The main facilitating factors were the model itself, which was understood and accepted (the work of coherence), regular structured team meetings (the work of cognitive participation) and positive feedback from patients (the work of reflexive monitoring). However, implementation was hindered by a lack of resources, underdeveloped IT systems and different perceptions of depression in the wider multidisciplinary team (the work of skill-set workability, relational and contextual integration). Consequently, the authors found evidence of 'strides towards utilizing' the model but not normalisation during the study period. The fact that some improvements were made should be acknowledged and celebrated as 'it seems unlikely that the changes reported by the clinicians would have occurred by itself within the primary care teams' (e.g. without external intervention).

Conclusion

This Chapter described a large number of factors that were perceived as facilitating or hindering the implementation of the TRM in general practice. Of these, four main factors were identified by the vast majority of participants as being essential for the successful implementation of the TRM. The first and most important factor is provision of adequate resources, and in particular allocating sufficient, protected time to perform trigger reviews. The second factor is legitimising the intervention through incentives and endorsements by the government or a professional body. An example would be the RCGP approving the TRM for specialty training, appraisal and revalidation purposes. Financially incentivizing the TRM by including it in the pay-for-performance QOF greatly facilitated both these factors ('resources' and 'legitimising'). The third factor is the characteristics of the participating clinicians, e.g. their degree of engagement, autonomy and pre-existing knowledge and skills. The fourth factor is the perceptions of the general practice teams about the TRM's utility.

Normalising innovations and interventions in health care require multiple approaches, which may sometimes appear contradictory: strong leadership is essential, but within a participatory culture; initiatives should have clear aims, but also be flexible and respond to local needs and contexts; performance should be monitored with regular critical feedback, yet without attaching blame; properly designing and planning an intervention is important, yet momentum and pace have to be maintained; clinician engagement is essential, yet some changes will of necessity go against the grain of existing professional cultures and customs. A key message from this study, and also from the international literature, is therefore that '*there is no magic bullet in improving quality in health care*' (346, 348). However, while it is true that implementing change is time consuming and normalisation depends on many, complex factors potentially hindering and facilitating at multiple levels, it is also true that improvement is possible! The final Chapter will therefore consider the likelihood of the TRM being normalised, but also whether the TRM *should* be normalised.

Chapter 10. Discussion

Chapters 6 to 9 described the main study findings, compared them with the international patient safety literature, and identified some of the important practical implications. This chapter begins by summarising the aims and the main findings of this study in Table 10.1. Next, four specific questions are raised in relation to this study, and the most likely answers are provided. The four questions are:

- 1. Should the TRM be normalised in general practice? This is arguably the most important issue to consider from the perspective of this study, which aimed to determine the usefulness of the TRM. The short answer is 'yes', and three reasons will be provided in justification.
- 2. What is the likelihood of the TRM being normalised in general practice? This question relates to the study aim of determining the barriers and facilitators to the implementation of the TRM. The study findings and international literature suggest that normalisation of the TRM would be very likely if all four of the following factors could be guaranteed: practice teams and clinicians have the necessary knowledge to apply the TRM effectively; there is senior leadership support for the TRM; adequate resources (including time) are provided; and it is formally integrated into existing general practice contexts such as GP specialty training (GPST) and appraisal and revalidation. However, given that there are currently no financial incentives or mandatory requirements for practices and individual GPs to use the TRM, normalisation seems unlikely.
- 3. What are the strengths and limitations of this study? This question will be answered by considering the strengths and limitations of the study in general, with additional subsections to consider the strengths and limitations of NPT and the TRM.
| Aims | Chapter | Findings |
|---|---------|---|
| To describe the patient
safety perceptions of
general practice
clinicians and staff | 6 | Participants perceived patient safety as
important, integral to care and amenable to
improvement. Medication and medication-related processes
and elderly, housebound patients were
considered to be particularly at risk for PSIs. All of the practice teams were already using a
range of improvement methods. They strongly
indicated that additional resources were critical
prerequisites for their future participation in
improvement initiatives. |
| To determine the
usefulness of the TRM
by describing the
outcomes from its
implementation | 7 | 47 Primary care clinicians performed trigger reviews of 1659 electronic medical records and detected 216 PSIs. A substantial minority of PSIs (29.2%) were associated with moderate or more substantial harm, the majority (54.8%) were rated as preventable or potentially preventable, and the most common type of PSI related to 'medication' (40.7%). Reviewers undertook specific patient safety-related actions during and after approximately two thirds of trigger reviews. |
| To explain how the TRM works | 8 | The TRM enabled the detection of PSIs through
a combination of three factors: structure;
mindset (knowledge); and opportunities. Detecting PSIs created potential learning
moments and the option to accept ownership of
the findings. Ownership of PSIs facilitated subsequent
reflection, learning and improvement actions. |

Table 10.1. Summary of the aims and main findings of this study

To identify and describe the main factors that facilitated or hindered the implementation of the TRM in general practice 9

Implementation of the TRM was facilitated when:

- Participants understood that it was compatible with their existing work, and perceived it as feasible and acceptable.
- Clinicians had the necessary knowledge to conduct effective trigger reviews
- PSIs were detected quickly, and the PSIs were unambiguous, serious, preventable and originated in primary care;
- Adequate resources and time were allocated;
- It was included in GP contexts such as QOF, appraisal and GPST
- Reviewers had senior leadership support and opportunities to enact change
- Participants perceived it as useful for them and others

4. Can recommendations be made as a result of the study? The answer is 'yes'. Recommendations can be made about implementation processes and improvement interventions in health care settings, in relation to the TRM as well as about current and future research priorities.

10.1. Should the TRM be normalised?

This section provides three reasons why the TRM should be normalised. They are:

- this study contributes further evidence that the TRM is in the main useful, acceptable and feasible as an approach to improve patient safety in UK general practice;
- (ii) For the overall safety of care to improve in health care, a range of complementary methods will be required; and
- (iii) The TRM provides clinicians and researchers with a unique perspective of patient safety incidents (PSI) and how they may be reduced.

10.1.1. The TRM is in the main a useful, acceptable and feasible approach to improve patient safety in UK general practice

All the clinician reviewers in this study were able to perform trigger reviews and submitted Summary Sheets. The vast majority of reviewers detected PSIs and the majority took or intended to take subsequent actions to improve care standards. They also reported learning points and some felt they were more vigilant for potential safety threats. It is highly unlikely that these positive outcomes would have occurred if the TRM had not been implemented. However, it is unclear whether the positive effects had a subsequent, measurable impact on clinical outcomes, and the cost-effectiveness of the method was not estimated. In addition, the safety 'mindset' of some reviewers was temporary and eroded over time because of the need to refocus their attention and efforts on other, competing priorities for their time.

The personal accounts provided by some of the study participants demonstrate how some PSIs can provoke a more powerful learning experience than others. Typically, the more memorable PSIs were the ones that had resulted in serious harm and were clearly preventable. Given that most of the PSIs detected in this study were of low to moderate severity and only about half were considered preventable, it may help to explain why the increased vigilance for safety threats was transient. However, as was described in Chapter 9, some participants described minor incidents that had caused them significant emotional distress and continued to exert a powerful influence on their clinical practice years later. The recommendation is therefore that clinicians perform trigger reviews regularly to help refresh their safety awareness. By the same reasoning, this recommendation would also apply to other improvement methods. The reviewers in this study felt two trigger reviews a year would be feasible, allow them time to disseminate and act on the findings *and* be frequent enough to retain the required knowledge to apply it effectively.

10.1.2. The overall safety of care can be improved, and this will require a range of complementary methods.

This study contributes to the growing evidence about the potential value of the TRM as an approach to improve patient safety in general practice (40, 41, 48, 349). Most Scottish general practices have now undertaken at least two trigger reviews, detected PSIs and implemented or considered a range of improvement actions. However, even in the unlikely scenario that every reported improvement action had been successful, the reality is that there will still be latent risks within all general practices, and as a result preventable PSIs *will* still occur in the future.

The lack of reliable measures of improvement is not unique to the TRM. Despite more than a decade of sustained attention and considerable investment and research in patient safety, there is still little reliable evidence of widespread reductions in harm rates in any health care setting (226, 350-352). Why is this? One answer that has been proposed is that, compared to traditional biomedical research and initiatives, fewer people have been working in this area for a relatively shorter period of time with only modest resources and without the advantage of many decades of previous evidence. In other words, 'we get what we pay for' (226).

Another potential reason, which is particularly relevant to this study, is that there are few interventions available that can reliable reduce PSIs. A systematic review by Lorincz et al in 2011 found no credible evidence of effective strategies to improve patient safety in primary care (35). The authors concluded that *'there are no magic bullets'*. It therefore seems obvious that the priority should be to develop new safety improvement methods, or at least further develop and test existing ones. But the question remains - which ones? Evidence-based interventions that target specific complications of care, or more general strategies like the TRM, with the potential to reduce a range of PSIs? Both approaches seem to have merit, but for now it is impossible to objectively recommend one over the other. All of the improvement methods and strategies for the general practice setting that were described in Chapter 3 have specific strengths, but also limitations. Two examples are provided to further illustrate this important issue.

The first example is the study by Tam et al who compared the strengths and weaknesses of chart review, patient surveys and voluntary reporting in detecting medication misadventures in general practice in Hong Kong (83). They found chart reviews uncovered significantly more preventable adverse drug events than incident reporting, but noted the 'pivotal' role voluntary reporting had for drawing attention to sentinel events. The second example is a study comparing clinical record review with four reporting systems in the Netherlands. These were: informal reporting by patients; formal complaints by patients; medicolegal claims; and voluntary incidents reported by healthcare professionals (353). They found only 18/498 (3.6%) of PSIs uncovered through record review were also detected by the reporting systems.

The two main findings and implications from these examples are consistent with the patient safety literature (84, 85, 237, 335, 349). The first finding is that clinical record review detects more PSIs than any other method. Based on this finding alone, the implication seems to be to select it in preference of the other methods. However, the second finding, that there is very little overlap between the PSIs detected with the different methods, negates this implication. For example, litigation and complaints are predominantly about communication breakdown and missed diagnostic opportunities, while the majority of PSIs detected through the TRM relate to medication-processes. This strongly supports the view that a mixed methods approach is required to improve patient safety.

It is now widely accepted that 'a single measure of safety is a fantasy' (354). Rather, the different methods are complementary to each other. The implication is that, if the overall safety of general practice is to be understood and improved, the best approach would be to combine the different methods (349, 355-358). Such an approach would also help to mitigate the potential risk of using a single improvement method or measure, which is that, 'given one measure of success, almost any group can be successful in the short term by optimizing that measure at the expense of other important measures' (359). This approach is already being taken in the UK through, for example, the introduction of the RCGP Patient Safety Toolkit (179, 180). The Toolkit contains a range of improvement methods for the general practice setting, including the TRM, thus offering a choice of methods to clinicians and practice teams.

10.1.3. The TRM provides a unique perspective of PSIs

Patient safety has been compared to an elephant in the well-known Indian fable of the blind men (296, 360). Just as each of the blind men 'saw' only a part of the elephant, specific improvement methods only produce evidence about certain aspects of patient safety. The moral of the fable is that, while each individual perspective may provide useful and unique insights, they are also limited and incomplete. Accordingly, the TRM is valuable because it increases our understanding about the prevalence and characteristics of PSIs, which are important components of the overall safety of care.

The types of PSIs uncovered by the TRM will most likely be familiar to primary care clinicians and are well known and described in the international patient safety literature. The important issue is that they had remained undetected until the TRM review was undertaken. Or, more likely, they may have been incidentally and opportunistically identified but were not shared or addressed because of a lack of time. As was described in Chapter 8, the TRM worked by creating opportunities and a 'mindset' for detecting and acting on PSIs.

A key finding from this study was that many participants associated the severity and preventability of PSIs with their usefulness (chapter 7). In other words, the outcomes of incidents, rather than the contributing factors to them, seem to influence the subsequent actions and learning of many individuals, teams and organisations. In reality, there can be as much or even more potential for learning and change in some near miss events (326), but this concept is challenging to convey to clinicians and staff. One way to address this issue may be through the culture of the organisation and will be considered on page 264.

Another implication is that a single approach or method should not be promoted as the definitive 'solution' to substandard care. Nor can any method be expected to detect *all* PSIs or create perfect patient safety. That is not to say that all methods are equally useful - they are not. For example, the TRM detects *more* PSIs than any other available metric. In this small study, PSIs were detected and reported at a rate of one incident for approximately every eight records that were reviewed, or approximately one incident for every two years of recorded care. While the precision of this rate is uncertain, it provides an estimate of the prevalence of PSIs in general practice that is comparable with larger studies (48). It also raises important questions. Are the estimated rates of PSIs 'high' or 'low' and do they justify further interventions?

These questions will be considered in more detail in subsequent sections. For now, normalisation of the TRM can be justified in part by the value of the evidence it produces. As our understanding of the nature and frequency of PSIs increase, important questions about the future of health care and the equitable and effective allocation of resources can be more objectively framed and considered.

10.2. Will the TRM be normalised?

The previous section considered the question whether the TRM should be normalised in general practice. The short answer was 'yes', because it is useful. However, while the 'usefulness' of an intervention is an important factor in determining whether it is normalised or not, there are other factors that are equally or even more important. This section therefore describes four additional requirements which, if met, will make normalisation of the TRM very likely. At the time of writing, only some of these requirements have been met. My prediction is therefore that normalisation of the TRM in UK general practice will either be a gradual and piecemeal process or may not happen at all.

10.2.1. Clinicians require sufficient knowledge to apply the TRM effectively

The success of improvement methods, including the TRM, is contingent on the critical assumption that clinicians possess the requisite *knowledge* - as defined in Chapter 8, page 200 - to effectively use them. Moreover, that they are able to apply the technique correctly, produce robust data, analyse and evaluate their findings and then plan and implement meaningful and sustainable improvements. A further assumption is that in those instances where this knowledge is lacking, health and educational authorities are able to up-skill the workforce on the scale necessary to support proposed initiatives. But how realistic are these assumptions? The TRM and this study provide a practical example in helping to answer this question.

Successful implementation of improvement interventions requires clinicians to have knowledge of three specific processes (361). In relation to the TRM, the knowledge is: (i) how to detect PSIs in order to identify a 'problem' or deficiency in care and then define it clearly and correctly; (ii) be able to analyse the problem and correctly identify the main contributing causes and decide whether they are sufficiently important to justify further action, e.g. rate the preventability and severity of PSIs; and (iii) be aware of, select and implement the most appropriate remedial action and evaluate its effectiveness. These processes are consecutive and interdependent, e.g. a process can only be undertaken if the previous one had been successfully completed. Performing a proper analysis is difficult though, and a skill that often requires training and practice (361). Even when clinicians *are* able to detect PSIs and identify contributing factors that may be amenable to action, they still need to choose or design and implement a feasible solution, evaluate the degree to which this addresses the perceived problem and mitigate any unintended consequences.

Many study participants were able to demonstrate knowledge of all these steps. However, as was described in Chapter 7, a small minority were unable to complete the first step of detecting PSIs, while a few more participants struggled with the third process, i.e. taking action to improve care. The assumption that all clinicians possess the necessary knowledge to successfully implement the TRM - and by extension other quality improvement interventions therefore seem overly optimistic.

On the other hand, the findings from this study indicate that overall, enough of the participating clinician reviewers had adequate knowledge after training to ensure normalisation of the TRM in general practice. Admittedly, normalisation would be more rapid if the knowledge of a minority of clinicians could be improved further. 'Knowledge' - or lack thereof - is also only one of the factors that determine the normalisation of an intervention. Even so, when safety improvement initiatives are evaluated, the knowledge of participants should routinely be considered as an important contributing factor to the observed outcomes (362).

Given the importance of knowledge, and the fact that some experienced clinicians already have many of the required core safety skills for improvement initiatives (363), should certain clinicians or professional groups be preferentially recruited and trained to use methods such as the TRM? While it is true that in this study the nursing group found fewer triggers and PSIs and were also less likely to document improvement actions or learning points compared with the medical groups, individual nurse reviewers did demonstrate the ability and knowledge to effectively implement the TRM. There was also one GP reviewer who did not detect PSIs. The implication is therefore that practices should select reviewers with the necessary *knowledge* to conduct TRM, as this is a more desirable characteristic than their professional roles, e.g. GP partner, salaried GP, GPST, practice nurse or nurse practitioner.

Providing suitable training for participants was identified as one of the facilitating factors of the implementation of the TRM in this study (Chapter 9). Training is therefore important and one of the recommendations that are made in relation to the TRM and future implementation initiatives in section 10.4, page 279. However, it is not necessarily synonymous with knowledge. A systematic review (n=39 studies) to determine the effectiveness of teaching quality improvement theory and methods to clinicians found that their self-reported knowledge and confidence improved, but there were little or no evidence that the training had any clinical benefits (364). One interpretation of these findings would be that for training to be truly effective, it needs to occur in conjunction with adequate resources, protected time and opportunities for reflection (365). In fact, it has been suggested that one of the best ways to improve patient safety in primary care may simply be for GPs 'to set aside time and space to be allowed to conduct the required, appropriate reflection *effectively*' (52).

10.2.2. Senior organisational and practice leadership support

The support of the practice leadership for the TRM was a key determinant of its successful implementation. In fact, this study would not have been possible without it. It was the practice manager or a senior GP in each practice who agreed to participate, who decided when and how much time and resources to allocate and who recruited additional team members. Furthermore, the practice leadership helped determine through which forums trigger review findings were disseminated and whether proposed improvement actions were implemented at the practice level. The implication is therefore that the practice leadership strongly influence not only whether an intervention is implemented, but also how and to what degree subsequent improvement and learning occur at a local level.

The powerful facilitating effect of leadership on the implementation and normalisation of interventions became even more apparent at regional and national levels (362, 366). Policymakers and senior healthcare executives design the financial (and other) incentives that are linked to the specific performance objectives they set. They also have the authority to mandate a wide range of activities deemed desirable. Individual practices and clinicians retain their clinical autonomy and freedom of choice, but only within predefined parameters. The visible and strong support of senior leadership for an intervention is therefore a key factor in determining whether it becomes normalised or not. This is why Buist and Middleton strongly recommend that clinicians '*in positions of responsibility, need to take ownership of the quality and safety agenda*' and '*claim the agenda*' (367).

The participants in this study were aware of the importance of leadership within their practices but also at higher levels. A GP partner explained the common perception that '*its decision makers actually who normalise things to a certain extent because they mainstream them and they bring them in and support them*' (GP08). This perception seems to be widely shared amongst health care professionals. For example, a qualitative study of chronic disease management in UK primary care found participants thought Telecare would only be normalised if it was made mandatory through a formal policy (268).

An important implication of these findings is that it is the practice leadership who ultimately creates the workplace culture (350, 368). Establishing a 'just' culture that enables the whole team to support and advance patient safety is, therefore, only possible with strong leaders. It is for the practice leadership -GPs, management and senior nursing staff - to facilitate and build a culture of trust that encourages effective team-working, collective learning from patient safety incidents and strong communication across the clinical disciplines and administrative staff. They have both the responsibility and the authority to ensure that there is a continued focus on improving the safety of patient care in essence, to establish safety as a cultural 'value' as well as a practice 'priority' (350). All participants in this study described the prevailing culture in their own teams and also in Scottish general practice as overall positive, strong and safetyorientated (Chapter 6). While it may be tempting to speculate whether or not the perceptions of safety culture reflect reality, it is actually the degree of variation between the different staff groups that is of more importance. A consistent finding of safety culture research - irrespective of industry or geographical setting - has been that the larger the variation in the perceptions of different staff groups is, the more likely PSIs become (311). The implication of the homogeneity of the safety culture perceptions in this study therefore suggest some measure of a protective effect against PSIs in general practice.

10.2.3. Normalisation requires adequate resources

A key finding from this study and from the comparable international literature is that normalisation of an intervention requires the provision of adequate resources, including time (347, 366, 369, 370). While this may seem self-evident, in practice many improvement interventions receive funding only for the implementation stage, and even then the initial investments are not always adequate. This study is no exception, as the professional fee offered to participating practices for implementing the TRM were not commensurate with the amount of time that they were expected to allocate for trigger reviews. It is therefore unsurprising than many interventions fail to become normalised despite evidence of their usefulness. On the other hand, interventions may become normalised if there are sufficient incentives to continue using them even when there is little or no evidence of their effectiveness.

The most common business model of general practice in the UK is that of independent contractors to the NHS. In order for practices to be viable business concerns, the partners and managers have to continually balance workload with available resources. They are able to adjust their workloads and increase resources to a degree, but patients and policy makers also exert significant influences - both positive and negative. The inherent challenges in this process can be conceptualised with an analogy of balancing a seesaw (Figure 10.1.).

The 'seesaw' analogy will be illustrated further with two practical examples from this study. The first example was provided by one of the practice managers (PM08). She explained that the practice team understood the importance of correct clinical coding and considered this a priority. There was agreement that the task should ideally be performed by a clinician, but the practice was unable to afford this option. Even if they could have afforded the additional clinical time, there was an even greater need for more consultations. The leaders therefore decided to assign the role of 'coder' to an administrative team member and provided the appointee with additional training and support. They acknowledged there may have been (and will likely be) some instances of incorrect coding, and that this may contribute to PSIs, but they perceived this as the only feasible option for them. Using the analogy of the see-saw, they reached a compromise between risk reduction and resource allocation.

The second example is the TRM and the Scottish QOF. Chapters 1, 2 and 9 described how financially incentivising the TRM through the QOF greatly facilitated its implementation. For practice teams, the relative increase in workload from implementing the TRM was 'balanced' by an increase in their resources, hence their willingness to implement it. When the QOF was discontinued in 2016, the 'resources' for the TRM were also removed (e.g. one end of the seesaw became unbalanced). The most likely response of most practices in Scotland will be to restore balance by reducing the workload, e.g. stop using the TRM. The main implication is therefore that the TRM will not be normalised. However, it is possible that the relative importance of the other facilitating factors, such as senior leadership support and contextual integration, may increase in their relative importance and be sufficient to still ensure normalisation. Nevertheless, even if this happens, normalisation will still require a much longer period of time and will likely occur in a piecemeal manner.





The two examples help to demonstrate the challenges clinicians and staff struggle with in modern health care systems to effectively balance opposing priorities. The importance of resource allocation in these instances cannot be overstated. The stark reality is that practice teams and individual clinicians are deciding - whether consciously or unconsciously - how much patient safety they can afford. If we accept that resources are finite, at least two important questions should be asked (371). The first question is how much safety can we afford in our current health care systems? The second question, strongly related to the first, is who should have the responsibility to make this decision?

These questions are challenging, emotive and currently shrouded in uncertainty (372). Very few patient safety studies evaluate cost-benefit/utility because it requires considerable resources, expertise and often data that are unavailable (373). A recent systematic review of those studies that did perform comparative economic analyses of patient safety improvement strategies in acute care settings found only four were 'economically attractive': pharmacist-led medication reconciliation; ICU intervention for central line-associated bloodstream infections; chlorhexidine for vascular catheter site care; and surgical sponge counts (374). When the same questions are asked about the primary care setting or in relation to this study there are simply no reliable answers. How much effort would be required to reduce the estimated rate of PSIs that were detected in this study? What is an achievable target? What are the opportunity costs to patients, practices and the NHS if practices decided to allocate time to this problem, rather than some other priority?

The different perspectives of why PSIs occur were described in Chapter 6. The main implication was that in socio-technical systems - such as general practice - the number and severity of PSIs may be reduced but can never be completely eliminated, irrespective of how much effort and resources are invested. In fact, 'efforts to reduce uncertainty towards zero result in increasing marginal costs with diminishing marginal returns for patient safety' (371). Vincent and Amalberti caution that 'primary care practitioners cannot (and emphatically should not) try to minimise all possible risk' as this would be 'completely unaffordable' (366). Only once this fact is publicly acknowledged and accepted,

will it be possible to have meaningful discussions of what feasible targets for patient safety should be. Until then, a succession of improvement initiatives may be decried as ineffective, incomplete or inadequate if they are evaluated against unrealistic targets and expectations.

Unfortunately, the notion that patient safety is imperfect is unpalatable, and for a minority inconceivable. So is acknowledging that resources are finite. For now the easiest, and ironically also the safest course of action for many policy makers and senior leadership in the NHS seems to be leaving the 'seesaw' for frontline staff to balance (366). Then, when the inevitable imbalance results in a 'fall' it is all too often individual clinicians who are held to account. In the longer term, the only sustainable solution will be for patients/society and policy makers to accept responsibility for their part in the 'seesaw' problem and actively contribute alongside clinicians to achieve the degree of balance that is acceptable to everyone (375). It *is* possible to improve the standards and safety of care, but it will require perseverance, ingenuity and realistic expectations from everyone (376) and solutions that can 'only be achieved at a national level' (366).

10.2.4. Normalisation is facilitated by contextual integration

One of the first tasks for researchers in implementation science is to describe the contexts within which they intend to implement an intervention, and consider how contextual factors may influence their proposed interventions (243). Chapter 9 described how implementation of the TRM was facilitated through the work of 'legitimisation' and its 'contextual integration' into existing general practice processes. In this section the potential influence of two particular contexts on further integrating, embedding and eventually normalising the TRM will be considered: GP specialty training (GPST) and medical appraisal and revalidation.

10.2.4.1. GP Specialty Training

UK GP Specialty Trainees are required to spend 18 months in a GP setting as part of a 3 year programme. The teaching required is governed by the Royal College of General Practitioners (RCGP) curriculum, and one area that is increasingly being highlighted is patient safety (240). Specific learning objectives are also defined which require the trainee to demonstrate a whole range of problemsolving skills aimed at improving the management of clinical risk and enhancing the patient experience of care. In response, medical educators have integrated patient safety-related topics and issues into undergraduate education and specialist training programmes.

As was described in chapter 7, the vast majority of trainees in this study were able to use the TRM to detect preventable PSIs directly related to issues within the practice, particularly in high risk elderly patient group (329). All participants were able to demonstrate some element of reflection, document potential learning needs and develop improvement action plans. The detected PSIs were directly related to issues within the practice which enabled focused discussion with educational supervisors and other colleagues, potentially facilitating rapid implementation of learning and remedial actions. The implications are that using the TRM during training may facilitate both formative and workplace case-based discussions and assessments. The TRM may therefore have a role to play in specialty training to help prepare GP trainees for the contractual and regulatory demands of independent clinical practice and future safety improvement expectations. However, in order for the TRM's potential value as an educational to be realised, further research would be required to explore and validate this application and educational supervisors would require additional training and support.

10.2.4.2. GP appraisal and revalidation

In terms of regulatory and educational policy in the United Kingdom, 'safety and quality' is one of four professional domains describing the expected duties and standards of every doctor registered with the General Medical Council (GMC) (238). All registered doctors are expected to participate in systems of quality assurance and improvement, perform regular reviews of and reflect on their performance and take action to improve the quality and safety of care they deliver if necessary. The TRM is well aligned with this expectation and, given the evidence presented here, can play an important role in helping to achieve this standard. In NPT terms the TRM is also legitimised through its formal inclusion in the RCGP Patient Safety Toolkit for general practice (180).

The study findings also demonstrate the complementary nature of the TRM in relation to established improvement methods in general practice. PSIs that are detected by trigger reviews can serve as topics for SEA and criterion audit. This is particularly helpful given that appraisal and revalidation requires GPs to analyse two significant events per year, with the GMC encouraging these events to be PSIs rather than broader quality of care issues. Identification and analysis of previously undetected PSIs is therefore particularly pertinent to improving the opportunity cost of SEA topics.

10.3. Strengths and limitations of the study

This section begins by considering the strengths and limitations of this study. The specific strengths and limitations associated with the NPT and the TRM in the context of implementation science and patient safety research are then described.

10.3.1. Study limitations

This study has at least five limitations. In addition, a number of specific limitations in relation to NPT and the TRM were identified (Sections 10.3.3 and 10.3.4)

The perceptions patients have of health care and their participation in research and improvement initiatives are widely recognised as important (377-380). The first limitation is therefore that patients could not be actively involved in this research, despite careful consideration of this option. However, all study participants recognised and supported the crucial roles and responsibility of patients in relation to the safety of their own care (Chapter 6).

The second limitation is that the *sampling strategy* was a pragmatic choice informed by time constraints, available resources and ready access to GPSTs and general practice teams through their association with NES. The study findings are therefore derived from the experiences and perceptions of participants who have previously demonstrated an interest in research and improvement initiatives. The sample may therefore not be representative of general practice or GPSTs in Scotland or, indeed, other countries in the UK or internationally. This potential limitation was mitigated to some degree by ensuring the sample included training and non-training practices, practice size varied from small (a single GP) to large (>10 GPs) and were spread across urban and semi-rural areas. The subsequent QOF data which were presented in Chapter 6 were aggregated from essentially all practices in the two NHS Boards from which the study sample was derived, and did not reveal significant differences compared with the study sample.

From a quantitative perspective, the size of the sample of records that were reviewed was determined beforehand by resource considerations. Applying the formula proposed by de Wet et al for determining the minimum number of records that have to be reviewed in order to ensure the detected PSI rate is reliable, suggests that the sample of records in this study was adequate to determine PSI rates with acceptable levels of precision, but would not have been adequately powered to detect a significant change over the study period (201).

The third limitation is in relation to the GPST component of the study. It is unknown why five of the trainees who attended training workshops were unable to undertake trigger reviews and submit completed summary sheets. Knowing 'why' may have provided us with greater insights into the overall utility of this method and associated implementation issues. However, it seems likely that the reason may have been timing, as the submission date for trigger reviews coincided with the final two weeks of their training programme. The fourth limitation is that all study data were reported, and therefore subjective. Consequently, it is unclear whether all the improvement actions clinicians indicated they considered were in fact undertaken. A related issue which will be discussed on page 278 is that inter-rater reliability was not measured. It is therefore possible - even likely - that if clinicians all reviewed the same samples of records, they may have detected different PSIs, provided higher or lower PSI ratings and considered or undertaken other improvement actions. However, while the IRR may affect the quantitative outcomes of this study to a degree, e.g. by over or underestimating the incidence of PSIs, it does not meaningfully affect the qualitative data or the main findings. While it must therefore be recognised that direct observation is the 'gold standard' of socio-technical studies, it is seldom feasible to do so. Instead, researchers 'do the best we could', and accept that, in many instances, 'interviews were the only window into events that happened far from the researcher's gaze' (271).

The fifth limitation is in relation to the analysis of the qualitative data. Despite a concerted effort to minimise researcher bias through the strategies that were described in Chapter 5, analysis of qualitative data is inevitably influenced by the previous experiences and other characteristics of the researchers. The findings must therefore retain some measure of subjectivity. However, the risk of incorrect assumptions and conclusions were reduced through a combination of reflection, rigorous application of a clearly described and acceptable analysis process and by evaluating the veracity of the results against the international literature. It could also be argued that the subjectivity of analyses is not only a limitation but, if accounted for, has the potential to enrich the data and make the findings more accessible to others.

The potential limitation of bias should not only be considered in relation to the methods of this study, but also to patient safety incidents in a more general manner. Patient safety narratives are 'constructed and re-constructed' by individuals and health care organisations and in the process their meaning change (381). The narratives of PSIs evolve through each process, starting with their detection, the way in which they are recorded and shared, whether and how further analyses are conducted, in the dissemination of identified learning

points and needs and finally whether they help to inform subsequent improvement actions. Waring described how the narratives of PSIs change through 'washing' of the original experiences of the patients and clinicians until - often unintentionally and unconsciously - their very 'form, meaning and content' have been 'standardised' in a way that is recognisable to organisations (381). It is unclear to what extent the same constructing and re-constructing of PSI narratives currently happen within practices and also in the wider primary care context. While the benefits and potential losses associated with this process are currently unknown, the intuitive implication seems to be that there is a need for a shared understanding of PSIs amongst all health care staff groups. In NPT terms, the recommendation is therefore to increase the work of coherence.

10.3.2. Strengths

In 2013, Schildmeijer et al's focus group study was the first known attempt to explore the strengths and limitations of clinicians working with the Global Trigger Tool (GTT) in secondary care settings (382). A unique strength of *this* study is that it is the first known attempt to investigate how the TRM is experienced and implemented in primary care by exploring the perceptions of clinicians and their general practice teams.

A second strength of this study is that it is informed by a validated theoretical framework, which is recommended for research in the discipline of implementation science (251). As was previously described in Chapter 4, NPT provides a conceptual framework and a socio-technical lens through which factors that hinder or facilitate normalisation of an intervention can be identified, described and understood.

A third strength is the study design, which is longitudinal, incorporated a mix of methods and collected qualitative data at two critical time points. The mixedmethods study design was selected because this approach is considered to be particularly useful for researching implementation processes and complex health care interventions and helped to enhance the reliability of the main findings. A fourth strength of this study is that the perceptions and experiences of the three different staff groups that were critical to the successful implementation of the TRM were considered. In order to reflect the reality of modern general practice, which is that it is delivered by teams, the qualitative data included 'matched' interviews within each practice, e.g. a practice nurse, GP and practice manager were interviewed, where available, using the same schedules. A further benefit of this approach is that it allowed the work of implementing the TRM to be investigated and understood at the levels of individuals and the practice teams (272). In addition, the interviewer had in-depth knowledge of the TRM and had good working relationships with the study participants. The interviews were candid, detailed and in-depth and it is evident from the transcripts that participants felt no obligation to report 'successes' with the TRM or the implementation process.

From a qualitative data analysis perspective, a key strength of the study is that the analysis were conducted using recognised methods in order to ensure robustness and transparency of the process (283, 285, 288). For example, the reflexivity, rigour and reliability of the findings were increased through data clinics and constant comparison (383). The perceptions of the vast majority of participants were highly congruent and independent of clinical, non-clinical, management or non-management stratification. The observed homogeneity is arguably a success factor in its own right, as it reflects high levels of coherence, which in turn increase the ability of teams to effectively enact interventions. In NPT terms, practice teams were able to successfully perform the necessary work of interactional workability.

The potential limitations of the sampling strategy have already been discussed. By comparison, and in terms of qualitative research, the sample is fairly large, thematic saturation was achieved and more interviews would not have materially strengthened the main findings.

10.3.3. The strengths and limitations of Normalisation Process Theory

Given the proliferation of available theories, each with a unique focus and intent, researchers are strongly advised to explicate their rationale for selecting a specific model, framework or theory (245). In the case of this study, the intervention - e.g. the TRM - had already been developed and tested beforehand. However, NPT was consciously selected to inform the implementation and evaluation of the TRM and incorporated into the study during the planning stage.

The rationale for choosing NPT was provided in Chapter 3, but there were two additional, more personal motivating factors that supported the decision, even if they did not determine the choice. The first factor is that one of my supervisors (COD) had extensive experience of NPT and was therefore able to provide 'expert' support throughout the study period. The second factor is that NPT appealed to me because of its emphasis on 'work' and its focus on the usefulness of interventions. On reflection, it now becomes clear that I was unconsciously performing the work of 'coherence' while deciding on a theory to support the study, asking questions like: how is NPT different to other theories (differentiation); what is the potential value of NPT (communal specification); and how does NPT fit with my previous research and clinical experience (internalization)?

Some researchers have reported struggling to apply the NPT framework because of difficulties in differentiating between its constructs and components (245, 275-277). Initially, differentiation was also challenging in this study, especially during the preliminary coding and before all the exemplar quotes for the framework had been selected. However, coding data to the framework became progressively quicker and easier with experience. My personal experience of using the NPT framework was similar to that of Macfarlane and O'Reilly-de Brun who studies the implementation of a language interpreting service in Irish general practice (384). NPT offers 'an organizing principle to think with' data and 'orienting principles and concepts'. A recent systematic review of the use of NPT found that the meanings researchers attributed to the constructs had high face validity (with a few exceptions) in terms of the reported analyses, interpretations and specific settings (245). But what are the implications if some researchers were to misunderstand the conceptual meanings of the constructs? The review found NPT frameworks were beneficial and helped to identify important issues *irrespective* of subjective judgements about which construct 'fitted best' to the data. The implication is that, while researchers should always aim to perform rigorous analyses of their data, the veracity of their findings is ultimately determined by its practical value (e.g. whether it has explanatory and predictive power).

The other concerns about the application of theoretical frameworks are that researchers may be constrained by theory, miss important findings or alternatively 'shoe horn' data into existing themes (384). NPT is no exception. It focuses on the 'work' study participants do rather than their attitudes or emotional experiences of interventions (245, 273). The concern whether important findings may have been missed in this study is at least partially addressed by having explicitly searched for data outside the framework and describing it. There is also evidence from the international literature that very little data fall outside the NPT framework, and the data that do are typically too generic, diffuse or vague to be meaningful or strictly technical and attitudinal in nature (385). In this study, the vast majority of data relevant to the factors that hindered or facilitated the implementation of the TRM fitted well within the NPT framework. However, the data about participants' perceptions of patient safety did not. NPT, like any other middle range theory, cannot and does not claim to be a 'theory of everything'. It is a heuristic device and not a 'conceptual straitjacket' (245). Data that were outside the NPT framework were therefore analysed thematically.

The close associations between some of the NPT constructs and components were described in Chapter 9. The associations have been conceptualized as a recursive loop by Bamford et al (270). They described how, when an intervention (in this instance, nutritional guidelines) failed to make sense (coherence) or engage staff (cognitive participation), some of them 'acted out' their resistance (collective action). As a result, the intervention did not produce results, which confirmed their opinion that it had no value (the work of reflexive monitoring).

Recursive loops were also evident in this study. Those clinicians who were concerned about latent safety risks in their systems (the work of coherence) were willing to apply the TRM and share their findings (cognitive participation). When the TRM was applied effectively, PSIs were detected which then, in many instances, led to learning and subsequent improvement actions (the work of collective action). As a result, there was evidence that the TRM was useful, and reviewers were therefore willing to continue using it (the work of reflexive monitoring).

10.3.4. Strengths and limitations of the Trigger Review Method

This study helps to demonstrate three key strengths of the TRM. The first strength of the TRM is the willingness of clinicians to disclose PSIs. This is a crucial finding, as it indicates at least some degree of clinician engagement. It also indicates that the vast majority of clinicians considered the possibility of medical errors and latent risks in their practice systems and proactively searched for them. Only a generation ago, the voluntary participation of clinicians in this kind of study, or a national patient safety programme would have been almost inconceivable. Now that the precedent has been established, i.e. that clinicians are willing to search for and report PSIs in general practice, the next systems issue is whether GPs are prepared to report PSIs originating in secondary care. It is known that some GP's will deal with such interface issues if they identify a significant event but this is neither compulsory nor adequately formalized.

The second strength of the TRM is its flexibility. It was identified as a facilitating factor of its implementation in this study, as was described in Chapter 9. The fact that the trigger review method can be customized for specific environments is one of the important reasons why it is considered valuable by researchers in the international patient safety community (85, 233, 236, 237, 242, 382).

The third strength of the TRM is that clinicians actively identify their own improvement priorities and decide whether and how to enact them. In this way, the TRM works to encourage 'ownership' of PSIs, which was discussed in Chapter 8. The importance of clinical ownership is well-known from the international literature. An example is the study by Sharek et al, who found that clinicians conducting 'internal' reviews in their own hospitals were significantly more likely to detect PSIs compared with a team of experienced external reviewers (386).

The best evidence we currently have suggest that improvement initiatives that are characterised by clinician and patient engagement and 'ownership' are more likely to be successful and become normalised compared with those that are reliant on 'top down' approaches (205). That does not mean that health care managers and leaders do not have important roles in improvement. Strong, visible support by leaders for improvement initiatives are an essential facilitating factor, as was explained on page 261. The most successful initiatives are those that synergistically blend clinician engagement and senior leadership.

In this study, all of the medical records were reviewed manually. The alternative option would have been to automate some or all of the clinical review processes. Whether this decision is considered a strength or limitation depends on the aims of the study and purpose of the researcher. We know from the patient safety literature that automation will detect many more triggers in a much shorter period of time than the manual process and often for a fraction of the cost (387, 388). However, while the results from automated reviews can be insightful, especially when investigating harm rates in larger organisations, they arguably do not confer the unique, personal and specific information about PSIs which prompted the majority of clinicians in this study to take further improvement actions. The findings from this study strongly suggest that it is the very act of searching for triggers in a systematic manner that creates the appropriate 'mindset' required by clinicians to detect PSIs (Chapter 8, page 202).

One of the main limitations of any clinical record review method, including the TRM, is the inherent subjectivity of the reported findings, which is a product of the degree of variation in reviewer characteristics (85, 241, 335, 389, 390). The trigger review findings that were submitted by the participating clinicians in this study were not independently verified. The inter-rater reliability (IRR) between reviewers was also not investigated. Consequently, the reliability of the findings cannot be quantified by, for example, Cohen's kappa statistic (391). The decision not to undertake formal testing of IRR was based on four considerations.

First, work conducted in the hospital setting shows that the IRR was variable despite using more than one reviewer (392, 393). For example, Forster et al estimated that reliable detection and categorisation of PSIs (e.g. a 95% chance that the PSI actually occurred) requires the agreement of at least *three* reviewers (394). The second reason for not conducting IRR testing is therefore that the available resources for this study were insufficient to do so.

Third, the current TRM process has been adapted over the course of several years to be as simple and easy to use as possible (Chapters 1and 3). The original, 2007 version of the TRM had seven steps (40). This version was not perceived as feasible or acceptable beyond the pilot stage, even to the initial group of clinicians that were strongly motivated to use it. In comparison, the perceptions of the vast majority of participants in this study, namely that the TRM is acceptability and feasible, facilitated its implementation and therefore help to justify the omission of IRR testing.

The fourth and final reason is that the study aimed to explore the potential of the TRM as an approach to improve the safety of care in general practice. For this purpose, a measure of the reliability of the numerical data is less important than whether clinicians considered and undertook improvement actions.

Finally, variation can extend beyond the main results of a study to involve the practical aspects of the review process itself. For example, the reported time reviewers required to complete a trigger review varied considerably in this study

(Chapter 7, page 174). Whether it was because of a lack of training, the complexities of the cases being reviewed, or some other factors remain unclear.

10.4. Recommendations

10.4.1. Implementation and improvement initiatives

In this sub-section, three recommendations about implementation and improvement initiatives are made. They are:

- Researchers and policy makers should pro-actively identify and address the main factors that are known to facilitate or hinder the implementation of improvement initiatives;
- II. The existing knowledge and 'engagement' of clinicians should be recognised and harnessed; and
- III. The lessons learnt from PSIs should be widely disseminated.

10.4.1.1. Identify and address the barriers and facilitators to improvement initiatives

The evidence from this study and the wider implementation science literature suggest that a small number of specific factors are instrumental in facilitating or hindering the implementation of most, if not all, complex healthcare interventions. These factors can be identified, described and understood and are amenable to intervention. Researchers should actively consider them from the earliest planning stages, through the implementation and evaluation process and on in an ongoing manner until normalisation eventually occur.

Devlin et al recently identified three key areas for researchers and policy makers to pro-actively consider and address in order to ensure future, large-scale initiatives are successfully implemented and normalised (347). They are: (i) time; (ii) what the authors refer to as 'readiness', which is the product of resources and clinician engagement; and (iii) issues related to information technology (IT). Although this study was smaller and had a shorter timeline, the equivalent of these three factors (four if the components of 'readiness' are counted separately) were also identified as the most important determinants of the successful implementation of the TRM.

The first factor is time. To be successful, implementation initially requires adequate time for planning and the necessary work of coherence and activation. Once the project is underway, time remains a critical factor and should be allocated to do the necessary work of implementation. If the aim is to normalise the intervention, then a minimum time of five years, ideally ten, should be commissioned.

The second factor is 'readiness'. The findings from this study add to the large body of evidence (366, 369) of the essential role provision of adequate resources have as a facilitator of improvement interventions. This issue was also described in Chapter 9 and section 10.2.3, page 263. The other component of 'readiness' is clinician engagement. For an improvement intervention to be successfully implemented, it is recommended that researchers and policy makers ensure local staff and clinicians understand and are prepared for the proposed change(s). In NPT terms, there is need for the work of coherence and cognitive participation, e.g. the intended users of the intervention need to understand what is required of them and be willing and able to do this.

This recommendation is strongly supported by evidence that coherence is an essential precursor to successful implementation (268, 376). Franx et al recommended, based on their research about the management of depression in primary care, that the problems of general practice teams should be addressed first and 'sensitising strategies' should be considered at local levels before any attempts are made to implement further improvement initiatives (276). Bamford et al and, more recently, Jeffries et al made a similar recommendation, namely that researchers and policy makers should ensure all potential stakeholders and users understand the rationale for and benefits of an intervention before attempting to implement it (270, 395). This study provides a further, practical example of the need for initial coherence work by researchers, policymakers, clinicians and staff.

The third factor is information technology (IT), which includes the related issues of information governance and the interoperability of systems. In NPT terms the work of reflexive monitoring, e.g. evaluating the effectiveness of an intervention, typically requires IT support and systems. It also requires the researchers and participants to agree outcome measures and processes for collecting the necessary data to evaluate the intervention. In this study, the implementation of the TRM was facilitated by the fact that practice teams already had adequate IT systems and were therefore able to guickly generate random patient lists for sampling. Because clinicians worked in the practice they were able to access electronic medical records without additional concerns which have to be addressed when external reviewers are involved. However, despite all patient identifiers intentionally being removed, a small minority of clinicians were still concerned how the findings may be used at regional or national levels (Chapter 9). Finally, implementation was facilitated because reviewers were able to enter data into simple trigger review Summary Sheets that were compatible with their existing software.

10.4.1.2. Harness existing knowledge and clinician engagement

The practice teams in this study participated voluntarily. They were able to provide many examples of improvement work they were already doing and demonstrated knowledge of a range of improvement methods. In addition, many participants clearly identified priority areas for interventions and described what they would require to undertake this work (Chapter 6). These findings seem at odds with the common perception that clinician engagement is challenging (346). The implication may be that the sentiment about clinician engagement should be further qualified, e.g. clinician engagement is challenging *when* they are expected to contribute to initiatives that do not seem to have value for them or their patients *and* they are not adequately supported to perform the work.

The second recommendation in relation to improvement initiatives is therefore to harness the existing knowledge of clinicians and staff and support and enable them to implement the changes they perceive as priorities by allowing some flexibility at local levels. From a regional and national perspective it is desirable - and necessary - to standardise processes and agree overarching objectives and outcomes. However, these requirements should not necessarily preclude frontline staff and clinicians to also plan, prioritise, design and test their own potential solutions (366).

This approach (local flexibility) is in stark contrast to the three common approaches taken by some initiatives. The first approach is to raise awareness of a problem, with the expectation that this will lead to improvement. A practical example would be to inform general practices that some investigation results of their patients are not appropriately managed, and as a result PSIs occurred. The second approach is to raise awareness of a problem and propose a solution. In the example of investigation management, a potential solution might be for practices to keep a log of all investigation requests and reports and regularly audit their performance. The third approach is to raise awareness of a problem, propose a solution and partly incentivise or mandate its application. In the investigation management an investigation management system, with the expectation that the practice will absorb the recurrent costs and increase in workload.

While this example is simplistic, it helps to explain why some initiatives are not normalised. In many instances clinicians are already aware of problems but, for one reason or the other are unable to implement a solution. Alternatively, clinicians may be aware of solutions, but lack the resources or influence to implement them. The challenge is therefore to select the 'right' approach in each instance, in order to appropriately harness existing knowledge and ensure clinician engagement. In this context 'right' may mean raising awareness of problems, proposing potential solutions, offering additional resources or a combination of all three approaches.

10.4.1.3. The lessons learnt from PSIs should be shared

The importance of learning from mistakes has been recognised for many years. Almost 30 years ago now Wu et al considered whether house officers learn from their mistakes (336). They concluded that doctors *do* learn from mistakes, and categorised the subsequent changes that clinicians make as a result of their learning in two main groups: constructive ('increased information seeking' and 'increased vigilance') and defensive ('keeping mistakes to self' and 'avoid similar patients'). These findings have been replicated many times since, including in this study. Chapter 8 summarised the wide range of learning needs and learning points reported by participants in this study as a result of conducting trigger reviews. It also described the great potential PSIs have for individual clinicians to learn and reflect on their own professional practice.

The answer to the question of whether (some) clinicians can learn from PSIs is therefore an indisputable 'yes'. However, a more appropriate question may be who should learn from PSIs, and more specifically, whether health care organisations learn from them (396)? Influential reports such as '*An organisation with a memory*' (27) clearly described the organisational imperative to learn from PSIs. The rationale is that when organisations fail to learn from mistakes, patients inevitably suffer the same preventable harm. However, detecting, reporting and even analysing PSIs do not automatically generate collective learning within practice teams or organisations. Only if the learning points from analyses are effectively shared, stored and can be retrieved as required, can the behaviour of teams and organisations change (337, 397). The implication is therefore that the findings from improvement initiatives should be disseminated to all relevant stakeholders, from the individual practice through to the national level to enable and maximize collective learning (21).

10.4.2. Recommendations relating to the TRM

This sub-section describes five specific recommendations in relation to the TRM. They are contingent on the assumption that eventual normalisation of the TRM is desirable. Many of the issues the recommendations relate to have been discussed already. However, it is worthwhile to consider them again, as they helped determine the successful implementation of the TRM and may be equally applicable to other improvement methods.

10.4.2.1. Incorporate the TRM into existing general practice contexts

This issue was previously discussed (Chapters 9 and Chapter 10). The main points are that the TRM will continue to feature in GP specialty training and as quality improvement method for appraisal and revalidation on a voluntary basis, but it needs to be incentivised locally or nationally for wider sustained implementation coverage. At deanery level in Scotland, the next steps will be to further refine TRM as a tool together with the associated training process and educational supporting materials. Exploring the potential need for e-learning and other interactive technology will also be necessary.

10.4.2.2. Provide training to clinicians before they apply the TRM

Training is an essential factor for successful implementation as, without it, participants are much less likely to have the necessary understanding and knowledge to complete trigger reviews. This finding is consistent with the international literature (398), which considers training to be of 'vital importance' for QI initiatives (227).

TRM training needs to be clear about the intended aims, benefits and practical application of the method. Specifically, the importance of the TRM as an approach to improve the safety of care needs to be emphasized. Reviewers need to understand that searching for triggers, detecting PSIs and 'ticking boxes' are intermediate outcomes to support the main purpose of the TRM, which is ultimately to improve the safety of care. The training should explicitly acknowledge that a small proportion of reviews may not detect PSIs and the potential reasons for this. Finally, training should emphasize the importance of identifying preventable system-causes of PSIs rather than a preoccupation with the perceived severity of PSIs or how many are detected.

The group teaching format was judged acceptable by GP trainees in this study, which increases the potential feasibility of delivering teaching to larger numbers of clinicians in comparison to one-to-one teaching (by another clinician) for specific aspects of the training curriculum. This is in keeping with other QI techniques such as criterion audit and SEA which can be taught by both clinicians and non-clinicians in large group settings but applied at the individual and practice-based levels.

10.4.2.3. Select reviewers with the knowledge and influence to enact change

The TRM should be applied by trained clinicians who have *sufficient influence* to ensure the findings are shared with their practice teams and who are capable to improve care for patients as well as at systems-level.

10.4.2.4. Allocate adequate resources and time to the TRM

This issue has previously been discussed. Of particular relevance to the TRM, is that reviewers should be allocated sufficient and protected time to allow them to conduct their trigger reviews, which include the completion of a trigger review summary sheet, in one session.

10.4.2.5. Disseminate the trigger review findings and learning points

The findings from trigger reviews should be disseminated to all relevant stakeholders, from the practice through to the national level to enable collective learning. This recommendation was justified and discussed in more detail on page 283.

10.4.3. Recommendations about future research and research priorities

This sub-section includes four recommendations in relation to future patient safety research, and potential research priorities in this discipline:

I. Patient safety research has practical value and should remain a priority area in general practice;

- One of the main research priorities is to develop and validate a range of effective and cost-effective improvement methods;
- III. Research is necessary to understand how a strong and positive safety culture can be built in diverse health care settings;
- If the value of 'hidden contributions' to patient safety are to be harnessed, more research will be necessary;

10.4.3.1. Patient safety research is a priority with practical value for general practice

The first, general recommendation is that patient safety should remain a priority research area (399, 400). There are many potential benefits, ranging from the philosophical pursuit of knowledge through to the practical aspects of helping to ameliorate or prevent patient harm. Three benefits in particular help to support this recommendation (47). The first benefit is that patient safety research increases our understanding of the nature and scale of PSIs as well as the associated safety threats. Being able to present reliable evidence of significant and systemic shortfalls in delivered care provides justification for policy makers and managers to allocate additional resources to help address deficiencies and allow them to distribute the resources in the most efficient manner.

The second benefit of patient safety research is that it helps to bridge the gap between 'what might work' which is based on intuition and expectation and what actually works in complex, dynamic and unique health care settings. Research methods are imminently suitable for developing, testing and rigorously evaluating complex health care interventions. Importantly, ineffective, unacceptable and unfeasible interventions can be identified before they are widely implemented.

The third benefit is the potential to measure health care performance. Reliable measures help staff set specific improvement targets, monitor their progress and identify when their aims have been achieved. Research can also help to identify and understand the factors that hinder or facilitate improvement processes.

10.4.3.2. Develop and validate a range of effective and cost-effective improvement methods

A systematic review and meta-analysis in 2006 found no evidence for the effectiveness of any intervention to reduce preventable drug-related PSIs and admissions, with the possible exception of pharmacist-led medication reviews (91). The need for a range of complementary methods to reliably improve care standards have previously been described on page 255. This recommendation is also supported by the wider patient safety research community (399, 401).

10.4.3.3. Understand how to build a strong and positive safety culture in health care

Different aspects of safety culture were considered in Chapters 2, 6 and 9. From a practical perspective though, the most important implication is that the prevailing safety culture (or conversely the lack of a safety culture) influences clinicians and staff to choose behaviours that enhance - or compromise - safety practices and thinking (171). This poses an important question: can a primary care organisation build a strong and positive safety culture that is capable of both predicting and avoiding patient safety incidents? The answer is a cautious and provisional 'yes', but will likely require many steps, considerable effort and research.

The very basic first steps and key requirements are to actually become aware of, measure and discuss the concept of a 'practice safety culture'. Perhaps more importantly, even if safety culture can be adequately measured, can we associate these metrics to clinical and organisational outcomes such as reductions in healthcare errors and avoidable patient harm, or improvements in the safety attitudes, knowledge and behaviours of the primary care workforce? The answers at this stage are unknown, but largely because the necessary depth of research and evaluation has yet to take place, hence the recommendation to prioritise safety culture as a research priority (350, 367, 402).
10.4.3.4. Harness the value of 'hidden contributions' to patient safety

Chapter 2 described a number of large-scale patient safety improvement programmes and initiatives. All of the programmes were endorsed by government or health care institutions, had clearly defined aims and interventions and were evaluated to some extent. However, there are many other patient safety-related activities in health care apart from these programmes. They often are unrecognised, unreported and unappreciated despite their considerable value.

One of the main mechanisms through which 'hidden contributions' are made to patient safety is through mitigation. Mitigating factors are defined as actions or circumstances that prevent or moderate the progression of an incident towards harming a patient (3). It has been estimated that mitigation may help to prevent from 8% (2) to as much as a quarter of health care errors in primary care (403). Practice staff are the most common mitigators, but patients and their families and other health care professionals such as pharmacists also make important contributions (404). Two exemplars are the recent ethnographic studies by Swinglehurst et al and Grant et al who found receptionists and administrative staff made important "hidden" and informal contributions to the safety of repeat prescribing in UK general practice (309, 369). Such hidden contributions warrant better identification and, potentially, utilisation in everyday practice.

Conclusion

In the last few decades there have been extraordinary advances in medical research and technology that have enabled clinicians to better diagnose, investigate and treat an ever-expanding list of pathological conditions, thereby further improving the overall health of individual patients and populations. At the same time, there has also been a growing acknowledgement that patient safety incidents (PSIs) commonly occur and that a substantial minority result in preventable, iatrogenic harm to patients.

This study is the first known attempt to investigate how the TRM is implemented and perceived in primary care from the perspective of general practice clinicians and staff. All of the study aims were achieved:

- The perceptions and understanding of general practice clinicians and staff of 'patient safety' were described in Chapter 6.
- The usefulness of the TRM was evaluated and the main outcomes from its implementation were described in Chapter 7.
- An explanation of how the TRM worked was provided in Chapter 8.
- The main factors that facilitated or hindered the implementation of the TRM in general practice were identified and described in Chapter 9.

In addition, four specific questions in relation to this study were considered in Chapter 10, and the potential answers as well as their implications for patient safety, implementation and improvement initiatives were described.

The findings from this study reaffirm the potential of the TRM to identify avoidable PSIs with potential educational and improvement value, particularly when it is applied to high risk groups of patients. In addition, most participants experienced the method as acceptable and feasible and perceived it as potentially useful. It is clear that the TRM is uncovering important patient safety concerns and also driving improvements in related care systems and processed at the individual practice level. The implication is that this is making a significant and demonstrable difference to patient care while impacting positively on local safety culture. On the evidence presented, normalisation of the TRM in general practice can therefore be recommended.

However, while the usefulness of an intervention is an important factor in determining whether it is normalised or not, the study findings also clearly indicate - consistent with the international literature - that there are other factors that are at least equally important. They are: that clinicians have the prerequisite knowledge to apply the method; practice and senior organisational leadership support; contextual integration; and allocation of adequate resources. At the time of writing, some of these requirements have been met. The vast majority of clinicians seem to have adequate knowledge to apply the TRM, and it has been included in the RCGP Patient Safety Toolkit. However, there is currently no formal mandate or financial incentives for general practice teams to perform regular trigger reviews. It therefore seems likely that the normalisation of the TRM in UK general practice will be gradual and piecemeal, if it happens at all. Nevertheless, the lessons learnt from this study can be incorporated in the ongoing efforts to further improve the safety of care in general medical practice.



Appendix 2. Invitation letter to general practices.





Financial reimbursement

We are able to provide each practice with a backfill participation fee of £813.03 (equivalent to three ٠ GP and Practice Nurse NES sessions each and a single Practice Manager session) immediately payable on agreement to participate. We appreciate that this contribution is more of a gesture of thanks and will not necessarily cover the full resource implications of participation.

Research Ethics

· The study was submitted to the West of Scotland Research Ethics Service. They considered the protocol and confirmed that it did not require to be submitted to an Ethics Committee. The study has been approved by the Research Ethics Committee, College of Medical, Veterinary and Life Sciences, University of Glasgow

Participation and Further Information

· If your practice would like to participate in this study or you would like further clarity or information before making a decision, please contact Dr Carl de Wet direct on 0141 223 1450 or email him on carl.dewet@nes.scot.nhs.uk

Thank you in anticipation for your assistance with this study.

Kind regards

Dr. Carl de Wet GP/Research

c.c. Linda Davidson, NHS GG&C

P -3 Dr. Paul Bowie Associate Adviser

NHS Education for Scotland Date Dear Doctor... Invitation to Practices to Participate in a Pilot Research Project: Trigger Review of Electronic Patient Records and Making General Practice Safer Why is the research important? It is now widely accepted that patients can suffer unintentional but preventable harm as a result of their health care. As a result, the Scottish Government has made improving patient safety a priority. The Scottish Patient Safety Programme (SPSP) initially focused on hospital care but is now being extended into primary care and will roll out on a national basis in March 2013. Improving and assuring the safety of patients is clearly desirable. However, very little is known about the extent of the safety threat or how this aim can be achieved in general practice. This pilot will help us to examine patient safety in our setting and allow practical improvement tools to be further developed and tested in partnership with frontline staff. The pilot will involve up to 40 GPSTs in NHS Greater Glasgow and Clyde and last between three and four months. How will you be expected to contribute? You will have to attend a short training session (around three hours) in the 'trigger review method'. The session will help to prepare you to review your own medical records in a rapid, structured and focused manner to detect potential safety-critical issues. Training will be delivered in a venue in Glasgow by two GP Associate Advisors from NHS Education for Scotland. After the training you will be expected to conduct one 'trigger review' of a small random sample of medical records (n=25) and summarize the main findings on the 'Trigger Review Summary Report'. The patient population that we recommend is those >75 years of age with COPD. The complete review process should take a maximum of four hours. We recommend that you consider sharing the findings with practice colleagues like your educational supervisor (if relevant) and jointly consider how they may inform your personal and professional learning needs and specific improvements in the practice. During the course of the study, we aim to hold a number of focus groups at convenient times and venues to discuss your experiences with the trigger review method. At the end of the study we will ask every participant to anonymously complete a survey which should last no more than ten minutes. The purpose of the survey is to explore in confidence participants' experiences and perceptions of the patient safety in primary care in general and the trigger review method in particular.

Appendix 3. Invitation letter to GP Specialist Trainees



What are the benefits of participating?

- You may uncover important opportunities to improve the quality and safety of the patient care you
 provide.
- You will be in a better position to engage with SPSP once it spreads into primary care in the near future.
- GPs may submit the 'trigger review findings' as supporting audit evidence for appraisal and revalidation
- Once trained, you will be able to demonstrate to others in the practice how to conduct trigger reviews
 of records
- Your overall contribution will help to inform the evolving patient safety agenda in primary care.

Research Ethics

The study was submitted to the West of Scotland Research Ethics Service. They considered the
protocol and confirmed that it did not require to be submitted to an Ethics Committee. The study has
been approved by the Research Ethics Committee, College of Medical, Veterinary and Life Sciences,
University of Glasgow.

Participation and Further Information

If you would like to participate in this study or you would like further clarity or information before
making a decision, please contact Dr Carl de Wet direct on 0141 223 1450 or email him on
carl.dewet@nes.scot.nhs.uk

Thank you in anticipation for your assistance with this study.

Kind regards

X

Dr. Carl de Wet GP/Research

P -3 Dr. Paul Bowie Associate Adviser

Appendix 4. Consent form for general practices

		Education for Scotland
Research proj	ect: A mixed-methods study to explore the safety of care delivered in general medical practice	health
	Consent form for GP practices	
PLEASE READ THIS	FORM CAREFULLY:	
 The purpose of received and re questions and/o 	this study has been explained to me and the practice team, I have ad a research information sheet and have had an opportunity to ask any or discuss any further concerns.	
 I understand that all information, 	at all information collected during the study will be confidential and that including the practice name, will be anonymised.	
 The practice tea of the premises 	am agrees to participate and to allow the researchers supervised access , medical records and safety-related meetings at arranged times.	
 The practice ag (PM) and practi 	rees to nominate a named general practitioner (GP), practice manager ice nurse (PN) to:	
 Receive a short 	training session (around 2-hours) in the 'trigger review method'.	
 Conduct two 'tri the course of th 	gger reviews' of a small random sample of medical records (n=25) over le study	
 Participate separation to them during to 	arately in two semi-structured interviews at a time and venue convenient the study period.	
 The estimated of May 2013 	duration of participation in the study will be from 1 st June 2012 to 31 st	
 I understand that time, without give 	at my participation is voluntary and that I am free to withdraw at any ving any reason, without my legal rights being affected.	
PLEASE TIC AGREE WITH TAKE PART	K THE BOX, SIGN AND DATE THE FORM TO INDICATE THAT YOU H EACH STATEMENT AND THAT THE PRACTICE IS WILLING TO IN THE STUDY.	
Name	(Please print)	
Signature		
Date	On behalt of practice	
Practice name:		

	Educatic for Scotlan
R	Research project: A mixed-methods study to explore the safety of health care delivered in general medical practice
	Consent form for individual participants
PLE	EASE READ THIS FORM CAREFULLY
•	The purpose of this study has been explained to me and I have received and read a research information sheet.
•	I understand that all the information I give is confidential and that all information I give will be anonymised.
	I agree that the interview can be audio-recorded.
	I understand that I am not obliged to answer all the research questions asked during the interview
•	I understand that the audio-recordings will be stored securely and only those working on the stud will have access to these.
•	I understand that the audio-recordings will be destroyed when the project has been completed.
•	I understand that the interview transcripts will be stored securely and that only those working on th study will have access to them.
	I understand that the information I give in the interview will be used for research purposes only.
•	I understand that my participation is voluntary and that I am free to withdraw at any time, withou giving any reason, without my legal rights being affected.
	PLEASE TICK THE BOX, SIGN AND DATE THE FORM TO INDICATE THAT YOU AGREE WITH EACH STATEMENT AND IS WILLING TO TAKE PART IN THE STUDY.
Na	me (Please print)
Sig	gnature
Dat	.te

Appendix 5. Consent form for individual study participants

Appendix 6. Schedule for interview one: general practitioners, practice

managers and practice nurses

 Hello Thank you for agreeing to meet Feel free to discuss any concer The aim of this interview is to d in general practice. 	t with me today. Please take a few m ms or ask for clarification before sign liscuss your awareness, perceptions a	ninutes to read through the consent form. ing. and personal experiences of patient safety
General		
Main question	Additional question	Clarifying question
What is your role in this practice?	How long have you been in this practice? How long have you worked in any primary care role?	
Patient safety and general pr	actice	
Main question	Additional question	Clarifying question
What does the term 'patient safety' mean to you?	What do the terms 'harm', 'error' and 'patient safety incident' mean to you?	
What do you think are the main patient safety threats in general practice?	Do you think any of the following are threats? Diagnostic errors Prescribing errors Primary-secondary care interface issues	Why do you think patient safety is a problem in general practice?
Personal experiences with pat	ient safety	
Main question	Additional question	Clarifying question
Why did you or your practice agree to participate in this project?	Do you have any previous patient safety experience?	
Have you or your family ever been involved in a patient safety incident in primary care?	Would you mind telling me a little about what happened?	
Are you aware of any recent patient safety incident in this practice?	or any incident you may have been involved in?	'Recent' could be interpreted as anytime in the last three months.
	If yes – would you mind telling me a little about what happened? If no – how would you have known or detected an incident if one had happened?	

Appendix 7. Schedule for interview two: general practitioners, practice

managers and practice nurses

 Introduction Hello Thank you for agreeing to me Feel free to discuss any cond The aim of this interview is to 	eet with me today. Please take a few cerns or ask for clarification before sig discuss your experience with the trig	minutes to read through the consent form. gning. ger review method (TRM).
General This section will only be used for t	hose respondents that did not partici	pate in interview one.
Main question	Additional question	Clarifying question
What is your role in your practice?	Provide the second s	
1. Coherence (i.e. 'mean	ing and sense making' by interv	viewees)
Main question	Additional question	Clarifying question
What are the benefits of the trigger review method (TRM)?	Who are likely to benefit? Is it clearly distinct from other interventions?	Patients? Staff? Are they likely to value the benefits?
Which findings or aspects of the TRM were unexpected?		
2. Cognitive participation	n (i.e. 'commitment and engage	menť)
Main question	Additional question	Clarifying question
What changes did you make as a result of the review findings?	If none, why?	It may be helpful to consider different levels, i.e. patient, practitioner, practice, primary care.
Who else could use the TRM?	Will they understand the rationale for the method? Will they be prepared to invest time and work in it?	
3. Collective action (i.e.	work participants do' to make 1	RM 'function'
Main question	Additional question	Clarifying question
What did you think of the training session?	Was the training adequate? How could the training be improved?	What was your experience of the provided learning resources, venue, presenter and presenting style?
How compatible is the TRM with your existing work?	What (if any) impact does it have on different professional groups?	Consider: division of labour, resources, power, responsibility.
How does the TRM fit with the overall ethos of general practice?		How does it fit the wider organisational goals?
Who did you share the findings with?	If no one, why?	
4. Reflexive monitoring	g ('reflect on and appraise' the	TRM)
Main question	Additional question	Clarifying question
How can the TRM be adapted o improved?	r	
What would help ensure that yo continue using it?	u Does the TRM have a role in appraisal, revalidation, educat and training?	ion
Thank you for your time and par	ticipation.	

Appendix 8. Trigger Review Summary Sheet

	otop one	: Pla	anr	nin	g	an	d	Pr	eļ	ba	rat	io	n									
Please complete:																						
Name of Reviewer:					Na	me	of F	rac	ctic	e		Γ										
Date of Review					Pro	ofes	sio	n (p	leas	e dr	rcle)		Sala	arie	d Gl	P-P	racti	ice l	Nurs	e e	aror	-
No. of Records Reviewed					Re (e.g	viev g. 3-	v Pe	erio nths	d)													
What Patient Group did you select records from?																						
	Step 7	[wo:	R	ev	ie	N	of	R	ec	or	ds	;										
Please aim to review 25 r time you find it in one of th	records from th he records.	e cho	sen	pa	tien	it gi	ou	ıp.	Tio	:k (one	bo	ж (~)	nex	ct to) ea	ich	tri	gge	er ea	ch
Trigger (A 'prompt' that may indicate a	a safety incident)																				То	al
≥3 consultations in	7 days		Γ												Γ							
New 'high' priority read	code added		Γ												Γ							_
New allergy read coo	de added		Γ								Γ	Γ		Γ	Γ	Γ						_
'Repeat' medication item	discontinued											Γ		Γ	Γ	Γ						_
Out of Hours/A&E att	tendance		Τ							Γ	Γ	Γ	Γ	Γ	Т	Γ			Γ	Γ		_
Hospital admiss	sion									Γ	Γ	Γ	Γ	Γ	Γ	Γ			Γ			_
Hb <10.0			T							Γ	Τ	Γ	Γ	Γ	Т	Γ			Γ	Γ		
eGFR reduction	1≥5											Γ	Γ	Γ	Т	Τ						
	the nationt ear																					
Please Driefly describe preventability of each incid Description of Detected Patie	dent using the sent Safety Incider	tety ii scales	ncid s be	lent Iow	s t an	that Id ti	i y her	ou 1 ac	de Id	ete the	cte tw	d. os	Ne: co	xt, res rity	ju s in	dge the	th 'pr	e ior bili	sev ity'	veri co Pl	tya lum RIOR	nd n. TY
Please briefly describe preventability of each inci	dent using the s	nts	ncid s be	lent	s f	that id ti	t y her	ou 1 ac	bb bb	the	cte tw	d. os		xt, res rity	ju s in P +	dge the reve	th 'pr	e ior bilit	sev ity' ty	veri co Pl	ty a lum RIOR	nd n. TY
Please briefly describe preventability of each inci	dent using the s	rety ii scales	ncid s be	lent	an	that id ti	t y	ou nac	de Id	the	cte tw	d. os		xt, res rity	ju s in IP +	dge the	th 'pr	e : ior bilit	sev ity' ty =	PI		nd n. TY
Please Drieny describe preventability of each incide Description of Detected Patie	ent Safety Incider	rety ii scales	ncid s be	lent	ran	that id ti	t y		de	the	cte tw	d. s		xt, re: rity	ju s in <u> </u> P + +	dge the	th 'pr	e : ior = =	se\ ity' = =	Veri CC PI		nd n. TY
Please briefly describe preventability of each inci	ent Safety Incider	nts	ncid s be	lent	an	that id ti	t yo her		əb bl	the	cte() tw	d. s		xt, res rity	ju s in + + +	dge the		e : ior = =	sev ity' = = =	PI		nd n. TY
Please Drieny describe preventability of each inci	ent Safety Incider	nts	ncid s be			that id ti	her		de Id	the	cte tw	d. s		xt, re: rity	juc s in + + +	dge the		e : ior = =	sev ity' = = =	PI		nd n. TY
Please brienty describe preventability of each incide preventability of each incide Description of Detected Patie	ent Safety Incider	nts	ncid s be			that id ti	her		ab bt	eteethe	cte(d. so s		xt, re: rity	ju s in + + +	dge		e : ior = = =	sev ity' ty = = =	PI		nd n. TY
Please briefly describe preventability of each incli Description of Detected Patie Description of Detected Patie Severity 3 1 Any incident with the Mild here incommune	Scale: potential to cause has	m m	ncid s be			that id ti	her		No	tpo	Perent Pe	d. So s		rity	juo s in + + +	dge the reve		e : ior = = =	sev ity' ty = = =	PI		nd n. TY
Please briefly describe preventability of each incle Description of Detected Patie Description of Detected Patie Severity 1 Any incident with the Mild ham: inconvenience, Mild ham: ensure no hi	Scale: potential to cause has am occurred.	m stigation	ncid s be			that d ti	reve	entaì	No ble a	t promotion of the second seco	Prevent	d. So	Ne: co eve	rity	jud s in + + + +	dge the reve	th 'pr enta	e : ior = = = =	sev ity' ty = = = =			nd TY
Please brieny describe preventability of each incie Description of Detected Patie Description of Detected Patie 1 Any incident with the 2 Mild harm: inconvenience, furt ensure no ha 3 Moderate harm: required interve d Prolonged, substantial or	Scale: potential to cause has her follow-up or inve arm occurred. ntion or duration for lo lay permanent harm, incl	m scales nts nts stigation onger the uding	1 to		1 2 3 4	P	reve	ental P	No ble a oter	t present	Prevention or igning	d. So	Ne: co eve	rity	juc s in + + + + ity S igins onds n pri id or ginat	dge the reve	th 'privental n sec re O re C re C re C	e : ior bilit = = = =	sev ity ty = = = = iary ot po	reri CO PI	ty a lumi	nd TY and
Pease Drietty describe preventability of each incid Description of Detected Patie Description o	Scale: potential to cause has her follow-up or inve am occurred. ntion or duration for le lay permanent harm, incl alization	m stigation onger ths uding	a to		1 2 3 4	P P	reve	entaì P	No ble a	t production of the second sec	P even origi ly pr	d. So	Ne: co eve	rity	juc s in + + + + ity s iginat	dge the reve	th 'privental ental	e : ior = = = = cond e in pr nary	sev ity ty ty = = = = iary ot p rima	ry care	ty a lum COR COR COR COR COR COR COR COR COR COR	nd TTY

oreb III					
A. Please describe any Actions/li	mprov	ement	ts ma	ade	DURING the review (e.g. updated coding or prescribing)
Use PRIORIITY scores to guide you	asar ou-ticl	esult o k as app	or the	e trig ate)	Jger review findings ?
Specific actions			,		Please describer
Significant event analysis	ŕ		–	Ť	
Audit			+		-
PDSA Cycle			+		-
Feed back to colleagues			+	-	4
Make a specific improvement(s)			_	<u> </u>	-
			_	<u> </u>	-
			_		-
Discuss with Educational Supervisor			\perp		
Update or develop a protocol					
C Diease describe identified Der	eonal	Drofe	eein	nal	or Dractice Team Learning Needs
	Jona	, 11010	3310		of Fluence Feath Learning Needs.
Personal:					
Professional:					
Practice Team:					
Please add any comments about	t the tr	igger	revie	w p	rocess:

Appendix 9. The NPT Framework and exemplar TRM quotes from this study

NPT constructs Exemplar quotes and components

Coherence: The work participants have to do to understand the TRM

- Differentiation Mistakes usually are only picked up when they've blown up, rather than going looking for them... This [the TRM] was better because actually I was fishing for things (GP06)
- CommunalI think it's useful as a learning tool to learn about your ownspecificationsystems and a way of trying to improve those systems and a wayof learning as a team with the results (GP05)
- Individual I was able, I felt empowered by looking at it, that I could look at specification the system, pick out errors that we hadn't really seen, bring it to somewhere to discuss, reflect on my own practice, and make some changes within that (GP06)
- Internalization I like nothing more than going back over notes, and reviewing and researching what we have or haven't done (GP06)

Cognitive participation: The work of building a TRM 'community of practice'

- Initiation I've been trying to start the ground level approach of saying 'this is how it should be used', you know, used formatively and using it to look at your systems as well, and things like that (GP05)
- Enrolment One of my partners has said to me: 'Do you mind if I do the next trigger tools? Cause I'm looking for something to do for an audit (GP05)

Activation That made us then do that as an SEA which made us decide 'well, ok, we all need whatever we do, we all need to do the same'...and it also made us do a search really for anyone with an HbA1c of over 6.5 who didn't have a diagnosis of diabetes (GP05)

Legitimation I'm not sure if I'd have gone back to [the TRM] if it had disappeared off the horizon. If it was out of QOF, but part of what we do, or it's made an RCGP approved tool or it was something that trainees do, then I could see why people would become lured to the idea that it is a good thing and it would've developed... cause it gives it fruits. There are fruits for your labour. Well, you have to justify the time in order to make it happen (GP06)

Collective action: Enactment of the TRM, e.g. how the work gets done

- Interactional I did it electronically from the outset because that's just much workability easier to store and save it (GP06)
- Relational GPs hopefully see it as a tool to help them and the second thing integration that'll help with that, is what's done with the results that come from it at trust level. And I think that needs to be made clear to GPs at the beginning [pause] so they don't feel that this is going to be something to punish them and just another tick box exercise but see it as something positive (GP05)
- Skill-setI would hope, I would have to say I would think, most GPs wouldworkabilitybe able to do that' [reflect on TRM findings] (GP05)
- Contextual So even having the lunchtime meeting, when we went over the trigger tool, that's done slightly on a squash and a squeeze it's not easily done. Protected learning time helps for that kind of process...if you're trying to make it part of the structure (GP06)

Reflexive monitoring: The work of appraising the TRM

Systematisation That paper copy I don't think can be found at this time, which is a bit disappointing cause I have seen it, it was done and we thought we'd sent it, or she'd sent it, but it hasn't transpired, I gather (GP06)
Individual I think it's been very useful for us...generates plenty of SEAs, has helped us refine some of our protocols and like for the diabetes thing we changed that. I think it does pick up [pause] incidents directly related to one patient (GP05)

Communal I think it can be a fantastic tool for appraisal (GP05)

appraisal

Reconfiguration I'm just trying to think if this would take away from it in any way, but is there a way to identify the triggers using some computer based template and then come up with x number of patients to review? (GP05)

- P1. Bowie P, Halley L, Gillies J, Houston N, **de Wet C**. Searching primary care records for predefined triggers may expose latent risks and adverse events. Clinical Risk. 2012;18(1).
- P2. **de Wet C**, Johnson P, O'Donnell C, Bowie P. *Can we quantify harm in general practice records? An assessment of precision and power using computer simulation*. BMC Med Res Methodol. 2013;13:39.
- P3. McKay J, **de Wet C**, Kelly M, Bowie P. Applying the Trigger Review Method after a brief educational intervention: potential for teaching and improving safety in GP specialty training? BMC Med Educ. 2013;13:117.
- P4. **de Wet C**, Bowie P. *Patient safety and general practice: traversing the tightrope*. Br J Gen Pract. 2014;64(621):164-5.
- P5. de Wet C, Kelly M, McKay J, Bowie P. The *Trigger Review Method*. In: Bowie P, de Wet C, editors. Safety and Improvement in Primary Care: The Essential Guide. London: Radcliffe Publishing Ltd; 2014.
- P6. **de Wet C**. The KM Seedat Lecture: Why patient safety incidents predictably occur in Family Medicine. Transactions. 2015;59(2):4.
- P7. **de Wet C**, Black C, Luty S, McKay J, O'Donnell CA, Bowie P. Implementation of the trigger review method in Scottish general practices: patient safety outcomes and potential for quality improvement. BMJ Qual Saf. 2016.
- P8. Bowie P, McKay J, McNab D, **de Wet C**. *The past, present and future of patient safety education and research in primary care*. Educ Prim Care. 2016;27(1):3-9.
- P9. **de Wet C**. *How primary care research contributes to improving the quality and safety of healthcare*. In: Goodyear-Smith F, Mash R, editors. International perspectives on primary care research. London: CRC Press; 2016. p. 8.
- P10. de Wet C, Bowie P, O'Donnell C. Implementation of the trigger review method in Scottish general practices: a qualitative study to identify barriers and facilitators using Normalisation Process Theory. Submitted for peer review.
- P11. de Wet C, Bowie P, O'Donnell C. 'The big buzz': A qualitative study of how safe care is perceived, understood and improved in general practice. Submitted for peer review.

Аррениіх 11.	Fresentations i		lay	
Date Eve Org	ent or ganisation	Venue	Delegates	Presenta

ctud ٨ 4:, 4-41 461

Date	Event or Organisation	Venue	Delegates	Presentation
26/10/17	GP17 - RACGP Annual Conference	Sydney	N/A	'The Big Buzz': How safe care is perceived, understood and improved in GP
11/04/17	Communication Hour Symposium	Griffith University	Medical students (years 3 and 4)	Normalisation Process Theory
15/09/16	'Subject Matter Experts' workshop	Gold Coast University Hospital; Australia	Managers and clinicians from the Specialist Outpatient Services (±20)	The value of NPT in planning and implementing change
05/08/15	Lunch-time lecture	Medical School; Durban; South Africa	Students, faculty (±20)	Perspectives of contributing factors to PSIs
01/08/15	Annual conference of the South African Academy of Family Physicians	Empangeni Hotel; Durban; South Africa	Family practitioners; medical superintendents; academic staff (±200)	The KM Seedat lecture: Perspectives of the contributing factors to PSIs
31/10/14	Workshop to plan the 'Harm in New Zealand general practice' study	Dunedin; New Zealand	Academic staff (±15)	TRM and general practice
13/06/14	SPSP-PC national Learning Event	Training centre; Glasgow; Scotland	Programme leads and participants (±90)	The TRM and QOF: Findings and recommendations
21/05/14	The Patient Safety Congress	Conference Centre; Liverpool; England	National and international delegates with an interest in safety	Improving patient safety in primary care in Scotland
09/05/14	SPSP-PC national	Crowne Plaza;	Programme leads	Qualitative findings

	Learning Event	Glasgow; Scotland	and participants (±90)	from this study
26/03/14	Quality Improvement workshop for GPSTs	NES office; Edinburgh	GPST (±40)	The potential of the NPT framework for QI projects
31/10/13	NES GP Educational Supervisor conference	Grand Central Hotel; Glasgow; Scotland	GP educational supervisors (±150)	Interactive workshops: the potential of TRM for GPST
16/10/14	ISQUA conference	Conference Centre; Edinburgh; Scotland	Delegates (±50)	The TRM and GPST
11/09/13	NES GP Appraiser conference	Grand Central Hotel; Glasgow; Scotland	GP appraisers (±150)	How to successfully incorporate the TRM in GP appraisal
28/08/13	AMEE	Prague; Czech republic		Workshop: The TRM and self-directed learning Presentation: Potential of TRM for teaching and improving safety in GP
19/06/13	NHS Lanarkshire PLT event	Hamilton Race Course; Scotland	All Lanarkshire GPs and staff	The Trigger Review Method
12/06/13	NHS Dumfries and Galloway educational event	Dumfries; Scotland	GPs, PMs, PNs (±180)	Patient safety and the TRM
22/05/13	Patient Safety Conference	Conference Centre; Birmingham; England	Delegates	Two posters: Monte Carlo simulation and the potential of the TRM for GPST
30/04/13	SPSP-PC Learning Event	AA football club; Scotland	A range of GP staff and clinicians (±250)	The TRM

15/03/13	SPSP-PC National Launch	Herriot Watt; Edinburgh; Scotland	A range of GP staff and clinicians (±250)	The TRM: Presentations, workshops
25/01/13	NADEGS conference	Stirling; Scotland	Staff from Scottish Medical schools	'There is safe and then there is safe': A qualitative exploration of patient safety in GP using NPT
24/09/12	Lothian Local Enhanced Patient safety workshop	Murrayfield; Edinburgh; Scotland	A range of GP staff and clinicians (±200)	The TRM
27/06/12	GGC Local Enhanced Patient safety Service workshop	Glasgow; Scotland	A range of GP staff and clinicians (±100)	Patient safety and general practice
31/05/12	The Scottish Practice Management Development Conference	Crieff Hydro; Scotland	GP Practice Managers	The Trigger Review Method, safety culture and patient safety
29/03/12	SIPC national learning event	Glasgow; Scotland	SIPC participants and board support staff	The Trigger Review Method
28/03/12	Lanarkshire PLT event	Lanark; Scotland	A range of GP staff and clinicians (±150)	The Trigger Review Method, safety culture and patient safety
01/03/12	MDDUS conference	Fairmont hotel; Scotland	A range of GP staff and clinicians	The Trigger Review Method, safety culture, never events and patient safety
20/01/12	NADEGS conference	Dundee; Scotland	Staff from Scottish Medical schools	The Trigger Review Method
05/2011	Patient safety congress	Birmingham; England	Conference delegates	Poster presentation of the TRM

References

1. Reason J. Human error: Models and management. BMJ. 2000;320:3.

2. Parnes B, Fernald D, Quintela J, Araya-Guerra R, Westfall J, Harris D, et al. Stopping the error cascade: a report on ameliorators from the ASIPS collaborative. Qual Saf Health Care. 2007;16(1):12-6.

3. Runciman WB, Hibbert PD, Thomson R, Van der Schaaf T, Sherman H, Lewalle P. Towards an international classification for patient safety: key concepts and terms. Int J Qual Health Care. 2009;21(1):9.

4. Fick DM, Cooper JW, Wade WE, Waller JL, Maclean JR, Beers MH. Updating the Beers criteria for potentially inappropriate medication use in older adults: results of a US consensus panel of experts. Arch Intern Med. 2003;163(22):2716-24.

5. May C, Finch T. Implementing, Embedding, and Integrating Practices: An Outline of Normalization Process Theory. Sociology. 2009;43(3):535-54.

6. Baxter R, Taylor N, Kellar I, Lawton R. What methods are used to apply positive deviance within healthcare organisations? A systematic review. BMJ Qual Saf. 2016;25(3):190-201.

7. O'Beirne M, Sterling PD, Zwicker K, Hebert P, Norton PG. Safety incidents in family medicine. BMJ Qual Saf. 2011;20(12):1005-10.

8. Pringle M, Bradley CP, Carmichael CM, Wallis H, Moore A. A study of the feasibility and potential of case-based auditing in primary medical care. Br J Gen Pract. 1995(Occasional Paper 70):21.

9. Amalberti R, Vincent C, Auroy Y, de Saint Maurice G. Violations and migrations in health care: a framework for understanding and management. Qual Saf Health Care. 2006;15 Suppl 1:i66-71.

10. Runciman WB, Sellen A, Webb RK, Williamson JA, Currie M, Morgan C, et al. Errors, incidents and accidents in Anasethetic practice. Anaesth Intens Care. 1993;21:14.

11. Brennan TA, Leape L, Laird N, Hebert L, RA. L, Lawthers A, et al. Incidence of adverse events and negligence in hospitalized patients: Results of the Harvard Medical Practice Study I. N Engl J Med. 1991;324(6):7.

12. Wilson RM, Runciman WB, Gibberd RW, Harrison BT, Newby L, Hamilton J. The Quality in Australian Health Care Study. Med J Aust. 1995;163:14.

13. Bhasale AL, Miller GC, Reid SE, Britt HC. Analysing potential harm in Australian general practice: an incident monitoring study. Med J Aust. 1998;169(2):4.

14. Vincent C, Neale G, Woloshynowych M. Adverse events in British hospitals: preliminary retrospective record review. BMJ. 2001;322:3.

15. Tehrani ASS, Lee H, Mathews SC, Shore A, Makary MA, Pronovost PJ, et al. 25-Year summary of US malpractice claims for diagnostic errors 1986-2010: an analysis from the National Practitioner Data Bank. BMJ Qual Saf. 2013;22(8):672-80.

16. Makeham MAB, Kidd MR, Saltman DC, Mira M, Bridges-Webb C, Cooper C, et al. The Threats to Australian Patient Safety (TAPS) study: incidence of reported errors in general practice. Med J Aust. 2006;185:4.

17. Tsang C, Majeed A, Banarsee R, Gnani S, Aylin P. Recording of adverse events in English general practice: analysis of data from electronic patient records. Inform Prim Care. 2010;18:8.

18. Landrigan CP, Parry GJ, Bones CB, Hackbarth AD, Goldmann DA, Sharek PJ. Temporal Trends in Rates of Patient Harm Resulting from Medical Care. N Engl J Med. 2010;363:11.

19. Tache SV, Sonnichsen A, Ashcroft DM. Prevalence of adverse drug events in ambulatory care: a systematic review. Ann Pharmacother. 2011;45(7-8):977-89.

20. Rosser W, Dovey S, Bordman R, White D, Crighton E, Drummond N. Medical errors in primary care: Results of an international study of family practice. Can Fam Physician. 2005;51:6.

21. Carson-Stevens A, Hibbert P, Williams H, Evans HP, Cooper A, Rees P, et al. Characterising the nature of primary care patient safety incident reports in the England and Wales National Reporting and Learning System: a mixed-methods agenda-setting study for general practice. Health Serv Deliv Res. 2016;4(27):106.

22. Panesar SS, deSilva D, Carson-Stevens A, Cresswell KM, Salvilla SA, Slight SP, et al. How safe is primary care? A systematic review. BMJ Qual Saf. 2016;25(7):544-53.

23. Jha AK, Prasopa-Plaizier N, Larizgoitia I, Bates DW. Patient safety research: an overview of the global evidence. Qual Saf Health Care. 2010;19(1):42-7.

24. Gaal S, Verstappen W, Wolters R, Lankveld H, van Weel C, Wensing M. Prevalence and consequences of patient safety incidents in general practice in the Netherlands: a retrospective medical record review study. Implement Sci. 2011;6:37.

25. Kohn LT, Corrigan JM, Donaldson MS. To Err Is Human: Building a Safer Health System. Washington (DC): The National Academies Press (US); 2000.

26. Committee on Quality of Health Care in America IoM. Crossing the quality chasm: a new health system for the 21st century. Washington (DC): The National Academies Press; 2001. 360 p.

27. Department of Health. An organisation with a memory: report of an expert group on learning from adverse events in the NHS. London: The Stationery Office; 2000.

28. Smith R. All changed, changed utterly: British medicine will be transformed by the Bristol case. BMJ. 1998;316.

29. Francis R. Report of the Mid Staffordshire NHS Foundation Trust Public Inquiry: Executive summary. London: The House of Commons; 2013.

30. Cosford P, Thomas J. Safer out of hours primary care. BMJ. 2010;341(3 July):3.

31. Vincent C. Patient Safety. London, UK: Elsevier Churchill; 2006. 254 p.

32. Hurwitz B, Sheikh A. Health care errors and Patient safety. West Sussex, UK: Wiley-Blackwell; 2009.

33. The Health Foundation. Evidence scan: Improving safety in primary care. London: The Health Foundation; 2011.

34. Kingston-Riechers J, Ospina M, Jonsson E, Childs P, Mcleod L, Maxted JM. Patient safety in primary care. Edmonton, AB: Canadian Patient Safety Institute and BC Patient Safety & Quality Council; 2010. 68 p.

35. Lorincz CY, Drazen E, Sokol PE, Neerukonda KV, Mezger J, Toepp MC, et al. Research in Ambulatory Patient Safety 2000-2010: A 10-year review. Chicago IL; 2010.

36. Whitlock EP, Lin JS, Chou R, Shekelle P, Robinson KA. Using existing systematic reviews in complex systematic reviews. Ann Intern Med. 2008;148(10):776-82.

37. Shojania KG, Sampson M, Ansari MT, Ji J, Doucette S, Moher D. How quickly do systematic reviews go out of date? A survival analysis. Ann Intern Med. 2007;147(4):224-33.

38. Agency for Healthcare Research and Quality. PSNet: Patient Safety Network 2017 [Available from: <u>https://psnet.ahrq.gov/</u>.

39. de Wet C, Kelly M, McKay J, Bowie P. The Trigger Review Method. In: Bowie P, de Wet C, editors. Safety and Improvement in Primary Care: The Essential Guide. London: Radcliffe Publishing Ltd; 2014.

40. de Wet C, Bowie P. The preliminary development and testing of a global trigger tool to detect error and patient harm in primary-care records. Postgrad Med J. 2009;85(1002):176-80.

41. Bowie P, Halley L, Blamey A, Gillies J, Houston N. Qualitative evaluation of the Safety and Improvement in Primary Care (SIPC) pilot collaborative in Scotland: perceptions and experiences of participating care teams. BMJ Open. 2016;6(1):e009526.

42. Healthcare Improvement Scotland. SPSP Primary Care. Scotland: Healthcare Improvement Scotland; 2016.

43. Information Services Division Scotland. Quality and Outcomes Framework: Prevalence, achievement, payment and exceptions data for Scotland, 2015/2016. Scotland: National Statistics; 2016.

44. May C, Finch T, Mair F, Ballini L, Dowrick C, Eccles M, et al. Understanding the implementation of complex interventions in health care: the normalization process model. BMC Health Serv Res. 2007;7:148.

45. Braithwaite J, Marks D, Taylor N. Harnessing implementation science to improve care quality and patient safety: a systematic review of targeted literature. Implement Sci. 2014;26(3):9.

46. May C. A rational model for assessing and evaluating complex interventions in health care. BMC Health Serv Res. 2006;6:86.

47. de Wet C. How primary care research contributes to improving the quality and safety of healthcare. In: Goodyear-Smith F, Mash R, editors. International perspectives on primary care research. London: CRC Press; 2016. p. 8.

48. de Wet C, Black C, Luty S, McKay J, O'Donnell CA, Bowie P. Implementation of the trigger review method in Scottish general practices: patient safety outcomes and potential for quality improvement. BMJ Qual Saf. 2016.

49. Kroll L, Singleton A, Collier J, Rees Jones I. Learning not to take it seriously: junior doctors' accounts of error. Med Educ. 2008;42(10):982-90.

50. Yu KH, Nation RL, Dooley MJ. Multiplicity of medication safety terms, definitions and functional meanings: when is enough enough? Qual Saf Health Care. 2005;14(5):358-63.

51. Elder NC, Pallerla H, Regan S. What do family physicians consider an error? A comparison of definitions and physician perception. BMC Fam Pract. 2006;7:73.

52. Jacobson L, Elwyn G, Robling M, Jones RT. Error and safety in primary care: no clear boundaries. Fam Pract. 2003;20(3):5.

53. Grober ED, Bohnen JMA. Defining medical error. Can J Surg. 2005;48(1):6.

54. Donaldson L. An international language for patient safety: Global progress in patient safety requires classification of key concepts. Int J Qual Health Care. 2009;21(1):1.

55. Nabhan M, Elraiyah T, Brown DR, Dilling J, LeBlanc A, Montori VM, et al. What is preventable harm in healthcare? A systematic review of definitions. BMC Health Serv Res. 2012;12:128.

56. Barach P, Johnson JK, Ahmad A, Galvan C, Bognar A, Duncan R, et al. A prospective observational study of human factors, adverse events, and patient outcomes in surgery for pediatric cardiac disease. J Thor Cardiov Surg. 2008;136(6):1422-8.

57. West DR, Wilson DP, Dickinson LM, Harris DM, Main DS, Westfall JM, et al. Relationship between patient harm and reported medical errors in primary care: a report from the ASIPS collaborative. In: Henriksen K, Battles JB, Keyes MA, Grady ML, editors. Advances in Patient Safety: New Directions and Alternative Approaches Volume 1: Assessment. Rockville (MD): Agency for Healthcare Research and Quality; 2008. p. 10. 58. Patanwala AE, Hays DP, Sanders AB, Erstad BL. Severity and probability of harm of medication errors intercepted by an emergency department pharmacist. Int J Pharm Pract. 2011;19(5):358-62.

59. Cheraghi-Sohi S, Singh H, Reeves D, Stocks J, Rebecca M, Esmail A, et al. Missed diagnostic opportunities and English general practice: a study to determine their incidence, confounding and contributing factors and potential impact on patients through retrospective review of electronic medical records. Implement Sci. 2015;10:105.

60. Bell BG, Campbell S, Carson-Stevens A, Evans HP, Cooper A, Sheehan C, et al. Understanding the epidemiology of avoidable significant harm in primary care: protocol for a retrospective cross-sectional study. BMJ Open. 2017;7(2):e013786.

61. Bates DW. Mountains in the clouds: patient safety research. Qual Saf Health Care. 2008;17(3):156-7.

62. Hippisley-Cox J, Vinogradova Y. Trends in Consultation Rates in General Practice 1995/1996 to 2008/2009: Analysis of the QResearch® database. The Information Centre for health and social care; 2009.

63. Hobbs FDR, Bankhead C, Mukhtar T, Stevens S, Perera-Salazar R, Holt T, et al. Clinical workload in UK primary care: a retrospective analysis of 100 million consultations in England, 2007-14. Lancet (London, England). 2016;387(10035):2323-30.

64. Information Services Division Scotland. Practice Team Information (PTI) Annual Update (2012/13). NHS National Services Scotland; 2013 29 October 2013.

65. Royal College of General Practitioners. Put patients first: Back general practice: Royal College of General Practitioners; 2017 [Available from: <u>http://www.rcgp.org.uk/campaign-home.aspx</u>.

66. Kostopoulou O, Delaney B. Confidential reporting of patient safety events in primary care: results from a multilevel classification of cognitive and system factors. Qual Saf Health Care. 2007;16(2):95-100.

67. Barnett K, Mercer SW, Norbury M, Watt G, Wyke S, Guthrie B. Epidemiology of multimorbidity and implications for health care, research, and medical education: a cross-sectional study. Lancet (London, England). 2012;380(9836):37-43.

68. Starfield B, Horder J. Interpersonal continuity: old and new perspectives. Br J Gen Pract. 2007;57(540):527-9.

69. Payne RA, Avery AJ, Duerden M, Saunders CL, Simpson CR, Abel GA. Prevalence of polypharmacy in a Scottish primary care population. Eur J Clin Pharmacol. 2014;70(5):575-81.

70. Panagioti M, Stokes J, Esmail A, Coventry P, Cheraghi-Sohi S, Alam R, et al. Multimorbidity and Patient Safety Incidents in Primary Care: A Systematic Review and Meta-Analysis. PLoS One. 2015;10(8):e0135947.

71. Sandars J, Esmail A. The frequency and nature of medical error in primary care: understanding the diversity across studies. Fam Pract. 2003;20(3):6.

72. Rubin G. Errors in general practice: development of an error classification and pilot study of a method for detecting errors. Qual Saf Health Care. 2003;12(6):443-7.

73. Woods DM, Thomas EJ, Holl JL, Weiss KB, Brennan TA. Ambulatory care adverse events and preventable adverse events leading to a hospital admission. Qual Saf Health Care. 2007;16(2):127-31.

74. Shojania KG, Dixon-Woods M. Estimating deaths due to medical error: the ongoing controversy and why it matters. BMJ Qual Saf. 2016.

75. Esmail A, Valderas JM, Verstappen W, Godycki-Cwirko M, Wensing M. Developing a research agenda for patient safety in primary care. Background, aims and output of the LINNEAUS collaboration on patient safety in primary care. Eur J Gen Pract. 2015;21 Suppl:3-7.

76. Kuo GM, Phillips RL, Graham D, Hickner JM. Medication errors reported by US family physicians and their office staff. Qual Saf Health Care. 2008;17(4):286-90.

77. Miller GC, Britt HC, Valenti LA. Adverse drug events in general practic patients in Australia. Med J Aust. 2006;184(7):5.

78. Barber ND, Alldred DP, Raynor DK, Dickinson R, Garfield S, Jesson B, et al. Care homes' use of medicines study: prevalence, causes and potential harm of medication errors in care homes for older people. Qual Saf Health Care. 2009;18(5):341-6.

79. Avery AJ, Ghaleb M, Barber N, Dean Franklin B, Armstrong SJ, Serumaga B, et al. The prevalence and nature of prescribing and monitoring errors in English general practice: a retrospective case note review. Br J Gen Pract. 2013;63(613):e543-53.

80. Howard R, Avery A, Bissell P. Causes of preventable drug-related hospital admissions: a qualitative study. Qual Saf Health Care. 2008;17(2):109-16.

81. Garfield S, Barber N, Walley P, Willson A, Eliasson L. Quality of medication use in primary care - mapping the problem, working to a solution: a systematic review of the literature. BMC Med. 2009;7:50.

82. Howard RL, Avery AJ, Slavenburg S, Royal S, Pipe G, Lucassen P, et al. Which drugs cause preventable admissions to hospital? A systematic review. Br J Clin Pharmacol. 2007;63(2):136-47.

83. Tam KWT, Kwok KH, Fan YMC, Tsui KB, Ng KK, Ho KYAH, et al. Detection and prevention of medication misadventures in general practice. Int J Qual Health Care. 2008;20(3):8.

84. Gurwitz JH, Field TS, Harrold LR, Rothschild JM, Debellis K, Seger AC, et al. Among Older Persons in the Ambulatory Setting Incidence and Preventability of Adverse Drug Events. JAMA. 2003;289(9):10.

85. Takata GS, Mason W, Taketomo C, Logsdon T, Sharek PJ. Development, testing, and findings of a pediatric-focused trigger tool to identify medication-related harm in US children's hospitals. Pediatrics. 2008;121(4):e927-35.

86. Avery AJ, Rodgers S, Franklin BD, Elliott RA, Howard R, Slight SP, et al. Research into practice: safe prescribing. Br J Gen Pract. 2014;64(622):259-61.

87. McCarthy L, Dolovich L, Haq M, Thabane L, Kaczorowski J. Frequency of risk factors that potentially increase harm from medications in older adults receiving primary care. Can J Clin Pharmacol. 2007;14(3):8.

88. Doubova SV, Reyes-Morales H, Torres-Arreola Ldel P, Suarez-Ortega M. Potential drug-drug and drug-disease interactions in prescriptions for ambulatory patients over 50 years of age in family medicine clinics in Mexico City. BMC Health Serv Res. 2007;7:147.

89. Stocks SJ, Kontopantelis E, Akbarov A, Rodgers S, Avery AJ, Ashcroft DM. Examining variations in prescribing safety in UK general practice: cross sectional study using the Clinical Practice Research Datalink. BMJ. 2015;351:h5501.

90. Zermansky AG, Petty DR, Raynor DK, Lowe CJ, Freemantle N, Vail A. Clinical medication review by a pharmacist of patients on repeat prescriptions in general practice: a randomised controlled trial. Health Technol Assess. 2002;6(20):93.

91. Royal S, Smeaton L, Avery AJ, Hurwitz B, Sheikh A. Interventions in primary care to reduce medication related adverse events and hospital admissions: systematic review and meta-analysis. Qual Saf Health Care. 2006;15(1):23-31.

92. Lainer M, Mann E, Sonnichsen A. Information technology interventions to improve medication safety in primary care: a systematic review. Int J Qual Health Care. 2013;25(5):9.

93. Lainer M, Vogele A, Wensing M, Sonnichsen A. Improving medication safety in primary care. A review and consensus procedure by the LINNEAUS collaboration on patient safety in primary care. Eur J Gen Pract. 2015;21 Suppl:14-8.

94. Weiner JP, Kfuri T, Chan K, Fowles JB. "e-latrogenesis": the most critical unintended consequence of CPOE and other HIT. J Am Med Inform Assoc. 2007;14(3):387-8; discussion 9.

95. Schiff GD. Diagnosis and diagnostic errors: time for a new paradigm. BMJ Qual Saf. 2014;23(1):1-3.

96. Graber ML, Trowbridge R, Myers JS, Umscheid CA, Strull W, Kanter MH. The next organisational challenge: finding and addressing diagnostic error. Joint Commission journal on quality and patient safety. 2014;40(3):9.

97. Singh H, Meyer AN, Thomas EJ. The frequency of diagnostic errors in outpatient care: estimations from three large observational studies involving US adult populations. BMJ Qual Saf. 2014;23(9):727-31.

98. Singh H, Thomas EJ, Khan MM, Petersen LA. Identifying Diagnostic Errors in Primary Care Using an Electronic Screening Algorithm. Arch Intern Med. 2007;167:7.

99. Graber ML. The incidence of diagnostic error in medicine. BMJ Qual Saf. 2013;22 Suppl 2:ii21-ii7.

100. Wallace E, Lowry J, Smith SM, Fahey T. The epidemiology of malpractice claims in primary care: a systematic review. BMJ Open. 2013;3(7).

101. Phillips RL. Learning from malpractice claims about negligent, adverse events in primary care in the United States. Qual Saf Health Care. 2004;13(2):121-6.

102. Moynihan R, Doust J, Henry D. Preventing overdiagnosis: how to stop harming the healthy. BMJ. 2012;344:e3502.

103. Newman-Toker DE. A unified conceptual model for diagnostic errors: underdiagnosis, overdiagnosis, and misdiagnosis. Diagnosis. 2014;1(1).

104. Singh H, Graber ML, Kissam SM, Sorensen AV, Lenfestey NF, Tant EM, et al. System-related interventions to reduce diagnostic errors: a narrative review. BMJ Qual Saf. 2012;21(2):160-70.

105. McDonald KM, Matesic B, Contopoulos-Ioannidis DG, Lonhart J, Schmidt E, Pineda N, et al. Patient Safety Strategies Targeted at Diagnostic Errors: A Systematic Review. Ann Intern Med. 2013;158:10.

106. Kostopoulou O, Delaney BC, Munro CW. Diagnostic difficulty and error in primary care - a systematic review. Fam Pract. 2008;25(6):400-13.

107. Ely JW, Kaldjian LC, D'Alessandro DM. Diagnostic errors in primary care: lessons learned. J Am Board Fam Med. 2012;25(1):87-97.

108. Schmidt HG, Mamede S, van den Berge K, van Gog T, van Saase JL, Rikers RM. Exposure to media information about a disease can cause doctors to misdiagnose similar-looking clinical cases. Acad Med. 2014;89(2):285-91.

109. Scott I. Errors in clinical reasoning: casues and remedial strategies. BMJ. 2009;339:4.

110. Sarkar U, Bonacum D, Strull W, Spitzmueller C, Jin N, Lopez A, et al. Challenges of making a diagnosis in the outpatient setting: a multi-site survey of primary care physicians. BMJ Qual Saf. 2012;21(8):641-8.

111. Norman G, Sherbino J, Dore K, Wood T, Young M, Gaissmaier W, et al. The etiology of diagnostic errors: a controlled trial of system 1 versus system 2 reasoning. Acad Med. 2014;89(2):277-84.

112. Singh H. Helping Health Care Organizations to Define Diagnostic Errors as Missed Opportunities in Diagnosis. Jt Comm J Qual Patient Saf. 2014;40(3):4.

113. Lo HG, Matheny ME, Seger DL, Bates DW, Gandhi TK. Impact of non-interruptive medication laboratory monitoring alerts in ambulatory care. J Am Med Inform Assoc. 2009;16(1):66-71.

114. Callen JL, Westbrook JI, Georgiou A, Li J. Failure to follow-up test results for ambulatory patients: a systematic review. J Gen Intern Med. 2012;27(10):1334-48.

115. Casalino L, Dunham D, Chin M, Bielang R, Kistner E, Karrison T, et al. Frequency of Failure to Inform Patients of Clinically Significant Outpatient Test Results. Arch Intern Med. 2009;169(12):8.

116. Elder NC, McEwen TR, Flach JM, Gallimore JJ. Management of test results in family medicine offices. Ann Fam Med. 2009;7(4):343-51.

117. Wahls TL, Cram PM. The frequency of missed test results and associated treatment delays in a highly computerized health system. BMC Fam Pract. 2007;8:32.

118. Hickner J, Graham DG, Elder NC, Brandt E, Emsermann CB, Dovey S, et al. Testing process errors and their harms and consequences reported from family medicine practices: a study of the American Academy of Family Physicians National Research Network. Qual Saf Health Care. 2008;17(3):194-200.

119. Bowie P, Price J, Hepworth N, Dinwoodie M, McKay J. System hazards in managing laboratory test requests and results in primary care: medical protection database analysis and conceptual model. BMJ Open. 2015;5(11):e008968.

120. Bowie P, Halley L, McKay J. Laboratory test ordering and results management systems: a qualitative study of safety risks identified by administrators in general practice. BMJ Open. 2014;4(2):e004245.

121. Elder NC. Laboratory testing in general practice: a patient safety blind spot. BMJ Qual Saf. 2015;24(11):667-70.

122. Bowie P, Forrest E, Price J, Verstappen W, Cunningham D, Halley L, et al. Good practice statements on safe laboratory testing: A mixed methods study by the LINNEAUS collaboration on patient safety in primary care. Eur J Gen Pract. 2015;21 Suppl:19-25.

123. Forster AJ, Murff HJ, Peterson JF, Gandhi TK, Bates DW. The Incidence and Severity of Adverse Events Affecting Patients after Discharge from the Hospital. Ann Intern Med. 2003;138:14.

124. Williams H, Edwards A, Hibbert P, Rees P, Prosser Evans H, Panesar S, et al. Harms from discharge to primary care: mixed methods analysis of incident reports. Br J Gen Pract. 2015;65(641):e829-37. 125. Perren A, Previsdomini M, Cerutti B, Soldini D, Donghi D, Marone C. Omitted and unjustified medications in the discharge summary. Qual Saf Health Care. 2009;18(3):205-8.

126. Witherington EM, Pirzada OM, Avery AJ. Communication gaps and readmissions to hospital for patients aged 75 years and older: observational study. Qual Saf Health Care. 2008;17(1):71-5.

127. Lindquist LA, Yamahiro A, Garrett A, Zei C, Feinglass JM. Primary care physician communication at hospital discharge reduces medication discrepancies. J Hosp Med. 2013;8(12):672-7.

128. Kwan JL, Lo L, Sampson M, Shojania KG. Medication Reconciliation During Transitions of Care as a Patient Safety Strategy: a systematic review. Ann Intern Med. 2013;158:10.

129. Gillespie U, Alassaad A, Henrohn D, Garmo H, Hammarund-Udenaues M, Toss H, et al. A Comprehensive Pharmacist Intervention to Reduce Morbidity in Patients 80 Years or Older: A Randomized Controlled Trial. Arch Intern Med. 2009;169(9):7.

130. Makeham MAB, Mira M, Kidd MR. Lessons from the TAPS study: Communication failures between hospitals and general practices. Aust Fam Physician. 2008;37(9):2.

131. Smith K. Effective communication with primary care providers. Pediatr Clin North Am. 2014;61(4):671-9.

132. Jackson PD, Biggins MS, Cowan L, French B, Hopkins SL, Uphold CR. Evidence Summary and Recommendations for Improved Communication during Care Transitions. Rehabil Nurs. 2016;41(3):135-48.

133. Gobel B, Zwart D, Hesselink G, Pijnenborg L, Barach P, Kalkman C, et al. Stakeholder perspectives on handovers between hospital staff and general practitioners: an evaluation through the microsystems lens. BMJ Qual Saf. 2012;Suppl 1:10.

134. Bell CM, Schnipper JL, Auerbach AD, Kaboli PJ, Wetterneck TB, Gonzales DV, et al. Association of communication between hospital-based physicians and primary care providers with patient outcomes. J Gen Intern Med. 2009;24(3):381-6.

135. Oduyebo I, Lehmann CU, Pollack CE, Durkin N, Miller JD, Mandell S, et al. Association of self-reported hospital discharge handoffs with 30-day readmissions. JAMA. 2013;173(8):624-9.

136. Coller RJ, Klitzner TS, Lerner CF, Chung PJ. Predictors of 30-day readmission and association with primary care follow-up plans. The Journal of pediatrics. 2013;163(4):1027-33.

137. Derkx H, Rethans JJ, Muijtjens A, Maiburg B, Winkens R, van Rooij H, et al. 'Quod scripsi, scripsi.' The quality of the report of telephone consultations at Dutch out-of-hours centres. Qual Saf Health Care. 2010;19(6):e1.

138. Smits M, Huibers L, Kerssemeijer B, de Feijter E, Wensing M, Giesen P. Patient safety in out-of-hours primary care: a review of patient records. BMC Health Serv Res. 2010;10:335.

139. Luu NP, Pitts S, Petty B, Sawyer MD, Dennison-Himmelfarb C, Boonyasai RT, et al. Provider-to-Provider Communication during Transitions of Care from Outpatient to Acute Care: A Systematic Review. J Gen Intern Med. 2016;31(4):417-25.

140. Dyer C. Bristol inquiry condemns hospital's "club culture". BMJ. 2001;323:1.

141. Department of Health. Building a safer NHS for patients: implementing an organisation with a memory. London: Department of health; 2001.

142. Dingwall R, Palmer C, Rowley E, Waring J, Murcott T. PS050: Synthesis of the outputs of research commissioned under the Patient Safety Research Portfolio. Nottingham: University of Nottingham Institute for Science and Society.; 2009.

143. The Scottish Government. Learning From Experience - How to Improve Safety for Patients in Scotland. Edinburgh: The Scottish Government; 2003.

144. National Patient Safety Agency. Seven steps to patient safety. London: National Patient Safety Agency; 2004.

145. The Health Foundation. Learning report: Safer Patients Initiative - Lessons from the first major improvement programme addressing patient safety in the UK. London: Health Foundation February 2011.

146. National Audit Office. A safer place for patients: Learning to improve patient safety. London: Department of Health; 2005.

147. NHS Quality Improvement Scotland. Safe Today - Safer Tomorrow. Scotland: NHS Quality Improvement Scotland; 2006.

148. Healthcare Improvement Scotland. SPSP Acute Adult - end of phase report. Scotland: Healthcare Improvement Scotland; 2016 August 2016.

149. The Health Foundation. Safer Patients Network: Evaluation - an independent evaluation of the Safer Patients Network improvement programme. London: Health Foundation; 2013.

150. Information Services Division Scotland. Quality and Outcomes Framework: General Practice 2016 [Available from: <u>http://www.isdscotland.org/Health-</u> <u>Topics/General-Practice/Quality-And-Outcomes-Framework/</u>.

151. Lester H, Campbell S. Developing Quality and Outcomes Framework (QOF) indicators and the concept of 'QOFability'. Qual Prim Care. 2010;18(2):103-9.

152. Steel N, Willems S. Research learning from the UK Quality and Outcomes Framework: a review of existing research. Qual Prim Care. 2010;18(2):117-25.

153. Downing A, Rudge G, Cheng Y, Tu YK, Keen J, Gilthorpe MS. Do the UK government's new Quality and Outcomes Framework (QOF) scores adequately measure

primary care performance? A cross-sectional survey of routine healthcare data. BMC Health Serv Res. 2007;7:166.

154. Campbell S, Reeves D, Kontopantelis E, Middleton E, Sibbald B, Roland M. Quality of Primary Care in England with the Introduction of Pay for Performance. N Engl J Med. 2007;357(2):10.

155. Oluwatowoju I, Abu E, Wild SH, Byrne CD. Improvements in glycaemic control and cholesterol concentrations associated with the Quality and Outcomes Framework: a regional 2-year audit of diabetes care in the UK. Diabet Med 2010;27(3):354-9.

156. Khunti K, Gadsby R, Millett C, Majeed A, Davies M. Quality of diabetes care in the UK: comparison of published quality-of-care reports with results of the Quality and Outcomes Framework for Diabetes. Diabet Med. 2007;24(12):1436-41.

157. Serumaga B, Ross-Degnan D, Avery AJ, Elliott RA, Majumdar SR, Zhang F, et al. Effect of pay for performance on the management and outcomes of hypertension in the United Kingdom: interrupted time series study. BMJ. 2011;342:d108.

158. Subramanian DN, Hopayian K. An audit of the first year of screening for depression in patients with diabetes and ischaemic heart disease under the Quality and Outcomes Framework. Qual Prim Care. 2008;16:4.

159. Walker S, Mason AR, Claxton K, Cookson R, Fenwick E, Fleetcroft R, et al. Value for money and the Quality and Outcomes Framework in primary care in the UK NHS. Br J Gen Pract. 2010;60(574):e213-20.

160. BBC News: Scotland politics. 'Outdated' QOF GP payment system scrapped in Scotland: BBC; 2015 [updated 01/10/2015. Available from: http://www.bbc.com/news/uk-scotland-scotland-politics-34419123.

161. Doran T, Kontopantelis E, Reeves D, Sutton M, Ryan AM. Setting performance targets in pay for performance programmes: what can we learn from QOF? BMJ. 2014;348:g1595.

162. Williams PH, de Lusignan S. Does a higher 'quality points' score mean better care in stroke? An audit of general practice medical records. Inform Prim Care. 2006;14(1):29-40.

163. Lester HE, Hannon KL, Campbell SM. Identifying unintended consequences of quality indicators: a qualitative study. BMJ Qual Saf. 2011;20(12):1057-61.

164. Reason J. The Human Contribution - Unsafe Acts, Accidents and Heroic Recoveries. Ashgate: Farnham; 2008.

165. Cooper MD. Towards a model of safety culture. Safety Science. 2000;36:26.

166. MacLean RHL. The Vale of Leven Hospital Inquiry Report: Executive Summary. Scotland: The Vale of Leven Hospital Inquiry; 2014 November 2014. 167. Flin R, Burns C, Mearns K, Yule S, Robertson EM. Measuring safety climate in health care. Qual Saf Health Care. 2006;15(2):109-15.

168. Kirk S, Parker D, Claridge T, Esmail A, Marshall M. Patient safety culture in primary care: developing a theoretical framework for practical use. Qual Saf Health Care. 2007;16(4):313-20.

169. Sexton JB, Helmreich RL, Neilands TB, Rowan K, Vella K, Boyden J, et al. The Safety Attitudes Questionnaire: psychometric properties, benchmarking data, and emerging research. BMC Health Serv Res. 2006;6:44.

170. de Wet C, Spence W, Mash R, Johnson P, Bowie P. The development and psychometric evaluation of a safety climate measure for primary care. Qual Saf Health Care. 2010;19(6):578-84.

171. Bowie P, de Wet C. Safety Culture. In: Bowie P, de Wet C, editors. Safety and Improvement in Primary Care: The Essential Guide. London: Radcliffe Publishing Ltd; 2014.

172. Wachter RM. Is Ambulatory Patient Safety Just Like Hospital Safety, Only without the "Stat"? Ann Intern Med. 2006;145:3.

173. Klemp K, Zwart D, Hansen J, Hellebek T, Luettel D, Verstappen W, et al. A safety incident reporting system for primary care. A systematic literature review and consensus procedure by the LINNEAUS collaboration on patient safety in primary care. Eur J Gen Pract. 2015;21 Suppl:39-44.

174. Dovey SM, Wallis KA. Incident reporting in primary care: epidemiology or culture change? BMJ Qual Saf. 2011;20(12):1001-3.

175. Jeffs L, Berta W, Lingard L, Baker GR. Learning from near misses: from quick fixes to closing off the Swiss-cheese holes. BMJ Qual Saf. 2012;21(4):287-94.

176. Olsen S, Neale G, Schwab K, Psaila B, Patel T, Chapman EJ, et al. Hospital staff should use more than one method to detect adverse events and potential adverse events: incident reporting, pharmacist surveillance and local real-time record review may all have a place. Qual Saf Health Care. 2007;16(1):40-4.

177. Wetzels R, Wolters R, van Weel C, Wensing M. Mix of methods is needed to identify adverse events in general practice: a prospective observational study. BMC Fam Pract. 2008;9:35.

178. Bowie P, Pope L, Lough M. A review of the current evidence base for significant event analysis. J Eval Clin Pract. 2008;14(4):520-36.

179. Bell BG, Spencer R, Marsden K, Perryman K, Campbell SM, Avery AJ. Building a Patient Safety Toolkit for use in general practice. InnovAiT. 2016;9(9):557-62.

180. Royal College of General Practitioners. Patient Safety Toolkit for General Practice 2017 [Available from: http://www.rcgp.org.uk/clinical-and-research/toolkits/patient-safety.aspx.

181. McKay J, Bradley N, Lough M, Bowie P. A review of significant events analysed in general practice: implications for the quality and safety of patient care. BMC Fam Pract. 2009;10:61.

182. Bowie P, McCoy S, McKay J, Lough JRM. Learning issues raised by the educational peer review of significant event analyses in general practice. Qual Prim Care. 2006;13:9.

183. Kluger AN, de Nisi A. The Effects of Feedback Interventions on Performance: A Historical Review, a Meta-Analysis, and a Preliminary Feedback Intervention Theory. Psych Bul. 1996;119(2):31.

184. Wu AW, Steckelberg RC. Medical error, incident investigation and the second victim: doing better but feeling worse? BMJ Qual Saf. 2012;21(4):267-70.

185. Scott SD, Hirschinger LE, Cox KR, McCoig M, Brandt J, Hall LW. The natural history of recovery for the healthcare provider "second victim" after adverse patient events. Qual Saf Health Care. 2009;18(5):325-30.

186. Sirriyeh R, Lawton R, Gardner P, Armitage G. Coping with medical error: a systematic review of papers to assess the effects of involvement in medical errors on healthcare professionals' psychological well-being. Qual Saf Health Care. 2010;19(6):e43.

187. O'Beirne M, Sterling P, Palacios-Derflingher L, Hohman S, Zwicker K. Emotional impact of patient safety incidents on family physicians and their office staff. J Am Board Fam Med. 2012;25(2):177-83.

188. Bowie P, McKay J, Dalgetty E, Lough M. A qualitative study of why general practitioners may participate in significant event analysis and educational peer assessment. Qual Saf Health Care. 2005;14(3):185-9.

189. Franklin BD, Shebl NA, Barber N. Failure mode and effects analysis: too little for too much? BMJ Qual Saf. 2012;21(7):607-11.

190. Gawande A. The checklist: if something so simple can transform intensive care, what else can it do? New Yorker (New York, NY : 1925). 2007:86-101.

191. Rosen MA, Pronovost PJ. Advancing the use of checklists for evaluating performance in health care. Acad Med. 2014;89(7):963-5.

192. Thomassen O, Storesund A, Softeland E, Brattebo G. The effects of safety checklists in medicine: a systematic review. Acta Anaesthesiol Scand. 2014;58(1):5-18.

193. Shekelle PG, Pronovost PJ, Wachter RM, McDonald KM, Schoelles K, Dy SM, et al. The Top Patient Safety Strategies That Can Be Encouraged for Adoption Now. Ann Intern Med. 2013;158(5):5.

194. Shapiro FE, Fernando RJ, Urman RD. Barriers to the implementation of checklists in the office-based procedural setting. J Healthc Risk Manag. 2014;33(4):35-43.

195. de Wet C, O'Donnell C, Bowie P. Never Events. In: Bowie P, de Wet C, editors. Safety and Improvement in Primary Care: The Essential Guide. London: Radcliffe Publishing Ltd; 2014.

196. Zaratkiewicz S, Whitney JD, Lowe JR, Taylor S, O'Donnell F, Minton-Foltz P. Development and Implementation of a Hospital-Acquired Pressure Ulcer Incidence Tracking System and Algorithm. J Healthc Qual. 2010;32(6):44-51.

197. de Wet C, O'Donnell C, Bowie P. Developing a preliminary 'never event' list for general practice using consensus-building methods. Br J Gen Pract. 2014;64(620):e159-67.

198. Brown J, Doloresco I, F., Mylotte JM. "Never events": not every hospital-acquired infection is preventable. Clin Infect Dis. 2009;49(5):4.

199. Macleod JB. Broadening never events: is it a plausible road to improved patient safety? Arch Surg. 2010;145(2):2.

200. Pronovost PJ, Goeschel CA, Wachter RM. The wisdom and justice of not paying for "preventable complications". JAMA. 2008;299(18):2197-9.

201. de Wet C, Johnson P, O'Donnell C, Bowie P. Can we quantify harm in general practice records? An assessment of precision and power using computer simulation. BMC Med Res Methodol. 2013;13:39.

202. Bowie P, de Wet C, McKay J. Criterion audit. In: Bowie P, de Wet C, editors. Safety and Improvement in Primary Care: the essential guide. London: Radcliffe Publishing Limited; 2014.

203. Lough JRM, McKay J, Murray TS. Audit and summative assessment: a criterion-referenced marking schedule. Br J Gen Pract. 1995;45:3.

204. Ivers N, Jamtvedt G, Flottorp S, Young JM, Odgaard-Jensen J, French SD, et al. Audit and feedback: effects on professional practice and healthcare outcomes. Cochrane Database Syst Rev. 2012(6):Cd000259.

205. Scott I. What are the most effective strategies for improving quality and safety of health care? Intern Med. 2009;39(6):389-400.

206. McKay J, Bowie P, Lough M. Variations in the ability of general medical practitioners to apply two methods of clinical audit: A five-year study of assessment by peer review. J Eval Clin Pract. 2006;12(6):622-9.

207. Bowie P, Cooke S, Lo P, McKay J, Lough M. The assessment of criterion audit cycles by external peer review - when is an audit not an audit? J Eval Clin Pract. 2007;13(3):352-7.

208. Hitchen L. England launches scheme to encourage use of "care bundles". BMJ. 2008;336(7639):294-5.

209. de Wet C, Bowie P. Care Bundles. In: de Wet C, Bowie P, editors. Safety and Improvement in Primary Care: the essential guide. London: Radcliffe Publishing Ltd; 2014.

210. Robb E, Jarman B, Suntharalingam G, Higgens C, Tennant R, Elcock K. Using care bundles to reduce in-hospital mortality: quantitative survey. BMJ. 2010;340:c1234.

211. Crunden E, Boyce C, Woodman H, Bray B. An evaluation of the impact of the ventilator care bundle. Nurs Crit Care. 2005;10(5):242-6.

212. Koehler BE, Richter KM, Youngblood L, Cohen BA, Prengler ID, Cheng D, et al. Reduction of 30-day postdischarge hospital readmission or emergency department (ED) visit rates in high-risk elderly medical patients through delivery of a targeted care bundle. J Hosp Med. 2009;4(4):211-8.

213. Guerin K, Wagner J, Rains K, Bessesen M. Reduction in central line-associated bloodstream infections by implementation of a postinsertion care bundle. Am J Infect Control. 2010;38(6):430-3.

214. Baldwin LN, Smith SA, Fender V, Gisby S, Fraser J. An audit of compliance with the sepsis resuscitation care bundle in patients admitted to A&E with severe sepsis or septic shock. Int Emerg Nurs. 2008;16(4):250-6.

215. Gao F, Melody T, Daniels DF, Giles S, Fox S. The impact of compliance with 6-hour and 24-hour sepsis bundles on hospital mortality in patients with severe sepsis: a prospective observational study. Crit Care. 2005;9(6):R764-70.

216. de Wet C, McKay J, Bowie P. Combining QOF data with the care bundle approach may provide a more meaningful measure of quality in general practice. BMC Health Serv Res. 2012;12:351.

217. Franx G, Meeuwissen JAC, Sinnema H, Spijker J, Huyser J, Wensing M, et al. Quality improvement in depression care in the Netherlands: the Depression Breakthrough Collaborative. A quality improvement report. Int J Integr Care. 2009;9:11.

218. Gould DA, Lynn J, Halper D, Myers SK, Simon L, Holmes H. The New York City Palliative Care Quality Improvement Collaborative. Jt Comm J Qual Patient Saf. 2007;33(6):10.

219. Nolan E, van Riper S, Talsma A, Mageno L, Richter A, Kearly G, et al. Rapid-cycle improvement in quality of care for patients hospitalized with acute myocardial
infarction or heart failure: moving from a culture of missed opportunity to a system of accountability J Cardiovasc Manag. 2005;16(1):6.

220. Varkey P, Sathananthan A, Scheifer A, Bhagra S, Fujiyoshi A, Tom A, et al. Using quality-improvement techniques to enhance patient education and counselling of diagnosis and management. Qual Prim Care. 2009;17(3):205-13.

221. de Wet C, Bowie P. The Plan-Do-Study-Act Method. In: de Wet C, Bowie P, editors. Safety and Improvement in Primary Care: the essential guide. London: Radcliffe Publishing Ltd; 2014.

222. Taylor MJ, McNicholas C, Nicolay C, Darzi A, Bell D, Reed JE. Systematic review of the application of the plan-do-study-act method to improve quality in healthcare. BMJ Qual Saf. 2014;23(4):290-8.

223. Auerbach AD, Landefeld CS, Shojania KG. The Tension between Needing to Improve Care and Knowing How to Do It. N Engl J Med. 2007;357(6):6.

224. Schouten LM, Hulscher ME, van Everdingen JJ, Huijsman R, Grol RP. Evidence for the impact of quality improvement collaboratives: systematic review. BMJ. 2008;336(7659):1491-4.

225. Baker R, Sullivan E, Camosso-Stefinovic J, Rashid A, Farooqi A, Blackledge H, et al. Making use of mortality data to improve quality and safety in general practice: a review of current approaches. Qual Saf Health Care. 2007;16(2):84-9.

226. Shojania KG, Thomas EJ. Trends in adverse events over time: why are we not improving? BMJ Qual Saf. 2013;22(4):273-7.

227. Woloshynowych M. Case record review of adverse events: a new approach. Qual Saf Health Care. 2003;12(6):411-5.

228. Baker GR, Norton PG, Flintoft V, Blais R, Brown A, Cox J, et al. The Canadian Adverse Events Study: the incidence of adverse events among hospital patients in Canada. Can Med Assoc J. 2004;170(11):9.

229. Davis P, Lay-Yee R, Briant R, Ali W, Scott A, Schug S. Adverse events in New Zealand public hospitals I: occurrence and impact. N Z Med J. 2002;115(1167):U271.

230. Unbeck M, Schildmeijer K, Henriksson P, Jurgensen U, Muren O, Nilsson L, et al. Is detection of adverse events affected by record review methodology? an evaluation of the "Harvard Medical Practice Study" method and the "Global Trigger Tool". Pt saf in Surg. 2013;7(10):12.

231. Naessens JM, O'Byrne TJ, Johnson MG, Vansuch MB, McGlone CM, Huddleston JM. Measuring hospital adverse events: assessing inter-rater reliability and trigger performance of the Global Trigger Tool. Int J Qual Health Care. 2010;22(4):9. 232. Jha AK, Kuperman GJ, Teich JM, Leape L, Shea B, Rittenberg E, et al. Identifying adverse drug events: development of a computer-based monitor and comparison with chart review and stimulated voluntary report. J Am Med Inform Assoc. 1998;5(3):305-14.
233. Resar RK. Methodology and rationale for the measurement of harm with trigger

234. Jick H. Drugs – Remarkably Nontoxic. N Engl J Med. 1974;291(16):824-8.

tools. Qual Saf Health Care. 2003;12(90002):39ii-45.

235. Classen DC, Pestotnik SL, Evans R, Burke JP. Computerized surveillance of adverse drug events in hospital patients. JAMA. 1991;266(20):2847-51.

236. Resar RK, Rozich JD, Simmonds T, Haraden CR. A trigger tool to identify adverse events in the intensive care unit. Jt Comm J Qual Patient Saf. 2006;32(10):585-90.

237. Griffin FA, Classen DC. Detection of adverse events in surgical patients using the Trigger Tool approach. Qual Saf Health Care. 2008;17(4):253-8.

238. General Medical Council. The Good medical practice framework for appraisal and revalidation. Manchester, UK: General Medical Council; 2013.

239. NHS Education for Scotland. Scottish Online Appraisal Resource (SOAR): Domain 2 - Safety and Quality: Trigger Review Documentation Scotland: NHS Education for Scotland; 2016 [Available from: <u>http://www.appraisal.nes.scot.nhs.uk/i-want-access-</u>to/toolkits/scottish-gp-appraisal-toolkit/domain-2/trigger-review-documentation.aspx.

240. Royal College of General Practitioners. 2 02 Patient Safety and Quality of Care 2017 [Available from: <u>http://www.rcgp.org.uk/training-exams/gp-curriculum-overview/online-curriculum/working-in-systems-of-care/2-02-patient-safety-and-</u> quality-of-care.aspx.

241. Rozich JD. Adverse drug event trigger tool: a practical methodology for measuring medication related harm. Qual Saf Health Care. 2003;12(>3):194-200.

242. Hibbert PD, Molloy CJ, Hooper TD, Wiles LK, Runciman WB, Lachman P, et al. The application of the Global Trigger Tool: a systematic review. Int J Qual Health Care. 2016.

243. Murray E, Treweek S, Pope C, MacFarlane A, Ballini L, Dowrick C, et al. Normalisation process theory: a framework for developing, evaluating and implementing complex interventions. BMC Med. 2010;8:63.

244. Woolf SH. The Meaning of Translational Research and why it Matters. JAMA. 2008;299(2):3.

245. McEvoy R, Ballini L, Maltoni S, O'Donnell C, Mair F, Macfarlane A. A qualitative systematic review of studies using the normalization process theory to research implementation processes. Implement Sci. 2014;9(2):13.

246. Eccles MP, Mittman BS. Welcome to Implementation Science. Implement Sci. 2006;1(1).

247. Eccles MP, Armstrong D, Baker R, Cleary K, Davies H, Davies S, et al. An implementation research agenda. Implement Sci. 2009;4:18.

248. Grol RPTM, Bosch MC, Hulscher MEJL, Eccles MP, Wensing M. Planning and studying improvement in patient care: the use of theoretical perspectives. Milbank Q. 2007;85(1).

249. Eccles M, Grimshaw J, Walker A, Johnston M, Pitts N. Changing the behavior of healthcare professionals: the use of theory in promoting the uptake of research findings. J Clin Epidemiol. 2005;58(2):107-12.

250. Michie S, Johnston M, Francis J, Hardeman W, Eccles M. From Theory to Intervention: Mapping Theoretically Derived Behavioural Determinants to Behaviour Change Techniques. Appl Psychol 2008;57(4):660-80.

251. Craig P, Dieppe P, Macintyre S, Michie S, Nazareth I, Petticrew M, et al. Developing and evaluating complex interventions: the new Medical Research Council guidance. BMJ. 2008;337:a1655.

252. Michie S, Abraham C. Interventions to change health behaviours: evidence-based or evidence-inspired? Psychol Health. 2004;19(1):29-49.

253. Wacker JG. A definition of theory: research guidelines for different theorybuilding research method in operations management. J Oper Manag. 1998;16.

254. Nilsen P. Making sense of implementation theories, models and frameworks. Implement Sci. 2015;10:53.

255. Greenhalgh T, Robert G, MacFarlane F, Bate P, Kyriakidou O. Diffusion of innovations in service organizations: systematic review and recommendations. Milbank Q. 2004;82(4).

256. Rogers EM. Diffusion of innovations. Fourth ed. New York: Free Press; 1995.

257. Prochaska JO, Velicer WF. The Transtheoretical Model of Health Behavior Change. Am J Health Promot. 1997;12(1):38-48.

258. Custers EJ. Thirty years of illness scripts: Theoretical origins and practical applications. Med Teach. 2015;37(5):457-62.

259. Lewis AP, Bolden KJ. General practitioners and their learning styles. Br J Gen Pract. 1989;39:3.

260. Norman G. Research in medical education: three decades of progress. BMJ. 2002;324(7353):1560-2.

261. Ajzen I. The theory of Planned Behavior. Organ Behav Hum Dec. 1991;50.

262. Gollwitzer PM, Sheeran P. Implementation intentions and goal achievement: a meta-analysis of effects and processes. Adv Exp Soc Psychol. 2006;38:49.

263. Webb TL, Sheeran P. Does changing behavioral intentions engender behavior change? A meta-analysis of the experimental evidence. Psychol Bull. 2006;132(2):249-68.

264. May CR, Mair F, Finch T, MacFarlane A, Dowrick C, Treweek S, et al. Development of a theory of implementation and integration: Normalization Process Theory. Implement Sci. 2009;4:29.

265. May C. Towards a general theory of implementation. Implement Sci. 2013;8(18):14.

266. de Bont A, Bal R. Telemedicine in interdisciplinary work practices: on an IT system that met the criteria for success set out by its sponsors, yet failed to become part of every-day clinical routines. BMC Med Inform Decis Mak. 2008;8:47.

267. Williams T, May C, Mair F, Mort M, Gask L. Normative models of health technology assessment and the social production of evidence about telehealth care. Health Policy. 2003;64(1):39-54.

268. May CR, Finch TL, Cornford J, Exley C, Gately C, Kirk S, et al. Integrating telecare for chronic disease management in the community: what needs to be done? BMC Health Serv Res. 2011;11:131.

269. Morrison D, Mair FS. Telehealth in practice: using Normalisation Process Theory to bridge the translational gap. Prim Care Respir J. 2011;20(4):351-2.

270. Bamford C, Heaven B, May C, Moynihan P. Implementing nutrition guidelines for older people in residential care homes: a qualitative study using Normalization Process Theory. Implement Sci. 2012;7(106):13.

271. Murray E, Burns J, May C, Finch T, O'Donnell C, Wallace P, et al. Why is it difficult to implement e-health initiatives? A qualitative study. Implement Sci. 2011;6(6):11.

272. Blakeman T, Protheroe J, Chew-Graham C, Rogers A, Kennedy A. Understanding the management of early-stage chronic kidney disease in primary care: a qualitative study. Br J Gen Pract. 2012:10.

273. Gallacher K, May CR, Montori VM, Mair FS. Understanding patients' experiences of treatment burden in chronic heart failure using normalization process theory. Ann Fam Med. 2011;9(3):235-43.

274. Furler J, Spitzer O, Young D, Best J. Insulin in general practice: Barriers and enablers for timely initiation. Aus Fam Phys. 2012;40(6):5.

275. Gunn JM, Palmer VJ, Dowrick CF, Herrman HE, Griffiths FE, Kokanovic R, et al. Embedding effective depression care: using theory for primary care organisational and systems change. Implement Sci. 2010;5:62. 276. Franx G, Oud M, de Lange J, Wensing M, Grol R. Implementing a stepped-care approach in primary care: results of a qualitative study. Implement Sci. 2012;7:8.

277. Atkins S, Lewin S, Ringsberg KC, Thorson A. Provider experiences of the implementation of a new tuberculosis treatment programme: a qualitative study using the normalisation process model. BMC Health Serv Res. 2011;11:275.

278. Leon N, Lewin S, Mathews C. Implementing a provider-initiated testing and counselling (PITC) intervention in Cape town, South Africa: a process evaluation using the normalisation process model. Implement Sci. 2013;8(97):17.

279. MacFarlane A, O'Reilly-de Brun M, de Brun T, Dowrick C, O'Donnell C, Mair F, et al. Healthcare for migrants, participatory health research and implementation science-better health policy and practice through inclusion. The RESTORE project. Eur J Gen Pract. 2014;20(2):148-52.

280. Michie S, Johnston M, Abraham C, Lawton R, Parker D, Walker A, et al. Making psychological theory useful for implementing evidence based practice: a consensus approach. Qual Saf Health Care. 2005;14(1):26-33.

281. Sturmberg JP, Martin CM. Complexity and health - yesterday's traditions, tomorrow's future. J Eval Clin Pract. 2009;15(3):543-8.

282. Brown C, Hofer T, Johal A, Thomson R, Nicholl J, Franklin BD, et al. An epistemology of patient safety research: a framework for study design and interpretation. Part 2. Study design. Qual Saf Health Care. 2008;17(3):163-9.

283. Kvale S. Interviews: An introduction to qualitative research interviewing. Thousand Oaks, California: Sage Publications; 1996.

284. de Wet C, Bowie P. Screening electronic patient records to detect preventable harm: a trigger tool for primary care. Qual Prim Care. 2011;19(2):115-25.

285. Saldana J. The coding manual for qualitative researchers. Second ed. London: Sage Publications; 2013.

286. Charmaz K. Grounded theory. In: Emerson RM, editor. Contemporary field research: perspectives and formulations. Second ed. Prospect Heighs, IL.: Waveland Press; 2001. p. 335-52.

287. Basit TN. Manual or electronic? The role of coding in qualitative data analysis. Educ Res. 2003;45(2):11.

288. Patton MQ. Qualitative research & evaluation methods. Third ed. Thousand Oaks, California: Sage Publications; 2002.

289. Sipe LR, Ghiso MP. Developing conceptual categories in classroom descriptive research: some problems and possibilities. Anthropol Educ Q. 2004;35(4):13.

290. Dovey SM. A preliminary taxonomy of medical errors in family practice. Qual Saf Health Care. 2002;11(3):233-8.

291. Makeham MA, Dovey S, County M, Kidd MR. An international taxonomy for errors in general practice: a pilot study. Med J Aust. 2002;177:5.

292. Jacobs S, O'Beirne M, Derflingher LP, Vlach L, Rosser W, Drummond N. Errors and adverse events in family medicine: Developing and validating a Canadian taxonomy of errors. Can Fam Physician. 2007;53(February):7.

293. Buetow S, Kiata L, Liew T, Kenealy T, Dovey S, Elwyn G. Patient error: a preliminary taxonomy. Ann Fam Med. 2009;7(3):223-31.

294. Klemp K, Dovey S, Valderas JM, Rohe J, Godycki-Cwirko M, Elliott P, et al. Developing a patient safety incident classification system for primary care. A literature review and Delphi-survey by the LINNEAUS collaboration on patient safety in primary care. Eur J Gen Pract. 2015;21 Suppl:35-8.

295. May CR, Johnson M, Finch T. Implementation, context and complexity. Implement Sci. 2016;11(1):141.

296. de Wet C. The KM Seedat Lecture: Why patient safety incidents predictably occur in Family Medicine. Transactions. 2015;59(2):4.

297. Gaal S, van Laarhoven E, Wolters R, Wetzels R, Verstappen W, Wensing M. Patient safety in primary care has many aspects: an interview study in primary care doctors and nurses. J Eval Clin Pract. 2010;16(3):639-43.

298. Christensen JF, Levinson W, Dunn PM. The heart of darkness: the impact of perceived mistakes on physicians. J Gen Intern Med. 1992;7(4):424-31.

299. Fischer MA, Mazor KM, Baril J, Alper E, DeMarco D, Pugnaire M. Learning from mistakes. Factors that influence how students and residents learn from medical errors. J Gen Intern Med. 2006;21(5):419-23.

300. Beasley JW, Wetterneck TB, Temte J, Lapin JA, Smith P, Rivera-Rodriguez AJ, et al. Information chaos in primary care: implications for physician performance and patient safety. J Am Board Fam Med. 2011;24(6):745-51.

301. Gaal S, Verstappen W, Wensing M. Patient safety in primary care: a survey of general practitioners in The Netherlands. BMC Health Serv Res. 2010;10:21.

302. Lillis S, Lord H, Ward D. Implementing incident management - reservations of practice staff. N Z Fam Physician. 2008;35(4):4.

303. Manwell LB, Williams ES, Babbott S, Rabatin JS, Linzer M. Physician Perspectives on Quality and Error in the Outpatient Setting. Wisc Med J. 2009;108(3):6.

304. Amalberti R, Brami JJ. 'Tempos' management in primary care: a key factor for classifying adverse events, and improving quality and safety. BMJ Qual Saf. 2011;21:729-36.

305. Brami J, Amalberti R, Wensing M. Patient safety and the control of time in primary care: A review of the French tempos framework by the LINNEAUS collaboration on patient safety in primary care. Eur J Gen Pract. 2015;21 Suppl:45-9.

306. Ely JW, Levinson W, Elder NC, Mainous AG, 3rd, Vinson DC. Perceived causes of family physicians' errors. The Journal of family practice. 1995;40(4):337-44.

307. Slight SP, Howard R, Ghaleb M, Barber N, Franklin BD, Avery AJ. The causes of prescribing errors in English general practices: a qualitative study. Br J Gen Pract. 2013;63(615):e713-20.

308. Arfanis K, Smith A. Informal risk assessment strategies in health care staff: an unrecognized source of resilience? J Eval Clin Pract. 2012;18(6):1140-6.

309. Grant S, Mesman J, Guthrie B. Spatio-temporal elements of articulation work in the achievement of repeat prescribing safety in UK general practice. Sociol Health Illn. 2016;38(2):306-24.

310. Hann M, Bower P, Campbell S, Marshall M, Reeves D. The association between culture, climate and quality of care in primary health care teams. Fam Pract. 2007;24(4):323-9.

311. de Wet C, Johnson P, Mash R, McConnachie A, Bowie P. Measuring perceptions of safety climate in primary care: a cross-sectional study. J Eval Clin Pract. 2012;18(1):135-42.

312. Parker D. Psychological contribution to the understanding of adverse events in health care. Qual Saf Health Care. 2003;12(6):453-7.

313. Henriksen K, Kaplan HC. Hindsight bias, outcome knowledge and adaptive learning. Qual Saf Health Care. 2003;12(Suppl II):6.

314. Gurses AP, Ozok AA, Pronovost PJ. Time to accelerate integration of human factors and ergonomics in patient safety. BMJ Qual Saf. 2012;21(4):347-51.

315. Woolf SH. A String of Mistakes: The Importance of Cascade Analysis in Describing, Counting, and Preventing Medical Errors. Ann Fam Med. 2004;2(4):317-26.

316. Burnett S, Franklin BD, Moorthy K, Cooke MW, Vincent C. How reliable are clinical systems in the UK NHS? A study of seven NHS organisations. BMJ Qual Saf. 2012;21(6):466-72.

317. Hollnagel E. The ETTO principle: Efficiency-Thoroughness Trade-off. Why things that go right sometimes go wrong. Farnham, UK: Ashgate; 2009.

318. Hollnagel E. FRAM: The Functional Resonance Analysis Method: Modelling complex socio-technical systems. Farnham, UK: Ashgate; 2012.

319. Smith MW, Davis Giardina T, Murphy DR, Laxmisan A, Singh H. Resilient actions in the diagnostic process and system performance. BMJ Qual Saf. 2013;22(12):1006-13.

320. Braithwaite J, Wears RL, Hollnagel E. Resilient health care: turning patient safety on its head. Int J Qual Health Care. 2015;27(5):418-20.

321. Nemeth C, Wears R, Woods D, Hollnagel E, Cook R. Minding the Gaps: Creating Resilience in Health Care. In: Henriksen K, Battles JB, Keyes MA, editors. Advances in Patient Safety: New Directions and Alternative Approaches (Vol 3: Performance and Tools). Rockville (MD): Agency for Healthcare Research and Quality; 2008.

322. Singh R, McLean-Plunckett EA, Kee R, Wisniewski A, Cadzow R, Okazaki S, et al. Experience with a trigger tool for identifying adverse drug events among older adults in ambulatory primary care. Qual Saf Health Care. 2009;18(3):199-204.

323. Morris CJ. Indicators for preventable drug related morbidity: application in primary care. Qual Saf Health Care. 2004;13(3):181-5.

324. Singh H, Giardina TD, Forjuoh SN, Reis MD, Kosmach S, Khan MM, et al. Electronic health record-based surveillance of diagnostic errors in primary care. BMJ Qual Saf. 2012;21(2):93-100.

325. Brenner S, Detz A, Lopez A, Horton C, Sarkar U. Signal and noise: applying a laboratory trigger tool to identify adverse drug events among primary care patients. BMJ Qual Saf. 2012;21(8):670-5.

326. Forster AJ, Fung I, Caughey S, Oppenheimer L, Beach C, Shojania KG, et al. Adverse Events Detected by Clinical Surveillance on an Obstetric Service. Obstet Gynecol. 2006;108:11.

327. Kaafarani HM, Rosen AK, Nebeker JR, Shimada S, Mull HJ, Rivard PE, et al. Development of trigger tools for surveillance of adverse events in ambulatory surgery. Qual Saf Health Care. 2010;19(5):425-9.

328. Wetzels R, Wolters R, van Weel C, Wensing M. Harm caused by adverse events in primary care: a clinical observational study. J Eval Clin Pract. 2009;15(2):323-7.

329. McKay J, de Wet C, Kelly M, Bowie P. Applying the Trigger Review Method after a brief educational intervention: potential for teaching and improving safety in GP specialty training? BMC Med Educ. 2013;13:117.

330. Manaseki-Holland S, Lilford RJ, Bishop JR, Girling AJ, Chen YF, Chilton PJ, et al. Reviewing deaths in British and US hospitals: a study of two scales for assessing preventability. BMJ Qual Saf. 2016;0:10.

331. Opondo D, Eslami S, Visscher S, de Rooij SE, Verheij R, Korevaar JC, et al. Inappropriateness of medication prescriptions to elderly patients in the primary care setting: a systematic review. PLoS One. 2012;7(8):e43617.

332. Billett S. Readiness and learning in health care education. Clin Teach. 2015;12(6):367-72.

333. Gans RO. Mentoring with a formative portfolio: a case for reflection as a separate competency role. Med Teach. 2009;31(10):883-4.

334. Moon JA. A handbook of reflective and experiential learning: theory and pracice. Abingdon: Routledge Falmer; 2004.

335. Sharek PJ, Horbar JD, Mason W, Bisarya H, Thurm CW, Suresh G, et al. Adverse events in the neonatal intensive care unit: development, testing, and findings of an NICU-focused trigger tool to identify harm in North American NICUs. Pediatrics. 2006;118(4):1332-40.

336. Wu AW, Folkman S, McPhee SJ, Lo B. Do house officers learn from their mistakes? JAMA. 1991;265(16):2089-94.

337. Drupsteen L, Guldenmund FW. What Is Learning? A Review of the Safety Literature to Define Learning from Incidents, Accidents and Disasters. J Conting Crisis Manage. 2014;22(2):81-96.

338. Newman MC. The emotional impact of mistakes on family physicians. Arch Fam Med. 1996;5(2):71-5.

339. Waterman AD, Garbutt J, Hazel E, Dunagan WC, Levinson W, Fraser VJ, et al. The emotional impact of medical errors on practicing physicians in the United States and Canada. Jt Comm J Qual Patient Saf. 2007;33(8):467-76.

340. Venus E, Galam E, Aubert JP, Nougairede M. Medical errors reported by French general practitioners in training: results of a survey and individual interviews. BMJ Qual Saf. 2012;21(4):279-86.

341. Seys D, Wu AW, Van Gerven E, Vleugels A, Euwema M, Panella M, et al. Health care professionals as second victims after adverse events: a systematic review. Eval Health Prof. 2013;36(2):135-62.

342. Ahmed M, Arora S, Carley S, Sevdalis N, Neale G. Junior doctors' reflections on patient safety. Postgrad Med J. 2012;88(1037):125-9.

343. Gallagher TH, Waterman AD, Ebers AG, Fraser VJ, Levinson W. Patients' and Physicians' Attitudes Regarding the Disclosure of Medical Errors. JAMA 2003;289(8):7.

344. Fisseni G, Pentzek M, Abholz HH. Responding to serious medical error in general practice - consequences for the GPs involved: analysis of 75 cases from Germany. Fam Pract. 2008;25(1):9-13.

345. Kaplan HC, Brady PW, Dritz MC, Hooper DK, Linam WM, Froehle CM, et al. The Influence of Context on Quality Improvement Success in Health Care: A Systematic Review of the Literature. Milbank Q. 2010;88(4):60.

346. Dixon-Woods M, McNicol S, Martin G. Ten challenges in improving quality in healthcare: lessons from the Health Foundation's programme evaluations and relevant literature. BMJ Qual Saf. 2012;21(10):876-84.

347. Devlin AM, McGee-Lennon M, O'Donnell CA, Bouamrane MM, Agbakoba R, O'Connor S, et al. Delivering digital health and well-being at scale: lessons learned during the implementation of the dallas program in the United Kingdom. J Am Med Inform Assoc. 2016;23(1):48-59.

348. Verstappen W, Gaal S, Esmail A, Wensing M. Patient safety improvement programmes for primary care. Review of a Delphi procedure and pilot studies by the LINNEAUS collaboration on patient safety in primary care. Eur J Gen Pract. 2015;21 Suppl:50-5.

349. Health Quality & Safety Commission. The global trigger tool: A review of the evidence. Wellington: Health Quality & Safety Commission; 2016 January 2016.

350. Gandhi TK, Berwick DM, Shojania KG. Patient Safety at the Crossroads. JAMA. 2016;315(17):1829-30.

351. Wachter RM, Pronovost PJ, Shekelle PG. Strategies to Improve Patient Safety: The Evidence Base Matures. Ann Intern Med. 2013;156:3.

352. Pronovost PJ, Wachter RM. Progress in patient safety: a glass fuller than it seems. Am J Med Qual. 2014;29(2):165-9.

353. Christiaans-Dingelhoff I, Smits M, Zwaan L, Lubberding S, van der Wal G, Wagner C. To what extent are adverse events found in patient records reported by patients and healthcare professionals via complaints, claims and incident reports? BMC Health Serv Res. 2011;11:49.

354. Vincent C, Burnett S, Carthey J. Safety measurement and monitoring in healthcare: a framework to guide clinical teams and healthcare organisations in maintaining safety. BMJ Qual Saf. 2014.

355. Wilson AR, Dowd BE, Kralewski JE. Patient Safety Research in Medical Group Practices: Measurement and Data Challenges. Henriksen K, Battles J, Marks E, editors. Rockville (MD): Agency for Healthcare Research and Quality (US); 2005.

356. Muething SE, Conway PH, Kloppenborg E, Lesko A, Schoettker PJ, Seid M, et al. Identifying causes of adverse events detected by an automated trigger tool through indepth analysis. Qual Saf Health Care. 2010;19(5):435-9.

357. Morimoto T. Adverse drug events and medication errors: detection and classification methods. Qual Saf Health Care. 2004;13(4):306-14.

358. O'Leary KJ, Devisetty VK, Patel AR, Malkenson D, Sama P, Thompson WK, et al. Comparison of traditional trigger tool to data warehouse based screening for identifying hospital adverse events. BMJ Qual Saf. 2013;22(2):130-8.

359. Shortell SM, Jones RH, Rademaker AW, Gillies RR, Dranove DS, Hughes EF, et al. Assessing the impact of total quality management and organizational culture on

multiple outcomes of care for coronary artery bypass graft surgery patients. Med Care. 2000;38(2):207-17.

360. Shojania KG. The Elephant of Patient Safety: What You See Depends on How You Look. Jt Comm J Qual Patient Saf. 2010;36(9):6.

361. Jørgensen K. A tool for safety officers investigating "simple" accidents. Safety Sci 2011;49(1):32-8.

362. Siriwardena AN. Quality improvement projects for appraisal and revalidation of general practitioners. Qual Prim Care. 2011;19(4):205-9.

363. Taylor-Adams S, Brodia A, Vincent C. Safety Skills for clinicians: an essential component of patient safety. J Patient Saf. 2008;4(3):7.

364. Boonyasai RT, Windish DM, Chakraborti C, Feldman LS, Rubin HR, Bass EB. Effectiveness of teaching quality improvement to clinicians: a systematic review. JAMA. 2007;298(9):1023-37.

365. Wittich CM, Reed DA, Ting HH, Berger RA, Nowicki KM, Blachman MJ, et al. Measuring reflection on participation in quality improvement activities for maintenance of certification. Acad Med. 2014;89(10):1392-7.

366. Vincent C, Amalberti R. Chapter 9: Safety strategies in primary care. Safer Healthcare: strategies for the real world. Online: Springeropen; 2016. p. 14.

367. Buist M, Middleton S. What went wrong with the quality and safety agenda? BMJ. 2013;347:f5800.

368. Apekey TA, McSorley G, Tilling M, Siriwardena AN. Room for improvement? Leadership, innovation culture and uptake of quality improvement methods in general practice. J Eval Clin Pract. 2011;17(2):311-8.

369. Swinglehurst D, Greenhalgh T, Russell J, Myall M. Receptionist input to quality and safety in repeat prescribing in UK general practice: ethnographic case study. BMJ. 2011;343:d6788.

370. Reed JE, Card AJ. The problem with Plan-Do-Study-Act cycles. BMJ Qual Saf. 2016;25(3):147-52.

371. Newman-Toker DE, McDonald KM, Meltzer DO. How much diagnostic safety can we afford, and how should we decide? A health economics perspective. BMJ Qual Saf. 2013;22 Suppl 2:ii11-ii20.

372. Asch DA, Pauly MV, Muller RW. Asymmetric Thinking about Return on Investment. N Engl J Med. 2016;374(7):606-8.

373. de Rezende BA, Or Z, Com-Ruelle L, Michel P. Economic evaluation in patient safety: a literature review of methods. BMJ Qual Saf. 2012;21(6):457-65.

374. Etchells E, Koo M, Daneman N, McDonald A, Baker M, Matlow A, et al. Comparative economic analyses of patient safety improvement strategies in acute care: a systematic review. BMJ Qual Saf. 2012;21(6):448-56.

375. Dixon-Woods M, Pronovost PJ. Patient safety and the problem of many hands. BMJ Qual Saf. 2016;25(7):485-8.

376. Scott I, Phelps G. Measurement for improvement: getting one to follow the other. Int Med. 2009;39:5.

377. Vincent CA. Patient safety: what about the patient? Qual Saf Health Care. 2002;11(1):76-80.

378. Schoen C, Osborn R, Huynh PT, Doty M, Zapert K, Peugh J, et al. Taking the pulse of health care systems: experiences of patients with health problems in six countries. Health Affair. 2005;Suppl Web Exclusives:W5-509-25.

379. Evans SM, Berry JG, Smith BJ, Esterman AJ. Consumer perceptions of safety in hospitals. BMC Public Health. 2006;6:41.

380. Rhodes P, McDonald R, Campbell S, Daker-White G, Sanders C. Sensemaking and the co-production of safety: a qualitative study of primary medical care patients. Sociol Health Illn. 2016;38(2):270-85.

381. Waring JJ. Constructing and re-constructing narratives of patient safety. Soc Sci Med. 2009;69(12):1722-31.

382. Schildmeijer K, Nilsson L, Perk J, Arestedt K, Nilsson G. Strengths and weaknesses of working with the Global Trigger Tool method for retrospective record review: focus group interviews with team members. BMJ Open. 2013;3(9):e003131.

383. Leung L. Validity, reliability, and generalizability in qualitative research. Journal of Family Medicine and Primary Care. 2015;4(3):324-7.

384. Macfarlane A, O'Reilly-de Brun M. Using a theory-driven conceptual framework in qualitative health research. Qual Health Res. 2012;22(5):607-18.

385. Mair FS, May C, O'Donnell C, Finch T, Sullivan F, Murray E. Factors that promote or inhibit the implementation of e-health systems: an explanatory systematic review. Bull World Health Organ. 2012;90(5):357-64.

386. Sharek PJ, Parry G, Goldmann D, Bones K, Hackbarth A, Resar R, et al. Performance characteristics of a methodology to quantify adverse events over time in hospitalized patients. Health Serv Res. 2011;46(2):654-78.

387. Stockwell DC, Kirkendall ES, Muething SE, Kloppenborg E, Vinodrao H, Jacobs BR. Automated Adverse Event Detection Collaborative: Electronic Adverse Event Identification, Classification, and Corrective Actions Across Academic Pediatric Institutions. J Patient Saf. 2013;9:8. 388. Cantor MN, Feldman HJ, Triola MM. Using trigger phrases to detect adverse drug reactions in ambulatory care notes. Qual Saf Health Care. 2007;16(2):132-4.

389. Thomas EJ, Studdert DM, Brennan TA. The Reliability of Medical Record Review for Estimating Adverse Event Rates. Ann Intern Med. 2002;136:5.

390. Elder NC, Vonder Meulen M, Cassedy A. The identification of medical errors by family physicians during outpatient visits. Ann Fam Med. 2004;2(2):125-9.

391. Viera AJ, Garret JM. Understanding interobserver agreement: the Kappa statistic. Fam Med. 2005;37(5):4.

392. Mattsson TO, Knudsen JL, Lauritsen J, Brixen K, Herrstedt J. Assessment of the global trigger tool to measure, monitor and evaluate patient safety in cancer patients: reliability concerns are raised. BMJ Qual Saf. 2013;22(7):571-9.

393. Schildmeijer K, Nilsson L, Arestedt K, Perk J. Assessment of adverse events in medical care: lack of consistency between experienced teams using the global trigger tool. BMJ Qual Saf. 2012;21(4):307-14.

394. Forster AJ, Taljaard M, Bennett C, van Walraven C. Reliability of the peer-review process for adverse event rating. PLoS One. 2012;7(7):e41239.

395. Jeffries M, Phipps DL, Howard RL, Avery AJ, Rodgers S, Ashcroft DM. Understanding the implementation and adoption of a technological intervention to improve medication safety in primary care: a realist evaluation. BMC Health Serv Res. 2017;17(1):196.

396. Hastie IR, Paice E. Commentary: Now the wrong question? Qual Saf Health Care. 2003;12(3):227-8.

397. Ramanujam R, Goodman PS. The challenge of collective learning from event analysis. Safety Sci 2011;49(1):83-9.

398. Classen DC, Lloyd RC, Provost L, Griffin FA, Resar R. Development and Evaluation of the Institute for Healthcare Improvement Global Trigger Tool. J Patient Saf. 2008;4(3):9.

399. Verstappen W, Gaal S, Bowie P, Parker D, Lainer M, Valderas JM, et al. A research agenda on patient safety in primary care. Recommendations by the LINNEAUS collaboration on patient safety in primary care. Eur J Gen Pract. 2015;21 Suppl:72-7.

400. Cresswell KM, Panesar SS, Salvilla SA, Carson-Stevens A, Larizgoitia I, Donaldson LJ, et al. Global Research Priorities to Better Understand the Burden of Iatrogenic Harm in Primary Care: An International Delphi Exercise. PLoS Medicine. 2013;10(11):e1001554.

401. Zegers M, Hesselink G, Geense W, Vincent C, Wollersheim H. Evidence-based interventions to reduce adverse events in hospitals: a systematic review of systematic reviews. BMJ Open. 2016;6(9):e012555.

402. Weaver SJ, Lubomski LH, Wilson RF, Pfoh ER, Martinez KA, Dy SM. Promoting a culture of safety as a patient safety strategy - systematic review. Ann Intern Med. 2013;158(5 (Part 2)):7.

403. Graham DG, Harris DM, Elder NC, Emsermann CB, Brandt E, Staton EW, et al. Mitigation of patient harm from testing errors in family medicine offices: a report from the American Academy of Family Physicians National Research Network. Qual Saf Health Care. 2008;17(3):201-8.

404. Franklin BD, Reynolds M, Sadler S, Hibberd R, Avery AJ, Armstrong SJ, et al. The effect of the electronic transmission of prescriptions on dispensing errors and prescription enhancements made in English community pharmacies: a naturalistic stepped wedge study. BMJ Qual Saf. 2014;23(8):629-38.